FINAL REPORT

Impacts



ISBN 978-1-925622-54-6

© Commonwealth of Australia 2017

All material presented in this publication is provided under a Creative Commons Attribution 4.0 Australia licence (www.creativecommons.org/licenses).

For the avoidance of doubt, this means this licence only applies to material as set out in this document.



The details of the relevant licence conditions are available on the Creative Commons website as is the full legal code for the CC BY 4.0 AU licence (www.creativecommons.org/licenses).

Contact us

Enquiries regarding the licence and any use of this document are welcome at:

Attorney-General's Department 3-5 National Circuit Barton, ACT 2600

Email: copyright@ag.gov.au

FINAL REPORT

Volume 3 Impacts

Content warning				
Content warning				
Content warning				
Content warning				
ontent warning				
ontent warning				
ontent warning				
ontent warning				
ontent warning				
ontent warning				
ontent warning				
ontent warning				
ontent warning				
Content warning				
	Content warning			
his volume contains information about child sexual abuse that may be distressing. We also rish to advise Aboriginal and Torres Strait Islander readers that information in this volume may ave been provided by or refer to Aboriginal and Torres Strait Islander people who have died.	rish to advise Aboriginal and Torr	es Strait Islander readers	that information in th	is volume may

Table of contents

Prefa	ace		1
	The F	Royal Commission	1
	Publi	c hearings	1
	Priva	te sessions	2
	Polic	y and research	3
	Com	munity engagement	3
	Diver	rsity and vulnerability	3
	Our i	nterim and other reports	4
	Defir	nition of terms	4
	Nam	ing conventions	5
	Struc	cture of the Final Report	5
Sum	mary		9
	Unde	erstanding impacts	9
	Child	10	
		cutional responses can have significant impacts	11
	Child	sexual abuse can have ripple effects	12
1	Intro	duction	14
	1.1	Overview	14
	1.2	Terms of Reference	15
	1.3	Links with other volumes	16
	1.4	Limitations of our work	16
	1.5	Key terms	18
	1.6	Structure of this volume	20
2	Unde	erstanding impacts	23
	2.1	Complex and profound impacts	23
	2.2	Impacts differ by individual	24
	2.3	Impacts change over time	25
	2.4	Factors that influence impacts	30

3	Impa	cts of child sexual abuse	73	
	3.1	Effects of trauma on children's development	77	
	3.2	Mental health	84	
	3.3	Interpersonal relationships	107	
	3.4	Physical health	115	
	3.5	Sexual identity, gender identity and sexual behaviour	122	
	3.6	Connection to culture	130	
	3.7	Spirituality and religious involvement	133	
	3.8	Interactions with society	138	
	3.9	Education, employment and economic security	147	
4	Impa	Impacts of institutional responses		
	4.1	Institutional betrayal	173	
	4.2	Continuation of abuse	178	
	4.3	Re-traumatisation	181	
	4.4	Fear, distrust and contempt	186	
	4.5	Ostracism	189	
	4.6	Appropriate responses and empowerment	191	
5	Rippl	Ripple effects		
	5.1	Victims' families	203	
	5.2	Other affected individuals	220	
	5.3	Communities	224	
	5.4	Australian society	231	

Preface

The Royal Commission

The Letters Patent provided to the Royal Commission required that it 'inquire into institutional responses to allegations and incidents of child sexual abuse and related matters'. In carrying out this task, the Royal Commission was directed to focus on systemic issues, be informed by an understanding of individual cases, and make findings and recommendations to better protect children against sexual abuse and alleviate the impact of abuse on children when it occurs. The Royal Commission did this by conducting public hearings, private sessions and a policy and research program.

Public hearings

A Royal Commission commonly does its work through public hearings. We were aware that sexual abuse of children has occurred in many institutions, all of which could be investigated in a public hearing. However, if the Royal Commission was to attempt that task, a great many resources would need to be applied over an indeterminate, but lengthy, period of time. For this reason the Commissioners accepted criteria by which Senior Counsel Assisting would identify appropriate matters for a public hearing and bring them forward as individual 'case studies'.

The decision to conduct a case study was informed by whether or not the hearing would advance an understanding of systemic issues and provide an opportunity to learn from previous mistakes so that any findings and recommendations for future change the Royal Commission made would have a secure foundation. In some cases the relevance of the lessons to be learned will be confined to the institution the subject of the hearing. In other cases they will have relevance to many similar institutions in different parts of Australia.

Public hearings were also held to assist in understanding the extent of abuse that may have occurred in particular institutions or types of institutions. This enabled the Royal Commission to understand the ways in which various institutions were managed and how they responded to allegations of child sexual abuse. Where our investigations identified a significant concentration of abuse in one institution, the matter could be brought forward to a public hearing.

Public hearings were also held to tell the stories of some individuals, which assisted in a public understanding of the nature of sexual abuse, the circumstances in which it may occur and, most importantly, the devastating impact that it can have on people's lives. Public hearings were open to the media and the public, and were live streamed on the Royal Commission's website.

The Commissioners' findings from each hearing were generally set out in a case study report. Each report was submitted to the Governor-General and the governors and administrators of each state and territory and, where appropriate, tabled in the Australian Parliament and made publicly available. The Commissioners recommended some case study reports not be tabled at the time because of current or prospective criminal proceedings.

We also conducted some private hearings, which aided the Royal Commission's investigative processes.

Private sessions

When the Royal Commission was appointed, it was apparent to the Australian Government that many people (possibly thousands) would wish to tell us about their personal history of sexual abuse as a child in an institutional setting. As a result, the Australian Parliament amended the *Royal Commissions Act 1902* (Cth) to create a process called a 'private session'.

Each private session was conducted by one or two Commissioners and was an opportunity for a person to tell their story of abuse in a protected and supportive environment. Many accounts from these sessions are told in a de-identified form in this Final Report.

Written accounts allowed individuals who did not attend private sessions to share their experiences with Commissioners. The experiences of survivors described to us in written accounts have informed this Final Report in the same manner as those shared with us in private sessions.

We also decided to publish, with their consent, as many individual survivors' experiences as possible, as de-identified narratives drawn from private sessions and written accounts. These narratives are presented as accounts of events as told by survivors of child sexual abuse in institutions. We hope that by sharing them with the public they will contribute to a better understanding of the profound impact of child sexual abuse and may help to make our institutions as safe as possible for children in the future. The narratives are available as an online appendix to Volume 5, *Private sessions*.

We recognise that the information gathered in private sessions and from written accounts captures the accounts of survivors of child sexual abuse who were able to share their experiences in these ways. We do not know how well the experiences of these survivors reflect those of other victims and survivors of child sexual abuse who could not or did not attend a private session or provide a written account.

Policy and research

The Royal Commission had an extensive policy and research program that drew upon the findings made in public hearings and upon survivors' private sessions and written accounts, as well as generating new research evidence.

The Royal Commission used issues papers, roundtables and consultation papers to consult with government and non-government representatives, survivors, institutions, regulators, policy and other experts, academics, and survivor advocacy and support groups. The broader community had an opportunity to contribute to our consideration of systemic issues and our responses through our public consultation processes.

Community engagement

The community engagement component of the Royal Commission's inquiry ensured that people in all parts of Australia were offered the opportunity to articulate their experiences and views. It raised awareness of our work and allowed a broad range of people to engage with us.

We involved the general community in our work in several ways. We held public forums and private meetings with survivor groups, institutions, community organisations and service providers. We met with children and young people, people with disability and their advocates, and people from culturally and linguistically diverse communities. We also engaged with Aboriginal and Torres Strait Islander peoples in many parts of Australia, and with regional and remote communities.

Diversity and vulnerability

We heard from a wide range of people throughout the inquiry. The victims and survivors who came forward were from diverse backgrounds and had many different experiences. Factors such as gender, age, education, culture, sexuality or disability had affected their vulnerability and the institutional responses to the abuse. Certain types of institutional cultures and settings created heightened risks, and some children's lives brought them into contact with these institutions more than others.

While not inevitably more vulnerable to child sexual abuse, we heard that Aboriginal and Torres Strait Islander children, children with disability and children from culturally and linguistically diverse backgrounds were more likely to encounter circumstances that increased their risk of abuse in institutions, reduced their ability to disclose or report abuse and, if they did disclose or report, reduced their chances of receiving an adequate response.

We examined key concerns related to disability, cultural diversity and the unique context of Aboriginal and Torres Strait Islander experience, as part of our broader effort to understand what informs best practice institutional responses. We included discussion about these and other issues of heightened vulnerability in every volume. Volume 5, *Private sessions* outlines what we heard in private sessions from these specific populations.

Our interim and other reports

On 30 June 2014, in line with our Terms of Reference, we submitted a two-volume interim report of the results of the inquiry. Volume 1 described the work we had done, the issues we were examining and the work we still needed to do. Volume 2 contained a representative sample of 150 de-identified personal stories from people who had shared their experiences at a private session.

Early in the inquiry it became apparent that some issues should be reported on before the inquiry was complete to give survivors and institutions more certainty on these issues and enable governments and institutions to implement our recommendations as soon as possible. Consequently, we submitted the following reports:

- Working With Children Checks (August 2015)
- Redress and civil litigation (September 2015)
- *Criminal justice* (August 2017)

Definition of terms

The inappropriate use of words to describe child sexual abuse and the people who experience the abuse can have silencing, stigmatising and other harmful effects. Conversely, the appropriate use of words can empower and educate.

For these reasons, we have taken care with the words used in this report. Some key terms used in this volume are set out in Chapter 1, 'Introduction' and in the Final Report Glossary, in Volume 1, *Our inquiry*.

Naming conventions

To protect the identity of victims and survivors and their supporters who participated in private sessions, pseudonyms are used. These pseudonyms are indicated by the use of single inverted commas, for example, 'Roy'.

As in our case study reports, the identities of some witnesses before public hearings and other persons referred to in the proceedings are protected through the use of assigned initials, for example, BZW.

Structure of the Final Report

The Final Report of the Royal Commission into Institutional Responses to Child Sexual Abuse consists of 17 volumes and an executive summary. To meet the needs of readers with specific interests, each volume can be read in isolation. The volumes contain cross references to enable readers to understand individual volumes in the context of the whole report.

In the Final Report:

The **Executive Summary** summarises the entire report and provides a full list of recommendations.

Volume 1, Our inquiry introduces the Final Report, describing the establishment, scope and operations of the Royal Commission.

Volume 2, Nature and cause details the nature and cause of child sexual abuse in institutional contexts. It also describes what is known about the extent of child sexual abuse and the limitations of existing studies. The volume discusses factors that affect the risk of child sexual abuse in institutions and the legal and political changes that have influenced how children have interacted with institutions over time.

Volume 3, Impacts details the impacts of child sexual abuse in institutional contexts. The volume discusses how impacts can extend beyond survivors, to family members, friends, and whole communities. The volume also outlines the impacts of institutional responses to child sexual abuse.

Volume 4, *Identifying and disclosing child sexual abuse* describes what we have learned about survivors' experiences of disclosing child sexual abuse and about the factors that affect a victim's decision whether to disclose, when to disclose and who to tell.

Volume 5, *Private sessions* provides an analysis of survivors' experiences of child sexual abuse as told to Commissioners during private sessions, structured around four key themes: experiences of abuse; circumstances at the time of the abuse; experiences of disclosure; and impact on wellbeing. It also describes the private sessions model, including how we adapted it to meet the needs of diverse and vulnerable groups.

Volume 6, Making institutions child safe looks at the role community prevention could play in making communities and institutions child safe, the child safe standards that will make institutions safer for children, and how regulatory oversight and practice could be improved to facilitate the implementation of these standards in institutions. It also examines how to prevent and respond to online sexual abuse in institutions in order to create child safe online environments.

Volume 7, *Improving institutional responding and reporting* examines the reporting of child sexual abuse to external government authorities by institutions and their staff and volunteers, and how institutions have responded to complaints of child sexual abuse. It outlines guidance for how institutions should handle complaints, and the need for independent oversight of complaint handling by institutions.

Volume 8, Recordkeeping and information sharing examines records and recordkeeping by institutions that care for or provide services to children; and information sharing between institutions with responsibilities for children's safety and wellbeing and between those institutions and relevant professionals. It makes recommendations to improve records and recordkeeping practices within institutions and information sharing between key agencies and institutions.

Volume 9, *Advocacy, support and therapeutic treatment services* examines what we learned about the advocacy and support and therapeutic treatment service needs of victims and survivors of child sexual abuse in institutional contexts, and outlines recommendations for improving service systems to better respond to those needs and assist survivors towards recovery.

Volume 10, *Children with harmful sexual behaviours* examines what we learned about institutional responses to children with harmful sexual behaviours. It discusses the nature and extent of these behaviours and the factors that may contribute to children sexually abusing other children. The volume then outlines how governments and institutions should improve their responses and makes recommendations about improving prevention and increasing the range of interventions available for children with harmful sexual behaviours.

Volume 11, Historical residential institutions examines what we learned about survivors' experiences of, and institutional responses to, child sexual abuse in residential institutions such as children's homes, missions, reformatories and hospitals during the period spanning post-World War II to 1990.

Volume 12, Contemporary out-of-home care examines what we learned about institutional responses to child sexual abuse in contemporary out-of-home care. The volume examines the nature and adequacy of institutional responses and draws out common failings. It makes recommendations to prevent child sexual abuse from occurring in out-of-home care and, where it does occur, to help ensure effective responses.

Volume 13, Schools examines what we learned about institutional responses to child sexual abuse in schools. The volume examines the nature and adequacy of institutional responses and draws out the contributing factors to child sexual abuse in schools. It makes recommendations to prevent child sexual abuse from occurring in schools and, where it does occur, to help ensure effective responses to that abuse.

Volume 14, Sport, recreation, arts, culture, community and hobby groups examines what we learned about institutional responses to child sexual abuse in sport and recreation contexts. The volume examines the nature and adequacy of institutional responses and draws out common failings. It makes recommendations to prevent child sexual abuse from occurring in sport and recreation and, where it does occur, to help ensure effective responses.

Volume 15, Contemporary detention environments examines what we learned about institutional responses to child sexual abuse in contemporary detention environments, focusing on youth detention and immigration detention. It recognises that children are generally safer in community settings than in closed detention. It also makes recommendations to prevent child sexual abuse from occurring in detention environments and, where it does occur, to help ensure effective responses.

Volume 16, Religious institutions examines what we learned about institutional responses to child sexual abuse in religious institutions. The volume discusses the nature and extent of child sexual abuse in religious institutions, the impacts of this abuse, and survivors' experiences of disclosing it. The volume examines the nature and adequacy of institutional responses to child sexual abuse in religious institutions, and draws out common factors contributing to the abuse and common failings in institutional responses. It makes recommendations to prevent child sexual abuse from occurring in religious institutions and, where it does occur, to help ensure effective responses.

Volume 17, Beyond the Royal Commission describes the impacts and legacy of the Royal Commission and discusses monitoring and reporting on the implementation of our recommendations.

Unless otherwise indicated, this Final Report is based on laws, policies and information current as at 30 June 2017. Private sessions quantitative information is current as at 31 May 2017.

Summary

As a victim, I can tell you the memories, sense of guilt, shame and anger live with you every day. It destroys your faith in people, your will to achieve, to love, and one's ability to cope with normal everyday living.¹

This volume describes what we learned during our inquiry about the impacts of child sexual abuse and the impacts of institutional responses to that abuse on victims and their families, as well as on other people in the institution, community and wider society.

In private sessions and public hearings, we heard many stories of profound and wide-ranging impacts on the lives of victims, in both their childhood and throughout their adult lives. We also heard stories that demonstrated survivors' strength, resilience and courage in the face of this adversity. These stories are at the heart of this volume, which bears witness to the pain and courage of many, the full extent of which has been buried or minimised for many decades. Research on the impacts of child sexual abuse, and the more limited research on the impacts of child sexual abuse in institutional contexts, supports what we learned directly from survivors, their families and others about child sexual abuse.

Understanding impacts

The impacts of child sexual abuse are different for each victim. For many victims, the abuse can have profound and lasting impacts. They experience deep, complex trauma, which can pervade all aspects of their lives, and cause a range of effects across their lifespans. Other victims do not perceive themselves to be profoundly harmed by the experience.

Some impacts on victims are immediate and temporary, while others can last throughout adulthood. Some emerge later in life; others abate only to re-emerge or manifest in response to triggers or events. As victims have new experiences or enter new stages of development over their life courses, the consequences of abuse may manifest in different ways.

For many survivors we heard from, the impacts of sexual abuse are experienced as cumulative harm, resulting from multiple episodes of sexual abuse and other types of child maltreatment over prolonged periods. During this inquiry, we heard from many survivors who were sexually abused in residential institutions — including orphanages, homes, missions and detention centres — and whose adverse life experiences before, during and following the abuse compounded its negative effects. For some, their vulnerability to sexual abuse and its adverse impacts was heightened by their loss of connection to family, culture and country. We heard that Aboriginal and Torres Strait Islander survivors have faced a heavier burden of cumulative harm due to a range of historical and contemporary factors. We also heard that because children with disability can face additional barriers to disclosure of child sexual abuse, they are vulnerable to further abuse and therefore cumulative harm.

Many complex and interconnected factors can influence the way that victims are affected by child sexual abuse. While no single factor can accurately predict how a victim will respond, some factors appear to influence either the severity or type of impacts they experience. These factors include:

- the characteristics of the abuse (such as the type, duration and frequency)
- the relationship of the perpetrator to the child
- the social, historical and institutional contexts of the abuse
- the victim's circumstances, experiences and characteristics (such as age, gender, disability, prior maltreatment, and experiences with disclosing the abuse).

The sources of strength and resilience that some victims draw on over the course of their lives play a key role in how they cope with and manage the effects of the abuse. We heard that these sources of strength and resilience include: strong relationships and social support from families, peers and others; therapeutic activities; education, work and leisure activities; spirituality; cultural connection; and a variety of inner resources, such as optimism and hope.

Child sexual abuse can affect many areas of a person's life

Child sexual abuse can affect many areas of a person's life, including their:

- mental health
- interpersonal relationships
- physical health
- sexual identity, gender identity and sexual behaviour
- · connection to culture
- spirituality and religious involvement
- interactions with society
- education, employment and economic security.

For some victims, child sexual abuse results in them taking their own lives.

The impacts of child sexual abuse most commonly described in research and in our private sessions and public hearings were mental health impacts. Of the survivors who provided information in private sessions about the impacts of being sexually abused, 94.9 per cent told us about mental health impacts. These impacts included depression, anxiety and post-traumatic stress disorder (PTSD); other symptoms of mental distress such as nightmares and sleeping difficulties; and emotional issues such as feelings of shame, guilt and low self-esteem. Notably, mental health issues were often described as occurring simultaneously, rather than as isolated problems or disorders.

After mental health, relationship difficulties were the impacts most frequently raised by survivors in private sessions, including difficulties with trust and intimacy, lack of confidence with parenting, and relationship problems. Education and economic impacts were also frequently raised.

For many people, these diverse impacts are interconnected in complex ways, making it difficult to isolate the specific impacts of child sexual abuse. These interconnected impacts can be experienced at the same time or consecutively, as a cascade of effects over a lifetime. For instance, we heard from many survivors that they developed addictions after using alcohol or other drugs to manage the psychological trauma of abuse, which in turn affected their physical and mental health, sometimes leading to criminal behaviour and relationship difficulties.

Part of the explanation for the profound and broad-ranging impacts of child sexual abuse lies in the detrimental impacts that trauma can have on the biological, social and psychological development of a child. Child sexual abuse can result in profound trauma, affecting the chemistry, structure and function of the developing brain and potentially interrupting normal psychosocial development at every critical stage of a child's formative years.

While the impacts of child sexual abuse in institutional contexts are similar to those of child sexual abuse in other settings, we learned that there are often particular effects when a child is sexually abused in an institution. These include impacts on spirituality and religious involvement, such as a loss of faith or a loss of trust in a religious institution, for those victims sexually abused in such settings. We also heard that distrust and fear of institutions and authority are particular features of the effects of child sexual abuse in an institutional context.

Institutional responses can have significant impacts

How institutions respond to child sexual abuse – including their reactions to disclosure, action taken following abuse, and broader prevention and protection measures – can have a profound effect on victims. Institutional responses have the potential to either significantly compound or help alleviate the impacts of the abuse. These include the responses of the institution where the abuse took place and the institutions that have authority over, or responsibility for, that institution. They include the responses of the police, criminal justice system, complaint and oversight bodies, support services and health services.

Throughout this inquiry we heard from many survivors about further impacts they experienced because institutions failed to respond appropriately to child sexual abuse. We also heard how some institutions responded in ways that were actively damaging – for example, by perpetuating the abuse or punishing victims for disclosing it.

Inappropriate or damaging responses by institutions can result in the sexual abuse continuing for the victim, as well as placing other children at risk. Victims and their families can be left feeling betrayed by the institutions they trusted, resulting in fear and distrust of, and contempt

for, institutions. Survivors told us that these responses can not only compound the impacts of the abuse, but cause additional impacts and re-traumatisation. We heard that some victims were ostracised by the institution because of disclosing the abuse.

Survivors' experiences of institutional responses were not universally bad. We heard of some responses that were a source of justice and support to survivors. Other survivors spoke of mixed experiences of institutional responses. We heard that appropriate responses can ensure children are safe and promote healing for victims, helping them to manage the effects of the abuse and move forward with their lives. However, there is a paucity of research on the impacts of appropriate responses to child sexual abuse.

Child sexual abuse can have ripple effects

In addition to affecting the victim, child sexual abuse has ripple effects that reach a wider network of people, including the victim's family, carers and friends, as well as other children and staff in the institution in which the abuse occurred, the community and wider society. These ripple effects can be long-lasting, even affecting future generations.

While the experiences of parents, carers, siblings, partners, extended family and children ('secondary victims') are different to those of the victims, secondary victims can be significantly affected by child sexual abuse and how institutions respond to it. They, too, can suffer adverse impacts on their mental health, relationships, family functioning, employment, financial security and social connectedness.

We heard how child sexual abuse can have intergenerational impacts. Children of some survivors have been exposed to the debilitating effects of trauma on their parents and families, including mental health and relationship difficulties, alcohol and drug abuse and family breakdown. In some cases, including for victims who were sexually abused in residential care and Aboriginal and Torres Strait Islander victims, these effects can span multiple generations, perpetuating cycles of disadvantage and trauma.

Other people with a connection to the institution where the sexual abuse occurred – such as other children and staff at the institution, whistleblowers, and families of perpetrators – can also experience significant effects, for example on their mental health, social connections and employment.

Further, entire communities can be affected. We heard in private sessions and public hearings of Aboriginal and Torres Strait Islander communities, and religious and other communities, such as schools, experiencing negative collective impacts.

The ripple effects of child sexual abuse have adverse and ongoing social, cultural and economic impacts on broader society, as individuals, families, communities, institutions and services struggle to provide support and respond to the needs of victims and others affected.

Endnotes

1 Name changed, private session, 'William Peter'.

1 Introduction

1.1 Overview

At the end of the day, what matters is how us, the survivors, perceive ourselves ... we don't need your pity. What we need is ... people to have an understanding of what we have had to endure.¹

This Royal Commission came about in part because various governments and parliaments of the day came to recognise that child sexual abuse within institutions has affected many people profoundly.

When a child is sexually abused, the effects can be devastating. For many, the impacts of the abuse last for their whole lives. When the abuse occurs in an institution and the institutional response is inadequate, victims can experience additional adverse impacts. While each person's story is unique, for many victims the abuse has created a complex constellation of mental health symptoms and associated negative outcomes that have changed their lives profoundly. The abuse can reach into all areas of a victim's life: mental health; interpersonal relationships; physical health; sexual identity, gender identity and sexual behaviour; connection to culture; spirituality; interactions with society; and education, employment and economic security. However, despite traumatic childhood experiences and inadequate responses of institutions, many survivors have never lost their capacity to love and care for others. In this inquiry, we have seen many examples of survivor strength and courage. We heard stories of survivors who, despite experiencing a difficult life journey, have led fulfilling lives.

For decades, institutions have failed to recognise the impacts of child sexual abuse in any meaningful way. Even in more recent years, many institutions have been slow to accept the evidence of such impacts — even after multiple abuse allegations had been made and victims had disclosed both the abuse and its consequences. We have heard that some people within both the legal and healthcare systems have not fully understood the impacts on victims of child sexual abuse in institutional contexts.

Understanding the complexity, interconnectedness and cumulative impacts of child sexual abuse in institutional contexts is essential for better responding to the needs of victims and survivors, as well as others affected by the abuse. Those working in institutions that care for children; the legal system, including the judiciary; and people working in the social support and welfare sector need to be especially aware of the impacts of child sexual abuse in institutional contexts, in order to respond to victims, and their families and friends, with compassion and care. In particular, when relevant, the decisions of the judiciary must be informed by, and consistent with, the best available research about these impacts. Any outcome that ignores this evidence may be unjust for victims. People in the community, including children and their families and friends, can also better support victims and others when they understand the potentially profound and wide-ranging impacts of child sexual abuse in institutional contexts.

It is also important for governments, institutions and service providers to understand that child sexual abuse has a negative effect on many others, including families, communities and the institutions themselves. Ultimately, child sexual abuse has an enormous social and economic cost to Australian society as a whole.

Insights from the personal accounts of survivors throughout this inquiry have already transformed our understanding of the impacts of child sexual abuse in institutional contexts. Through private sessions and public hearings, we heard directly from survivors and their families about the complex and varied ways in which they have been affected by child sexual abuse. We also saw firsthand how survivors' public accounts of their experiences have started to transform the wider community's understanding of these impacts.

This volume and the Final Report aim to contribute further to this understanding.

1.2 Terms of Reference

Our Terms of Reference acknowledged the cost of child sexual abuse for individuals, the economy and society. They directed us to inquire into 'what institutions and governments should do to address, or alleviate the impact of, past and future child sexual abuse and related matters in institutional contexts'. To make appropriate recommendations on how to help victims, we needed to understand how they and their families have been, and still are, affected by what happened to them.

Our Terms of Reference also directed us to have regard to 'the experience of people directly or indirectly affected by child sexual abuse and related matters in institutional contexts'. We are committed to sharing these experiences, including the effects of child sexual abuse and institutional responses to it, with the Australian community. In this way, the experiences of victims can be validated, in contrast to the dismissal, denial and lack of recognition many victims and others told us they experienced. Although there were many commonalities, each person we heard from had their own story to tell about how the abuse had affected them, and their way of managing its effects.

1.3 Links with other volumes

This volume examines what we have learned during our inquiry about the impacts of child sexual abuse in institutional contexts.

It draws on discussion in Volume 2, *Nature and cause*, which discusses the nature, extent and causes of child sexual abuse in institutional contexts. It also links closely with Volume 4, *Identifying and disclosing child sexual abuse*. These volumes provide more detailed discussion about institutional, social and historical, and perpetrator and victim-related factors that can influence the way that victims are affected by child sexual abuse.

Volume 5, *Private sessions* discusses some of the common themes relating to the wellbeing and life outcomes of survivors who attended our private sessions. It is an important complement to this volume and provides more examples of what survivors told us in private sessions about how they have been affected by child sexual abuse.

Volume 9, Advocacy, support and therapeutic treatment services discusses the advocacy, support and treatment needs of primary and secondary victims of child sexual abuse. Understanding the complex and diverse ways that victims have been affected by child sexual abuse is key to providing appropriate advocacy, support and therapeutic treatment. Volume 3, Impacts has informed the development of recommendations in Volume 9, Advocacy, support and therapeutic treatment services.

Volumes 11–16 provide further information about how victims have been affected by child sexual abuse in institutional contexts. While Volume 3, *Impacts* addresses the impacts on victims of child sexual abuse in all institutional contexts, these volumes consider the particular impacts in certain institutions in more detail:

- Volume 11, Historical residential institutions
- Volume 12, Contemporary out-of-home care
- Volume 13, Schools
- Volume 14, Sport, recreation, arts, culture, community and hobby groups
- Volume 15, Contemporary detention environments
- Volume 16, Religious institutions.

1.4 Limitations of our work

Our work on the impacts of child sexual abuse in institutional contexts was limited by the available research and evidence.

1.4.1 Limited information on the impacts of child sexual abuse in institutional contexts

We faced challenges in examining the impacts of child sexual abuse in institutional contexts due to the limited available research and evidence. While there have been numerous studies on the impacts of child sexual abuse in the past, the focus has been on familial or other forms of extra-familial child sexual abuse, with little attention given to abuse in institutional contexts. Moreover, most research on child sexual abuse in institutions has focused on the impacts of historical abuse, predominantly in religious settings.² The impacts on secondary victims have also received little attention in existing research. The effects of institutional responses to sexual abuse on victims has likewise received limited research attention before the Royal Commission. Notwithstanding these limitations, we were able to use various information sources to inform our understanding of impacts, including our public hearings, private sessions, and policy and research program (see Preface to this volume).

1.4.2 Challenges in researching impacts

We had to be aware of methodological challenges to understanding the impacts of child sexual abuse, including:

- Many victims of child sexual abuse either do not fully disclose the abuse, or delay disclosing the abuse for many years (see Volume 4, *Identifying and disclosing child sexual abuse*). As a result, many experiences of child sexual abuse and the subsequent impacts are not taken into account in research.
- It can be difficult to distinguish the effects of child sexual abuse from other influences on victims' lives. While some studies have tried to account for the likely influence of other factors besides the sexual abuse itself, these studies often have limitations. More recent research has used more rigorous designs that consider these confounding factors and have more standardised measures.³ Where possible, we have referred to studies that have sought to account for these other factors. However, few studies focusing on child sexual abuse in institutional contexts use rigorous designs.
- Research on the impacts of child sexual abuse often draws on samples selected from a group of individuals or settings, such as universities or psychiatric settings, which may not be representative of the general population. This may mean the research underestimates or overestimates the impacts of abuse.
- Information from our private sessions suffers from similar limitations. Many victims
 who came to us were exposed to other forms of maltreatment before, during, and
 after the sexual abuse, which have added to the complexity of the impacts they have
 experienced. It is difficult to isolate the specific impacts of sexual abuse. Further,
 people attending our private sessions provided valuable information about the impacts
 of abuse, but, as a self-selected group may not be representative of the broader
 population of victims and survivors.

1.5 Key terms

The inappropriate use of words to describe child sexual abuse and the people who experience the abuse can have silencing, stigmatising and other harmful effects. Conversely, the appropriate use of words can empower and educate.

For these reasons, we have taken care with the words used in this report. Some key terms used in this volume are set out below. A complete glossary is contained in Volume 1, *Our inquiry*.

Children with harmful sexual behaviours

We use the term 'children with harmful sexual behaviours' to refer to children under 18 years who have behaviours that fall across a spectrum of sexual behaviour problems, including those that are problematic to the child's own development, as well as those that are coercive, sexually aggressive and predatory towards others. The term 'harmful sexual behaviours' recognises the seriousness of these behaviours and the significant impact they have on victims, but is not contingent on the age or capacity of a child.

The term 'children with harmful sexual behaviours' is used when referring to the general group of children with sexual behaviour problems.

At times, we use more specific terms:

- 'Problematic sexual behaviours' refers to sexual behaviours that fall outside the normal or age-appropriate range for younger children. These may or may not result in harm to another person. Problematic sexual behaviours in young children may be an indicator of them having been harmed and may place the child displaying such behaviours at risk of sexual exploitation.
- 'Sexual offending' refers to sexual behaviours that fall within the definition of a sexual offence, where the child could be held criminally responsible for their conduct. In Australia, children aged 10 and over may be charged with a sexual offence.

Perpetrator

We use the term 'perpetrator' for an adult who has sexually abused a child.

Secondary victims

The term 'secondary victims' is used in this volume to refer to people who are affected by the sexual abuse perpetrated against the primary victim (the child who is sexually abused). 'Secondary victims' can include partners, parents, children (including children born as a result of the abuse), siblings and extended family. The impacts of sexual abuse can also be felt by a wider range of people, including whistleblowers and other people (including other children) within the institution where the abuse occurred. There may also be collective trauma impacts for entire communities.

Trauma

'Trauma' refers to an event that is extremely harmful or distressing, such as experiencing or being threatened with sexual violence. The word also refers to a person's psychological response to the distressing event, immediately and over the medium and long term.⁴ In reference to child sexual abuse, we use trauma to describe experiences of abuse and institutional responses, as well as the ongoing impact that they have on the survivors' psychological wellbeing. Traumatic events 'involve threats to life or bodily integrity, or a close encounter with violence and death' which can 'overwhelm the ordinary human adaptations to life' and 'confront human beings with the extremities of helplessness and terror'.⁵ The overwhelming distress associated with the traumatic event can cause a range of ongoing psychological problems, including depression, anxiety, nightmares and flashbacks, hyper-arousal (heightened anxiety and alertness) and hyper-vigilance, hypo-arousal (delayed or weakened physical and cognitive responses) and dissociation, feelings of helplessness, problems with concentration and an exaggerated startle response.⁶ Experiences of trauma can inhibit survivors' capacity to regulate their emotional states.⁷

When diagnosing individuals with post-traumatic stress disorder (PTSD), clinicians use a specific set of criteria related to the type of experience and its ongoing psychological consequences. Our use of the word 'trauma' does not necessarily imply a formal diagnosis.

Victim and survivor

We use the terms 'victim' and 'survivor' to describe someone who has been sexually abused as a child in an institutional context. We use the term 'victim' when referring to a person who has experienced child sexual abuse at the time the abuse occurred. We use the term 'survivor' when referring to a person who has experienced child sexual abuse after the abuse occurred, such as when they are sharing their story or accessing support. Where the context is unclear, we have used the term 'victim'.

We recognise that some people prefer 'survivor' because of the resilience and empowerment associated with the term.

We recognise that some people who have experienced abuse do not feel that they 'survived' the abuse, and that 'victim' is more appropriate. We also recognise that some people may have taken their lives as a consequence of the abuse they experienced. We acknowledge that 'victim' is more appropriate in these circumstances. We also recognise that some people do not identify with either of these terms.

When we discuss quantitative information from private sessions in this volume, we use the term 'survivor' to refer both to survivors and victims who attended a private session and those (including deceased victims) whose experiences were described to us by family, friends, whistleblowers and others. This quantitative information is drawn from the experiences of 6,875 victims and survivors of child sexual abuse in institutions, as told to us in private sessions to 31 May 2017.

1.6 Structure of this volume

Chapter 2 describes some common themes that are important for understanding the impacts of child sexual abuse in institutional contexts, and the impacts of the institutional response. It discusses:

- how child sexual abuse can have complex and profound impacts
- how impacts may differ by individual
- how impacts can change over time
- some common factors that may influence how victims are affected by child sexual abuse
- sources of strength and resilience that survivors draw on to help them to cope with, and manage, these impacts.

Chapter 3 discusses the impacts of child sexual abuse that were commonly heard by Commissioners, including those particular to institutional contexts. It describes the immediate and long-term effects of such abuse on:

- children's development
- mental health
- interpersonal relationships
- physical health
- sexual identity, gender identity and sexual behaviour
- connection to culture
- spirituality and religious involvement
- interactions with society
- education, employment and economic security.

Chapter 4 addresses the impacts of institutional responses to child sexual abuse on victims. Commonly reported impacts of poor institutional responses included:

- feeling betrayed by the institution
- continued sexual abuse
- re-traumatisation
- fear, distrust of, and contempt for, institutions
- ostracism.

Chapter 4 also canvasses the way that appropriate responses can help victims to manage the impacts of child sexual abuse.

Chapter 5 outlines what we learned about the ripple effects of child sexual abuse in institutional contexts, which extend beyond primary victims to others. It discusses some of the commonly reported impacts on:

- victims' families, including parents, partners, siblings and children
- other people associated with the institution where the abuse occurred, including other children in the institution, staff in the institution, whistleblowers and the families of people who have sexually abused children
- communities, including religious and cultural communities
- Australian society.

Endnotes

- 1 Transcript of S Hallett, Case Study 43, 7 September 2016 at C18134:37–41.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 27.
- J Cashmore & R Shackel, *The long-term effects of child sexual abuse,* Australian Institute of Family Studies, Melbourne, 2013, pp 4–5.
- J Ford & C Courtois, 'Defining and understanding complex trauma and complex traumatic stress disorders' in C Courtois and J Ford (eds), *Treating complex traumatic stress disorders: An evidence-based guide*, The Guildford Press, New York, 2009, pp 14–15; J Atkinson, J Nelson, R Brooks, C Atkinson & K Ryan, 'Chapter 17: Addressing individual and community transgenerational trauma' in P Dudgeon, H Milroy and R Walker (eds), *Working together: Aboriginal and Torres Strait Islander mental health and wellbeing principles and practice*, Commonwealth of Australia, Canberra, 2014, p 291.
- 5 J Herman, *Trauma and recovery*, Basic Books, New York, 1992, p 33.
- P Ogden, K Minton & C Pain, *Trauma and the body: A sensorimotor approach to psychotherapy*, WW Norton & Company, New York, 2006, p 35; B Perry, 'Applying principles of neurodevelopment to clinical work with maltreated and traumatized children' in N Webb (ed), *Working with traumatized youth in child welfare*, The Guilford Press, New York, 2006, p 33; H Bath, 'The three pillars of trauma-informed care', *Reclaiming Children and Youth*, vol 17, no 3, 2008, p 18.
- 7 BA van der Kolk, 'Developmental trauma disorder', Psychiatric Annals, vol 35, no 5, 2005, p 403.

2 Understanding impacts

Chapter 2 describes some common themes that are important for understanding both the impacts of child sexual abuse in institutional contexts and the impacts of the institutional response to the abuse. It discusses:

- how child sexual abuse can have complex and profound effects on victims
- how impacts may differ by individual
- how impacts can emerge after 'trigger' events or at different stages of life, and can compound over time
- some common factors that may influence how victims are affected by child sexual abuse
- sources of strength and resilience that survivors draw on to help them to cope with, and manage, these impacts.

These themes underpin the remaining chapters in this volume, which examine the different types of impacts experienced by victims of child sexual abuse and others, and the institutional responses to that abuse.

2.1 Complex and profound impacts

Child sexual abuse can have a profound and lasting effect on a person's life. As one survivor told the Royal Commission:

As a victim, I can tell you the memories, sense of guilt, shame and anger live with you every day. It destroys your faith in people, your will to achieve, to love, and one's ability to cope with normal everyday living. It has [been] and is an enormous struggle to stay on top of life.²

Research published by the Royal Commission suggests that victims of child sexual abuse in institutional contexts can experience profound impacts, in similar ways to victims abused in other contexts.³ We heard in private sessions and public hearings that for many people, these impacts reached into all areas of their lives: mental and physical health, interpersonal relationships, culture, spirituality, education, and social and economic wellbeing. Many of these impacts are illustrated in Volume 5, *Private sessions* which discusses the wellbeing outcomes of survivors who attended private sessions.

As Dr Bruce Perry, Adjunct Professor at the Department of Psychiatry and Behavioural Sciences at Northwestern University in the USA, told us in *Case Study 57: Nature, cause and impact of child sexual abuse in institutional contexts (Nature, cause and impact of child sexual abuse)*, a range of physical, mental and social health problems are associated with child sexual abuse:

I think one of the big issues in our field broadly is recognising the role that sexual abuse can play in the manifestation of a whole variety of physical and mental health and social problems. In fact, if you look at just about any of the major health problems, mental health problems, social problems, you will find that individuals who are struggling with those issues have a much higher percentage of having been sexually abused than the comparison populations.⁴

For many people, these impacts are interconnected in complex ways, making it difficult to isolate the specific impacts of child sexual abuse. As child and family psychiatrist Dr Carolyn Quadrio explained in *Case Study 28: Catholic Church authorities in Ballarat (Catholic Church authorities in Ballarat)*:

It's not like you can isolate it like – just one organ; not like you've just got kidney damage but your liver is working fine. It's like every part of your person is affected by that [abuse], so it's really difficult to separate out. It's almost impossible to know what could have been if it hadn't been for the trauma, because every part of your being has been affected by it.⁵

These interconnected impacts can be experienced at the same time or consecutively, as a cascade of effects over a lifetime. For instance, some victims reported developing substance addictions after using alcohol or other drugs to manage the psychological trauma of abuse, which in turn affected their physical and mental health.

Part of the explanation for these effects lies in the detrimental impacts that interpersonal trauma can have on the biological, social and psychological development of the child (see Section 3.1). Many survivors told us that the abuse interfered with the way they developed emotionally. As one survivor said: 'I would prefer to be like, if he came up to me and he stabbed me in the leg ... Then stitched it up, there's a scar there ... But he's stabbed me within the basis of my human makeup'.⁶

The effects of interpersonal trauma on children are discussed in more detail in Chapter 3.

2.2 Impacts differ by individual

The effects of child sexual abuse are not uniformly felt among victims or across individuals' life courses. There is a complex association between a victim's experiences of sexual abuse in childhood, their reaction to the abuse and their wellbeing throughout their life. While some victims experience the consequences of abuse as deep, complex trauma that affects all aspects of life, others may not perceive themselves to be profoundly harmed by the experience. For example, in a private session we heard from a survivor, 'Warwick John', who recognised that the abuse he experienced was a serious crime and an abuse of trust. But he also considered what happened to him to be relatively 'trivial' and says that has meant he came out of it 'relatively unscathed'. He attributed his resilience to being brought up in a supportive family.⁸

Studies on the impacts of child sexual abuse indicate that not all adult survivors exhibit detrimental life outcomes. One study on the long-term impact of child abuse in religiously-affiliated institutions found that 12 per cent of a group of 76 adult males who were abused in such institutions showed an absence of mental health problems. A systematic literature review of research studies on resilience among victims of child sexual abuse across all settings also showed that some victims displayed a 'normal level of functioning' and/or an absence of psychological disorders at the time of the study.

Further, it is important to understand that while many victims will experience significant negative impacts at some stages of their lives, many will also lead fulfilling lives. Commissioned research involving qualitative and quantitative analysis of the private sessions we conducted found that:

some victims who suffered prolonged abuse, sometimes by multiple abusers in different institutions, had gone on to have relatively high levels of wellbeing, with good interpersonal relationships, productive employment and reasonable mental health.¹²

Survivor Ms Mary Farrell-Hooker echoed what many other survivors told us, when she gave evidence in *Case Study 7: Child sexual abuse at the Parramatta Training School for Girls and the Institution for Girls in Hay (Parramatta Training School for Girls)*. She said:

Just because you are a child abuse survivor doesn't make you a mental case ... My life has been affected greatly and yet, somehow, I have ended up with a loving husband, two beautiful children and four grandchildren, and we celebrate 30 years of marriage this May.¹³

Section 2.4.6 and Volume 5, *Private sessions* provide further examples of what survivors told us about the sources of strength and resilience which have helped them lead positive and fulfilling lives.

2.3 Impacts change over time

The impacts of child sexual abuse are variable, manifesting differently in each individual and changing over time. ¹⁴ Some impacts, such as certain physical injuries, are immediate and temporary. Others, such as the effects on mental health, can last well into adulthood and pervade all aspects of victims' lives. As one survivor told us, it is 'like a disease... it never goes away'. He said, 'it's always there ... it affects you on a daily basis'. ¹⁵ Some impacts emerge later in life; others lessen, only to re-emerge or manifest in response to triggers or events throughout life.

2.3.1 Triggering events

Some victims do not experience psychological problems until an event later in their life triggers them. Triggers' refer to objects, experiences or events that cause a victim to recall, often in a sensory manner, a previous traumatic memory. When asked about the short- and long-term effects of child sexual abuse during the case study on *Catholic Church authorities in Ballarat*, Dr Quadrio gave evidence that:

about 20 to 40 per cent of children who have been abused won't show any symptoms at all, and that's because some of them are what we describe as 'resilient': children who somehow survive trauma ... But some of those apparently non-symptomatic children become symptomatic later on. That's called the sleeper effect: that they look fine at the time and then some years later something else triggers it.¹⁸

Delayed symptoms can persist into adulthood. While some survivors appear to be functioning well in some or most aspects of life, they may still be deeply affected by the abuse, and may experience an acute episode following a specific event or trigger, including an encounter with the perpetrator. As one survivor told us:

I was at ... the Dawn Service ... with some of my friends and I saw him and I collapsed on the ground and I was dry-retching and stuff like that ... and then I had to grab [my friend] to stop [my friend] going and smacking his head in. And then I just ... freaked out.¹⁹

These triggers vary between victims because the context of the abuse, and the way that victims experience trauma, differ. In private sessions and public hearings survivors spoke of triggers that included: certain smells, objects or locations associated with the perpetrator or the abuse; other family members disclosing sexual abuse; media reports about child sexual abuse or the perpetrator; and invasive or intrusive medical procedures. The response of certain football (soccer), cricket and tennis organisations to allegations of child sexual abuse, told us his trauma was triggered by the sight of a brown handkerchief because the perpetrator would use one when with his victims. He said the sight of a brown handkerchief continued to make him freeze and would 'put me back to when I was 12 – when I was 13, 14, 15 and those dates'. He would then go on to re-live what had happened, waking up with cold sweats and night terrors.

'Cheryl Valerie' told us in a private session that when she was 17 years old, in the 1960's, she was subjected to numerous invasive gynaecological examinations, without her consent, by up to a dozen male medical students and doctors at a time, during her pre-natal care in a public hospital. Some years ago, being ordered to change into a medical gown for a medical scan triggered a flashback, and she was given Valium so that the scan could be carried out. 'Cheryl Valerie' told us that she has not been back for further follow-up scans, even though she is supposed to. 'I could not, it just brings it all back again.'²³

Other triggers relate to stressful life events, such as job loss, assault, relationship issues, major financial problems and serious illness or injury. The Royal Commission itself has acted as a trigger for some of the victims who came forward.²⁴ Similarly, investigations into abuse can trigger further trauma for victims, as well as potentially assisting with recovery.²⁵

2.3.2 Life stages

As victims have new experiences or enter a new stage of development over their life courses, the consequences of abuse may manifest in various ways.²⁶ The specific impacts at different life stages are discussed throughout Chapter 3.

In childhood, the impacts of sexual abuse can differ according to each critical stage of development. For example, in a pre-verbal child, hyper-vigilance, nightmares and hyper-arousal may be common trauma responses.²⁷ As a child grows older and enters puberty, questions of identity, self-image and interpersonal relationships may come to the fore. Adolescents can present with anxiety, depression, self-harming behaviours and drug dependency.²⁸ The specific impacts of sexual abuse during the stages of childhood are discussed more fully in Section 3.1.

Victims may feel few impacts of the child sexual abuse until they experience a major life event in adulthood, such as forming an intimate relationship or having a child themselves.²⁹ In a private session, 'Carol Jane' told us she had been sexually abused by her school teacher at the age of 11. After disclosing to her parents while still a child, she didn't speak about the abuse again until she told her husband after reading about the work of the Royal Commission. For many years, she thought she had dealt with the abuse and that it had not had any lasting effects. This changed when she became a parent. She said, 'When I became a mother is when it all came back to me ... [The children] weren't allowed to go to sleepovers – like if the kids were picked up from school, brought home, that was lovely'. Changes in that routine – 'If they went somewhere with somebody else' – made her fearful and anxious. 'I've never gone to the doctor and been treated but I do suffer terrible anxiety'.³⁰ Some survivors told us that their trauma was triggered when their child reached a certain age, or reached the age they were when they were abused.³¹

Negative impacts on employment and economic security also become evident in adulthood. Some survivors told us they were unable to settle into a job or career path, often because they had difficulties working under authority as a result of their experiences of sexual abuse by a trusted adult.

Some victims do not make connections until later in life between their experiences of child sexual abuse and issues they go on to deal with, including addiction, relationship breakdown and mental health issues. It is only when they seek help from support services, for example drug and alcohol services, or experience other significant life events, for example incarceration, that they start to question the source of their difficulties.³² For further details see Volume 9, *Advocacy, support and therapeutic treatment services*.

In older age, physical and psychological impacts can emerge or intensify. While there is little research on the effects of child sexual abuse on adults aged 65 and over,³³ the research that does exist suggests that the effects of child sexual abuse can reach into old age.³⁴ One review of the effects of childhood abuse considered the possibility that older adults may suffer continued, and possibly increased, difficulty in coping with unresolved childhood sexual trauma.³⁵ It suggested the difficulties may be compounded because of the loss of peer support, physiological changes in later life, and the loss of roles and resources that at one time served to distract the survivor and alleviate the stress arising from unresolved abuse issues.³⁶

A number of older survivors spoke to us in private sessions about memories of abuse returning as they aged. As 'Clementine' told us, 'the pain doesn't go away now that I'm older, it's worse because I don't work and I have more time to think'.³⁷

In the *Nature, cause and impact of child sexual abuse* case study, Dr Philomena Horsley, Research Fellow at La Trobe University, described the particular vulnerabilities of survivors of child sexual abuse as they become frail:

As people age and become older, particularly those in their 70s and 80s and 90s, there's a physical frailty, there's often a cognitive impairment of some kind, or moving into that space, as well as, in a social sense, greater isolation from networks, particularly if one is in care or going into care, and so lack of support. I think what can happen is as those feelings of vulnerability and lack of safety occur in someone's life, it can re-trigger memories, either specific memories or just a more generalised sense of feeling vulnerable, not feeling safe, feeling a lack of control about their physical surroundings, whether they're in a family home or whether they're in an aged-care facility or whether they're in a prison, for instance, which I think is a really under-recognised area. People become less physically mobile, so [have] less [of] a sense of empowerment in themselves.³⁸

Older survivors entering aged care facilities face particular challenges relating to the loss of privacy and loss of control. As Dr Horsley described:

What happens particularly in aged care settings and in many cases hospital settings where a lot [of] older people spend significant periods of time, is they lose their privacy. They often have intrusive physical examinations, they may be catheterised, they may have a range of UTIs [urinary tract infections], which is really common in older people, all of which involve other people touching their genitals or their body generally, who they don't know or have a relationship with.³⁹

For example, Dr Horsley described how in one aged care facility, in order to manage the incontinence of residents, staff would go around at night while people were asleep and put their hands down into their nappies to check whether or not they were wet. She said, 'You can imagine the experience that that would be for somebody who wakes or is subconsciously experiencing that'.⁴⁰ Fear and distrust of aged care is discussed further in Chapter 3.

Significant events at various life stages may also mitigate the effects of abuse. Some survivors said that having children or grandchildren, getting a job or establishing a close relationship signified a turning point towards recovery. The positive role of these events in survivors' lives is highlighted in Section 2.4.6.

2.3.3 Cumulative harm

As discussed in Volume 2, *Nature and cause*, we heard numerous accounts of children suffering multiple episodes of sexual abuse, many over prolonged periods. In private sessions, 92.3 per cent of people told us about the number of times they were sexually abused. Eighty-five per cent of these survivors told us they were sexually abused multiple times. Many victims told us they had experienced other forms of maltreatment in childhood – including physical and emotional abuse and neglect – before, during or after the sexual abuse. This includes children, particularly in out-of-home care, who may have experienced sexual abuse by family members before entering care. For these victims, the impacts of child sexual abuse are experienced as 'cumulative harm', defined as 'the effects of multiple adverse or harmful circumstances and events in a child's life'. These experiences can be profound and exponential, and can diminish a child's sense of safety, stability and wellbeing. Leading the service of the sexual abuse are experienced as 'cumulative harm', stability and wellbeing. Leading the sexual abuse are experienced as 'cumulative harm', stability and wellbeing. Leading the sexual abuse are experienced as 'cumulative harm', stability and wellbeing. Leading the sexual abuse are experienced as 'cumulative harm', stability and wellbeing. Leading the sexual abuse are experienced as 'cumulative harm', stability and wellbeing. Leading the sexual abuse are experienced as 'cumulative harm', stability and wellbeing. Leading the sexual abuse are experienced as 'cumulative harm', stability and wellbeing. Leading the sexual abuse are experienced as 'cumulative harm', stability and wellbeing. Leading the sexual abuse are experienced as 'cumulative harm', stability and wellbeing. Leading the sexual abuse are experienced as 'cumulative harm', stability and 'cumulative harm', stability and 'cumulative harm' and 'cumulative harm' are the sexual abuse and 'cumulative harm' are the sexual abuse are experienced as 'cumulative harm' are th

There is a growing body of research that shows the impact of such cumulative harm on the developing brain.⁴³ It shows how chronic stress sensitises neural pathways and overdevelops regions in the brain involved in anxiety and fear responses. Meanwhile, other neural pathways and regions in the brain are underdeveloped. Whereas brief stress promotes healthy regulatory abilities, repeated exposure is damaging and can interfere with a child's ability to monitor and regulate their emotions, behaviours and thoughts.⁴⁴

Some survivors we heard from have been particularly vulnerable to cumulative harm. In our private sessions with survivors in correctional institutions, we heard how multiple adversities in childhood, including domestic violence, family breakdown, racism, abuse and neglect, set them on a difficult life path, often into out-of-home care and correctional facilities, where further abuse occurred, compounding their trauma. We also heard about cumulative harm experienced by children with disability who face particular challenges in disclosing abuse, which may mean that the abuse is undetected and continues (see Section 2.4.5).

The cumulative effects of child sexual abuse and other maltreatment were also particularly evident in what we heard from survivors of abuse in out-of-home care. The cumulative harm many children experienced before entering care adversely affected their development and their capacity to form trusting relationships conducive to disclosing any subsequent sexual abuse and promoting their safety. Unlike children victimised in other institutional settings, many victims of child sexual abuse in out-of-home care faced additional ongoing disadvantages because their connections to family, community, culture and country were often severed.

Aboriginal and Torres Strait Islander survivors face a higher burden of cumulative harm from a range of factors most notably as a result of colonisation and the racially discriminatory policies of 'protectionism' and assimilation that followed.⁴⁵ This historical legacy – including the intergenerational transmission of grief and trauma⁴⁶ as well as cultural,⁴⁷ economic⁴⁸ and health impacts⁴⁹ – are, as concluded in *Bringing them home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families (Bringing them home)*, profound life-stressors in Aboriginal and Torres Strait Islander communities today.⁵⁰ High levels of interpersonal,⁵¹ as well as institutionalised, racism⁵² may also compound the impacts of sexual abuse for Aboriginal and Torres Strait Islander survivors.

2.4 Factors that influence impacts

As discussed above, the impacts of child sexual abuse in institutional contexts are not uniform among victims and can also vary throughout individuals' lives. The available research on child sexual abuse, which reflects what we were told in private sessions and public hearings, indicates that complex and connected factors influence the way victims are affected.⁵³ These factors include the:

- characteristics of the abuse, including the type, duration and frequency
- relationship of the perpetrator to the victim
- institutional contexts in which the abuse occurred
- social and historical contexts in which the abuse occurred
- individual characteristics, circumstances and experiences of the victim
- sources of strength and resilience available to each individual victim.

2.4.1 Characteristics of the abuse

Research has long suggested that the characteristics of child sexual abuse contribute to its impacts on victims.⁵⁴ While there is no single abuse-related factor that accurately predicts how a victim will respond to child sexual abuse, some factors appear to influence either the severity or the type of impact experienced.

Type, duration and frequency

Research has linked child sexual abuse involving physical violence and/or penetration, and child sexual abuse that occurs frequently and over long periods, with worse outcomes for victims than abuse without these features.⁵⁵

A number of studies on the impacts of child sexual abuse have indicated that sexual abuse with penetration leads to worse mental health outcomes for victims. Studies on child sexual abuse in institutional contexts have also made this association. For example, a study of the impacts of abuse in the Catholic Church in Austria found an association between post-traumatic stress disorder (PTSD) and sexual abuse involving penetration. Stexual abuse of children that involves penetration, especially younger children, can also cause severe physical injury (see Section 3.4). However, research suggests that penetration is only one of several aspects of abuse that influences the severity of outcomes for victims. The betrayal of trust and high levels of violence may also be factors associated with the seriousness of impacts.

While penetration may increase the risk of worse health outcomes, the absence of penetration does not mean that a victim suffers lesser impacts. We heard in private sessions from many survivors who were subjected to non-penetrative sexual abuse and suffered debilitating long-term impacts.⁵⁹

Similarly, while the frequency and duration of abuse are related to more psychosocial symptoms than abuse without these features, 60 we heard from survivors who suffered a single episode of abuse and had been severely affected. For example, 'Hallie' said a single incident of abuse at a migrant facility when she was seven significantly affected her life. She told us that the perpetrator was an older man at the facility, who lured her towards him on the pretext that he had locked himself out of his room and then sexually abused her. She said that throughout her life she has had difficulties forming intimate relationships and has suffered from self-doubt and low expectations of herself. Her fear of child sexual abuse perpetrators has made her reluctant to have children, in case they become victims. She said that she also finds medical procedures, such as pap smears and internal examinations, 'petrifying' and has changed doctors many times over the years to avoid them. 61

Other forms of maltreatment

We heard in private sessions that child sexual abuse is often accompanied by psychological or emotional abuse, and sometimes physical abuse and neglect. Research suggests that sexual abuse accompanied by other forms of maltreatment may be linked to greater mental health problems than child sexual abuse alone.⁶²

In private sessions, 57.2 per cent of survivors told us they had experienced another form of maltreatment connected to sexual abuse. Of this group, the most common form of co-occurring maltreatment experienced was emotional abuse (80.7 per cent), followed by physical abuse (64.4 per cent), witnessing the sexual or other abuse of children (18.1 per cent), neglect (15.7 per cent), and child labour (11.3 per cent). We heard how many victims of sexual abuse also experienced other harmful practices within the institution where they were sexually abused, including bullying, hazing, dehumanisation and, for Aboriginal and Torres Strait Islander children in particular, institutionalised racism (see Volume 2, *Nature and cause*).

The experiences of physical abuse were particularly evident in some of the children's residential institutions before the 1990s, which have been the focus of several of our public hearings. For example, in *Case Study 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland (The Salvation Army boys' homes, Australia Eastern Territory)*, we heard evidence of brutal physical abuse of boys in each of the four Salvation Army homes. Sexual abuse by officers or employees in the four homes was often accompanied by physical violence or the threat of violence.⁶³ Volume 11, *Historical residential institutions* discusses the experiences of children in these types of institutions.

In Case Study 40: The response of the Australian Defence Force to allegations of child sexual abuse, we heard evidence that victims were subjected to severe and degrading forms of sexual abuse, which occurred in the context of violent physical assaults. Sometimes this resulted in serious physical injuries.⁶⁴ CJA, who enlisted in the Navy in 1967, and was sent to the Navy's Junior Recruit Training Establishment at HMAS *Leeuwin* in Fremantle, Western Australia,⁶⁵ told us:

I was also a victim of 'midnight raids' which occurred while we slept. On these occasions I was woken up by punches to the face and the body by my attackers. I was sometimes beaten on the genitals or had my penis rubbed until I had an erection. On some occasions, base staff participated in these bashings. The night attacks, like the gauntlets, were spasmodic and irregular. I recall losing a tooth after one of these raids.⁶⁶

He said that the physical and sexual abuse that he suffered at HMAS *Leeuwin* has had a deep and far-reaching impact on his life.⁶⁷

Grooming

The grooming tactics that adult perpetrators or children with harmful sexual behaviours use to sexually abuse children and to prevent detection may also influence the way that victims are affected by child sexual abuse. As discussed in Volume 2, *Nature and cause*, grooming is designed to establish an emotional connection and build trust with a child or significant people in the child's life.⁶⁸ A perpetrator who is grooming a child may establish friendships and build special and exclusive relationships with the child or others over a long time. When a victim comes to the realisation later in life that what they regarded as affectionate attention from a caregiver was instead sexual abuse, it can be bewildering and lead to profound distrust about adult love and sexuality.⁶⁹ Studies of abuse in religious institutions show that being singled out by a leader, mentor or role model initially makes victims feel special, but later makes them feel conflicted and confused, and ultimately betrayed.⁷⁰

Professor Anne-Marie McAlinden, Director of Research, School of Law, Queen's University Belfast, giving evidence in the *Nature, cause and impact of child sexual abuse* case study, pointed out that grooming can work explicitly to prohibit disclosure by making the child feel that they will betray the abuser if they disclose. Disclosure is also discouraged when children are made to feel complicit in the sexual activity, such as by exposing them to pornography. This causes them to feel shame or guilt, and that they have acquiesced to the abuse in some way.⁷¹ She said:

in terms of adults, the therapeutic work that has been done with adolescent and adult victims has shown that grooming has a deeply emotional impact [and] psychological harm on the victim. Irrespective of the actual sexual abuse itself, the grooming itself is very harmful and it is something that needs unpacking in therapeutic work. It can last for years, if that's not dealt with, in terms of the emotional damage and the self-blame and the difficulty in forming adult relationships.⁷²

Some survivors in our private sessions told us that perpetrators made them feel special and loved, which led to conflicted feelings of guilt and betrayal when the grooming progressed to sexual abuse. For example, 'Sophia' told us that she was abused by a Catholic priest who was working in her local parish and who visited her family home. On one of his visits, which her family used to see as an honour, he asked for her help in the presbytery. She told us he then compelled her to perform oral sex on him, saying she was special and was making him happy. She said the abuse was repeated on multiple occasions. She said she did not tell her parents because she felt ashamed, guilty, and thought she would not be believed. 'Sophia' told us, 'What he did however was take a naïve, trusting 14-year-old girl and twist her mind so much that almost 40 years later she still feels that if only I had resisted him more it would not have happened'.⁷³

Similar conflicted feelings were reported in public hearings by victims of grooming. For example, in *Case Study 34: The response of Brisbane Grammar School and St Paul's School*, BQG told us he was sexually abused by Kevin Lynch, a school counsellor at Brisbane Grammar School.⁷⁴ We heard that Lynch effectively became a father figure to BQG.⁷⁵ He said, 'He became my dad. I felt really special and that the abuse was the price to me of feeling special. He opened his heart and he took in another son – or at the time, that's how it felt'.⁷⁶ In *Case Study 2: YMCA NSW's response to the conduct of Jonathan Lord*, we heard evidence from several parents about the betrayal their children felt when they were sexually abused by Jonathan Lord, who was working at YMCA NSW. The mother of one of Lord's victims, AH, said that he had conflicting emotions because he had thought Lord was his friend.⁷⁷ The mother of another victim, BA, said her son was devastated by his betrayal by a very important friend and is now mistrustful of adults and their motives.⁷⁸

Grooming can also mean that the abuse continues for longer, with greater negative impacts, because it is intended to both overcome the child's resistance and prevent the child from disclosing.⁷⁹

The impacts of grooming can extend to families, colleagues and others in institutional settings who may also have been groomed by perpetrators (see Volume 2, *Nature and cause*). We heard that parents can suffer guilt and feelings of betrayal, with the betrayal leading them to question their judgments and lose their trust in people. In *Case Study 37: The response of the Australian Institute of Music and RG Dance to allegations of child sexual abuse (<i>Centres for performing arts*), we heard from BZR, mother of BZP, who developed a close relationship with principal dance instructors Grant Davies (the perpetrator), and his sister, seeing them as her own 'big kids'.⁸⁰ BZR said:

Finding out about Grant was very tough. I lost trust in someone I knew so well not just as a dance teacher but as a father figure and a hero for BZP. I felt betrayed and felt like he had broken something inside of her. ... Personally I felt angry and guilty at myself for letting Grant into our lives and for not seeing him for what he was. I wonder if he manipulated me or whether I had allowed myself to be manipulated by him.⁸¹

Grooming can also affect a family because survivors may feel angry with and betrayed by their parents or carers for allowing the grooming to happen. We heard from some survivors that this had led to fractures in families.

Online image-based sexual abuse

As discussed in Volume 2, *Nature and cause*, the online environment, including social media and the internet, can be used to groom, sexually abuse and harass children.⁸² Online sexual abuse can include image-based abuse: the non-consensual capture and/or distribution of explicit or sexually-abusive images, including photographs, videos or computer-generated or altered images. This can also include the non-consensual sharing of intimate images that a child has previously shared consensually. Perpetrators can also use images of a child to groom the child for abuse that is perpetrated in person. Further, sexually explicit and non-sexually explicit images of children can be used in child sexual exploitation material distributed online, including through live streaming.⁸³

In a recent survey on the extent, nature and impacts of image-based abuse among youths and adults (aged 16 to 49) in Australia, victims reported high levels of psychological distress consistent with a diagnosis of moderate to severe depression and/or anxiety disorder.⁸⁴ In 2016, a large United States survey of young adults⁸⁵ who had been the victims of threats to expose sexual images reported the personal and psychological toll of image-based abuse, with 24 per cent seeing a medical or mental health practitioner.⁸⁶ Other impacts included leaving school or changing schools or experiencing other school-related problems, such as being blamed, shamed and bullied by peers.⁸⁷ Other research on the role of emerging technologies in young people's experiences of sexual violence suggests that the public nature of online sexual abuse can cause shame and humiliation.⁸⁸ Victims may also experience ongoing anxiety because of the permanence and pervasiveness of online images.⁸⁹

We heard in private sessions from survivors about the effects on them when explicit or sexually abusive images had been distributed online by the abusers, either adult perpetrators or other children. In some cases, the images were not actually of the victim, but had been artificially and digitally created. In a private session, 'Jodie' told us that she experienced image-based abuse by an adult perpetrator, 'John Maxwell', when she was a junior sports player in her teens.90 She said she started to receive large numbers of text messages from boys she didn't know in her sports social circle. It later emerged that someone had been impersonating her and sending images of naked, faceless young women to a large number of boys via text and email, requesting explicit images from them in return. She said that for over a year she was tormented by calls from other players that consisted of physical threats, hysterical laughter and 'the worst possible name-calling'. She withdrew from her sport. She told us that two years later, after an investigation by the Australian Federal Police, a sports coach, 'John Maxwell', whom she had only met once several years before, was identified as responsible. She believed that he had targeted her to use as bait to facilitate his access to boys. 'Jodie' wrote to the Commission: When it ended, and I mean legally ended, I was a scarily mature 18 year old, with depression. There was no physical element in my experience with 'John Maxwell' and I always considered myself one of the unbelievably lucky ones because of this'. But 'Jodie' is 'horribly saddened' that she considered herself lucky despite being 'humiliated, objectified, bullied, isolated and used as a sexual pawn'.91

Exposure to pornography may also have negative effects on children. In private sessions we heard that perpetrators had deliberately exposed children to pornography as part of the sexual abuse. A growing body of literature and commentary suggests that pornography may have negative impacts on some young people's attitudes to sex, sexuality and relationships.⁹²

Volume 6, Making institutions child safe, further discusses online child sexual abuse.

2.4.2 Relationship of perpetrator to victim

Research suggests that the nature of the relationship between a person who sexually abuses a child and the victim may contribute to the effects of the abuse.⁹³

Trusted adult or authority figure

Research on child sexual abuse in all settings suggests that abuse by trusted adults who are close to the child can increase the impacts of the abuse.⁹⁴ Children need to maintain attachment relationships for their survival, and it can lead to trauma when those relationships are betrayed as a result of abuse.⁹⁵ In institutional settings, child sexual abuse by a trusted and admired role model or spiritual leader has been noted across qualitative research studies to disorient the victim, leaving them with a profound sense of betrayal and powerlessness.⁹⁶

As discussed in Volume 2, Nature and cause, we heard in private sessions that while adult perpetrators of child sexual abuse held a wide variety of paid and voluntary positions in institutions, most held some position of leadership or authority, such as people in religious ministry, teachers and residential care workers. Common to most of these positions were access to children, power and authority over children, trust from children, and an expectation of obedience from children. For example, in Case Study 21: The response of Satyananda Yoga Ashram at Mangrove Mountain to allegations of child sexual abuse by the ashram's former spiritual leader in the 1970s and 1980s (Satyananda Yoga Ashram), the devotion to the gurudisciple relationship that was required in the practice of Satyananda Yoga ultimately culminated in a complete and unquestioning trust by both adults and children in the erratic and irrational actions of Akhandananda as the guru. 97 Many survivors have described to us the power exercised by perpetrators, either because of their personal characteristics, such as age, gender or competency, or because of their role or status. In the Centres for performing arts case study we were told by alleged victims that dance instructor and alleged perpetrator Grant Davies was charming, charismatic, motivational and encouraging. Research across a variety of settings describes the powerful charisma that perpetrators often possess.98

In religious settings, the power of perpetrators is enhanced by their representation of God, and their religious rituals and deportment.⁹⁹ For example, 'Bridget', who was raised Catholic in a family of generations of devout Catholics, spoke to us about 'the authority that a priest holds' and the 'fear they inspire':

As a little child you know, being brought up Catholic, their word is sacred. They hold the truth, they are respected, and you don't have anything on that. And when they treat you, like he treated me, you do feel like you are the ultimate dirt because this is the man of God doing this to you. You are nothing. To have that as a little child, that takes away anything, everything. It colours everything and every other person that you see and that you meet.¹⁰⁰

Dr David Ranson, Vicar General, Diocese of Broken Bay, and Parish Priest Wahroonga, New South Wales, described in *Case Study 50: Institutional review of Catholic Church authorities* the effect the exalted position of a Catholic priest could have on children:

Little children of two or three come to mass every weekend with their families. They see me in this exalted position and they naturally, through their childlike sense of wonder, equate me with God. This is an extraordinarily frightening proposition, and it has taught me the hurt and the extraordinary damage that is created in the experience of abuse — it is not just an individual abusing another individual; it is God who has abused, in the mind of the child, and this sets up irreparable damage.¹⁰¹

The combined effect of spiritual manipulation and the use of threats, coercion and bullying may have a distinct and discernible effect on the spiritual wellbeing of victims. ¹⁰² In public hearings and private sessions, survivors gave evidence about their spiritual manipulation by religious representatives who sexually abused them. For example, in *Case Study 4: The experiences of four survivors with the Towards Healing process,* one survivor said that she was told that if she loved God it would be okay to have sex with the perpetrator 'because he was God's representative'. ¹⁰³ We heard from Mr Joseph Kiernan in *Case Study 26: The response of the Sisters of Mercy, the Catholic Diocese of Rockhampton and the Queensland Government to allegations of child sexual abuse at St Joseph's Orphanage, Neerkol (St Joseph's Orphanage, Neerkol),* who said he was sexually abused by Father Reginald Durham, and was told not to tell anybody about it. ¹⁰⁴ Father Durham said to Mr Kiernan, 'you can't say anything, we're doing God's work'. ¹⁰⁵

The spiritual impacts of child sexual abuse in a religious context are discussed further in Chapter 3 and in Volume 16, *Religious institutions*.

Another child

The impacts of children's harmful sexual behaviours on victims can be as serious as those of adult-perpetrated abuse. Dr Wendy O'Brien, lecturer in criminology at Deakin University, gave evidence in *Case Study 45: Problematic and harmful sexual behaviours of children in schools* (*Harmful sexual behaviours of children in schools*) that the harm to a child who experiences harmful sexual behaviour from another child is profound. She said that as with adult-perpetrated sexual abuse, the longer that it continues, the more entrenched the harm.¹⁰⁶

While there is limited research on the impacts of harmful sexual behaviours on children in institutional contexts, ¹⁰⁷ many survivors told us in private sessions about the severe and complex effects the abuse has had on their lives. ¹⁰⁸ They described similar types of impacts to those described by victims of adult-perpetrated sexual abuse. For example, 'Bree Amy' told us she was eight years old when a boy in his mid-teens in the Anglican children's home where she lived sexually assaulted her. The abuse, which she said occurred on several occasions, left 'Bree Amy' with significant mental health issues, including agoraphobia, anxiety and substance misuse. She spent time in a psychiatric facility after becoming suicidal. 'I still to this day cannot sleep in the dark. I can't sleep without the door open so I have an escape. I slept in with my parents up until I was 17 years old'. She said she began drinking heavily in her teens to help her sleep. ¹⁰⁹

As is the case with abuse by trusted adults, abuse by an older child who is in a position of authority or serves as a role model may result in feelings of betrayal in the victim. Volume 10, *Children with harmful sexual behaviours* discusses how children in leadership positions or who have been afforded a level of authority by an institution may exploit greater access to, and power over other children with whom they have contact. In the *Harmful sexual behaviours of children in schools* case study, we heard evidence from CLG that he was mentally abused and physically assaulted, including by being 'blackballed' (putting black boot polish on his genitals)¹¹⁰ and raped, by older boys at The King's School boarding school in Sydney as a junior student in the 1970s.¹¹¹ He told us that as a child he developed PTSD and had a major break-down that led to attempted suicide.¹¹² In evidence he said that as an adult he has continued to deal with severe PTSD and has been 'on suicide watch' several times.¹¹³

In the case study, Dr O'Brien told us that children develop with respect to their networks and not in isolation – as members of community, families and peer groups. ¹¹⁴ If people in these support networks invalidate their disclosure, for example by minimising the abuse, this has significant impacts on them. ¹¹⁵ In some cases, such as sibling abuse, the sexual abuse breaks bonds of attachment, leading to feelings of betrayal:

Those bonds exist naturally in sibling groups and to an extent they exist naturally in peer groups as well. This is part of why we don't see disclosures sometimes, because a child may be reluctant to sever a bond with somebody that they trust and love, and it's also part of the reason why harm can be so profound in these contexts, because trust has been violated, profound trust. At a point at which children's attachments are so important, when those attachments are broken and violated, the betrayal is enormous.¹¹⁶

In this case study we also heard from school psychologist, Ms Kate Lumsdaine, about how a survivor, CLA, felt betrayed and humiliated after being sexually abused by others of his age at school who he thought were his friends.¹¹⁷

Many survivors who were sexually abused by other children told us in private sessions of their feelings of shame, fear and concern about reporting the abuse. ¹¹⁸ Further, we heard from some survivors that, as children, they did not understand that they had been abused, or that the other child's sexual behaviour was harmful, until many years later. ¹¹⁹

See Volume 10, Children with harmful sexual behaviours, for more information.

2.4.3 Institutional contexts

The impacts of child sexual abuse in institutional contexts are often similar to those experienced by victims of abuse in other settings. Yet research, together with what we have been told, suggests that abuse in institutions has specific impacts that can be related to the:

- character and culture of the institution
- type of institution
- institutional responses to child sexual abuse.¹²⁰

These factors may be strongly influenced by community-level factors, including social attitudes towards children in general and certain groups of children, and gender (see Volume 2, *Nature and cause*).¹²¹

Volume 5, *Private sessions* further details how victims' experiences of sexual abuse, including their vulnerability to abuse and the impacts of that abuse, differed depending on the institutional context and the period in which the abuse occurred.

Character and culture of institutions

The character and culture of the institution where abuse occurs are critical influences on how the abuse affects a victim. In Volume 2, *Nature and cause* we identify some of the factors related to institutional culture that increased the risk of child sexual abuse occurring, for example, failing to listen to, or value, children; encouraging secrecy and isolation; normalising harm or being insular and cut-off from the outside world. ¹²² Such cultural factors in turn influence the way that victims are affected by the abuse, creating barriers to disclosure and increasing feelings of betrayal, powerlessness and shame.

A commissioned review of research on the impacts of child sexual abuse in institutions described the influence of institutional culture on the severity of impacts as follows:

Institutional contexts of privacy, power and control create a climate conducive to abuse that is more severe, more likely to occur over longer periods of time and more likely to involve multiple offenders, all factors known to be associated with adverse outcomes for victims/survivors.¹²³

In one study, victims of abuse who had lived in 'closed' or 'total institutions' reported that the sense of powerlessness, helplessness and betrayal they experienced there may have exacerbated the impacts of the abuse. ¹²⁴ These highly controlled, residential institutions were often large, densely populated and physically, socially and culturally isolated from the broader community. ¹²⁵ In effect, they were closed to the outside world. They typically exhibited hierarchical and authoritarian features, with formalised, strict rules and procedures. ¹²⁶ Children who were mandatorily kept in these institutions described feeling powerless to escape or believing that whatever happened at the institution was 'normal', contributing to the harm they experienced. ¹²⁷ For example, 'Cassandra' told us she was removed from her mother when she was about one year old and placed in a Western Australian Aboriginal mission. ¹²⁸ She told us that the missionaries at the mission flogged and beat the children so regularly that 'to us it was just like normal'. The normalisation of violence had a lifelong impact on 'Cassandra' and many other children we heard from. 'Cassandra' told us that when she left the mission and was sexually assaulted, 'I didn't know it was a reportable offence... And when it happened again I just thought it was normal'. ¹²⁹

In another example, we heard that victims of child sexual abuse in institutions that were seen as having clear moral authority – such as those run by churches, or prestigious schools – felt powerless and ashamed. In the absence of any alternative understanding, abuse in these contexts also left victims feeling responsible in some way for that abuse, ¹³⁰ and made disclosure more difficult. ¹³¹ For some, these institutions also held significant meaning and value for their families in terms of their educative, religious, spiritual, reputational, moral and guardianship authority. Commissioned research suggests that these factors can influence the victim's experience of abuse and subsequent disclosure, as well as the impacts on the families of the victims. ¹³²

Type of institution

During our inquiry we heard from victims of child sexual abuse and their families across a wide variety of institutional types, including residential institutions, schools, sport and recreation institutions, religious institutions and detention environments (see Volume 2, *Nature and cause*). Across all institutional types we heard similar stories from survivors and families about the wide-ranging and potentially profound impacts of the abuse on their lives.

However, it is clear that some of the effects of child sexual abuse on victims are closely related to the type of institution more broadly, including its function or the activity it provided. For example, a child who has been abused in a place of worship may experience discernible spiritual impacts, in a way not experienced by a child abused in another context.

In Volumes 12–16, we examine some of the most common impacts related to specific institutions.

In Volume 12, Contemporary out-of-home care, we discuss how victims of child sexual abuse in out-of-home care are likely to have already experienced severe abuse and neglect leading to their removal from families, and that the impact of compounding experiences of abuse may result in complex trauma and cumulative harm. We heard that victims can experience feelings of betrayal and loss of trust when abused in this context. The sexual abuse can also exacerbate placement instability, as children may be removed from a placement as a result of disclosure or because carers are unable to manage the ways in which children express complex trauma. The experiences of sexual abuse, and a poor institutional response to that abuse, can compound other adverse experiences in childhood, setting some children on a pathway to drug and alcohol abuse, homelessness and criminal behaviour. We also heard that experiences of abuse and placement in care can have intergenerational effects.

In Volume 13, *Schools*, we examine educational impacts of child sexual abuse in schools. We heard that when a child is abused in a school context, they commonly experience unhappiness at school, learning difficulties and a decline in academic performance. Victims may avoid school altogether, or leave school early as a result. They may also be reluctant to engage with education later in life due their experiences of abuse in this context.

In Volume 14, Sport, recreation, arts, culture, community and hobby groups, we examine some of the impacts on victims who were sexually abused in sport and recreation organisations. We heard that as a result of the abuse some victims become disengaged from the sport or recreation activity that they once enjoyed, and can be isolated from the sport and recreation community. The abuse can also affect their mental and emotional health and interpersonal relationships. In addition, families, carers and others connected to the victim, and the sport and recreational organisation, can be affected.

In Volume 15, Contemporary detention environments, we discuss some of the impacts of child sexual abuse in youth detention and immigration detention. Victims of sexual abuse in youth detention often experience the impacts of abuse as cumulative harm, many having experienced other maltreatment and neglect before their placement in detention. Victims' experiences of sexual abuse in youth detention can play a role in their subsequent criminal behaviour, and can also lead to significant distrust and hatred of institutions and authority.

Victims of child sexual abuse in immigration detention face challenges related to the refugee experience and the conditions of immigration detention. They are vulnerable to cumulative harm, due to their experiences of multiple adversities before, and throughout, their refugee journey. The impacts of sexual abuse in the immigration detention context may be compounded by the continuing detention of victims and their families following the abuse, particularly where it is indefinite or prolonged or involves poor conditions and treatment. Victims also may have limited access to therapeutic treatment to aid in recovery following abuse. Children and their families may fear and distrust institutions and authority as a result of their experiences. Once released from immigration detention, victims can continue to experience mental health and other impacts of the abuse if they are not adequately addressed.

In Volume 16, *Religious institutions* we examine some of the impacts on victims and their families particular to sexual abuse in religious institutions. These include a loss of faith, spiritual confusion, ostracism from the religious community, family breakdown and a loss of trust in authority.

Some of these effects are also discussed in Chapters 3, 4 and 5.

Institutional responses

If an institution responds inappropriately or negatively to child sexual abuse, it can cause or exacerbate adverse effects for victims, resulting in a range of impacts, including:

- feelings of being betrayed by the institution
- a continuation of the sexual abuse
- re-traumatisation
- fear, distrust of, and contempt for, institutions, which may leave victims feeling less able to access support services provided by other institutions (for example, health services)
- ostracism, including of their families, particularly when the institution occupies a central place in the community.

When an institution ignores allegations of abuse, it can also put other children at risk.

The complicity of some institutions in creating situations and settings where abuse can occur, and in concealing abuse or failing to deal appropriately with disclosures, can lead to a sense of institutional betrayal for victims. This has been associated with increased anxiety, trauma symptoms and dissociation.¹³³ Victims may also suffer additional impacts that are specific to the institution involved. For instance, victims told us about having a crisis of faith or abandoning their religion after church authorities responded negatively.

Conversely, an appropriate response, such as believing a victim and taking appropriate action, can help stop abuse, minimise further impacts for the victim and promote healing. The impacts of institutional responses are discussed in more detail in Chapter 4.

2.4.4 Social and historical contexts

The social and historical contexts in which child sexual abuse occurs can exert an important influence on the way victims are affected. Community attitudes, and the policies and laws they reflect, can influence the effects on victims. They can shape the settings in which abuse occurs, the way that victims are abused, whether victims are able to disclose, and institutional and community responses to the abuse. These community attitudes include attitudes towards children, social awareness of child sexual abuse, the extent to which an institution is perceived to be a source of authority in a community, gender stereotypes, racism and homophobia or other discriminatory community attitudes. These attitudes, and the policies and laws they have influenced, have changed over time.

Children placed in Australia's residential institutions – including orphanages, missions, children's homes, and boarding, reformatory and industrial schools – in the decades up to 1990 were often marginalised and among society's most powerless groups. Research suggests that society stigmatised these children, viewing them as inferior and in need of institutional care or intervention. This, together with their isolation, contributed to their vulnerability to sexual abuse, and also made it easier for officials to discount, disbelieve or deny disclosures of their sexual or other abuse. As discussed below, this has implications for whether or not the abuse continued, as well as whether a victim received support to cope with its impacts.

Gender stereotypes can also influence the experience of, and reaction and response to, child sexual abuse. Research suggests that society's general disbelief that males could be victims of sexual abuse has influenced the meaning that men attribute to the abuse, their likelihood of disclosing it and the availability and access of appropriate supports. As discussed in Volume 4, *Identifying and disclosing child sexual abuse*, men may be reluctant to disclose abuse out of fear of being judged in accordance with gendered stereotypes associated with masculinity, homophobia or fears that they will become perpetrators. For female victims, attitudes to female sexuality and virginity, including patriarchal attitudes, can have a stigmatising effect and make disclosure difficult. These types of gendered responses can impact on a victim's identity and trajectory throughout their lives. For example, male victims may suffer impacts on their masculine identity and, especially for male victims of male perpetrators, confusion about their sexual identity. Female victims may experience disempowerment and re-victimisation later in life.

There are particular impacts for Aboriginal and Torres Strait Islander children who were abused in missions and related institutions that were administered under racially discriminatory 'protection' laws and assimilation policies. Under these laws, in place until the late 1960s, 140 children were forcibly removed from their families to prevent them from learning their own culture. 141 In 1997, the *Bringing them home* report found that these 'Institutionalised Indigenous children faced a hazard over and above that experienced by institutionalised non-Indigenous children. This was the continual denigration of their Aboriginality and that of their families'. 142 The report noted the daily experience of racism as part of life for Aboriginal and Torres Strait Islander peoples. Research suggests that for Aboriginal and Torres Strait Islander peoples the 'context of the historical legacy of colonisation' is inextricably entwined with the historical and contemporary experiences of child sexual abuse in institutions. 143

Community attitudes towards people with disability have also influenced the way that survivors with disability are affected by child sexual abuse. This is discussed in Section 2.4.5.

Volume 2, *Nature and cause* describes further the changing social and historical context of child sexual abuse. Volume 5, *Private sessions* describes what we heard in private sessions about the role of community attitudes in shaping the responses of institutions and families to child sexual abuse.

2.4.5 Victims' characteristics, circumstances and experiences

While all victims of child sexual abuse may be vulnerable to profound and broad-ranging impacts, research suggests that certain victim-related factors may influence the extent and type of impacts experienced. At Rather than any single factor having an influence, these victim-related factors are likely to interact with other factors to influence the outcomes for each individual.

Age and developmental stage at the time of abuse

The age and developmental stage of a child at the time of sexual abuse may influence the way in which victims experience the effects of abuse. As discussed in Volume 2, *Nature and cause*, we heard from survivors who were abused at a variety of ages and stages of development. The abuse, and the impacts of that abuse, took place in a context specific to their psychological, emotional and physical development.

Research is inconclusive on whether a child's age or developmental stage at the time they are abused is related to the level of emotional disturbance they experience. Some research associates poorer long-term outcomes for victims when child sexual abuse begins at an early age and lasts through more than one developmental stage. On the other hand, another study suggests that perpetrators may use more force with adolescent victims than with younger

children, which may be associated with long-term psychological distress.¹⁴⁷ Research suggests that while there is some indication of a link between a child's age at the time of abuse and the extent of their emotional and behavioural disturbances, this factor needs to be separated from the duration of the abuse, which may also be linked to greater psychological distress.¹⁴⁸

In the *Nature, cause and impact of child sexual abuse* case study, Dr Bruce Perry told us that 'in general, the younger you are and the more pervasive the abusive experience, the more the impact is on the physiology of the body and the brain'. However, Dr Perry warned that some of these changes are not necessarily seen until a child gets older, which can lead to a set of confounding issues. 150

Research in this area is in its early stages.¹⁵¹ However, some research suggests that certain types of mental health problems may be more likely to develop when abuse occurs at particular developmental stages. For example, higher dissociative symptoms and greater levels of amnesia may be associated with sexual abuse at an earlier age.¹⁵²

We heard in private sessions that victims of sexual abuse experience feelings of confusion and guilt when the sexual abuse extends from adolescence into their adult years, and this may impact on their ability to disclose. In *Case Study 8: Mr John Ellis's experience of the Towards Healing process and civil litigation,* Mr John Ellis gave evidence that he was sexually assaulted by Father Aidan Duggan from the age of 13, and that it continued into his early adult years. He told us that following the breakdown of his first marriage, he attended a peer ministry for divorced, separated and widowed persons. It was then that he first recognised Father Duggan's conduct had been abusive and wrong:

the realisation dawned on me for the first time that what Father Duggan had done was actually abuse. I had appreciated up until then only that I had felt ashamed, embarrassed and uncomfortable about the sexual contact with Father Duggan. However, that was because I had thought that my behaviour was wrong or abnormal and that I was homosexual or bisexual. I had also recognised the sexual contact as something which I had kept secret.¹⁵⁴

Until recognising Father Duggan's conduct as abusive, Mr Ellis had thought that, 'I had been a willing participant in the sexual interactions and that other people wouldn't understand what had happened'. ¹⁵⁵

We heard in private sessions from other survivors who had been sexually abused as adolescents, and into their adult years, about the difficulties of disclosing because they feared being judged or disbelieved, and because of their feelings of self-blame and shame. 'Cindy' told us she first met 'Rex Slater' when she was 14, when he became her solicitor after she badly injured her foot in an accident. ¹⁵⁶ A friendship developed over the years, which 'Cindy' said was well beyond the bounds of a solicitor/client relationship. She told us that he sexually abused her when she was 17 and that she continued a sexual relationship with him for five years. She said, 'I think I felt indebted for

all he'd done for me and feared losing his friendship which had become crucial for me over the traumatic years ... As if I would be invisible without him'. She ended the relationship when she met her future husband at university, not telling him about her history with the perpetrator until many years later. When she was 30 she studied child sexual abuse, and only then understood what had happened to her. She became angry, depressed and agoraphobic, hiding from her friends. Eventually 'Cindy' began seeing a counsellor. 'This abuse by 'Slater' was still so taboo for me I couldn't tell a counsellor at that stage because I felt so culpable, because I felt I was at an age where I should've known what was happening. I felt it was my fault'. 157

Gender

Overall, research confirms that both male and female victims of child sexual abuse suffer adverse long-term impacts.¹⁵⁸ In private sessions, many of the most common long-term impacts on mental health and relationships, such as depression, anxiety and difficulties with trust and intimacy, were described in similar numbers by male and female victims.

While these similarities exist, male and female victims can be affected in different ways, and gender can play a role in how they are abused. Some research on the sexual abuse of children across all settings shows that the sexual abuse of boys may involve more physical violence and physical harm than the sexual abuse of girls. ¹⁵⁹ In addition, the research suggests that adolescent boys may be more likely than adolescent girls to be abused by multiple perpetrators. ¹⁶⁰ Other research on child sexual abuse in the Anglican Church suggests that ongoing abuse may be more common for male victims than for female victims. ¹⁶¹ These features of violence, duration and multiple perpetrators have been linked to more negative mental health outcomes. ¹⁶²

Research also suggests that there are some gender-specific impacts. For example, female victims may experience gynaecological problems and altered child-bearing patterns. For male victims, a qualitative study suggests that child sexual abuse can have powerful effects on their experience of masculinity. It suggests that the social pressure to be 'appropriately masculine' — stoic, strong and dominant — is at odds with the pain, vulnerability and helplessness associated with child sexual abuse, for and as a result many male victims struggle with their masculinity. While child sexual abuse may leave many male victims feeling inadequate and 'unmasculine', others deny these feelings and adopt 'hyper-masculine' attributes, and some experience problems with anger and rage (see Chapter 3).

Other impacts, while not exclusive to either female or male victims, may be reported in higher numbers by one gender, and manifest in specific gendered ways. For example in private sessions, more male survivors than female survivors described impacts such as alcohol abuse, aggression, criminality and confusion about sexual identity. These effects are discussed in more detail in Chapter 3.

Further, gender roles and social expectations may influence how male and female victims disclose abuse, and deal with its aftermath. Social attitudes relating to masculinity and homophobia, attitudes to female sexuality and virginity, and the shame of male victims abused by females make it more difficult for victims to disclose (see Volume 4, *Identifying and disclosing child sexual abuse*). Research indicates that male victims generally take longer to disclose than female victims. ¹⁶⁶ In Volume 4, we discuss how males may be reluctant to disclose because of factors related to male socialisation, such as an over-emphasis on self-reliance, contempt for victims in general, contempt for homosexuals, ideas about sexual prowess, and a masculine obsession with heterosexuality and independence. ¹⁶⁷ Social views that perpetrators are exclusively male, or that sex of any kind with a female 'should be every man's dream' can make male victims of abuse by female perpetrators reluctant to disclose. ¹⁶⁸ As some research suggests, the act of disclosure is beneficial for the mental health of victims, ¹⁶⁹ and not disclosing may mean an increased risk of mental health impacts.

Some research also suggests that there may be differences in how male or female victims respond to the trauma throughout their life course. It suggests that female victims stereotypically become more entrenched in an internalised struggle with the effects of the abuse, 170 resulting in greater contact with mental health services. Men may suppress the effects through externalised behaviours, 171 such as violent outbursts.

Prior maltreatment and trauma

As described above, there is evidence of cumulative harm for victims of child abuse who have experienced prior maltreatment.¹⁷² Maltreatment may include neglect, physical, sexual and emotional abuse, such as exposure to domestic violence.¹⁷³ Research suggests that children's experiences before entering foster care and residential care may be associated with both greater risk of abuse in care and higher rates of PTSD and psychosocial difficulties in adult life.¹⁷⁴

Many survivors we heard from in private sessions and public hearings told us they had adverse childhood experiences before being sexually abused in an institution. This included sexual and physical abuse in the home, witnessing domestic violence, family breakdown and neglect. Some told us that experiences of domestic violence – either witnessing violence in their family or being assaulted by family members – left deep scars, which were compounded by the effects of sexual abuse in an institutional context. We heard from survivors who were placed in out-of-home care because of abuse and neglect at home. We were told that in care, where they should have been protected from further maltreatment, some children were further abused. For example, in *The Salvation Army boys' homes, Australia Eastern Territory* case study, survivors told us they had been placed in The Salvation Army homes as a result of abuse or neglect at home – because they and their siblings had been beaten by a parent, ¹⁷⁵ a parent had passed away, or their parents could not look after them. ¹⁷⁶ Many spent their childhood years in boys' homes and orphanages. ¹⁷⁷ We heard that these children, who were already vulnerable, were subject to physical and sexual abuse when they arrived at an institution that was intended to protect them. ¹⁷⁸

Some studies have suggested that perpetrators of extra-familial abuse may target children who have previously been victims of abuse and neglect in the belief that they are less likely to disclose the sexual abuse,¹⁷⁹ making these children vulnerable to cumulative harm. These victims felt that nobody would believe them.¹⁸⁰ Some felt too ashamed, seeing themselves as 'disgusting' or 'bad'.¹⁸¹ As one survivor told us in the *Parramatta Training School for Girls* case study, 'I was a State Ward and it seemed like anyone could do anything to me'.¹⁸²

Disability

Children with disability face significant and unique issues that may influence the way they are affected by child sexual abuse.

We have been told by advocacy groups there is a misconception in the community that children with disability, particularly cognitive impairment, will not be harmed by sexual abuse. ¹⁸³ The information available to us from research, private sessions and our inquiries indicates this is misguided, and could perpetuate the harm caused to children with disability who have been abused, and to their families.

While research into the impacts of child sexual abuse on people with disability is limited, it is clear that victims with intellectual disability experience a range of psychological symptoms similar to those reported by sexual abuse victims in the general population, including PTSD, depression, psychosis, dissociative symptoms, self-harm and alcohol abuse.¹⁸⁴

Survivors with disability and their parents told us in private sessions and public hearings about a range of profound impacts of child sexual abuse. For example, in *Case Study 41: Institutional responses to allegations of the sexual abuse of children with disability*, we heard about a boy living with moderate autism who used a QWERTY keyboard to communicate. His mother told us after he disclosed and reported child sexual abuse, he experienced impacts ranging from anxiety, fear of men and aggressive behaviour, PTSD, mistrust of police, feelings of marginalisation, loss of academic focus, and a continuing fear of retaliation by the perpetrator.¹⁸⁵

Research suggests there may also be particular, and more severe impacts, for children with disability, depending on a range of factors. One recent study found higher rates of penetrative abuse of children with intellectual disability who had been sexually abused and more occasions of ongoing (repeated) sexual abuse, than among other children who had been sexually abused. This is consistent with other studies which indicate that children with disability who are sexually abused can experience ongoing forms of 'contact' abuse (such as penetrative abuse or more violent sexual assault). As discussed above, penetrative abuse and repeated abuse have been associated in research with worse mental health outcomes for victims. The study also found higher rates of conduct disorders among child sexual abuse victims with intellectual disability than victims without intellectual disability. Volume 5, *Private sessions*, outlines what we heard in private sessions about the nature of the sexual abuse that survivors with disability experienced as children.

We explain in Volume 4, *Identifying and disclosing child sexual abuse*, how children with disability can face additional barriers to disclosure. We heard in private sessions that if the sexual abuse was not detected children lived with ongoing vulnerability to further abuse and therefore cumulative harm. For example, 'Jeanette', who was born deaf, was not taught sign language and had limited written English. She told us she was unable to communicate when she was being abused by one of the teachers at her school. With no means of telling her parents or six siblings, they couldn't understand why she would often be upset and refuse to go to school.¹⁸⁹ She told us that she continued to be abused.

In a submission to *Issues paper 9: Addressing the risk of child sexual abuse in primary and secondary schools*, advocacy organisation Children with Disability Australia told us that the expression of distress in children and young people with high communication needs is often not recognised: 'Certain behaviour, such as repeated head banging or nail biting, may indicate distress but is often misattributed to disability, meaning the cause of distress is not identified'.¹⁹⁰

Another stakeholder organisation, People with Disability Australia, told us that trauma can be compounded for children with disability if behavioural disclosures or symptoms of trauma are interpreted as 'challenging behaviours'. They said the response to this can be to use 'restrictive practices' such as solitary confinement, seclusion, forced sedation, chemical restraint or physical restraint, and that these actions may make a child more vulnerable and more traumatised. The Disability Reform Council has approved a national framework to reduce restrictive practices, given they infringe basic human rights. He heard that more work is needed to identify best practice responses to behaviours that a child or young person with disability may present, due to the trauma they may have suffered.

Research also suggests that negative community attitudes that promote stigma about, and discrimination towards, children with disability can play a role in how they are affected by sexual abuse. This is because such attitudes may lead adults to be dismissive or disinterested and fail to respond effectively.¹⁹⁴

Survivors also talked to us about other circumstances, such as family relationships and a family's loyalty to an institution, which can affect a child's decision whether to disclose the abuse or the ability of parents to recognise a child's unusual behaviour (see Volume 5, *Private sessions*). Volume 4, *Identifying and disclosing child sexual abuse*, discusses further barriers to disclosing and identifying child sexual abuse for people with disability.

Experiences of disclosing abuse

The impacts of child sexual abuse on victims may also be affected by their experiences of disclosure in both childhood and adulthood – including by the reaction of parents, friends and community members. Research on whether the impacts of disclosure itself – as distinct from the impacts of the abuse – can be positive or negative is inconclusive. However, it suggests disclosure is likely to play a key role in either helping a victim heal and recover, or compounding the adverse effects of abuse. He adverse effects of abuse.

In Volume 4, *Identifying and disclosing child sexual abuse*, we discuss how disclosure can be a critical step in beginning the process of recovery from the trauma of sexual abuse. Some research suggests that the process of disclosing abuse may improve the mental health of victims, ¹⁹⁷ possibly mitigating adverse impacts. It can also help to ensure the child is safe from further abuse. Many survivors described in private sessions the powerful effects of disclosing the abuse, especially on their mental health. For some, disclosing to a Commissioner in a private session was their first experience of being heard and believed. As one survivor in a private session said, 'In a way it's a good thing, just talking to you ... It's just like someone has lifted a brick off my shoulders'. ¹⁹⁸ For some victims the need to disclose was so profound that their final wish was to disclose before dying. Commissioners have held private sessions at hospital bedsides and by telephone for aged victims in palliative care.

The consequences of delaying disclosure have also been highlighted. Some studies indicate that such a delay, or non-disclosure, may exacerbate the impacts of child sexual abuse. ¹⁹⁹ A 2013 study on the experiences of male survivors of child sexual abuse in the United States, of whom 62 per cent were sexually abused by a clergy member, found that those who waited longer before telling someone about the sexual abuse were more likely to have experienced mental distress including symptoms of depression, anxiety, somatisation (physical symptoms) and suicidality. ²⁰⁰

However, the impacts of disclosure on a victim depend greatly on the response of the person to whom they are disclosing. Evidence suggests that dismissive, disbelieving, hostile, non-protective or non-supportive responses from others can increase the risk of negative outcomes. Popular Negative responses may in turn deter further disclosure and result in feelings of isolation and distress. Poor responses by others to disclosure may further traumatise victims, leading to 'secondary wounding'. Research conducted for the Royal Commission showed that the response of parents to a disclosure of abuse can play an especially important role in how the victim experiences the impacts of that abuse.

We heard from survivors who had disclosed, including as children and as adults, and who described their experience as distressing because of a poor response they received. Victims' experiences of disclosure are discussed in more detail in Volume 4, *Identifying and disclosing child sexual abuse*. The impacts of poor institutional responses are discussed in Chapter 4.

2.4.6 Sources of strength and resilience

The sources of strength and resilience that survivors draw on over the course of their lives can play a key role in how they are able to cope with, and manage the effects of, the abuse. A survivor's ability to draw on these sources of strength and resilience can be limited by the extent of their trauma; the intersections of economic disadvantage, racism and discrimination; cultural barriers; disability; and community resources. Still, in private sessions, most survivors described to us sources of strength – whether personal or environmental – they had drawn on at various stages of their lives.

Research increasingly points to the role of resilience in explaining why some victims display clinical symptoms of child sexual abuse at certain times and others do not.²⁰⁶ While definitions of resilience have changed over time, and continue to be contested,²⁰⁷ resilience can generally be understood as the concept of children displaying 'adaptive or competent functioning despite exposure to high levels of risk or adversity'.²⁰⁸ Recent conceptions of the term emphasise that resilience:

- is not static, but may wax and wane throughout an individual's life course²⁰⁹
- is not all-encompassing; some children may display more resilience in some aspects of their lives and less in others²¹⁰
- may be a more common response to adversity than was once considered.²¹¹

The concept of resilience as a process, rather than an end point, resonates with what survivors and their families and friends told us in private sessions and public hearings. Commissioners heard many stories of survivor resilience, strength and achievements in the face of adversity. However, these stories were not confined to those who consider themselves unharmed by the sexual abuse. We also heard from survivors who have been profoundly affected by the abuse at certain times, but who nevertheless have demonstrated resilience at other times. Further, some have suffered profound effects in certain areas of life, but exhibit resilience in others. We heard from many survivors that they have drawn on a wide variety of internal and external sources of strength to manage the impacts of the abuse. Some of the most common sources of strength are discussed in more detail in Volume 5, *Private sessions*.

In research, there is some evidence on the specific factors that support resilience for victims of child sexual abuse, often termed 'protective factors'. Reviews of literature on child sexual abuse show that a 'triad of factors' related to an individual's disposition, family support and external support systems plays a role in supporting resilience in victims. These factors are likely to work in combination, rather than in isolation. According to these reviews, the factors with the best research support were those associated with family and other relationships; social support from the wider environment; education; and a range of inner resources, such as optimism and hope, active coping strategies and externalisation of blame. Other factors include self-esteem, religion and spirituality, leisure and cultural activities and individual employment and socio-economic status. For a range of social and historical reasons, some children

have less access to these protective factors than others, having fewer social, economic and educational resources.

The following section describes some of these protective factors, illustrated by what we heard from survivors in private sessions and from survivors, their families and experts in public hearings.

Strong relationships and social support

Research on resilience factors for victims of child sexual abuse clearly shows that social support from various sources, including family, partners and peers, can be a buffer against the adverse effects of the abuse. This has particularly important ramifications for children in out-of-home care and others who are disconnected from these supports. The sources of social support may differ according to the victim's developmental stage. Findings indicate that younger children who are the victims of child sexual abuse draw largely on the support of their parents or caregivers, whereas adolescents depend on their friends and peers, and adult survivors on their partners. The sources of social support may differ according to the victim's developmental stage.

In the *Nature*, cause and impact of child sexual abuse case study, Dr Graham Gee, a psychologist from the Victorian Aboriginal Health Service, told us that his research has identified that strong relationships and strong social connections are key factors that support resilience:

One of the key factors that enable people to resist the impact of child sexual abuse, for example, is if they've been fortunate enough, or for whatever reason, to maintain strong relationships and strong social connections, despite their experiences. That social support and having at least one or two really strong, trusting relationships is one of the biggest factors in helping someone to be resilient to trauma ...²¹⁸

Overall, family members – parents, grandparents, partners, siblings and children – were identified as the main sources of social support by survivors in private sessions. Survivors said they drew on supportive relationships to help them cope, even if the supporters did not know about the abuse or only found out later.²¹⁹ In public hearings, many survivors told us about the importance of family support in their lives. In the *St Joseph's Orphanage, Neerkol* case study, AYB told us:

Through all the hurt and pain of my life I have been accompanied by my husband and family and the personal support of beautiful friends who were able to hear my pain and shame, yet, they have continued to love me. I owe them my life.²²⁰

Many survivors told us that the support of parents in particular helped them to manage the impacts of abuse, in both childhood and adulthood. Growing up in a stable family has been identified as a protective factor for adolescents and adults.²²¹ Secure attachment to family, such as a sense of acceptance and belonging, has also been positively correlated with resilience in some studies.²²²

Adult survivors spoke frequently about the emotional and psychological support of their partners. In addition to having supportive partners, the satisfaction with and quality of a partner relationship play a role in positive outcomes for adult survivors of child sexual abuse. Sometimes these supportive relationships have saved lives. Stew', for example, described his wife as 'the strongest thing in my life', and 'Rodney Adam' said in a private session, 'I can truly say my wife saved my life'. Nessa' told us, 'My husband always has a glass half full attitude. When I first met him, I didn't have a glass. Now I've got a glass'.

In Case Study 43: The response of Catholic Church authorities in the Maitland – Newcastle region to allegations of child sexual abuse by clergy and religious (Catholic Church authorities in Maitland – Newcastle), survivor Mr Gerard McDonald told us:

My wife Sharon is my rock. She gets frustrated because I can't sit down and talk to her about it. I want her to know that I wouldn't be here without her today. Even though I can't talk directly to her about what happened, her love and understanding has kept me alive. It's for her and my son Ben that I keep going.²²⁷

A number of adult survivors told us that they drew strength from their role as parents or carers. Research suggests that for some survivors having children is associated with positive outcomes in dealing with the impact of child sexual abuse.²²⁸ Survivors told us that becoming a parent made them want to stay well so they could look after and protect their children and make their children's lives better than their own. 'May' echoed what many survivors told us, 'My kids are the reason I put one foot in front of the other each day'.²²⁹

Survivors similarly described becoming a grandparent, foster carer or a carer of an elderly parent as a source of strength that helped them to manage the impacts of abuse.

'Luisa': becoming a parent and looking after others has helped her to cope

We heard that 'Luisa's' mother caught her husband abusing 'Luisa' when she was six years old. 'Luisa' told us that she was abused again later by her male teacher.

'I can remember just being difficult and it stayed with me for a very long time', she said. 'Through my high school years I was wagging school, smoking. Wasn't interested in education at all. And any authority figure, I didn't want to listen to'. As she grew older, 'Luisa' began to self-medicate with drugs and alcohol. Relationships with men failed.

She told us that she reached a turning point with the birth of her second child. 'I was 25 years old and I realised I couldn't be living on the edge, with drug abuse, alcohol, everything. It was just to blank everything out', she recalled. Even more changed for 'Luisa' with the birth of her third child. 'That's when I realised I wanted a future for them. I had to get my shit together'.

'Luisa' went on to build a successful career in management in the public sector. Her children have grown into healthy adults and have left home, which she found hard. She told us that she is happier now her mother is moving in. 'I know that's ridiculous, but again I have that drive, that focus, looking after someone', she said.

Having shared her story, 'Luisa' said that she would like to put her abuse behind her:

It's so easy to be consumed by everything and blame everyone else. At the end of the day we have a choice. You either choose for it to consume you or you choose to have a better life. My children were my driving point and my focus. I wanted to give them a better life than what I had.²³⁰

We also heard of the importance of strong relationships outside of the family. This was especially so for the many survivors we heard from who had been separated from their family and placed in institutions, or had experiences of family conflict and breakdown. As a result, from an early age, they could not turn to family members for support.

Survivors told us in private sessions about the support they received from friends and others in their wider social networks, including teachers, co-workers, foster carers and neighbours. For example, one survivor told us about a staff member at the institution where she was abused who was kind to her and maintained contact with her after she left.²³¹

In submissions to our issues papers, we also heard a lot about the value of peers – both child and adult peers – in providing support to victims²³² (see Volume 9, *Advocacy, support and therapeutic treatment services*). For adolescents, peer support can be a particularly important source of strength.

Pets were also important for helping some victims. 'Maeve' said that, after a bad day:

'Minty' got up on the bed, over to me, a paw on my shoulder and his chin on my cheek and we stayed there for hours. And after that I didn't have the heart not to let him on the bed. He was it – he stopped all the fellas. The thing is, if I was such a bad person, was so evil, why would a dog be like that with me?²³³

Therapeutic activities

Survivors frequently cited counselling and other therapeutic activities when asked what had helped them to cope with and manage the impacts of abuse. Therapeutic activities can be a key mechanism through which victims recognise and draw on their strengths and resilience. The role that therapeutic treatment services, and advocacy and support services, can play in helping victims to cope with the impacts of abuse is discussed in Volume 9, *Advocacy*, *support and therapeutic treatment services*.

Many survivors in private sessions said counselling in particular had helped them.²³⁴ These survivors found counselling had helped them to process their experiences, find some degree of closure, and make them feel stronger. Some people said it helped them to manage relationships or learn coping strategies.²³⁵ 'Gladys' said that getting counselling was one of the best decisions she has ever made. 'With my counsellor's help I'm finally feeling strong enough to stand up and be heard', she said.²³⁶

We also heard that therapy has helped survivors to make meaning of the abuse, by giving them an understanding of the dynamics of abuse and why it happened, and the impacts of trauma on development. For example, 'Marjorie' told us she was sexually abused by a teacher at a Pentecostal school at the age of six. 'Marjorie' started working with a specialised trauma therapist one year before her private session. She told us, 'He was the first person to say, "you've got early childhood trauma". Everything that I tell him, he's like, "that's part of what you've been through". He's really articulated all the things that I've never been able to actually understand about myself'. She said she is now learning to cope with the things that have happened to her: 'I never grew healthy structures of how to deal with things, because it was stunted. It's like I'm reprogramming myself, how to respond healthily'.²³⁷

Doctors, social welfare workers and therapeutic groups have also helped victims to cope, survive and heal. 'Shevonne' told us in private sessions:

I am pretty much where I am now [because of my social worker]. She never gave up on me. She's the only person who ... through all my tantrums and drugs and toxic relationships, [she] never gave up, ever, ever, ever. [She was] my only constant.²³⁸

Meditation was another coping technique commonly mentioned by survivors. 'Shevonne' told us, 'I had quite intensive therapy for probably only six months, but I learned a lot. It wasn't just psychotherapy. I learned strategy. I learned mindfulness, I learned things like that'.²³⁹

Several older survivors spoke about the benefits to them of attending support groups made up of other survivors with similar experiences. As discussed in Volume 9, *Advocacy, support and therapeutic treatment services*, these can be a powerful tool to break isolation and assist recovery. For example, in the *Catholic Church authorities in Maitland – Newcastle* case study, Mr Peter Russ described to us the assistance he has received from Survivors & Mates Support Network. ²⁴⁰ He said, 'I have attended their eight-week course in Newcastle twice and cannot speak highly enough of their ongoing support and encouragement. They are a life-saving service to survivors'. ²⁴¹

In Aboriginal and Torres Strait Islander community consultations, we also heard that collective healing approaches were a source of strength and resilience.²⁴² For many of the survivors who participated in these consultations, an important aspect of healing was about restoring relationships in families and communities. Survivors emphasised that sexual and other forms of abuse happened in the context of 'protectionist' and assimilation policies which intentionally broke up these relationships. Further, institutions lied to many children forcibly removed from their families, telling them their families 'don't want you anymore'.²⁴³ We heard that as well as the disconnection from family and community, the stigma attached to being a victim of sexual abuse was a further trauma. In one consultation, the importance of collective healing approaches was described to us:

The strength that comes through collective healing is really about community healing ... its reconnecting with the communities ... So part of it is going back to those communities ... restoring our family structures and that's restoring the community.²⁴⁴

Collective healing approaches are discussed in more depth in Volume 9, *Advocacy, support* and therapeutic treatment services.

We also heard from some survivors that being able to disclose to an authoritative forum, such as the Royal Commission, has been a source of strength and healing. 'Dale' for example, still finds it difficult to talk about his experiences but told us he is committed to meeting the challenge. He told us:

I'm breaking my silence to an authority figure, the Royal Commission, and it's a long time coming and I really want to have a witness, and this is part of my empowerment, self-empowerment, self-healing process, because this thing hasn't gone away after 40 years and in fact it's imploding more and more as the years go on, despite having my own family now.²⁴⁵

In Aboriginal and Torres Strait Islander community consultations, survivors told us that sharing their stories was also an important part of their recovery, so the wider community and Australian population know what has happened and can acknowledge the extent of harm done.²⁴⁶

Education, work and leisure activities

Survivors told us in private sessions and public hearings that education, work and leisure activities have been sources of strength – distracting them from the trauma of abuse, boosting their self-esteem, giving them skills, and 'proving' the abusers and others wrong.

Research has identified that education is well established as a factor which supports resilience in victims of child sexual abuse.²⁴⁷ Various studies suggest that educational engagement, positive performance at school, and positive feelings towards school have contributed to resilience in both adolescents and adults.²⁴⁸

We heard that education played a key role for survivors who felt they had missed out on schooling because of the deficiencies of the institution they attended, or because they had disengaged from school as a result of the abuse. In a private session, 'Lucy' told us that she was abused by several priests on three separate occasions during the 1960s, and suffered physical and psychological abuse from the religious sisters at her Catholic boarding school. She had been 'kicked out' of that school and several other Catholic institutions when, at about age 15, she took matters into her own hands and enrolled herself in the local public high school. There she had a 'fantastic time'. 'Lucy' said she loved learning, describing it as her 'salvation'. She still loves to learn and spends her days working in the education field.²⁴⁹

Some survivors coped with the impacts of abuse as adults by throwing themselves into work or single-mindedly pursuing a successful career. Current paid employment and/or high socioeconomic status are suggested in some research to be protective against mental disorders. In the *Brisbane Grammar School and St Paul's School* case study, BQK, who gave evidence that he saw a school counsellor, Mr Lynch, every day for about one year and was sexually abused by him during those sessions, said that he has been able to 'channel the fear and anger caused by the abuse into a drive to succeed in the business I created'. Similarly, some survivors told us that they gained strength by succeeding in sports, working on the land, working to be self-sufficient and working to pay the rent or mortgage. A number of survivors told us they had dedicated themselves to helping others who suffered child abuse or faced other adversities by working and volunteering for non-government welfare organisations, in law or in social work.

However, we heard that in some cases, using work as an escape and coping strategy had led to adult survivors over-achieving, which affected their family lives and health. For example, 'Sebastian' said, 'I had pushed all this down into my subconscious for a long time. I dealt with life by either working or drinking. If I had a day off, I would be feeling guilty because I had a day off, I kid you not'.²⁵²

Engaging in leisure activities, such as sport or cultural activities, is also associated with resilience, helping to prevent mental disorders and boosting self-esteem.²⁵³ Survivors commonly told us in private sessions of their involvement in a variety of activities that made them feel good and helped them cope with the effects of abuse. Some activities involved nature, such as gardening or caring for animals, while others were physical, such as sport, martial arts or yoga. Creative pursuits, such as music or art, were also commonly cited as ways of coping with trauma. 'Jed' told us he took an occasional day off work for painting lessons. Eventually this helped him to give up drinking alcohol: 'I took on art and just found another way. I just didn't have a lot of time for [drinking] because I'd wasted all those years'.²⁵⁴

We also heard from some survivors that engaging in certain activities gave them a feeling of control and power. In a private session, 'Gabrielle' told us she became addicted to adrenaline as a coping mechanism: running, riding horses and driving cars as fast as she could. However, she told us it was a 'short-term fix'.²⁵⁵

As with educational activities and work, leisure activities can also give survivors of child sexual abuse a sense of gaining skill or mastery over something, which contributes to their personal agency and self-esteem. This can be especially important when a victim has experienced denigration by a perpetrator. For example, one survivor of sexual abuse, 'Ethan', told us he tried to channel his anger into sport, and for many years he was a champion in his chosen sport. He did very well for a while, and earned a good living from it.²⁵⁶

Spirituality

Some studies have found that spirituality or being part of a religious group are factors supporting the resilience of victims and survivors.²⁵⁷ This is reflected in what some survivors told us in private sessions. They spoke about how their faith, and spiritual and religious beliefs and practices, had helped them cope with the abuse.

'Roddick', who told us he was sexually abused by several brothers at a Catholic boys' home, described how his faith remained important to him:

I like reading the Bible. It's strengthened my faith. It's not [God's] fault, it's their fault. [God] gave us free will and we're free to choose what we will. That's helped me. I treat others the way I like to be treated. Religion is man's way to God but Christianity is God's way to man. I don't want to sound like a Bible basher, but that's kept me sane. 258

We also heard from 'Joni' in a private session, who told us she had kept the abuse buried deep for most of her life. After 40 years she dug it up again because she wanted to reclaim her faith. Abuse, she said, 'takes away something that your spirit so deeply needs. You can get by physically, mentally, logically; you can put all those parts of the puzzle together, but not that spiritual side'. 'Joni' approached the Anglican Church and ended up speaking to three female priests. She said the help that they have given her has been vital to her recovery.²⁵⁹

Cultural connection

Before colonisation, Aboriginal and Torres Strait Islander peoples successfully managed trauma and traumatic events that affected their communities.²⁶⁰ Aboriginal and Torres Strait Islander cultures have been dynamic, enabling Aboriginal and Torres Strait Islander peoples to survive despite the magnitude of collective trauma they face.²⁶¹ Research has found that connection to strong culture is associated with better emotional, social and physical health of Aboriginal and Torres Strait Islander peoples.²⁶²

In the *Nature, cause and impact of child sexual abuse* case study, we heard that strong cultural connections provided an important source of resilience for Aboriginal and Torres Strait Islander survivors of child sexual abuse.²⁶³ Giving evidence in that public hearing, Dr Gee from the Victorian Aboriginal Health Service told us:

With my mob of Stolen Generation members in particular, it has been the fact that, despite being institutionalised and experiencing abuse, they've somehow managed to maintain a connection to their cultural connections and to their family members and communities. That's a massive source of resilience that sometimes differentiates between those who have been really damaged and are on really long journeys of recovery.²⁶⁴

Aboriginal and Torres Strait Islander survivors told us in private sessions how cultural connection gave them strength. 'Donald Steven' told us about finding 'strength and wisdom' in his culture.²⁶⁵ In her private session, 'Brydie' told us how connecting with her culture and experiencing traditional Aboriginal healing in the bush with other older Aboriginal women and participating in 'women's sacred stuff' gave her strength and played a role in her healing.²⁶⁶ She thinks that younger Aboriginal women might benefit from this kind of healing too. 'They [the women] give you their energy, which is so kind of them ... And they also centre your energy, so that you can think straight'.²⁶⁷

'Evelyn': connecting with her Aboriginal culture

'Evelyn' told us that connecting with her Aboriginal culture has been a big part of her healing. She explained:

Today I am a sober woman of almost 30 years now but the healing journey took me to dealing with the issues relating to the low self-esteem, the feeling of being inferior to others, excessive and volatile anger that lead to behaviours that were of a violent nature, the tormented mind that led me to being suicidal at different times in my life.

'Evelyn' told us that she was recently able to stand on her father's land for the first time:

I experienced my own cultural spirituality from the land that belonged to my family/ kin group. The land held within the natural landscape something that was excitingly sacred ... I started to understand some of what Aboriginal spirituality meant and now I was standing in the soil where my father was born. It was where my life also started. I experienced for the first time the stillness of the early morning and was witness to the rising of the morning star. I was able to absorb the sacredness of what I was part of for all these years and which was denied for most my childhood and adult life. The trip to my homelands was awesome in its beauty. Today I hold these experiences sacred and holy. I had to redefine "who I was." I have come to accept myself as I am, just to be me with all my faults and to share a common humanity with others.²⁶⁸

Inner resources

Research has identified some internal factors associated with resilience in victims of child sexual abuse.²⁶⁹ These include:

- understanding and managing emotions, interpersonal competence and trust.
 One review explained that as sexual abuse is a traumatic event where trust is often shattered, re-establishing interpersonal and emotional competence afterwards may play a role in preventing the adverse effects of the abuse²⁷⁰
- optimism and hope. Studies show that qualities of optimism and expectancy about the future were predictors of resilient outcomes in both adolescents and adults²⁷¹
- active coping. Some studies suggest that active coping, such as seeking social support and problem-focused coping, is associated with greater resilience for children, adolescents and adults²⁷²
- external attribution of blame. Studies suggest that this can have a positive effect for victims of child sexual abuse because it reduces the perceptions of self-guilt and shame, which are often debilitating emotional impacts.²⁷³

Survivors in private sessions often identified personal qualities, including the above, which they felt helped them to cope with the impacts of abuse at various stages. For example, many people spoke of their own inherently optimistic qualities, which they harnessed to focus on the future. 'Lachlan' told us, 'I always describe my personality as being pragmatic but optimistic at the same time. And I don't like to hold regrets. You can't change the past, all you can do is influence the future'. Our commissioned analysis of private sessions found that taking a positive view of the future, while accepting the trauma of the past, was frequently mentioned by victims as a coping strategy. Yolume 5, *Private sessions* explores this in further detail.

Other survivors told us that they drew on the wisdom of others to move forward. 'Rainey' described how the lessons she learned from her father gave her the resilience and toughness she needed to deal with the abuse. 'Your life is governed by the choices you made yesterday', she said. 'The choices you make determine the pathway. Whether you go on a narrow road or a four-lane highway. But the choices are there. ... That's what I was taught to believe. Those are the things that have helped me'.²⁷⁶

Many survivors found that their determination and fighting spirit provided them with a way forward. They described personal qualities that helped them to cope, such as having a 'black sense of humour', having a fighting spirit, being determined and stubborn, helping others, and having a sense of justice.²⁷⁷ 'Stanley Peter' echoed the words of many survivors when he spoke of not wanting to 'let them beat me', and he told us he believed that the abusers would win if he took his own life.²⁷⁸

Sometimes survivors have channelled this fighting spirit and external attribution of blame into pursuing justice for themselves and for others who have been victims.²⁷⁹ In a private session, 'Ross William' told us, 'The more I've got rid of the guilt and the shame the more I've been able to be an advocate, be outspoken, be vocal about what happened'. Now 'Ross William' uses his skills and his strength to advocate for other survivors of child sexual abuse. 'I've got a voice and I have to use it. I'm obligated to use it for those who can't speak or aren't here to speak anymore'.²⁸⁰

Other survivors told us that working to help others heal has been a way to cope with their own abuse. In the *Satyananda Yoga Ashram* case study, Jyoti told us:²⁸¹

I eventually went back to study and completed an Associate Diploma in Social Science/Community Development, and later followed my passion to become an art therapist and counsellor to help women and children who have suffered from abuse, depression and anxiety. I loved doing that work and did it for 15 years ... I chose to work in that field because it was my way of coping with the injustice I had experienced.²⁸²

Endnotes

- J Cashmore & R Shackel, *The long-term effects of child sexual abuse,* Australian Institute of Family Studies, Melbourne, 2013, p 2.
- 2 Name changed, private session, 'William Peter'.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors:*A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017; I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017.
- 4 Transcript of B Perry, Case Study 57, 30 March 2017 at 27699:23–31.
- 5 Transcript of C Quadrio, Case Study 28, 25 May 2015 at C8494:2–9.
- 6 Name changed, private session, 'Allan Peter'.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 45–6.
- 8 Name changed, private session, 'Warwick John'.
- I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 67–8; DA Wolfe, KJ Francis & AL Straatman, 'Child abuse in religiously-affiliated institutions: Long-term impact on men's mental health', *Child Abuse & Neglect*, vol 30, no 2, 2006, p 207.
- DA Wolfe, KJ Francis & AL Straatman, 'Child abuse in religiously-affiliated institutions: Long-term impact on men's mental health', *Child Abuse & Neglect*, vol 30, no 2, 2006, p 207.
- M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 487. This literature review surveyed research which variously used child, adolescent and adult samples. The rates of resilience in child and adolescent samples ranged from 10 per cent to 53 per cent, and in adult samples, it ranged from 15 per cent to 47 per cent. The wide variation in resilience rates of victims of child sexual abuse, as well as in the rates seen following other types of child maltreatment, is considerably influenced by a number of factors related to conceptual and methodological issues.
- 12 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 21.
- 13 Transcript of M E Farrell-Hooker, Case Study 7, 27 February 2014 at 5021:13–19.
- 14 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions*, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 177, 198; T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings*, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 70.
- Name changed, private session, 'Rodney'.
- I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 171; GG Peacock & ML Holland, *Emotional and behavioral problems of young children: Effective interventions in the preschool and kindergarten years*, Guildford Press, New York, 2003, p 149.
- 17 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 171; GG Peacock & ML Holland, *Emotional and behavioral problems of young children: Effective interventions in the preschool and kindergarten years*, Guildford Press, New York, 2003, p 149.
- 18 Transcript of C Quadrio, Case Study 28, 25 May 2015 at C8456:3–11.
- 19 Name changed, private session, 'Penelope Sue'.
- 20 R Allagia, 'Many ways of telling: Expanding conceptualisations of child sexual abuse disclosure', *Child Abuse & Neglect*, vol 28, 2004, p 1220.
- For example: Name changed, private session, 'Colin David'; Name changed, private session, 'Earl'; Name changed, private session, 'Ronald Arthur'.
- Transcript of T A Quagliata, Case Study 39, 5 May 2016, 18584:10–29.
- Name changed, private session, 'Cheryl Valerie'.
- I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 174. For some people, publicity about the Royal Commission's focus was a trigger for memories of the sexual abuse. For others, triggers occurred in coming forward to tell their story. The Royal Commission provided counselling support and referral for those impacted by the Commission's work.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 74; DA Wolfe, PG Jaffe, AW Leschied & BL Legate, 'Assessing historical abuse allegations and damages', *Child Abuse & Neglect*, vol 34, no 3, 2010, p 142.
- 26 R Gordon, *The passage of trauma through life*, Australian Centre for the Study of Sexual Assault, Melbourne, 2010.
- L Bromfield & R Miller, *Cumulative harm: Best interests case practice model: Specialist practice resource,* Victorian Government Department of Human Services, Melbourne, 2012, p 22.

- Transcript of M Watson, Case Study 57, 30 March 2017 at 27701:40–42.
- 29 M Griffiths, Submission to the Royal Commission into Institutional Responses to Child Sexual Abuse, *Issues Paper No 10: Advocacy and Support and Therapeutic Treatment Services*, 2016.
- Name changed, private session, 'Carol Jane'.
- I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 155.
- 32 For example: Name changed, private session, 'Andy Trevor'; Name changed, private session, 'Cormick'.
- B Draper, JJ Pfaff, J Pirkis, J Snowdon, NT Lautenschlager, I Wilson & OP Almeida, 'Long-term effects of childhood abuse on the quality of life and health of older people: Results from the Depression and Early Prevention of Suicide in General Practice Project', *Journal of the American Geriatrics Society*, vol 56, no 2, 2008, p 262; CT Allers, KJ Benjack & NT Allers, 'Unresolved childhood sexual abuse: Are older adults affected?', *Journal of Counseling & Development*, vol 71, no 1, 1992. p 14.
- B Draper, JJ Pfaff, J Pirkis, J Snowdon, NT Lautenschlager, I Wilson & OP Almeida, 'Long-term effects of childhood abuse on the quality of life and health of older people: Results from the Depression and Early Prevention of Suicide in General Practice Project', *Journal of the American Geriatrics Society*, vol 56, no 2, 2008.
- 35 CT Allers, KJ Benjack & NT Allers, 'Unresolved childhood sexual abuse: Are older adults affected?', *Journal of Counseling & Development*, vol 71, no 1, 1992, p 14.
- 36 CT Allers, KJ Benjack & NT Allers, 'Unresolved childhood sexual abuse: Are older adults affected?', *Journal of Counseling & Development*, vol 71, no 1, 1992, p 14.
- Name changed, private session, 'Clementine'.
- Transcript of P Horsley, Case Study 57, 30 March 2017 at 27788:10–14.
- 39 Transcript of P Horsley, Case Study 57, 30 March 2017 at 27789:41–27790:1.
- Transcript of P Horsley, Case Study 57, 30 March 2017 at 27789:31–39.
- L Bromfield & R Miller, *Cumulative harm: Best interests case practice model: Specialist practice resource,* Victorian Government Department of Human Services, Melbourne, 2012, p 5.
- 42 L Bromfield & R Miller, *Cumulative harm: Best interests case practice model: Specialist practice resource,* Victorian Government Department of Human Services, Melbourne, 2012, p 5.
- RL Gaskill & BD Perry, 'Child sexual abuse, traumatic experiences, and their impact on the developing brain' in P Goodyear-Brown (ed), *Handbook of child sexual abuse: Identification, assessment, and treatment*, John Wiley & Sons, New Jersey, 2012, pp 33–37.
- L Bromfield & R Miller, Cumulative harm: Best interests case practice model: Specialist practice resource, Victorian Government Department of Human Services, Melbourne, 2012, p 7.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 19–25.
- S Silburn, S Zubrick, JD Maio, C Shepherd, J Griffin, F Mitrou, R Dalby, C Hayward & G Pearson, *The Western Australian Aboriginal child health survey: Strengthening the capacity of Aboriginal Children, families and communities*, Curtin University of Technology and Telethon Institute for Child Health, Perth, 2006, pp xvii, 595.
- Human Rights and Equal Opportunity Commission, *Bringing Them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, pp 121, 202; K Menzies & L McNamara, 'Addressing violence and oppression: Debates and challenges' in B Fawcett and F Waugh (eds), *Addressing violence, abuse and oppression,* Routledge, London, 2008, p 42; P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, *Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 22–24.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 22.
- 49 Z Sarnyai, M Berger & I Jawan, 'Allostatic load mediates the impact of stress and trauma on physical and mental health in Indigenous Australians', *Australasian Psychiatry*, vol 24, no 1, 2016, p 74.
- Human Rights and Equal Opportunity Commission, *Bringing Them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, p 214.
- A Ferdinand, Y Paradies & M Kelaher, *Mental health impacts of racial discrimination in Victorian Aboriginal communities: The localities embracing and accepting diversity (LEAD) experiences of racism survey,* The Lowitja Institute, Melbourne, 2012.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, *Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 33.
- I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions*, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 48; T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 82.

- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 75.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 75–6; K Kendall-Tackett, L Williams & D Finkelhor, 'The impact of sexual abuse on children: A review and synthesis of recent empirical studies', *Psychological Bulletin*, vol 113, no 1, 1993, p 171; B Lueger-Schuster, V Kantor, D Weindl, M Knefel, Y Moy, A Butollo, R Jagsch & T Gluck, 'Institutional abuse of children in the Austrian Catholic Church: Types of abuse and impact on adult survivors' current mental health', *Child Abuse & Neglect*, vol 38, no 1, 2014, p 10.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 75–6; D Finkelhor, 'Early and long-term effects of child sexual abuse: An update', *Professional Psychology: Research and Practice*, vol 21, no 2, 1990; MC Cutajar, PE Mullen, JRP Ogloff, SD Thomas, DL Wells & J Spataro, 'Psychopathology in a large cohort of sexually abused children followed up to 43 years', *Child Abuse & Neglect*, vol 34, no 11, 2010, p 820.
- B Lueger-Schuster, V Kantor, D Weindl, M Knefel, Y Moy, A Butollo, R Jagsch & T Gluck, 'Institutional abuse of children in the Austrian Catholic Church: Types of abuse and impact on adult survivors' current mental health', *Child Abuse & Neglect*, vol 38, no 1, 2014, p 10.
- D Finkelhor, 'Current information on the scope and nature of child sexual abuse', *The Future of Children*, vol 4, no 2, 1994, p 42.
- 59 For example: Name changed, private session, 'Mae'.
- B Lueger-Schuster, V Kantor, D Weindl, M Knefel, Y Moy, A Butollo, R Jagsch & T Gluck, 'Institutional abuse of children in the Austrian Catholic Church: Types of abuse and impact on adult survivors' current mental health', *Child Abuse & Neglect*, vol 38, no 1, 2014; J Steel, L Sanna, B Hammond, J Whippie & H Cross, 'Psychological sequelae of childhood sexual abuse: Abuse-related characteristics, coping strategies, and attributional style', *Child Abuse & Neglect*, vol 28, no 7, 2004, p 795.
- 61 Name changed, private session, 'Hallie'.
- S Roth, E Newman, D Pelcovitz, BA van der Kolk & FS Mandel, 'Complex PTSD in victims exposed to sexual and physical abuse: Results from the DSM-IV field trial of posttraumatic stress disorder', *Journal of Traumatic Stress*, vol 10, no 4, 1997, p 551; DJ Higgins & MP McCabe, 'Multiple forms of child abuse and neglect: Adult retrospective reports', *Aggression and Violent Behaviour*, vol 6, 2001, p 576.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No. 5: Response of the Salvation Army to child sexual abuse at its boys homes in New South Wales and Queensland*, Sydney, 2015, Finding 2, p 38.
- 64 Exhibit 40-0001, 'Statement of CJA', Case Study 40, STAT.0993.001.0001 R at 0003 R, 0004 R.
- 65 Transcript of 'CJA', Case Study 40, 21 June 2016 at 19282:13–16; Exhibit 40-0001, 'Statement of CJA', Case Study 40, STAT.0993.001.0001 R at 0002 R.
- 66 Transcript of 'CJA', Case Study 40, 21 June 2016 at 19284:19–29; Exhibit 40-0001, 'Statement of CJA', Case Study 40, STAT.0993.001.0001 R at 0004 R.
- 67 Transcript of 'CJA', Case Study 40, 21 June 2016 at 19297:8–10; Exhibit 40-0001, 'Statement of CJA', Case Study 40, STAT.0993.001.0001 R at 0014 R.
- AM McAlinden, 'Grooming' and the sexual abuse of children: Institutional, internet and familial dimensions, Oxford University Press, Oxford, 2012, p 24; P O'Leary, E Koh & A Dare, Grooming and child sexual abuse in institutional contexts, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 16, 23.
- 69 E Olafson, 'Child sexual abuse: Demography, impact, and interventions', *Journal of Child & Adolescent Trauma*, vol 4, no 1, 2011, p 13.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 76.
- 71 Transcript of A McAlinden, Case Study 57, 29 March 2017 at 27608:17–23.
- 72 Transcript of A McAlinden, Case Study 57, 29 March 2017 at 27608:32–41.
- 73 Name changed, private session, 'Sophia'.
- 74 Transcript of BQG, Case Study 34, 3 November 2016 at 12020:24–12023:39.
- 75 Transcript of BQG, Case Study 34, 3 November 2016 at 12023:28–29.
- 76 Transcript of BQG, Case Study 34, 3 November 2016 at 12023:36–39.
- 77 Exhibit 2-0004, 'Statement of AS', Case Study 2, STAT.0031.001.0001 at 0012_R.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 2: YMCA NSW's response to the conduct of Jonathan Lord*, Sydney, 2014, p 17.
- AM McAlinden, *'Grooming'* and the sexual abuse of children: Institutional, internet and familial dimensions, Oxford University Press, Oxford, 2012, p 24, citing Sutton and Jones 2004.
- 80 Transcript of BZR, Case Study 37, 7 March 2016 at 16794:21–22.
- 81 Transcript of BZR, Case Study 37, 7 March 2016 at 16810:6–19; Exhibit 37-0011, 'Statement of BZR', Case Study 37, STAT.0878.001.0001 R at 0024 R, 0025 R.
- 82 S Greijer & J Doek, *Terminology guidelines for the protection of children from sexual exploitation and sexual abuse,* ECPAT International, Luxembourg, 2016, p 23.

- 83 S Greijer & J Doek, Terminology guidelines for the protection of children from sexual exploitation and sexual abuse, ECPAT International, Luxembourg, 2016, pp 22–3.
- N Henry, A Powell & A Flynn, Not just 'revenge pornography': Australians' experiences of image-based abuse. A summary report, RMIT University, Melbourne, 2017, p 5.
- J Wolak & D Finkelhor, Sextortion: Findings from a survey of 1,631 victims, Crimes Against Children Research Center, University of New Hampshire, Durham, 2016, p 8. Respondents to the survey were 18–25 years old. Forty six per cent of respondents were 17 years or younger when the threats began, p 8.
- J Wolak & D Finkelhor, Sextortion: Findings from a survey of 1,631 Victims, Crimes Against Children Research Center, University of New Hampshire, Durham, 2016, p 31.
- J Wolak & D Finkelhor, Sextortion: Findings from a survey of 1,631 victims, Crimes Against Children Research Center, University of New Hampshire, Durham, 2016, pp 31–3.
- N Bluett-Boyd, B Fileborn, A Quadara & S Moore, *The role of emerging communication technologies in experiences of sexual violence: A new legal frontier?*, Australian Institute of Family Studies, Melbourne, 2013, p 36.
- 89 N Bluett-Boyd, B Fileborn, A Quadara & S Moore, *The role of emerging communication technologies in experiences of sexual violence: A new legal frontier?*, Australian Institute of Family Studies, Melbourne, 2013, p 36.
- 90 Names changed, private session, 'Sally Anne and Jodie'.
- 91 Names changed, private session, 'Sally Anne and Jodie'.
- 92 M Flood, 'The harms of pornography exposure among children and young people', Child Abuse Review, vol 18, 2009, p 384.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 74.
- 94 K Kendall-Tackett, L Williams & D Finkelhor, 'The impact of sexual abuse on children: A review and synthesis of recent empirical studies', *Psychological Bulletin*, vol 113, no 1, 1993, p 170; D Finkelhor, 'Current information on the scope and nature of child sexual abuse', *The Future of Children*, vol 4, no 2, 1994, p 46; J Cashmore & R Shackel, 'Gender differences in the context and consequences of child sexual abuse', *Current Issues in Criminal Justice*, vol 26, no 1, 2014, p 77.
- 95 CP Smith & JJ Freyd, 'Institutional betrayal', *American Psychologist*, vol 69, no 6, 2013; CP Smith & JJ Freyd, 'Dangerous safe havens: Institutional betrayal exacerbates sexual trauma', *Journal of Traumatic Stress*, vol 26, no 1, 2013, p 119.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 75.
- 97 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 21: The response of the Satyananda Yoga Ashram at Mangrove Mountain to allegations of child sexual abuse by the ashram's former spiritual leader in the 1970s and 1980s*, Sydney, 2016, p 54.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 75.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 75; D Palmer, *The role of organisational culture in child sexual abuse in institutional contexts,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 68.
- 100 Name changed, private session, 'Bridget'.
- 101 Transcript of Dr D Ranson, Case Study 50, 13 February 2017 at 25254:21–30.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 76–7.
- 103 Transcript of J K Isaacs, Case Study 4, 9 December 2013 at 2353:7–8.
- 104 Transcript of JA Kiernan, Case Study 26, 15 April 2015 at C7418:25–29.
- 105 Transcript of JA Kiernan, Case Study 26, 15 April 2015 at C7418:25–26.
- Transcript of Dr W O'Brien, Case Study 45, 20 October 2016 at 21658:14–16.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 30.
- 108 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions*, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 13–15.
- 109 Name changed, private session, 'Bree Amy'.
- 110 Transcript of CLG, Case Study 45, 25 October 2016 at 21982:17.
- 111 Transcript of CLG, Case Study 45, 25 October 2016 at 21982:14–26.
- 112 Transcript of CLG, Case Study 45, 25 October 2016 at 21984:44–21985:10.
- 113 Transcript of CLG, Case Study 45, 25 October 2016 at 21987:22–3.
- 114 Transcript of Dr W O'Brien, Case Study 45, 20 October 2016 at 21660:3–7.
- 115 Transcript of Dr W O'Brien, Case Study 45, 20 October 2016 at 21660:3–14.
- 116 Transcript of Dr W O'Brien, Case Study 45, 20 October 2016 at 21667:2–12.
- 117 Transcript of K Lumsdaine, Case Study 45, 20 October 2016 at 21713:21–2.

- For example: Name changed, private session, 'Barry Stephen'; Name changed, private session, 'Stuart Andrew'; Name changed, private session, 'Krispin'.
- 119 For example: Name changed, private session, 'Fleur'; Name changed, private session, 'Clark James'.
- T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 45, 73–4; J Breckenridge & G Flax, Service and support needs of specific population groups that have experienced child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 23. For a discussion of institutional culture, see D Palmer, The role of organisational culture in child sexual abuse in institutional contexts, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016.
- D Palmer, *The role of organisational culture in child sexual abuse in institutional contexts,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 38.
- See D Palmer, *The role of organisational culture in child sexual abuse in institutional contexts,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 70.
- DA Wolfe, KJ Francis & AL Straatman, 'Child abuse in religiously-affiliated institutions: Long-term impact on men's mental health', *Child Abuse & Neglect*, vol 30, no 2, 2006.
- D Palmer, *The role of organisational culture in child sexual abuse in institutional contexts,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 36–7.
- 1Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 16; D Palmer, *The role of organisational culture in child sexual abuse in institutional contexts,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 36, 46–7.
- J Breckenridge & G Flax, Service and support needs of specific population groups that have experienced child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 23.
- 128 Name changed, private session, 'Cassandra'.
- 129 Name changed, private session, 'Cassandra'.
- A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 13.
- J Breckenridge & G Flax, Service and support needs of specific population groups that have experienced child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 23.
- A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 67.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 74.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 71.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 71.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 75.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 80.
- 138 R Alaggia, 'Disclosing the trauma of child sexual abuse: A gender analysis', Journal of Loss and Trauma, vol 10, 2005, p 465.
- 139 R Alaggia, 'Disclosing the trauma of child sexual abuse: A gender analysis', Journal of Loss and Trauma, vol 10, 2005, p 464.
- Human Rights and Equal Opportunity Commission, *Bringing Them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, pp 34–5.
- Human Rights and Equal Opportunity Commission, *Bringing Them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, p 121; P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, *Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 9.
- Human Rights and Equal Opportunity Commission, *Bringing Them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, p 200.

- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017. See also J Breckenridge & G Flax, Service and support needs of specific population groups that have experienced child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 30.
- J Cashmore & R Shackel, *The long-term effects of child sexual abuse*, Australian Institute of Family Studies, Melbourne, 2013, p 5; T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/ survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 77–80.
- 145 CS Widom & AM Ames, 'Criminal consequences of childhood sexual victimization', *Child Abuse & Neglect*, vol 18, no 4, 1994, p 306.
- E Olafson, 'Child sexual abuse: Demography, impact, and interventions', *Journal of Child & Adolescent Trauma*, vol 4, no 1, 2011.
- J Steel, L Sanna, B Hammond, J Whippie & H Cross, 'Psychological sequelae of childhood sexual abuse: Abuse-related characteristics, coping strategies, and attributional style', *Child Abuse & Neglect*, vol 28, no 7, 2004, p 795.
- J Cashmore & R Shackel, 'Gender differences in the context and consequences of child sexual abuse', *Current Issues in Criminal Justice*, vol 26, no 1, 2014, p 78.
- 149 Transcript of B Perry, Case Study 57, 30 March 2017 at 27701:5–8.
- 150 Transcript of B Perry, Case Study 57, 30 March 2017 at 27700:45–27701:5.
- 151 Transcript of B Perry, Case Study 57, 30 March 2017 at 27700:39–42.
- JA Chu, LM Frey, BL Ganzel & JA Matthews, 'Memories of childhood abuse: Dissociation, amnesia, and corroboration', American Journal of Psychiatry, vol 156, no 5, 1999, p 753; JG Allen, Traumatic relationships and serious mental disorders, John Wiley & Sons, Chichester, 2002, p 163.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 8: Mr John Ellis's experience of the Towards Healing process and civil litigation*, Sydney, 2015, p 20.
- 154 Transcript of J A Ellis, Case Study 8, 10 March 2014 at 5323:37–45.
- 155 Transcript of J A Ellis, Case Study 8, 10 March 2014 at 5324:1–3.
- 156 Name changed, private session, 'Cindy'.
- 157 Name changed, private session, 'Cindy'.
- J Cashmore & R Shackel, 'Gender differences in the context and consequences of child sexual abuse', *Current Issues in Criminal Justice*, vol 26, no 1, 2014, p 83; SR Dube, RF Anda, CL Whitfield, DW Brown, VJ Felitti, M Dong & WH Giles, 'Long-term consequences of childhood sexual abuse by gender of victim', *American Journal of Preventive Medicine*, vol 28, no 5, 2005.
- J Cashmore & R Shackel, 'Gender differences in the context and consequences of child sexual abuse', *Current Issues in Criminal Justice*, vol 26, no 1, 2014, p 78.
- J Cashmore & R Shackel, 'Gender differences in the context and consequences of child sexual abuse', *Current Issues in Criminal Justice*, vol 26, no 1, 2014, p 78.
- P Parkinson, K Oates & A Jayakody, *Study of reported child sexual abuse in the Anglican Church,* Anglicare Australia, Canberra, 2009, p 5; Transcript of P Parkinson, Case Study 52, 17 March 2017 at 26640:32–45.
- J Cashmore & R Shackel, 'Gender differences in the context and consequences of child sexual abuse', *Current Issues in Criminal Justice*, vol 26, no 1, 2014 p 78.
- JL Davis & PA Petretic-Jackson, 'The impact of child sexual abuse on adult interpersonal functioning: A review and synthesis of the empirical literature', *Aggression and Violent Behavior*, vol 5, no 3, 2000, pp 311–2; ED Reissing, YM Binik, S Khalif, D Cohen & R Amsel, 'Etiological correlates of vaginismus: Sexual and physical abuse, sexual knowledge, sexual self-scheme, and relationship adjustment', *Journal of Sex & Marital Therapy*, vol 29, no 1, 2003.
- D Lisak, 'The psychological impact of sexual abuse: Content analysis of interviews with male survivors', *Journal of Traumatic Stress*, vol 7, no 4, 1994, p 537; J Cashmore & R Shackel, 'Gender differences in the context and consequences of child sexual abuse', *Current Issues in Criminal Justice*, vol 26, no 1, 2014, p 79.
- D Lisak, 'The psychological impact of sexual abuse: Content analysis of interviews with male survivors', *Journal of Traumatic Stress*, vol 7, no 4, 1994, p 537.
- J Cashmore & R Shackel, 'Gender differences in the context and consequences of child sexual abuse', Current Issues in Criminal Justice, vol 26, no 1, 2014, p 80; P Parkinson, K Oates & A Jayakody, Study of reported child sexual abuse in the Anglican Church, Anglicare Australia, Canberra, 2009, pp 31–2; G Priebe & CG Svedin, 'Child sexual abuse is largely hidden from the adult society: An epidemiological study of adolescents' disclosures', Child Abuse & Neglect, vol 32, no 12, 2008, p 1098; P O'Leary & J Barber, 'Gender differences in silencing following childhood sexual abuse', Journal of Child Sexual Abuse, vol 17, no 2, 2008; Queensland Crime Commission & Queensland Police Service, Project AXIS Child Sexual Abuse in Queensland: The Nature and Extent, Queensland Crime Commission and Queensland Police Service, Brisbane, 2000, p xii.
- 167 P O'Leary & J Barber, 'Gender differences in silencing following childhood sexual abuse', *Journal of Child Sexual Abuse*, vol 17, no 2, 2008.
- 168 G Priebe & CG Svedin, 'Child sexual abuse is largely hidden from the adult society: An epidemiological study of adolescents' disclosures', Child Abuse & Neglect, vol 32, no 12, 2008, p 1105; E Teram, C Stalker, A Hovey, C Schachter & G Lasiuk, 'Towards malecentric communication: Sensitizing health professionals to the realities of male childhood sexual abuse survivors', Issues in Mental Health Nursing, vol 27, no 5, 2006, p 503.

- CM Arata, 'To tell or not to tell: Current functioning of child sexual abuse survivors who disclosed their victimization', Child Maltreatment, vol 3, no 1, 1998; J Cashmore & R Shackel, 'Gender differences in the context and consequences of child sexual abuse', Current Issues in Criminal Justice, vol 26, no 1, 2014, 81; SE Palmer, RA Brown, NI Rae-Grant & MJ Loughlin, 'Responding to children's disclosure of familial abuse: What survivors tell us', Child Welfare, vol 78, no 2, 1999, p 263.
- M Knefel & B Lueger-Schuster, 'An evaluation of ICD-11 PTSD and complex PTSD criteria in a sample of adult survivors of childhood institutional abuse', *European Journal of Psychotraumatology*, vol 4, 2013; E Romano & RVD Luca, 'Male sexual abuse: A review of effects, abuse characteristics, and links with later psychological functioning', *Aggression and Violent Behavior*, vol 6, no 1, 2001.
- E Romano & RVD Luca, 'Male sexual abuse: A review of effects, abuse characteristics, and links with later psychological functioning', *Aggression and Violent Behavior*, vol 6, no 1, 2001.
- DJ Higgins & MP McCabe, 'Multiple forms of child abuse and neglect: Adult retrospective reports', *Aggression and Violent Behaviour*, vol 6, 2001; J Cashmore & R Shackel, *The long-term effects of child sexual abuse*, Australian Institute of Family Studies, Melbourne, 2013, p 5.
- 173 The Victorian Royal Commission into Family Violence revealed the profound effects of domestic violence on children, either as targets of violence or indirectly through exposure to family violence or its effects in the home. Royal Commission into Family Violence, Royal Commission into Family Violence: Report and Recommendations, Victorian Government, Melbourne, 2016, p 35.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors:*A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 72.
- 175 Transcript of R Carlile, Case Study 5, 28 January 2014 at T3903:34–37.
- 176 Transcript of M Stiles, Case study 5, 3 February 2014 at T4231:45–T4232:7.
- 177 For example, Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 5: Response* of The Salvation Army to child sexual abuse at its boys homes in New South Wales and Queensland, Sydney, 2015, p 19.
- 178 For example: Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 5: Response* of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland, Sydney, 2014, pp 17–19.
- B Gallagher, 'The extent and nature of known cases of institutional child sexual abuse', *British Journal of Social Work*, vol 30, no 6, 2000, p 807.
- Transcript of J Mulquiney, Case Study 7, 27 February 2014 at 4995:17–24; Exhibit 7-0011, 'Statement of Diane Chard', Case Study 7, STAT.0149.001.0001_R at 0004_R; Exhibit 7-0012, 'Statement of Janet Mulquiney', Case Study 7, STAT.0155.001.0001 at 0007; Exhibit 7-0013, 'Statement of Mary Farrell-Hooker', Case Study 7, STAT.0151.001.0001_R at 0007_R; Exhibit 7-0019, 'Statement of Yvonne Kitchener', Case Study 7, STAT.0152.001.0001_R at 0005_R; Exhibit 7-0020, 'Statement of Coral Campbell', Case Study 7, STAT.0148.001.0001 at 0006; Exhibit 7-0016', Statement of Robyne Stone', Case Study 7, STAT.0147.001.0001_R at 0005_R.
- Transcript of F Hillery, Case Study 7, 26 February 2014 at 4881:23–27; Transcript of W Robb, Case Study 7, 28 February 2014 at 5083:10–21; Exhibit 7-0012, 'Statement of Janet Mulquiney', Case Study 7, STAT.0155.001.0001 at 0005; Exhibit 7-0006, 'Statement of Robin Kitson', Case Study 7, STAT.0162.001.0001_R at 0007_R. Transcript of R Carlile, Case Study 5, 28 January 2014 at T3903:34–37.
- 182 Transcript of M E Farrell-Hooker, Case Study 7, 27 February 2014, at 5011:2–3.
- 183 Children with Disability Australia, Submission to the Royal Commission into Institutional Responses to Child Sexual Abuse, *Issues Paper No 10: Advocacy and Support and Therapeutic Treatment Services*, 2015, p 5.
- J Breckenridge & G Flax, Service and support needs of specific population groups that have experienced child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 40.
- 185 Transcript of CIF, Case Study 41, 12 July 2016 at 20193:46–20194:37; Exhibit 41-0006, 'Statement of CIF', Case Study 41, STAT.1032.001.0001_R at 00017_R.
- J Breckenridge & G Flax, Service and support needs of specific population groups that have experienced child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 39, 40, 42.
- 187 I Hershkowitz, ME Lamb & D Horowitz, 'Victimisation of children with disabilities', *American Journal of orthopsychiatry*, vol 77, no 4, 2007; MH Kvam, 'Is sexual abuse of children with disabilities disclosed? A retrospective analysis of child disability and the likelihood of sexual abuse among those attending Norwegian hospitals', *Child Abuse & Neglect*, vol 24, no 8, 2000; S Akbas, A Turla, K Karabekiroglu, O Pazvantoglu, T Keskin & O Boke, 'Characteristics of sexual abuse in a sample of Turkish children with and without mental retardation, referred for legal appraisal of the psychological repercussions', *Sexuality & Disability*, vol 27, 2009, pp 205–13.
- J Breckenridge & G Flax, Service and support needs of specific population groups that have experienced child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 41.
- Name changed, private session, 'Jeanette'. See also: Name changed, private session, 'Carly'.
- 190 Children With Disability Australia, Submission to the Royal Commission into Institutional Responses to Child Sexual Abuse, Issues Paper No 9: Risk of Child Sexual Abuse in Schools, 2015, p 9.
- People with Disability Australia, Submission to the Royal Commission into Institutional Responses to Child Sexual Abuse, Issues Paper No 10: Advocacy and Support and Therapeutic Treatment Services, 2015, p 6.

- Transcript of F Hand, Case Study 41, 19 July 2015 at 20624:34–47; Exhibit 41-0030, 'Statement of Felicity Hand', Case Study 41, STAT.1047.001.0001_R at 0003_R. The current Commonwealth position is that approval for restrictive practices is managed through state and territory government processes, with further oversight of restrictive practices to be undertaken by disability service providers under the National Disability Insurance Scheme Quality and Safeguarding Framework. See Council of Australian Governments Disability Reform Council, NDIS Quality and Safeguarding Framework, Department of Social Services, Canberra, 2016, p 17.
- 193 Transcript of M Bowden, Case Study 41, 19 July 2016 at 20632:9–27.
- 194 G Llewellyn, S Wayland & G Hindmarsh, *Disability and child sexual abuse in institutional contexts,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 31; S Robinson, *Enabling and protecting: Proactive approaches to addressing the abuse and neglect of children and young people with disability,* Children With Disability Australia, Clifton Hill, Vic, 2012, pp 5, 10.
- 195 C Esposito, Child sexual abuse and disclosure: What does the research tell us?, NSW Government, Family and Community Services, 2015, p 40; SSS Tang, JJ Freyd & M Wang, 'What do we know about gender in the disclosure of child sexual abuse?', Journal of Psychological Trauma, vol 6, no 4, 2008, p 4.
- 196 C Esposito, *Child sexual abuse and disclosure: What does the research tell us?*, NSW Government, Family and Community Services, 2015, p 40.
- 197 CM Arata, 'To tell or not to tell: Current functioning of child sexual abuse survivors who disclosed their victimization', *Child Maltreatment*, vol 3, no 1, 1998; J Cashmore & R Shackel, 'Gender differences in the context and consequences of child sexual abuse', *Current Issues in Criminal Justice*, vol 26, no 1, 2014, p 81.
- 198 Name changed, private session, 'Magdalena'.
- 199 C Esposito, Child sexual abuse and disclosure: What does the research tell us?, NSW Government, Family and Community Services, 2015, p 40; SE Ullman, 'Relationship to perpetrator, disclosure, social reactions, and PTSD symptoms in child sexual abuse survivors', Journal of Child Sexual Abuse, vol 16, no 1, 2007, pp 21, 32; SD Easton, 'Disclosure of child sexual abuse among adult male survivors', Clinical Social Work Journal, vol 41, 2013, p 351.
- SD Easton, 'Disclosure of child sexual abuse among adult male survivors', Clinical Social Work Journal, vol 41, 2013, p 351.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 79.
- D Tener & S Murphy, 'Adult disclosure of child sexual abuse: A literature review', *Trauma, Violence & Abuse*, vol 16, no 4, 2015, p 397.
- J Astbury, Child sexual abuse in the general community and clergy-perpetrated child sexual abuse, Australian Psychological Society (unpublished), 2013, p 12; T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 79.
- 204 A Matsakis, *I can't get over it: A handbook for trauma survivors*, 2nd edn, New Harbinger Publications, Inc, Oakland, 1996, pp 90–1.
- A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 35–40.
- M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015; Child Family Community Australia, *Is Resilience Still a Useful Concept When Working with Children and Young People?*, Australian Institute of Family Studies, Melbourne, April 2012.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 46; Child Family Community Australia, *Is resilience still a useful concept when working with children and young people?*, Australian Institute of Family Studies, Melbourne, April 2012.
- 208 Child Family Community Australia, *Is resilience still a useful concept when working with children and young people?*, Australian Institute of Family Studies, Melbourne, April 2012.
- 209 VL Banyard & LM Williams, 'Women's voices on recovery: A multi-method study of the complexity of recovery from child sexual abuse', *Child Abuse & Neglect*, vol 31, 2007, p 287.
- 210 S Luthar, 'Resilience in development: A synthesis of research across five decades' in D Chicchetti and D Cohen (eds), Developmental psychopathology: Risk, disorder and adaptation, John Wiley & Sons, New York, 2006, p 743; Child Family Community Australia, Is resilience still a useful concept when working with children and young people?, Australian Institute of Family Studies, Melbourne, April 2012.
- AS Masten, 'Ordinary magic: Resilience processes in development', Amercian Psychologist, vol 56, no 3, 2001, p 227.
- G Pérez-Fuentes, M Olfson, L Villegas, C Morcillo, S Wang & C Blanco, 'Prevalence and correlates of child sexual abuse: A national study', *Comprehensive Psychiatry*, vol 54, no 1, 2013.
- 213 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015; C Marriott, C Hamilton-Giachritsis & C Harrop, 'Factors promoting resilience following childhood sexual abuse: A structured, narrative review of the literature', *Child Abuse Review*, vol 23, 2014.
- 214 C Marriott, C Hamilton-Giachritsis & C Harrop, 'Factors promoting resilience following childhood sexual abuse: A structured, narrative review of the literature', *Child Abuse Review*, vol 23, 2014, p 26.

- 215 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015; C Marriott, C Hamilton-Giachritsis & C Harrop, 'Factors promoting resilience following childhood sexual abuse: A structured, narrative review of the literature', *Child Abuse Review*, vol 23, 2014.
- 216 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 489.
- 217 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 488; T Moore, M McArthur, D Noble-Carr & D Harcourt, *Taking us seriously: children and young people talk about safety and institutional responses to their safety concerns,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2015, p 32.
- 218 Transcript of Dr G Gee, Case Study 57, 31 March 2017 at 27811:1–15.
- 219 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 147.
- 220 Transcript of AYB, Case Study 26, 14 April 2015 at C7328:3–7; Exhibit 26-0001, 'Statement of AYB', Case Study 26, STAT.0531.001.0001 R at 0020 R.
- 221 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 486.
- 222 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 485.
- 223 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 486.
- Name changed, private session, 'Stew'.
- Name changed, private session, 'Rodney Adam'.
- Name changed, private session, 'Nessa'.
- Transcript of G J McDonald, Case Study 43, 31 August 2016 at 17602:11–16; Exhibit 43-001, 'Statement of Gerard McDonald', Case Study 43, STAT.1168.001.0001 at 0013.
- 228 C Marriott, C Hamilton-Giachritsis & C Harrop, 'Factors promoting resilience following childhood sexual abuse: A structured, narrative review of the literature', *Child Abuse Review*, vol 23, 2014, p 29.
- Name changed, private session, 'May'.
- Name changed, private session, 'Luisa'.
- Name changed, private session, 'Calista Ann'.
- 232 M Griffiths, Submission to the Royal Commission into Institutional Responses to Child Sexual Abuse, *Issues Paper No 10: Advocacy and Support and Therapeutic Treatment Services*, 2015; Heal for Life Foundation, Submission to the Royal Commission into Institutional Responses to Child Sexual Abuse, *Issues Paper No 10: Advocacy and Support and Therapeutic Treatment Services*, 2015; Knowmore, Submission to the Royal Commission into Institutional Responses to Child Sexual Abuse, *Issues Paper No 10: Advocacy and Support and Therapeutic Treatment Services*, 2015.
- Name changed, private session, 'Maeve'.
- 234 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 155.
- I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 155.
- 236 Name changed, private session, 'Gladys'.
- 237 Name changed, private session, 'Marjorie'.
- 238 Name changed, private session, 'Shevonne'.
- Name changed, private session, 'Shevonne'.
- 240 Transcript of P J Russ, Case Study 43, 8 September 2016 at 18228:33–38; Exhibit 43-029, 'Statement of Peter John Russ', STAT.1175.001.0001 at 0004, 0005.
- 241 Transcript of P J Russ, Case Study 43, 8 September 2016 at 18228:33–38; Exhibit 43-029, 'Statement of Peter John Russ', STAT.1175.001.0001 at 0004, 0005.
- 242 Royal Commission Aboriginal and Torres Strait Islander community consultations, 2014–2017.
- 243 Royal Commission Aboriginal and Torres Strait Islander community consultations, 2014–2017.
- 244 Royal Commission Aboriginal and Torres Strait Islander community consultations, 2014–2017.
- Name changed, private session, 'Dale'.
- 246 Royal Commission Aboriginal and Torres Strait Islander community consultations, 2014–2017.
- 247 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 488; Transcript of Dr G Gee, Case Study 57, 31 March 2017 at 27811:15–18.
- 248 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 488.
- Name changed, private session, 'Lucy'.
- 250 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 486.

- 251 Transcript of BQK, Case Study 34, 3 November 2015 at 12011:5–11.
- Name changed, private session, 'Sebastian'.
- 253 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 486.
- Name changed, private session, 'Jed'. See also: Name changed, private session, 'Brandon Jesse'.
- Name changed, private session, 'Gabrielle'.
- Name changed, private session, 'Ethan'.
- 257 C Marriott, C Hamilton-Giachritsis & C Harrop, 'Factors promoting resilience following childhood sexual abuse: A structured, narrative review of the literature', *Child Abuse Review*, vol 23, 2014, p 30; RG Tedeschi & LG Calhoun, *Trauma and transformation: Growing in the aftermath of suffering*, Sage, Thousand Oaks, 1995, p 117.
- Name changed, private session, 'Roddick'.
- Name changed, private session, 'Joni'.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 19; S Zubrick, C Shepherd, P Dudgeon, G Gee, Y Paradies, C Scrine & R Walker, 'Social determinants of social and emotional wellbeing' in P Dudgeon, H Milroy and R Walker (eds), Working together: Aboriginal and Torres Strait Islander mental health and wellbeing principles and practice, 2nd edition, Commonwealth of Australia, Canberra, 2010, p 83.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 19; see also State Government of Victoria, The State of Victoria's children 2009: Aboriginal children and young people in Victoria, Department of Education and Early Childhood Development, Melbourne, 2009, p 45.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 35; S Silburn, S Zubrick, JD Maio, C Shepherd, J Griffin, F Mitrou, R Dalby, C Hayward & G Pearson, The Western Australian Aboriginal child health survey: Strengthening the capacity of Aboriginal children, families and communities, Curtin University of Technology and Telethon Institute for Child Health, Perth, 2006, p 369. See also AM Dockery, 'Culture and wellbeing: The case of Indigenous Australians', Social Indicators Research, vol 99, no 2, 2010; S Garnett, B Sithole, P Whitehead, C Burgess, F Johnston & T Lea, 'Healthy country, healthy people: Policy implications of links between Indigenous human health and environmental condition in tropical Australia', Australian Journal of Public Administration, vol 68, no 1, 2009; C Burgess, F Johnston, D Bowman & P Whitehead, 'Healthy country: Healthy people? Exploring the health benefits of Indigenous natural resource management', Australian and New Zealand Journal of Public Health, vol 29, no 2, 2005; K Rowley, K O'Dea, I Anderson, R McDermott, K Saraswati, R Tilmouth, I Roberts, J Fitz, Z Wang, A Jenkins, J Best, Z Wang & A Brown, 'Lower than expected morbidity and mortality for an Australian Aboriginal population: 10 year follow-up in a decentralised community', The Medical Journal of Australia, vol 188, no 5, 2008.
- 263 Transcript of Dr G Gee, Case Study 57, 31 March 2017 at 27811:20–27.
- 264 Transcript of Dr G Gee, Case Study 57, 31 March 2017 at 27811:20–27.
- Name changed, private session, 'Donald Steven'.
- Name changed, private session, 'Brydie'.
- Name changed, private session, 'Brydie'.
- Name changed, private session, 'Evelyn'.
- 269 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 479.
- 270 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 488.
- 271 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 479; C Marriott, C Hamilton-Giachritsis & C Harrop, 'Factors promoting resilience following childhood sexual abuse: A structured, narrative review of the literature', *Child Abuse Review*, vol 23, 2014, p 27.
- 272 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 479; C Marriott, C Hamilton-Giachritsis & C Harrop, 'Factors promoting resilience following childhood sexual abuse: A structured, narrative review of the literature', *Child Abuse Review*, vol 23, 2014, p 27. See also Transcript of Dr G Gee, Case Study 57, 31 March 2017 at 27811:1–15.
- 273 M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, p 485.
- Name changed, private session, 'Lachlan'.
- I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 186.
- Name changed, private session, 'Rainey'.
- For example: Name changed, private session, 'Antony'; Name changed, private session, 'Lesley'; Name changed, private session, 'Tricia'; Name changed, private session, 'Wayne Christopher'; Name changed, private session, 'Peggy'.

- Name changed, private session, 'Stanley Peter'. 278
- Name changed, private session, Stanley Peter.

 I Katz, A Jones, BJ Newton & E Reimer, Life journeys of victim/survivors of child sexual abuse in institutions:

 An analysis of Royal Commission private sessions, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 191–2.

 Name changed, private session, 'Ross William'.

 Transcript of Jyoti, Case Study 21, 2 December 2014 at 10916:41–44. 279
- 280

72

- 281
- 282 Transcript of Jyoti, Case Study 21, 2 December 2014 at 10916:43–10917:5.

3 Impacts of child sexual abuse

As discussed in Chapter 2, the Royal Commission heard that the impacts of child sexual abuse can differ for each individual and across a lifetime. Each story we were told was unique, reflecting the type of sexual abuse, when and where the victim was abused and by whom, and their individual circumstances

Although each individual is affected differently, some impacts are commonly experienced by survivors of child sexual abuse in institutional contexts. This chapter describes the common impacts we heard about in private sessions, public hearings and the available research on child sexual abuse. It begins by describing how sexual abuse can result in profound trauma, potentially interrupting normal psychosocial development at every critical stage of a child's formative years. It then describes the many areas of victims' lives that can be affected by sexual abuse in institutional contexts, including:

- mental health
- interpersonal relationships
- physical health
- sexual identity, gender identity and sexual behaviour
- connection to culture
- spirituality and religious involvement
- interactions with society
- education, employment and economic security.

Many of the commonly identified impacts of child sexual abuse in institutional contexts are also experienced by victims of child sexual abuse in familial and other contexts. However, research suggests there may also be distinct impacts when children are sexually abused in institutional settings. These include impacts on spirituality and religious involvement, such as a loss of faith and a loss of trust in the religious institution, for those victims sexually abused in a religious context. Distrust and fear of institutions and authority are also distinctive effects on some victims of child sexual abuse in institutional contexts.

In addition, inadequate institutional responses to child sexual abuse can have distinctive effects on victims, for example feelings of institutional betrayal, a continuation of abuse and re-traumatisation. Institutions need to be aware of their potential to either significantly compound or help alleviate the impacts of the abuse. The impacts of institutional responses to abuse are discussed in Chapter 4.

The impacts on secondary victims, including family members, children and staff in the institution, as well as communities, are discussed in Chapter 5.

What survivors told us in private sessions about how they were affected by child sexual abuse is an important source of information for this chapter. Of 6,875 survivors who attended private sessions as at 31 May 2017, a total of 6,412 (93.3 per cent) provided information about the impacts of being sexually abused as a child. For a range of reasons, some survivors in private sessions provided limited or no detail on impacts. Some chose not to discuss the impact of the abuse. Others said that they were affected by the abuse but provided no additional detail, while some told us they did not know what the impact of abuse had been on their lives.

Throughout this chapter, when we discuss the percentages of private sessions attendees who told us about specific types of impacts, we present these as a proportion of only those who provided information about impacts. As a result, we may overestimate the proportion of all private sessions attendees who experienced these impacts. Other limitations of private sessions information are discussed in Chapter 1.

Despite these limitations, we consider that private sessions information is a good indication of the common impacts of child sexual abuse in institutional contexts. The prevalence of the impacts of child sexual abuse we heard about in private sessions is supported by what we learned through public hearings and research.

The most common impacts discussed by survivors who attended private sessions are presented in Table 3.1. Mental health issues, such as low self-esteem, anxiety and depression were most commonly spoken about in private sessions, with more than nine in 10 survivors telling us about these issues. The next most frequently raised concerns were relationship difficulties, including difficulties with trust and intimacy, lack of confidence with parenting and marital problems. Following this, concerns related to education and employment were frequently described. Volume 5, *Private sessions*, describes in further detail what survivors told us about the impacts of their experiences of sexual abuse.

Table 3.1 – Impacts of child sexual abuse on survivors, information from private sessions, May 2013 – May 2017

Impact	Number affected (6,412 survivors) ^a	Proportion (%)
Mental health issues	6,088	94.9
Relationship issues	4,332	67.6
Education and/or economic issues	3,569	55.7
Sexual behaviour issues	1,554	24.2
Involvement in crime	1,456	22.7
Physical health issues	461	7.2
Direct consequences (eg pregnancy, sexually transmitted infection, injury)	440	6.9
Other	32	0.5

^a A total of 6,412 survivors provided information on how they were affected.

Note: Percentages are presented as a proportion of those survivors who provided information in private sessions about the impacts of child sexual abuse.

While this chapter describes a wide range of effects on victims across distinct areas of their lives, people mostly told us they experienced multiple and complex effects at any one time, and – as shown in the example of BYC – these were often interconnected. The combined effect of multiple impacts over a lifetime can be debilitating.

Survivor BYC: impacts on mental health, employment and family relationships

BYC gave evidence in *Case Study 36: The response of the Church of England Boys' Society and the Anglican Dioceses of Tasmania, Adelaide, Brisbane and Sydney to allegations of child sexual abuse (Church of England Boys' Society)* about his sexual abuse by Simon Jacobs, a leader in the St Ives branch of the Church of England Boys' Society (CEBS) in the 1970s.² BYC joined the St Ives CEBS branch when he was seven years old. At that time, Jacobs was 13 years older than BYC and was a leader of the 'Pages' group that BYC attended, running their weekly activities as well as excursions and camps. Over time, Jacobs became a close friend of BYC's family and eventually started driving BYC home from weekly Friday evening CEBS meetings. BYC said that he was first sexually abused by Jacobs in May 1977, when he was 10 years old, during an overnight trip to Young in New South Wales with his family and Jacobs. BYC said that from that time until 1981 Jacobs continued to sexually abuse him.

In early 1987, BYC first disclosed the abuse to his parents, as well as to the Reverend Jobbins, who at that time was the rector at St Swithun's Anglican church in Pymble. BYC said that Reverend Jobbins told him to 'let sleeping dogs lie'. BYC first reported the sexual abuse to police in October 1988, but charges against Jacobs were dismissed. However, in 2009, Jacobs was convicted of sexual offences against BYC and other boys.

BYC told us about the profound and interconnected impacts of the abuse over the course of his life:

I have had severe depression and very low self-esteem since I was 12 years old, I have been diagnosed with PTSD and severe anxiety. I have been on mood-stabilising medication since 1998. I see my GP once or twice a week, and both my psychiatrist and social worker once a month. I have been hospitalised in a psychiatry unit on at least five occasions. I have attempted suicide several times. I constantly experience flashbacks of the abuse, triggered by media and certain places and events. I find life difficult because I don't cope with stress, change or sudden events, and I am often unable to manage everyday living. Emotionally, I have not developed beyond my early teens, the age I was at the time of the abuse.

After a short time, people can tell that I have psychological issues and am not normal. I don't see a future for myself. If I did not have my dog, I would often see no reason to get out of bed. I have to get up to feed him, take him outside, and play with him.

I am very isolated and not able to have relationships. I have very few friends and I find it hard to trust people. I feel resentment and anger at any dominating authority.

My relationship with my parents up until 2000 was very difficult and I was not able to live at home between 1998 and 2000. I blamed my parents for what had happened with Jacobs. I often lose my temper with Mum and I don't know how she copes with it. She has been a fantastic support to me and has acted as a buffer between me and the Church, Centrelink and the New South Wales Department of Housing.

My schooling, education and professional life have been affected. I developed issues with authority as a teenager and was disruptive in class. I had to repeat my HSC, and it took me 10 years to complete my Bachelor of Arts degree. I worked primarily in social service jobs ... and I had a few other jobs temporarily, but nothing for more than a year. I have not worked since 2007. I have been on the Disability Pension since 1998. I live below the poverty line, in subsidised rental accommodation that is very unstable and precarious.

I feel the abuse has affected every facet of my life.³

3.1 Effects of trauma on children's development

Experiences in an individual's childhood can shape their health and wellbeing throughout their entire life. Therefore, it is important to understand how sexual abuse can affect the emotional, social and physical development of the child.

As discussed in Chapter 2, while not all victims are affected by child sexual abuse the same way, many survivors told us their experiences were traumatic, with profound, complex and long-lasting effects. There is strong evidence that early onset trauma caused by adverse childhood events, including sexual abuse, can have a lasting impact both on childhood development and on the formation of a secure child-caregiver attachment, as well as on subsequent interpersonal relationships. The potential long-term impact of the sexual abuse is likely to be mediated by victim, perpetrator and institutional factors, and the broader social and historical contexts in which the sexual abuse took place. Significantly, the impacts of childhood trauma manifest differently according to individual vulnerabilities at particular critical phases of development.

Developmentally appropriate early intervention that is both trauma-informed and recoveryoriented can assist children who have been subjected to sexual abuse to heal and recover.⁶

3.1.1 Understanding childhood trauma

Traumatic events, such as child sexual abuse, are understood as 'extraordinary, not because they occur rarely, but rather because they overwhelm the ordinary human adaptations of life'. Experiencing these traumatic events increases a child's susceptibility to developing a range of neurological, social and physical health difficulties across their lifespan. The roots of these complex trauma-related problems lie in a child's adaptive responses or reactions to threat. Psychiatrist Professor Judith Herman explains how a child trapped in an abusive environment, and in the absence of adult care and protection, is faced with a formidable task:

She must find a way to preserve a sense of trust in people who are untrustworthy, safety in a situation that is unsafe, control in a situation that is terrifyingly unpredictable, power in a situation of helplessness. Unable to care for or protect herself, she must compensate for the failures of the adult care and protection with the only means at her disposal, an immature system of psychological defences.⁸

These immediate adaptive responses to threat usually include either hyper-arousal (activating a 'flight or fight' response) or dissociation, involving withdrawal from the outside world to an inner world.⁹

Knowledge about the effects of childhood trauma is increasing rapidly. Child maltreatment can produce lasting alterations in the endocrine, autonomic and central nervous systems. Neuroimaging studies have mapped visual evidence of disturbances to the 'emotional brain', particularly the hippocampus and amygdala, areas of the brain involved in processing memory and emotion, such as fear. Changes in the central nervous system can cause hyper-vigilance, a state of sensory sensitivity accompanied by an exaggerated intensity of behaviours used to detect threats. Children may also repeatedly re-experience the terror of the original traumatic event (often through nightmares and flashbacks), and experience feelings of helplessness, as well as problems with concentration and an exaggerated startle response. Children exposed to perpetrator tactics that involve secrecy, complicity and menace, may come to view adults as potential sources of threat rather than comfort and support.

One of the most significant consequences of early trauma is disruption to the development of emotional regulation. Psychiatrist and trauma expert Dr Bessel van der Kolk has argued that 'at the core of traumatic stress is a breakdown in the capacity to regulate internal states', which may include fear, anger and sexual impulses. A child who cannot regulate internal states may have lifelong difficulties tolerating or regulating distress, behaviour and impulses. This may result in self-destructive, self-harming behaviours, as well as excessive risk taking and thoughts of suicide in later life. 14

Recent research on trauma also suggests that other significant consequences of childhood trauma caused by sexual abuse can include changes in memory function and concentration, distortion of self-perception caused by internalising shame and guilt, and a compromised capacity to trust others and maintain personal safety. Some theorists have also identified links between complex developmental trauma and increased risk of re-victimisation and poly-victimisation (co-occurrence of different forms of abuse) over an individual's lifetime.

Children and adults surviving traumatic events are frequently diagnosed with post-traumatic stress disorder (PTSD).¹⁷ PTSD describes the development of a specific set of characteristic symptoms following exposure to one or more traumatic events. These symptoms include intrusive memories or dreams related to the traumatic event, avoidance of stimuli associated with the traumatic event, negative changes in cognition and mood, and hyper-arousal.¹⁸ In *Case Study 28: Catholic Church authorities in Ballarat (Catholic Church authorities in Ballarat)* child and family psychiatrist Dr Carolyn Quadrio described 'classic PTSD' as consisting of 'hyperarousal and hyper-vigilance ... being preoccupied with memories of the trauma ... [and] while preoccupied with the trauma ... at the same time you often have blanking it out as well'.¹⁹ As discussed in this chapter, many victims who are sexually abused as children experience these symptoms in adulthood. Research shows that PTSD is also increasingly recognised in children who have been sexually and/or physically abused.²⁰

However, over recent decades, several researchers and practitioners have suggested that inadequate diagnostic criteria for PTSD have created barriers to understanding the effects of traumatic experiences, such as child sexual abuse. These experiences are interpersonal, pervasive and sometimes associated with multiple types of prolonged abuse. ²¹ 'Complex trauma' or 'complex PTSD' has been increasingly used to describe the complicated range of symptoms that can result from these experiences, including child sexual abuse (although such usage of these terms is not yet outlined systematically in diagnostic classifications). ²² These symptoms can extend beyond recognisable mental health issues and surface as problems with relationships and identity, ²³ physical symptoms or substance abuse issues, which have often developed out of a survivor's attempts to cope with the trauma. ²⁴ Some trauma theorists have also suggested that the failure of practitioners to recognise complex trauma as a distinct entity may result in survivors being diagnosed with a 'bewildering array' of conditions and disorders that are stigmatising and pathologising. ²⁵ Importantly, if practitioners miss the underlying cause of the problems experienced by these survivors, the treatment may be fragmented and ineffective. ²⁶

When children suffer repeated and complex trauma, as is the case with many instances of child sexual abuse, there are particular effects. This is because, unlike adults, children who experience repeated and complex trauma are still developing.²⁷ The impact of this trauma at key development stages may have formative impacts that it may not have for an adult. Psychiatrist Professor Herman observes that 'repeated trauma in adulthood erodes the structures of the personality already formed', whereas repeated trauma in childhood 'forms and deforms personality'.²⁸

The developmental aspects of trauma for victims of childhood sexual abuse were explained by Dr Quadrio in giving evidence in the *Catholic Church authorities in Ballarat* case study:

... [A child's] developmental sequence is disrupted by the abuse and, if the abuse continues over a long period of time, every new phase of development is being affected by the ongoing trauma. So, it ends up being what we call developmental trauma or complex PTSD, because it affects their personality. So now you're not dealing with just symptoms, like symptoms of depression. You're dealing with characterological disturbance, and so every aspect of a child's function becomes disturbed; their feelings, their thinking, their memory, their concentration. What's very damaging is that a child is supposed to be developing a sense of who I am and what kind of person I am and a sense of understanding the world and other people in it. And so, obviously, the child's self-perception is very damaged, all those feelings of shame, for example, are very damaging to the sense of self.²⁹

3.1.2 Effects of trauma on the developing brain

Trauma affects more than just the emotional and psychosocial development of the child. Increasingly, research suggests that trauma affects the chemistry, structure and function of the developing human brain, especially when it is repeated or ongoing.³⁰ Early life trauma affects a person's ability to process and regulate emotion, with potential impacts on empathy and social interaction. In particular, it affects the hypothalamic-pituitary-adrenal (HPA) axis, which regulates anxiety and manages the 'freeze, fight or flight' response to stress.³¹

Very young children are at particular risk of suffering lasting effects of trauma precisely because their brains are still developing and also because brain development is profoundly guided by experience. The developing brain changes physically in response to repeated experiences. Just as this process can create positive learning and memory, so it can create negative learning and memory due to neglect, trauma and chaos. When a child is threatened and activates a stress response in a repetitive or extremely prolonged fashion, the neural networks involved will alter the baseline activity and reactivity of the stress response systems. The brain will then reset, acting physiologically as if the individual is under threat, even when the threat no longer exists. The brain becomes poorly developed and functionally disorganised, rendering the child less able to respond emotionally, verbally or intellectually to normal experiences, let alone traumatic ones. The stress response systems are provided to respond emotionally, verbally or intellectually to normal experiences, let alone traumatic ones.

Recent research suggests that child sexual abuse may also be associated with cognitive risk factors and cognitive changes.³⁶ Hyper-vigilance and hyper-arousal may result in severe limitations in the child's ability to concentrate and learn in the classroom. Dr Quadrio's evidence in the *Catholic Church authorities in Ballarat* case study emphasised this point:

If a child is continually suffering hyper-arousal, that can have a very damaging impact on their entire function, because it means that they won't be concentrating at school because their thoughts are racing. They won't be attentive because their attention's flickering all the time. They won't be sleeping so well. That will affect their growth; that will affect their energy levels.³⁷

3.1.3 The role of attachment

'Attachment' describes an individual's emotional ties with another. Attachment theorists have suggested that attachment involves forming secure, responsive and flexible bonds between a child and a caregiver, such as a parent.³⁸ These bonds exert a profound influence on the social, cognitive and emotional development of the child, particularly in the 'sensitive' or 'critical' period when the social brain is developing during the first years of life.³⁹

Secure attachment not only serves the primary function of preserving life, but also forms the basis of the child's ability to regulate emotions. Children who are soothed and comforted when they experience distress, gradually learn to comfort themselves by evoking mental images of their caretaker. It provides the secure base from which the child can confidently explore the environment and, as Professor Herman said in her research on the matter, 'permits the development of a self-identity as a person worthy of love and care and a capacity to love and care for others'.⁴⁰

When caretaking relationships are disrupted, such as when a caregiver sexually abuses a child, secure attachment is compromised, with profound impacts on a child's ability to connect with others. As a result, the child may be less able to learn about themselves, other people or the world around them. They may find it hard to understand the rules of relationships or manage and regulate strong emotions, or even basic bodily functions, such as breathing and heart rate.⁴¹

As discussed in Chapter 2, we heard in private sessions and case studies about numerous victims whose caregiving relationships had been severed or disrupted at an early age and who were later sexually abused in institutions. Without earlier experiences of secure attachment, children cannot draw on such attachment relationships to help regulate their emotions and protect themselves from child sexual abuse and its impacts. Research suggests that the trauma of child sexual abuse maybe mediated by a positive or supportive parent-child relationship.⁴²

3.1.4 Effects of trauma at different developmental stages

Social development in humans is ongoing and cumulative, with earlier phases of development providing the building blocks for later development. However, while the path of development is sequential and predictable, there is variation in what is considered 'normal' development. Unravelling the impacts of child sexual abuse means understanding that the experience of this abuse can interrupt normal psychosocial development at each critical stage of a child's formative years, affecting their ability to consolidate skills and progress to the next developmental stage.

Research suggests that the trauma of sexual abuse acts on children in different ways according to their stage of development.⁴³ For example, an infant who has not yet developed verbal skills may not hold a concrete narrative memory of the abuse, although the memory may be stored in their sensory systems (sight, smell, sound, taste and touch). Children at this stage of development may manifest their trauma physically. For example, they may have difficulty sleeping and eating, and may be hard to console.

A preschool child with verbal skills may regress to earlier stages of development, causing them to fear being alone and near unfamiliar adults, or have difficulties with sleeping and learning. They may not want to engage with peers or may lose skills they had mastered, such as bladder or bowel control.⁴⁴ They may become quiet and withdrawn. For example, in *Case Study 19: The response of the State of New South Wales to child sexual abuse at Bethcar Children's Home in Brewarrina, New South Wales (Bethcar Children's Home), AIQ, who believes she was two or three years old when she was first sexually abused, told us:*

The sexual abuse was horrible. I had no one I could talk to about it, so I just bottled it up inside and blocked it out of my mind. This caused me to become a timid, shy and weak child. I always kept to myself...⁴⁵

As a child gets older and progresses to upper primary school, questions of identity, self-image and interpersonal relationships come to the fore. Children at this stage of development will often believe the sexual abuse they are experiencing is their fault and may develop a negative self-image, and internalise their feelings. They may suffer disturbances in specific skills, including social and communication skills, memory and the ability to make sense of the behaviours of others. In Case Study 32: The response of Geelong Grammar School to allegations of child sexual abuse of former students, BIZ told us he was sexually abused during 1985 and 1986 at the age of 12 or 13 by live-in boarding house assistant, Phillippe Trutmann. Giving evidence, BIZ said:

After I was abused, I withdrew a lot. I remember I used to listen to music and walk around the grounds a lot. My self-esteem and confidence dropped to new lows and I started feeling depressed at school and stopped trusting people and persons in authority. This impacted on my school work and I found it difficult to concentrate.⁴⁷

At school, children who have been sexually abused may also externalise their distress in disruptive, angry or aggressive behaviour, or exhibit signs of hyper-vigilance and lack of concentration. Eating problems, fear and anxiety, and non-participation in school and social activities may emerge.⁴⁸

The tendency to use 'externalising' behaviours, such as angry and violent behaviour, are common responses to trauma in adolescence. Feelings of shame, embarrassment, resentment, anger and pain can increase with age and maturity. Ongoing childhood abuse reinforces and consolidates self-blame and perceptions of lack of worth. In adolescence, this may lead to a vulnerability to depression, anxiety, stress disorders, self-injury and suicidal ideation.⁴⁹ For example, 'Myles' told us that he was raped by Brother 'Theo Macken' at the age of 14. Ashamed 'Myles' withdrew from his friends: 'I always felt scared that they might find out what had happened and treat me differently'. He believed that he was 'Macken's' only victim and he became a bully, lashing out at other kids. 'Myles' told us he hated himself. He began wetting the bed, burning photos of himself, drinking too much, and engaging in risky behaviours. He tried to take his own life twice. 'Myles' became confused about his sexuality and wondered whether he was gay. He told us his concept of himself 'as a sexual being was totally destroyed ... I still struggle with sex, sexuality, intimacy'.⁵⁰

The tasks of managing puberty, hormonal change and burgeoning sexuality, combined with a desire for intimacy, are especially problematic for a child who has been sexually abused. Adolescent children may have problems with interpersonal relationships and regulating emotions, and may experience ongoing vulnerability to stress and further sexual abuse or other forms of abuse. The repertoire of defence mechanisms and attempts at self-soothing that a child may have previously used, can change with adolescence to include the abuse of alcohol or other drugs, the use of food for comfort or punishment, social withdrawal, isolation and impulsivity, as well as having sex at an early age or having many sexual partners. Mr Manny Waks, giving evidence in *Case Study 22: The response of Yeshiva Bondi and Yeshivah Melbourne to allegations of child sexual abuse made against people associated with those institutions,* told us about the effects that sexual abuse had on him as a young teenager. He described it as a change 'from a happy, positive, reasonably well-behaved boy to an angry, rebellious teenager with a substance abuse problem'. He continued, 'I rejected the religion in which I had been raised, lost all focus on my studies and became alienated from my family and community'. Se

Children of all ages traumatised by sexual abuse can exhibit developmentally inappropriate sexual behaviours.⁵³ These behaviours can also occur in a child as a response to other traumatic experiences. Research suggests that even though many young people who have experienced trauma will not display inappropriate sexualised behaviour, of those who do display these behaviours, a large proportion have experienced trauma.⁵⁴

Inappropriate sexual behaviours can manifest in different ways, depending on the age of the victim. In early childhood, concerning sexual behaviours can include 're-enacting' the abuse, inappropriate touching, sexualised play with toys, sexually explicit drawings, or sudden and intense masturbation.

As a child develops, these sexual behaviours may manifest as aggressive, explicit or exploitative sexualised engagement with other children, sexualised behaviour with adults, or excessive concern with genitalia and adult sexual behaviour. In adolescent children, the effects of sexual abuse may include sexually exploitative interactions with younger children, or potentially risky sexual behaviours, including engaging in unprotected sexual intercourse or sex with multiple partners.⁵⁵

In private sessions, we heard from some parents about their children's sexualised behaviour following sexual abuse. In a private session, 'Coralie' and 'Jack' told us about their son 'Josh', who has a moderate to severe intellectual disability and limited descriptive speech. We heard that after being sexually abused, 'Josh' began displaying some very out-of-character behaviours: 'He would greet men by touching their genital areas and he had problematic sexualised behaviours at school. He had been in trouble on several occasions for touching other children'. ⁵⁶

We also heard from adult survivors about their inappropriate sexual behaviours when they were children, including harmful sexual behaviours. Many told us that they were unaware at the time that their behaviour was inappropriate. For example, we heard from 'Gil James' whose early experience of abuse in his home left him vulnerable to both further harm and harming others. He grew up in the 1980s in Melbourne and when he was in primary school he was sexually abused by older girls in the playground, including one who performed oral sex on him. He said he thought the behaviour was normal. At the age of 11, he was placed in foster care. While in foster care, 'Gil James' sexually abused a younger girl in the home. He told us that at the time he had no idea of the gravity of what he was doing as he was simply mimicking what had been done to him.⁵⁷

However, while a few survivors told us that they exhibited harmful sexual behaviours in childhood, research shows that the vast majority of victims of child sexual abuse do not go on to perpetrate sexual abuse. ⁵⁸ See Volume 10, *Children with harmful sexual behaviours* for more detail.

It is important that parents, carers, professionals and institutions who work with children are alert to the trauma of child sexual abuse, and its manifestations at various developmental stages, to enable them to respond appropriately to children's needs. An understanding of this can also assist us with identifying possible sexual abuse, especially in the absence of a disclosure from a child. How the physical, behavioural and emotional symptoms of trauma may indicate abuse is discussed in Volume 4, *Identifying and disclosing child sexual abuse*. However, while adults need to keep child safety – including the possibility of sexual abuse – in mind when they notice changes in a child, they should not jump to conclusions, but rather carefully consider what they know about the child and the context in which they interact with that child. There are many possible reasons for a change in a child or young person's behaviour or emotions, making it difficult to always establish a causal link between these changes and child sexual abuse. Further, as discussed above, while some children who have been sexually abused will manifest signs of trauma, others may not show any physical or emotional signs.

3.2 Mental health

I have been diagnosed by a psychiatric specialist with chronic dysthymia, episodes of major depression in partial remission, post-traumatic stress disorder, and substance abuse in remission. I have also been diagnosed with generalised anxiety disorder and chronic insomnia. I have suffered significantly from these disorders. I have regular suicidal ideation which requires medication to control. Even with medication I still think about suicide to the point that it is almost a part of me. I require [sleeping tablets] to sleep at night and can only sleep anywhere between two to six hours. Without medication I sleep less than one hour a night.⁶⁰

Ongoing mental health issues were the most commonly described impacts of child sexual abuse that victims identified in private sessions. Of the survivors who told us about impacts, 94.9 per cent spoke about their mental health-related issues following abuse, including depression, anxiety and PTSD, as well as conditions such as schizophrenia and bipolar disorder. Other symptoms of mental distress included nightmares and sleeping difficulties, and emotional issues such as feelings of shame, guilt and low self-esteem. Notably, survivors often had multiple mental health disorders or issues at the same time, rather than in isolation.

An association between child sexual abuse and the adverse mental health consequences that many victims experience has been consistently described in research on child sexual abuse, ⁶¹ as well as in the limited number of studies specific to child sexual abuse in institutional contexts. ⁶² For example, one Australian study suggested that more than one in five victims of child sexual abuse in the study had contact with a public mental health facility in their lifetime, compared with less than one in 10 with no history of sexual abuse. ⁶³

As discussed in Chapter 2, not all victims of child sexual abuse develop mental health issues or adjustment difficulties in adulthood. Some victims may experience few impacts in the short term.⁶⁴ For some, problems may increase with age.⁶⁵ It is also increasingly understood that the impacts of trauma, such as child sexual abuse, may differ over a lifetime, emerging in response to triggers or at various transition points.

This section describes some of the most commonly described impacts on mental health, including:

- emotional issues, such as low self-esteem, shame and guilt and self-blame
- depression
- anxiety
- PTSD
- eating disorders
- other diagnosed disorders
- sleeping difficulties
- suicidality
- self-harm
- alcohol abuse, drug abuse and gambling problems.

3.2.1 Fmotional issues

Emotional issues were often among the most debilitating mental health impacts of child sexual abuse survivors described in private sessions. They included feelings of:

- fear
- low self-esteem and self-worth
- shame and humiliation
- guilt and self-blame
- anger
- grief, sadness, emptiness and loss.

The association between child sexual abuse and emotional issues – particularly shame, self-blame and anger – is reflected in research on the impacts of child sexual abuse, including in institutional contexts. ⁶⁶ Some research suggests that issues of trust, shame, self-blame and low self-esteem and their effect on emotions and behaviour are key factors in understanding the link between child sexual abuse and adult outcomes. ⁶⁷ While the emotional issues described here are treated separately, they may manifest in devastating combinations affecting many aspects of a survivor's life. ⁶⁸

Fear

For many victims, the most overwhelming emotion at the time of sexual abuse was fear. Some survivors told us they were so frightened when it happened that they were 'frozen'. 'Jarrod' told us:

I froze. I froze. For some time, I froze. I can't recall how long. In my mind's eye I froze for nearly 48 years now. And it's just thawing out. My mind's eye is just thawing out to tell this story. It's not a story, it's a fact.⁶⁹

Some survivors told us that experiencing intense fear or terror had lingering effects in their lives. As 'Lillian Jane' told us in a private session, 'I've noticed that I've never reached my full potential because I've always been scared to step out ... You live in fear. You live in fear your whole life. I still live in fear'. We also heard the constant fear that some children experienced over long periods has meant that, as adults, they try to avoid certain situations that might trigger the fear. Many victims said that they re-experienced the terror through nightmares and flashbacks.

Certain events reminiscent of the sexual abuse can also bring back the fear. 'Chelsea' told us that lying on her back triggers fear because she lay on her back and was crushed during the abuse.⁷¹ Some survivors told us they were still terrified of the perpetrators, despite them either being dead or in gaol. We heard how 'Janis', abused by her principal and deputy principal when she was five, continues to wake in terror every morning, feeling 'like they're coming to get me, and I often end up throwing up'.⁷²

Low self-esteem and self-worth

One of the most common emotional issues spoken about by survivors was low self-esteem. Of survivors who discussed impacts in private sessions, 29.3 per cent described experiencing low self-esteem as a result of the sexual abuse. 'Lennox' told us that he was sexually abused by his teacher, who was also a cadet leader at the high school. He said:

I always had very low self-esteem, always thinking I'm one of the worst people around – never had any liking for myself whatsoever. Just the idea of hopelessness, and – that word – useless. And that's been all through my life.⁷³

Another survivor, 'Margo', told us that she was sexually abused by her Sunday school teacher over three years. As a child, she concluded she was worthless as an explanation for why this terrible thing had happened to her. She said she asked herself over and over, 'Why is he allowed to do such awful things to me?' Eventually 'Margo' settled on her answer, and it's stuck with her ever since: 'I believe that I am unworthy of God ... that I'm unworthy of safety and protection, that I'm unworthy of love'. She told us she has suffered 'post-traumatic stress disorder, depression, suicide attempts, dissociation, issues with drug and alcohol, issues with poor eating because I don't believe I'm worth feeding'.

Survivors told us about intense self-hatred: 'If I'm not good enough to be kept by my own mother, I'm not good enough for anything else either';⁷⁶ 'I don't have a hatred of myself anymore, I don't have that shame around what was done to me anymore. But for the first 30 years, yeah I did. And I treated myself very, very badly as a result';⁷⁷ and 'That's what I got from [the school]. I ... hated myself and I didn't want to be on the planet'.⁷⁸

For some victims, this led to low levels of self-confidence about their appearance. For example, in the *Bethcar Children's Home* case study, AIQ told us in evidence:

I hate people seeing me naked. I can't stand my partner coming into the bathroom if I am undressed. I do not care about my appearance and usually wear ugly clothing. I see myself as unattractive.⁷⁹

KR, the mother of KE, observed similar behaviour in her daughter in *Case Study 6: The response* of a primary school and the *Toowoomba Catholic Education Office to the conduct of Gerard Byrnes*. In evidence she told us that, in public, KE covers her body under bulky clothes, wearing 'jumpers and baggy pants on sweltering hot days'. KR said she suspects that this also is a defence mechanism.⁸⁰

Shame and humiliation

Research on the impacts of child sexual abuse emphasises the acute feelings of shame that many victims said they experienced. Shame can be described as a feeling of humiliation or distress, about behaviour seen as disgraceful or dishonourable. Shame bears the fear of being judged and socially disapproved of, and is inwardly directed, causing a survivor to have low self-worth. While victims of child sexual abuse in other settings also commonly recount experiencing shame, as well as guilt (see below), research suggests that there are distinct aspects of victims' experiences of institutional sexual abuse that appear to serve as ongoing reminders of betrayal, deprivation and disadvantage.

Commissioners heard from many survivors about their feelings of shame, as well as humiliation and embarrassment, stemming from the sexual abuse.

'Davey' told us that he was sexually abused by a carer at the age of eight, and then by a cadet officer between the ages of 15 and 17. He told us:

I live life with the thoughts of my past childhood on a weekly and sometimes daily basis which often leads me to sad thoughts, feelings of shame and disgust, embarrassment and anger – I get angry at myself for not having done something much earlier in my life.⁸⁴

Survivors also told us they felt ashamed because they believed they should have stopped the abuse or told somebody, finding it hard to forgive themselves for not acting, despite their being a child at the time. 'Luke Anthony' told us about sexual abuse at an Air Force base, 'After it happened ... I just felt really ashamed and guilty. I should have said, "Stop", but I didn't'.85

Feelings of shame may contribute to problems with relationships. ⁸⁶ 'Roddick' said the thing he found most difficult was forgiving himself. 'I look at myself and see a male whore. It's ingrained in your mind: "It's your fault", "You're a dirty bugger", "You're causing us to do this"'. At various times throughout his life, 'Roddick' has contemplated suicide. On one occasion, he veered his car into the path of a semi-trailer, changing course only at the last moment. 'I thought, "What are you doing, stupid?" and stopped'. He attributed the breakup of his first marriage to his constant feelings of shame which manifested in arguments and difficulty communicating with his wife.⁸⁷

Many survivors also cited feelings of shame as a barrier to disclosure. WP described this feeling in Case Study 12: The response of an independent school in Perth to concerns raised about the conduct of a teacher between 1999 and 2009 (Perth independent school). He told us:

The sexual abuse I suffered was something I was, prior to making the disclosure, something I was going to take to my grave because I was so embarrassed and there was so much shame and confusion and I was just so sad and full of all kinds of emotions.⁸⁸

Volume 4, *Identifying and disclosing child sexual abuse* discusses in more detail how shame and embarrassment, among other factors, can be barriers to victims telling someone about the sexual abuse.

Associated with shame is the fear of being stigmatised by others because of the sexual abuse. Some survivors told us they did not tell anyone about the abuse because they thought they would be singled out or stigmatised by others. Giving evidence in the *Sporting clubs and institutions* case study, BXI told us he was sexually abused by the volunteer groundskeeper, who was also a volunteer coach, at the local cricket club when he was 10. BXI described how he felt when he was first 'touched inappropriately':

The rest of the day I was in shock, I was almost frozen. The whole day felt surreal. It seemed like the world had just stopped. I had no idea what it meant. I didn't fully understand what had happened, and was confused and had a lot of 'what ifs' and why would someone do that to me. I felt like I didn't want to tell anyone what had occurred, I didn't want to be the boy who cried wolf and be scrutinised by the community.⁸⁹

In Case Study 57: Nature, cause and impact of child sexual abuse in institutional contexts (Nature, cause and impact of child sexual abuse) we heard from Ms Karen Menzies, lecturer in Indigenous Health, University of Newcastle, about the complexities of shame experienced by Aboriginal and Torres Strait Islander survivors. O Ms Menzies told us that for Aboriginal and Torres Strait Islander victims: 'The shame moves from being a sense of embarrassment to being a very internalised sense of feeling that they are a complete failure or that they are somehow unworthy'. Like many survivors, Aboriginal and Torres Strait Islander children were encouraged to internalise responsibility for the abuse. However, for Aboriginal and Torres Strait Islander survivors, the experience of shame was felt at both individual and community levels, whereby shame was cast against the whole community. Bringing them home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families (Bringing them home) explains how authorities and 'carers' dismissed allegations of sexual abuse, labelling Aboriginal people in general as promiscuous and immoral. In our community consultations we heard from women who had been sexually assaulted as children. One woman told us, 'It felt as if being Aboriginal was a disgusting thing and we were only "good for" being abused'.

Research describes how shame is also a collective legacy stemming from the violence of colonisation and the use of sexual violence to collectively humiliate, degrade and disrupt sophisticated social relations between Aboriginal and Torres Strait Islander peoples.⁹⁴

Guilt and self-blame

Many survivors coming to us expressed complex feelings of guilt and self-blame. Guilt refers to the feeling of having done something wrong or having failed in an obligation.⁹⁵ Of survivors who spoke about impacts in private sessions, 16.4 per cent described feelings of guilt. This is consistent with research that suggests that feelings of guilt and self-blame are common among both male and female victims of child sexual abuse.⁹⁶

For some survivors, guilt and self-blame were as associated with disclosure of sexual abuse as with the abuse itself. As children, some survivors were told by the person who abused them, or a person they disclosed to, not to speak about the abuse to anyone. This was a powerful influence on whether they felt in some way responsible for the abuse. Others disclosed, only to be branded a troublemaker. As research suggests, perpetrators of child sexual abuse can frequently convince the child that they are to blame for letting the abuse happen.⁹⁷ Victims may feel additional guilt if they were made to perform sexual acts on the perpetrator or another child.

Clinical reports offer explanations for why some male victims have such strong feelings of guilt and self-blame. Male victims who responded physiologically with pleasure or arousal during the sexual abuse may conclude that they had enjoyed and sought the experience. Male victims who identified positive aspects associated with the abuse – such as affection – may believe that, as children, they had instigated or desired the experience. In other cases, male victims may blame themselves for not being able to defend themselves against the perpetrator or child with harmful sexual behaviours.⁹⁸

We heard from a number of survivors in private sessions and case studies who felt intense shame, guilt and self-blame because they had responded physiologically with pleasure or arousal when they were being sexually abused. For example 'Fletcher' told us in a private session that he was sexually abused by his music tutor at school. He told us, 'It all happened so quick'. Afterwards, 'Fletcher' remembered 'fear, guilt, all sorts of terrible feelings. And I blame myself, because I enjoyed it'. Some survivors also told us that because they experienced feelings of pleasure during the abuse, they were confused about whether what they had experienced was abusive. This confusion can act as a barrier to disclosure (see Volume 4, *Identifying and disclosing child sexual abuse*).

Other survivors told us they felt guilty because they did not defend themselves physically at the time. Giving evidence in *Case Study 35: Catholic Archdiocese of Melbourne*, BVD told us:

As I got older, I began to question myself: 'How could you be so stupid? Why didn't you bash him?' All of these things were going through my head ... Then I became ashamed and embarrassed. Self-loathing is a very powerful and draining emotion that has consumed me for the majority of my life.¹⁰⁰

Survivors also described feelings of guilt associated with whether they reported the sexual abuse, and feelings of worry or anxiety about whether the perpetrator or child with harmful sexual behaviours went on to sexually abuse other children.

Some survivors told us they felt guilty about disclosing sexual abuse to their families, particularly their parents, because of the distress caused. For example, one survivor said he believed his mother's suicide was directly related to his disclosure as an adult, and so he did not tell anyone else about the abuse until he spoke to the Royal Commission.¹⁰¹

Victims can also feel guilty for reporting the sexual abuse because of their close ties to the perpetrator, child with harmful sexual behaviours, or the institution itself. Some survivors told us about their feelings of guilt and betrayal for reporting the abuse because of their close ties to the religious institution where the abuse took place. For example, in *Case Study 29: The response of the Jehovah's Witnesses and Watchtower Bible and Tract Society of Australia Ltd to allegations of child sexual* abuse (*Jehovah's Witnesses*), BCB told us:

I have huge feelings of guilt about coming forward with my story. I feel as though I am betraying the Jehovah's Witnesses and bringing reproach on to them. When I speak to officers at the Royal Commission, my chest gets tight, I have heart palpitations, and I have difficulty breathing because of my anxiety about the betrayal.¹⁰²

Anger

A number of survivors told us that anger and rage dominated their emotions. This anger could be directed at the perpetrator or child with harmful sexual behaviours, or at the institution, but it could also manifest in family and work life.

BXI, giving evidence in the *Sporting clubs and institutions* case study, described the 'endless rage' he felt as a result of being sexually abused by the volunteer groundskeeper and cricket coach between the ages of 10 and 14:

By the time I was 18 years old, I had an incredible amount of rage inside of me. At that point in my life, I was getting into a lot of fights for no apparent reason. I put it down to possibly needing to vent frustration. It started to get to the point where I wanted to physically hurt Bob [the alleged perpetrator] and I had started to put a plan together of what I was going to do to him, which I found quite disturbing. It made me angry that I was in this position. Instead, I put everything into Rugby League ... Sometimes my uncontrolled aggression spilled into my footy for all the wrong reasons. ¹⁰³

Of those survivors who told us about impacts in private sessions, 20.4 per cent spoke about aggression as an outcome of abuse. More male (26.4 per cent) than female survivors (9.3 per cent) discussed aggression as an outcome. This is consistent with research that suggests men may be more likely than women to manifest externalising symptoms of trauma, such as anger, aggression and 'acting out', whereas women may manifest internalising symptoms such as anxiety. However, in our private sessions and case studies, we also heard stories of manifestations of anger, aggression and rage in some female survivors. In *Case Study 26: The response of the Sisters of Mercy, the Catholic Diocese of Rockhampton and the Queensland Government to allegations of child sexual abuse at St Joseph's Orphanage, Neerkol (St Joseph's Orphanage, Neerkol)* we heard evidence from AYB who said she was groomed and regularly sexually abused from the age of 11 by Father Reginald Durham at the Neerkol presbytery and other locations. AYB said:

My husband, children and extended family were also violated by Father Durham. They have suffered so much by his betrayal. There were many times they bore the brunt of my anger and rage, which saw me turn into a screaming, yelling madwoman, consumed with anger, shame, resentment and so much worthlessness. My inner being was cut to the core. My physical, emotional, spiritual and psychological wellbeing was shattered. 106

Recurrent feelings of anger could be triggered in adulthood by contact with the institution when survivors were pursuing redress or participating in legal proceedings. Although anger may have been a self-protective coping mechanism for survivors, ultimately it could be unhelpful, harming their relationships with others. We also heard from prisoners in private sessions how their anger and rage had contributed to their committing violent offences (see Section 3.8.3).

Grief, sadness, emptiness and loss

Grief, sadness, emptiness and loss were also frequently mentioned as emotional responses to child sexual abuse. In particular, adult survivors expressed great sadness at the loss of their childhood, and continued to grieve for who they could have been if not for the abuse.

We heard in a private session from survivor 'Lisa Michelle', who told us she suffered childhood physical and sexual abuse. She said that she regrets the 'huge loss of potential' the most. 'Lisa Michelle' said it gives her some solace to know that her own children got a good education and have gone on to succeed in life, but the feeling is bittersweet. 'I'm really happy about that but I'm sad for the young girl that I was because I think I could have been something unbelievable'.¹⁰⁸

For many victims, the enduring effects of child sexual abuse have meant that they never feel 'at peace'; they are tormented by memories of the abuse or suffer the emotional effects of the abuse every day. Others spoke about how they were 'dead inside', emotionless or unable to experience joy or love.

Survivor 'Grant Lee' told us in a private session held in prison that although he has a 'beautiful' partner who he knows loves him, he feels that he cannot love her. 'I don't feel love at all. It's like, I don't feel love against anybody though'. He told us how he ended up in a youth detention centre at the age of 16, after a turbulent childhood moving around various foster homes. We heard that 'Grant Lee' was befriended at the centre by a pastor, who then sexually abused him a number of times at the centre and his house. As he grew older, 'Grant Lee' said he got into more serious trouble with police, and ended up in adult prison, where he was drugged and raped by his cellmate. 'Grant Lee' told us he can feel his mental health declining again. He feels numb, and can't trust anyone: 'My cellmate reckons I'm cold-hearted, I've got no emotion there inside he reckons'. 109

These emotional effects have left victims lonely and isolated, and have affected their ability to form and maintain intimate and affectionate relationships. The impact of child sexual abuse on intimate relationships is explored further in Section 3.3.1.

Some survivors described the sadness of 'losing' their childhood self. For example, Mr Scott Hallett told us in *Case Study 43: The response of Catholic Church authorities in the Maitland – Newcastle region to allegations of child sexual abuse by clergy and religious (Catholic Church authorities in Maitland – Newcastle)* of his repeated sexual abuse by a priest when he was an altar boy. He said that the priest 'stole his innocence' and that 'once a child is broken, they can't be fixed'.¹¹⁰

Similarly, survivor 'Dalton John' told us in a private session, 'I feel that I lost most of my childhood to the hands of abuse. I feel deeply regretful and saddened that I will never be able to get back the years I lost ...'111

3.2.2 Depression

Emotionally I don't feel close to anyone and I feel like I am dead and emotionless on the inside. I am currently suffering from depression and post-traumatic stress syndrome. I am on a disability pension. My doctor has diagnosed the cause of these illnesses as Pastor Frank's abuse.¹¹²

Depression was one of the most common psychological impacts survivors described in private sessions, and survivors and families mentioned it numerous times when giving evidence in public hearings. Of survivors who spoke about impacts in private sessions, 44.0 per cent had suffered from depression.

The link between depression and child sexual abuse is supported by research, which describes it as both a short- and long-term outcome of abuse. For example, a longitudinal New Zealand study found the prevalence of depression for victims of penetrative abuse was approximately double that for people who had not experienced abuse. It is also reflected in research

on the effects of child sexual abuse in institutional contexts.¹¹⁵ One study found that children in foster care who were sexually abused self-reported substantially higher rates of depression (67.6 per cent) than children in foster care who did not suffer maltreatment (25.3 per cent).¹¹⁶

Survivors in private sessions and public hearings described their depression as a 'disease' and a 'burden'. One parent described in a private session his daughter's depression: 'She went to sleep and woke up a different person the day after she was abused, and will never be the same again. She's suffered depression, she's been suicidal'. 119

In Case Study 13: The response of the Marist Brothers to allegations of child sexual abuse against Brothers Kostka Chute and Gregory Sutton (Marist Brothers), AAJ told us that the sexual abuse he suffered at the hands of Brother Kostka prevented him from having an enjoyable and normal life as a teenager, a young man and older adult.¹²⁰ He spoke about his depression:

Sometimes it would be deep, dark and black, and I couldn't get out of it, and I'd go and get help. I had this anxiety, which was a real problem to me, panic attacks, the waking in the morning with a sense of dread and apprehension.¹²¹

Depression was frequently experienced by survivors in addition to other mental health disorders, such as anxiety and PTSD.¹²² This confirms research which indicates that where trauma plays a significant role in the development of depression, this results in post-traumatic depression — a state of depression intertwined with fearfulness, anxiety, and a high level of reactivity.¹²³

3.2.3 Anxiety

Research suggests that child sexual abuse in institutional contexts, and in all settings, may be associated with anxiety¹²⁴ and the development of specific anxiety disorders.¹²⁵ In line with this research, of the survivors who discussed impacts with us in a private session, 35.1 per cent told us about anxiety. They also told us about common symptoms of anxiety including panic attacks, hyper-vigilance and sleep difficulties.

Some survivors said they developed specific anxiety disorders, such as obsessive compulsive disorder (OCD), resulting from the sexual abuse. For example, 'Ernst' told us in a private session that his OCD manifested in a desire to feel clean:

It's the guilt. I felt guilt and disgusting. I'm a chronic hand washer you know, I think it's probably 'cause I felt dirty all my life. I shower two or three times a day and my partner says to me, "Why do you have to shower all the time?" I do that because - I don't know why, because I felt dirty I guess. 126

Anxiety can be completely debilitating for victims, affecting their ability to function in a normal environment. As one survivor told us in a private session: 'I fell to pieces. I couldn't sleep. My blood pressure was up. I was taking sleeping pills, Valium and anti-anxiety medication. I was a wreck'.¹²⁷

'Sophia', who told us she was sexually abused as a teenager by a Catholic priest on a number of occasions, described how she becomes anxious when she loses control – for example when she is lost:

I have an obsessive need for control. I try and control every aspect of my life because the one time I lost control, look what happened. I get incredibly anxious if I get lost in the car or something, because I don't have control. I always need to keep it together.¹²⁸

Panic attacks were also frequently mentioned by survivors in private sessions, sometimes manifesting as specific panic disorders. 'Alanna', who was born into a strict Catholic family in regional Queensland in the late 1970s, told us she was sexually abused between the ages of six and 11 by a young Catholic priest who had befriended her parents. What started as an 11-year-old girl's terror and confusion grew into the complex mental illness that 'Alanna' has carried her whole adult life:

For as long as I can remember I have been painfully sad, I have suffered from major depression all my life, post-traumatic stress, anorexia, debilitating anxiety, panic attacks ... self-harming, sexual dysfunction, confusion, traumatic nightmares, phantom body pain and general malaise. I often have thoughts of suicide.¹²⁹

Anxiety can stay with a victim throughout their adult life, affecting their relationships and employment. Mr Graeme Frazer, giving evidence in *Case Study 40: The response of the Australian Defence Force to allegations of child sexual abuse (Australian Defence Force)*, told us he suffered bastardisation and sexual and physical abuse as a junior recruit at the Navy's HMAS *Leeuwin*, a training school in Western Australia in the 1960s. He told us he believes the abuse led to his social anxiety disorder, which makes him extremely uncomfortable in groups of people. This affected his employment prospects after he left the defence force:

My anxiety was so severe that job interviews were distressing and it took 17 interviews before I was accepted for a position as a social worker. My anxiety causes me severe distress in social situations, and therefore I try to avoid them. Examples of positions I have had to relinquish because of my social anxiety disorder include an Honorary Fellow at a university college in Hobart and a Bench Justice at Hobart Magistrates Court. I also had to cease a doctorate at the University of Tasmania as it required communicating with many people. 132

3.2.4 Post-traumatic stress disorder (PTSD)

PTSD was the most common mental health diagnosis that survivors spoke about in private sessions. It was also mentioned frequently by victims in public hearings.

This is in line with research that shows PTSD as the most commonly reported impact of child sexual abuse. ¹³³ For example, one meta-analysis reported that child sexual abuse victims have a 143 per cent increase in the risk of developing PTSD, compared with the general population. ¹³⁴ Although reported rates of PTSD for victims of child sexual abuse in different settings (intrafamilial, institutional and other extra-familial) appears similar, some research suggests the dynamics of abuse experienced in institutional settings may increase both the likelihood and the severity of PTSD. ¹³⁵

As discussed in Section 3.1, PTSD describes the development of characteristic symptoms after exposure to one or more traumatic events. These can include:

- intrusive re-experiencing and flashbacks
- hyper-arousal and reactivity (such as hyper-vigilance and sleep disturbance)
- numbing and persistent avoidance of stimuli associated with the traumatic event
- negative alterations in cognitions and mood (such as an inability to recall important aspects of the traumatic event, or persistent negative self-beliefs or emotional states).¹³⁶

These symptoms can cause daily torment to many victims throughout their adult lives. For example, as 'Claire Grace' told us:

I wonder what would have been my potential if I wasn't abused. What could I have achieved if I was not burdened or tormented daily by post-traumatic stress? To not wake up in the morning crippled by an inferiority complex, anger, rage, hurt, sensitivity, blame, paranoia ... To sit comfortably in the moment, to feel at peace.¹³⁷

Re-experiencing aspects of the traumatic event or subsequent outcomes is seen as the hallmark symptom of PTSD.¹³⁸ This includes intrusive and recurrent recollections of the event, recurrent distressing dreams, and a sense of reliving the experience, for example, through flashbacks.¹³⁹ Many survivors in private sessions spoke about these flashbacks.

'Noeline' told us in a private session that she began to experience flashbacks to her childhood sexual abuse after she started seeing a psychologist to deal with the effects of a sexual assault she suffered as an adult. She said, 'It was like a little movie in my head that I couldn't stop, all these horrible smells and senses. I ended up fainting ... I couldn't shut the door on it at all, it was non-stop'. 140

Other survivors described hyper-vigilance. 'Trina Beth' told us she finds herself on high alert much of the time and, because of this hyper-vigilance, she is often exhausted and stressed.¹⁴¹ Another survivor, 'Roy Leonard' told us he has ongoing issues with anger and paranoia and is hyper-vigilant about the people his nieces and nephews spend time with. He said:

If I had any inclination or any thought that something bad was happening to them like happened to me I'd become very defensive and I would lash out. I would say that would be a part of what happened to me. It would make [me] very angry, very angry. 142

As discussed in Section 3.1, some researchers suggest the complexity of symptoms exhibited by many victims of child sexual abuse cannot be explained by a diagnosis of PTSD, and suggest that 'complex PTSD' or 'complex trauma' is a better diagnosis. ¹⁴³ Some survivors told us in private sessions that they had complex PTSD.

3.2.5 Eating disorders

We heard in private sessions and public hearings from some survivors who had developed eating disorders – such as anorexia, bulimia and overeating – in the years after they were sexually abused. Eating disorders can play a significant role in attempts to regulate traumarelated distress. Anorexia (weight loss caused by self-starvation) may be one way for a victim to feel more in control. On the other hand, bulimia (binge eating often associated with purging) may be an extreme attempt at distraction from unbearable emotional distress. Like alcohol abuse and other drug abuse, eating disorders can be highly damaging to health.

Some survivors in private sessions told us they developed eating disorders during childhood, and it was only as adults that they realised this may have been part of their response to the sexual abuse. We heard that 'Marjorie', who said she was sexually abused daily by her teacher from kindergarten to Grade 3 during the 1980s and began to 'comfort eat' during her childhood as a coping strategy: 'I remember just saying, "Why do I eat the way I do, why do I have so many issues with my weight?"' As no one, including her mother, acknowledged the abuse, 'Marjorie' found it hard to recognise the effects it had. 'Obviously, the way that I lived my life showed that something had happened, but I wasn't aware of what the trauma effects are on a child'. ¹⁴⁶

Another survivor, 'Charlene', told us that after being abused by her Grade 1 and 2 teacher she developed an eating disorder in her teens. This developed into anorexia, which persisted for some years. She then became morbidly obese in her early 20s. 'When I can't cope with things, I think ... I'm a big fat cow; if I wasn't a big fat cow I'd be alright'. Since then, 'Charlene' has been admitted to hospital for self-harm, has overdosed and has undergone electroconvulsive therapy. Although she is on medication, she still struggles with anxiety and panic attacks. Her mother and her counsellor provide support, and she is treated by a psychiatrist.¹⁴⁷

Other survivors told us about eating disorders that manifested later in adult life, often occurring alongside other psychological problems such as anxiety or depression. One survivor, AOE, gave evidence in *Case Study 20: The response of The Hutchins School and the Anglican Diocese of Tasmania to allegations of child sexual abuse at the school* about being sexually abused at school in the 1960s: by the Headmaster in Grade 6; by his music teacher in either the end of Grade 6 or during Grade 7; and then by his physical education teacher in Grade 8 or 9.¹⁴⁸ He described a range of impacts, including an eating disorder:

I have ... lost all self-respect and have eating disorders. On the day I met with officers of the Royal Commission it would have been six weeks since I've eaten a proper meal and I've been vomiting from anxiety over the preceding weekend.¹⁴⁹

The association between child sexual abuse and eating disorders apparent in survivors who attended private sessions and public hearings is consistent with research on the impacts of child sexual abuse.¹⁵⁰ The link between child sexual abuse in institutional contexts and eating disorders was also noted in submissions to the Commission to Inquire into Child Abuse in Ireland.¹⁵¹

3.2.6 Other diagnosed disorders

Aside from PTSD, and eating disorders, some survivors in private sessions spoke about other mental health disorders which research has associated with child sexual abuse. ¹⁵² These include:

- mood disorders, such as bipolar disorder
- psychotic disorders, such as schizophrenia
- personality disorders, such as borderline personality disorder
- dissociative disorders.

However, it is important to note that while early life trauma such as child sexual abuse is a clear risk factor for the development of mental health disorders (or 'psychopathology') later in life, it is not the sole factor. An individual's vulnerability to developing these disorders is made up of several other risk factors, including genetics.¹⁵³

Research suggests that victims of child sexual abuse may be at particular risk of developing borderline personality disorder (BPD),¹⁵⁴ which is characterised by strong feelings of insecurity and impulsivity.¹⁵⁵ People with BPD often have difficulty relating to others, and have a distorted and unstable self-image. One study indicated that BPD was found seven times more often among female victims of child sexual abuse than among the general population.¹⁵⁶

We heard from a number of survivors in private sessions who have been diagnosed with BPD. Many experienced it in addition to a range of other mental health issues.

'Ryan Ross' was placed into a Salvation Army boys' home in South Australia when he was 12. There, he told us, he was sexually abused by two older residents when he was 14. He told us that he has sought counselling since the early 2000s. He has been diagnosed with paranoia, schizophrenia, anti-social personality traits and BPD. 'Ryan Ross' has an acquired brain injury. He suffers from nightmares, sometimes has suicidal thoughts and has self-harmed.¹⁵⁷

Dissociative disorders have also been linked to experiences of trauma, including child sexual abuse. Dissociation involves altered states of consciousness in the face of overwhelming stress. These altered states can take many forms, ranging from feeling 'spacey', unreal or even outside your body, to periods of amnesia and abrupt shifts in identity. For some, dissociation may be a diagnosed disorder. Dissociation

Dissociation can be a way of coping with the stress of sexual abuse, and is a common survival mechanism for children who suffer trauma. As psychiatrist Professor Herman has described, children faced with the terror of abuse may exhibit a 'frozen watchfulness', where they are quiet and immobile, despite being in a constant state of hyper-arousal. ¹⁶¹ The younger the child, and the less chance they have to fight or flee a situation, the more likely that they will employ dissociative responses. ¹⁶² For example, 'Lloyd Edward' told us in a private session that he was sexually abused by Father 'Carlisle' in the early 1970s when he offered to give 'Lloyd Edward' some treatment for his damaged legs. He recalled:

I was too scared to flinch or say anything so I justified to myself that it must be necessary and it would end soon ... I cannot remember everything as I seemed to go into a trance where I disconnected from what was happening and I only recall snippets of what happened. 163

Professor Herman has pointed out that dissociating 'offers a means of mental escape at the moment when no other escape is possible, it may be that this respite from terror is purchased at far too high a price'. There is evidence that children who enter a dissociative state at the time of the traumatic event may be among those most likely to develop long-lasting trauma reactions. The second state of the traumatic event may be among those most likely to develop long-lasting trauma reactions.

A number of survivors in private sessions told us they had experienced dissociation and dissociative disorders, including dissociative identity disorder (DID), throughout their lives. People who have DID – formerly called multiple personality disorder – experience two or more distinct personality states that repeatedly take control of their behaviour. 166

'Dion' told us that he was seeing a counsellor and found this helpful in managing the dissociative episodes he sometimes experienced. 'Sometimes I'm just not there. I can disappear from myself for a couple of days. It's not a nice thing to have to live with. It's scary. It's very scary. I just did the best I could'. 167

Dissociative disorders overlap with PTSD, which can include dissociative symptoms. For example, flashbacks, experienced by those with PTSD, are dissociative states that involve alterations in consciousness, memory, identity and perception. However, unlike PTSD, exposure to a traumatic event is not necessary for a diagnosis of dissociative disorder, although trauma is a common cause of such a condition. ¹⁶⁸

3.2.7 Sleeping difficulties

Sleeping difficulties are a common effect of child sexual abuse, including in institutional contexts, both as an immediate response and a long-term problem. Victims told us of sleeping difficulties caused by persistent nightmares, 'night terrors' (that is, feelings of extreme terror when asleep, with dramatic presentation), hyper-vigilance, fear and anxiety. Of the survivors who discussed impacts with us in a private session, 13.5 per cent described nightmares. As discussed in Section 3.2.4, nightmares and other intrusive thoughts are recognised as symptoms of the hyper-vigilance that is a marker of PTSD.

Nightmares can cause victims to experience extreme distress and panic. As 'Edna' described:

For years, I've had recurring nightmares where I feel like someone is sitting on my bed and leaning over me. It's so real I can feel their breath on the back of my neck. I wake in a panic too scared to move, with my heart racing, and I'm trying to scream but no sound will come out. For more mornings than I care to count, I just stand in the shower and let the tears fall.¹⁷⁰

Some survivors told us that the way in which they were sexually abused was a direct cause of anxiety-related difficulties with sleep. Survivors who were placed in residential institutions told us that they, and other children, were sexually abused during the night, either being taken out of bed or assaulted in their beds.¹⁷¹ For some, the fear associated with this affected their sleeping patterns, which persisted into adulthood. Giving evidence in *Case Study 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland (The Salvation Army boys' homes, Australia Eastern Territory)*, one survivor, EP, described the effects of being taken out of bed and sexually abused by older boys:

You were on the defensive all the time; you were on the lookout all the time. You could never sleep through the whole night, you'd lie there waiting for someone to come and get you; even now I still can't sleep.¹⁷²

Two other survivors giving evidence in this case study told us they still had constant nightmares or they experienced screaming and thrashing in bed.¹⁷³ One said he had to close his bedroom door to sleep. Another said he could not sleep in the same room as his partner and had learned to sleep lightly because of his fear of being attacked while he slept.¹⁷⁴

As a result of their difficulties sleeping, victims told us they were prescribed sleeping tablets and self-medicated with alcohol and other drugs. 'Renny' told us in a private session that he was sexually abused repeatedly by a teacher in Grade 3. At the age of 12 he discovered that smoking cannabis helped control his terrifying nightmares and flashbacks. 'Renny' said he left home when he was 17 years old, specifically so he could increase his intake of alcohol and other drugs to help dull his panic attacks, anxiety, flashbacks and nightmares.¹⁷⁵

3.2.8 Self-harm

Research has consistently found an association between child sexual abuse and self-harm.¹⁷⁶ Of those survivors who discussed impacts in a private session, 13.0 per cent of female survivors and 7.1 per cent of male survivors spoke to us about self-harming episodes, in particular physically injuring themselves in the years immediately after the abuse or following a trigger event. Self-harm can take many forms including cutting, burning, banging the head and ingesting.¹⁷⁷ In the *Catholic Church authorities in Ballarat* case study, BAS told us that as an adult he had self-harmed by scrubbing himself with steel wool to 'scrub away the man'. It was only going to a men's group that got him 'out of there'.¹⁷⁸

In Case Study 10: The Salvation Army's handling of claims of child sexual abuse 1989 to 2014, JH gave evidence that her daughter was sexually abused when she was eight or nine years old by a captain in The Salvation Army. From that time, her daughter started to pick at her face and arms. 'I have photographs prior to 1989 and after and you can see by looking at them that she had started to damage herself'.¹⁷⁹

While knowledge is growing about the risk factors that increase the likelihood of children self-harming, much less is known about why and how children do so. Self-harm is distinct from attempting suicide, as it is intended to provide temporary respite from emotional pain rather than bringing permanent escape through death. Self-harm can have multiple meanings for victims, including self-soothing, punishing oneself, dulling the emotional pain, distracting from the overwhelming effects of emotional pain or being used as a call for help. It has been described as a way of coping with feeling numb or suffering intense pain, distress or unbearably negative feelings, thoughts or memories and can become habitual. Survivors reflected on these feelings in private sessions. Lucy' described in a private session what compelled her to self-harm:

When this distress started to become overwhelming, then I would have this 'I've got to cut myself, I've got to cut myself and let the poison out and I'll be okay'. So I did do some self-harming. I didn't cut myself — well, actually I did, but it wasn't on my arms. Your rational mind is saying, 'No, no, no, you don't need to do this to yourself, this is ridiculous, you can't be doing this', but then ... it's so overwhelming I thought, 'Okay, so I'll control it, I'll press things into myself to give me the pain without the injury', it was very clever. So I used to press forks and the ends of pencils and stuff into my flesh to snap me out.¹⁸⁴

While self-harm may help victims regulate unbearable emotional distress, it can end up being counterproductive, as it can exacerbate or precipitate the painful emotional states it seeks to mitigate. 185

3.2.9 Suicidality

We heard about a number of victims of child sexual abuse in institutional contexts who had attempted to take their own lives or had died by suicide. Thoughts of suicide and attempted suicides were some of the most common outcomes of child sexual abuse described in private sessions. Of those survivors who described impacts in a private session, 19.8 per cent told us they had suicidal thoughts and 16.4 per cent said they had attempted suicide. We also heard from family members or others, who told us about victims who had taken their own lives. These victims represent just under one per cent of all private sessions where information about impacts was provided.

Research suggests a link between child sexual abuse across all settings and suicidal ideation, suicide attempts and actual suicide. 186 For example, one longitudinal study showed that 'exposure to child sexual abuse is related to clear increases in the risks of later mental health problems', including suicide attempts, although adverse factors in childhood, such as physical abuse, problematic parent-child attachment and parental history of illicit drug use may also have had an influence. 187 Another study of Australian high school students showed that 61 per cent of victims of child sexual abuse had suicidal thoughts compared to 21 per cent of students who had not been victims of child sexual abuse, and 28 per cent had attempted suicide. 188 According to another study, victims of child sexual abuse are 18 times more likely than people in the general population to die as a result of suicide and accidental fatal overdose. 189 As discussed in Chapter 1, it is difficult for researchers to distinguish between outcomes that have emerged from a person's experiences of child sexual abuse and those that stem from other influences in that person's life. These difficulties are as complex, if not more so, for research on suicide. For example, Dr Margaret Cutajar, Senior Psychologist at the Victorian Institute of Forensic Medical Health, explained in the Nature, cause and impact of child sexual abuse case study that most victims in her research had taken their own lives a considerable length of time after the sexual abuse. This means that there are potentially more influences in those people's lives that need to be taken into account when trying to determine causal factors for the suicides. 190

Some survivors in private sessions and public hearings told us they had attempted to take their own lives in the years immediately after the sexual abuse. Some survivors contemplated taking their own lives as young as age nine or 10. For example, 'Adeline' told us in a private session:

I remember drinking some poison out of a bottle in the shed and I also ... there was a fire going at the back of our home and I walked through that, and I had third-degree burns. It was like I just ... I didn't fit in ... and I just tried to end my life so many times and so it just went on and on ...¹⁹¹

Some survivors told us that the effects of child sexual abuse they suffered as young children stayed with them, leading to attempts to end their lives in their teenage years. For example, 'Simeon Paul' told us he was sexually abused by his primary school teacher when he was nine years old. He said of the abuse, 'The after-effects are devastating. I've lived a half-life since then'. He told us, 'I've been suicidal since I was 19'. 192

Another survivor in the *Jehovah's Witness* case study, BCG, described the overwhelming nature of her feelings that led to her attempt to take her own life in her teenage years. She described being sexually abused by her father, who was an Elder at her congregation of Jehovah's Witnesses, and taking part in a traumatic interview with the Committee of Elders about the abuse:

During my teens, I was at times depressed and suicidal and this became worse after my father's sexual abuse of me while my mother was away at Expo'88. I also attempted suicide several months after the Committee Meetings in 1989 as a result of my experience of the Committee Meetings with the Elders. I couldn't bear the judgment of those around me, the public vilification and ostracism. I wanted to dig a hole and die. 193

Survivor BAP, who gave evidence in the *Catholic Church authorities in Ballarat* case study, did not realise his behaviour was suicidal until later in life. He described his risk-taking behaviour, which he now realises were attempts to take his own life:

At 19, I lost my driving licence because I had been drinking and driving. One day I drove very fast down Eureka Street in Ballarat. I drove about halfway, hit a telephone pole, lost control of the car and crashed through the front fence of a house. The ambulance and police could not believe that I lived through such a crash. The only injury that I got was some glass in my head. They said the only thing that saved me was that I didn't have my seatbelt on and was thrown into the back seat. The front of the car was crushed. It took me a long time to realise that these accidents were actually suicide attempts. I think I was trying to be heard. My mum said to me, 'BAP you've got nine lives, you must be a cat'. I couldn't even tell her why I was doing these things. 194

We also heard from parents and other family members about victims who had taken their own lives at a young age. In some cases it was unclear to them why the child had taken their life until later, when other evidence about sexual abuse by the same perpetrator came to light. For example, Mrs Audrey Nash spoke in the *Catholic Church authorities in Maitland – Newcastle* case study about the suicide of her son Andrew at the age of 13, which she told us she now believes was because he had been sexually abused at Marist Brothers Hamilton, New South Wales:

When Brother Romuald was charged in 2013 I read a media article that mentioned that he had sexually abused some boys at Bar Beach. Reading that article brought back to mind the incident that occurred about six months before Andrew committed suicide, when he came home late and told me he had been there ... Many of the boys that Brother Romuald was convicted of sexually abusing were in Andrew's class at school, and I have no reason to

believe that Andrew wasn't targeted by Brother Romuald, just the same as the other boys who were sexually abused. In February 2013, a school friend of Andrew's wrote to me saying he believed Andrew had been sexually abused at Marist Brothers Hamilton, and that is why Andrew committed suicide ... I now believe that Andrew was sexually abused and that he took his own life because of the abuse.¹⁹⁵

We also heard of many attempted suicides and suicides later in life, after years of suffering the impacts of child sexual abuse. Research shows the risk of suicidal behaviour for victims of child sexual abuse may be greater at certain points in the life cycle. For example, one Australian study on the risk of suicide and victims of child sexual abuse showed that death as a result of suicide predominantly occurred when the victims were aged in their 30s, similar to the peak time for suicide in the general population. For some victims, it may be that the prospect of certain life changes leads them to consider taking their own life. For example, older survivors told us they feared being re-institutionalised, dependent and vulnerable when they grow older, because for them these are powerful triggers for the sexual abuse they experienced in their childhood (see Section 3.8.1).

'Clusters' of suicides were reported in some locations where many children had been sexually abused by one or more perpetrators over years. For instance, giving evidence in the *Catholic Church authorities in Ballarat* case study, witnesses said they knew a number of victims of child sexual abuse at St Alipius in Ballarat in the 1970s who had taken their own lives. ¹⁹⁸ One witness showed the Commissioners a photograph of 33 boys in his Grade 4 class at St Alipius in 1974, of whom 12 were dead. He believed suicide to be the cause of their deaths. ¹⁹⁹

These suicides affected others who were sexually abused during the same time.

The suicides have left an impact on me, too. I have been and continue to be impacted by the grief and loss of losing other victims of [the Brother] to suicide, some of whom were my family, brothers and cousins. I have had to bury people as a result of the trauma caused by these crimes. I have had to go to funerals where the person took their own life. I knew they were childhood sexual assault victims of [the Brother]. I live with this knowledge and grief every day.²⁰⁰

Similarly, we were told in the *Catholic Church authorities in Maitland – Newcastle* case study that several former students at Marist Brothers Hamilton had taken their own lives, which a witness believed was because of the sexual abuse they had suffered at the school.²⁰¹

How child sexual abuse has affected communities is discussed further in Chapter 5, and in Volume 16, *Religious institutions*.

3.2.10 Alcohol abuse, drug abuse and gambling problems

Survivors told us they have used alcohol and other addictive substances to manage the effects of child sexual abuse, leading to their overuse. Alcohol abuse was one of the most common mental health and wellbeing problems described by survivors in private sessions. Of all survivors who spoke to us about impacts in a private session, 31.1 per cent told us about alcohol abuse, and 27.8 per cent spoke about illicit drug use. A small number told us they had overused prescription medications at some stage of their lives.

Research has long suggested a relationship between child abuse and subsequent alcohol and substance abuse. ²⁰² Using population-based studies to compare to the general population, some research suggests that child sexual abuse victims may be up to eight times more likely than non-abused people to develop alcohol-related and/or drug-related disorders. ²⁰³ Research also suggests that these problems have an earlier age of onset in sexual abuse victims. ²⁰⁴ One study found that female victims were 88 times more likely than the general Australian population aged 15–64 to die from an accidental drug overdose, and male victims were 38 times more likely to die from an accidental drug overdose. ²⁰⁵

The abuse of alcohol or other drugs may be a problem more common among male victims.²⁰⁶ One Australian study on the outcomes for sexually abused children found that generally males were more likely than females to abuse alcohol or drugs. However, it also found that child sexual abuse was a particularly strong risk factor for females abusing alcohol. The rate of alcohol abuse for female victims was almost nine times higher than for females in the community, compared to 3.4 times higher for male victims.²⁰⁷

In private sessions and public hearings, victims told us how alcohol and other drugs were used as a way of coping with the pain of the sexual abuse in their early years. In the *Australian Defence Force* case study, CJV told us that ever since his second year at Balcombe Barracks, Victoria, when he was 16 years old, he started to drink heavily. He told us this helped him to cope with the effects of the abuse.²⁰⁸ CJV still drinks heavily today.²⁰⁹ We also heard from Mr Angus Ollerenshaw in *Case Study 23: The response of Knox Grammar School and the Uniting Church in Australia to allegations of child sexual abuse at Knox Grammar School in Wahroonga, New South Wales,* who told us, 'I have suffered great amounts of mental pain that I tried to cover up with, at times, drugs'. He started to take drugs when he was 13 in 2005 and moved into 'more heavy drugs' at 15, continuing to use drugs on and off until late 2013.²¹⁰

Some victims spoke about starting to use alcohol and other drugs in their teenage years as part of 'acting out' or rebelling in response to the sexual abuse. For example 'Zahara' told us in a private session, 'I did the rebellious teenager and did the drugs and alcohol and early teen pregnancy'.²¹¹

Victims also continued to rely on alcohol and drugs to cope with the impacts of sexual abuse into adulthood, well after they left the institution. Giving evidence in the *Bethcar Children's Home* case study, AIH told us that after she left Bethcar, 'I turned to alcohol to solve my problems and to make the pain go away; however, it just got worse and worse'.²¹² She went on to say, 'I trust nobody because adults in my life have taken advantage of me, and the only time I can cope with life is when I am drunk or using drugs'.²¹³

Some victims said that the perpetrator introduced them to alcohol and other drugs – such as tobacco – when they were children, as part of the grooming process or to facilitate the abuse. In one private session, we heard that after this type of experience the victim then used alcohol to cope with the abuse. 'Gabriel' told us the perpetrator offered him a drink when they were alone on a weekend 'posting' in cadets. He described how the perpetrator gave him Scotch whisky, which made him feel sick. 'Gabriel' then lay down on one of the beds. The next thing he remembers is waking to find the perpetrator and another man standing over him. He said they then raped him. Thereafter, his life 'spiralled into using alcohol'.²¹⁴

However, a number of victims said that while alcohol and other drugs were sometimes helpful to numb the pain in the short-term, excessive use of alcohol and drugs had affected them negatively later in life. Many victims told us that their excessive use of alcohol or other drugs led to a breakdown of relationships with others (including with their parents, partners and children), physical illness, unemployment and criminal behaviour. For example, in the *Catholic Church authorities in Maitland – Newcastle* case study CNQ told us:

I have had difficulty with relationships and have been through two failed marriages. I have trouble with intimacy and I am often uncomfortable with sexual relationships. Usually these issues, combined with my alcohol and substance abuse problem have resulted in my relationships breaking up.²¹⁵

'Odie', who told us he was sexually abused by religious sisters and a priest at an orphanage, experienced troubled years after he left the institution. He lived on the streets, was a heavy user of drugs and alcohol and found himself in court on multiple occasions. Over time, he could see a pattern. 'Odie' said, 'It's like a cycle. Every three or four years I just break down. And I've always treated it with drugs and alcohol'. 216 In 2006, he got drunk and went to a party where he ended up stabbing someone. It proved a watershed moment:

Alcohol's played a major role in my life ... I've always woken up the next day with someone else's blood on my hands. It's never mine. I was getting to the point where I was actually fearing that I'm going to kill somebody one day. I decided then that drugs and alcohol's not for me. I'm sick of that way of life. I needed to change. It was either that or I'm going to end up in an early grave.²¹⁷

Survivor 'Norm' told us in a private session that, on turning to drugs as a result of abuse, '... I lost who I was, and this was the thing: I became somebody that I didn't really want to become'. ²¹⁸

We also heard from some survivors, mostly men, about the connection between child sexual abuse and the development of gambling problems later in life. In a submission to the Royal Commission, Mr Mark Griffiths, psychologist and Chair of the Clinical Advisory Board of Survivors & Mates Support Network (SAMSN), described how some victims use gambling to cope with the effects of sexual abuse, with many developing compulsive gambling habits:

When the groups for male survivors started in the early 90s it was unusual to hear that participants had gambling problems. However, over subsequent years it has evolved that every group will include men who have experienced problems with compulsive gambling. Much like drug use, survivors use gambling to temporarily calm themselves, in this case diverting their attention from their internal struggles and emotional distress. A client once told me that when he was in a TAB he experienced 'an overwhelming feeling of peace'. When I asked him what he meant he told me, 'when I am in there focusing on the screens, and on my bets – that's the one time I don't think about the abuse'.²¹⁹

3.3 Interpersonal relationships

Survivors spoke to Commissioners about the difficulties they have experienced with interpersonal relationships, including with intimate partners, other family members and friends. Consistent with what we were told, research associates child sexual abuse with negative impacts on social and interpersonal relationships.²²⁰ Perhaps this is to be expected given that the abuse often occurs in the context of a close, personal relationship where a sense of trust and safety has been established. Child and adolescent psychiatrist, Dr Bruce Perry, giving evidence in the *Nature*, cause and impact of child sexual abuse case study, explained how child sexual abuse in an institutional context makes a victim less willing to trust others, which impacts on their capacity to establish and maintain relationships:

Everything that we learn, the entire way we heal, is all in the context of relationships, and if the very vehicle that we use to learn, to grow, to develop, to heal, to engage, to teach — all the stuff that we do as a parent — if that is corrupted by the process of sexual abuse and you can't trust and you don't feel safe, it makes it difficult for you to make your way through life. If the institution — you know, these institutions are either ignoring, colluding, rejecting or even attacking you because of your disclosure, and you see this and you feel it, it makes you much, much, much less willing to trust.²²¹

Compounding this loss of trust, the secrecy and sometimes the fear associated with hiding the abuse from others create a sense of guilt, shame and confusion that can disrupt the way the child views the world.²²² In adulthood, this worldview affects the way that survivors understand the motives of others and how they handle stressful life events.²²³ This may strongly influence their relationships with others throughout their lives, especially intimate partners.

Despite the pervasive negative effects on personal relationships, many survivors in private sessions and public hearings, told us about the importance to them of close relationships with parents, partners, siblings, children, extended family, and friends. As discussed in Chapter 2, they drew on supportive relationships to help them cope, even if the supporters did not know about the abuse or only found out later. Many reflected on how their relationships with others had given them meaning, and helped motivate them to seek healing and recovery. These sources of strength are also discussed in more detail in Volume 5, *Private sessions*.

3.3.1 Intimate relationships

Difficulty in forming and maintaining intimate relationships was a pervasive impact described to us by many survivors. Of survivors who discussed impacts in private sessions, 24.2 per cent described having relationship problems and 45.2 per cent spoke of having difficulties with trust and intimacy. Many survivors who gave evidence during public hearings also identified these problems.²²⁴ For example, in *Case Study 30: The response of Turana, Winlaton and Baltara, and the Victoria Police and the Department of Health and Human Services Victoria to allegations of sexual abuse (Youth detention centres, Victoria)*, one survivor told us, 'I am currently married, but I have very poor intimacy abilities. I don't enjoy having sex; it feels functional and awkward. And, although my marriage is strong, my wife and I struggle with the intimacy'.²²⁵

Research supports what we have heard from survivors of child sexual abuse about the difficulties they have experienced functioning in intimate relationships. ²²⁶ Studies suggest survivors of child sexual abuse are less satisfied with their current intimate relationship or marriage, less likely to marry and more likely to divorce than people who were not sexually abused as children. ²²⁷ In evidence to other government inquiries, victims of institutional child sexual abuse have described lifelong isolation and loneliness, even in the presence of others or in the context of supposedly happy relationships. The inability to trust, a sense of profound anger and powerful feelings of guilt and shame frequently compromised their relationships and were described as an ongoing barrier between them and others. ²²⁸

Consistent with research, survivors in private sessions and public hearings often described what they saw as a connection between their lack of trust stemming from the child sexual abuse and their inability to sustain long-term intimacy. Some victims have been unable to have committed relationships at all. As one male survivor said, he avoids relationships because '... if I don't get close to people, they can't hurt me'. ²²⁹ Similarly, BQG told us in the *Brisbane Grammar School and St Paul's School* case study, 'I don't trust anyone and I don't let people in or let them get close. I have a fear of being abandoned. I have had a lifetime of hollow relationships'. ²³⁰

The stress of coping with the impacts of the sexual abuse – including mental illness and other negative outcomes – also takes its toll on intimate relationships. Some survivors told us that as a result of these difficulties their relationships broke down completely. For example, in the *Catholic Church authorities in Ballarat* case study, BAB said:

I think that my actions of holding on to this internally have actually led to a psychological barrier, which now means that my marriage is almost collapsing. I think it's become more debilitating the longer it's gone on. I have withdrawn more and it is to a point where it is really affecting my relationship with my wife.²³¹

The impacts of child sexual abuse are sometimes realised only when a victim enters a committed relationship. 'Vittorio' told us that it was only when he was in his 20s that he finally recognised that sexual abuse by a priest when he was an altar boy had affected his life. He realised he wasn't close to his family, women or people in general. 'Vittorio' didn't have serious relationships until he met his partner when he was in his 40s. The relationship became rocky and they went to see a counsellor, where the issue of the sexual abuse surfaced. Although he and his partner remain friends, the relationship ended. 'Vittorio' told us, 'It just hasn't been a good part of my life, the last four years ... Sometimes I can cope with it, sometimes I can't. I still haven't got a lot of friends and I spend most of my time alone'. 232

We also heard how a survivor's ability to form and maintain intimate relationships can be related to difficulties with physical intimacy – issues which stem from the sexual abuse when they were children. Survivors in private sessions and public hearings told us that they experienced a range of sexual problems or dysfunction that they attributed to the impacts of the sexual abuse they experienced as a child, and which affected their relationships. These included difficulties with arousal and avoidance of, or fear of, sexual intimacy. Other sexual impacts, such as having sex at an early age, unprotected sex, multiple partners, and hypersexuality and engaging in sex work, are discussed separately in Section 3.5.3.

Research has suggested a relationship between child sexual abuse and a victim developing sexual problems or sexual dysfunction.²³³ One analysis of research on the impacts of child sexual abuse in intra-familial and other extra-familial settings identifies that 'while around a third of female victims in community samples report problems of sexual dysfunction, almost two-thirds of clinical samples of female victims report problems'.²³⁴ Studies of institutional sexual abuse suggest that sexual dysfunction may be a common impact for male victims of historical child sexual abuse in religious institutions.²³⁵

Some survivors in told us in private sessions that they avoided sex or were fearful of sex. Of the survivors who discussed impacts with us in a private session, 6.7 per cent told us that they avoided or were fearful of sex. Female survivors (9.4 per cent) were slightly more likely than male survivors (5.2 per cent) to describe avoidance or fear of sex. A smaller percentage of survivors, female (2.0 per cent) and male (1.9 per cent), told us they experienced difficulties with sexual arousal.

Some survivors told us they avoided sex because they feared triggering memories of the sexual abuse. This avoidance in turn affected their relationships. For example, 'Francesca', told us that after she was sexually abused by a female camp leader when she was in primary school, consensual sexual activity would trigger her into feeling as though she was being abused. This reaction affected her relationships. 'Sexually, it had a huge impact, most definitely. And that was a factor that led to the breakdown of my marriage'.²³⁶

Avoidance behaviours were not limited to female survivors of child sexual abuse. For example, 'Cedric' told us that he was accustomed to shutting down when something sexual was happening to him. He found that 'whenever I got to a point of sexual intimacy my libido would freeze and I could not consummate'.²³⁷

In the *St Joseph's Orphanage, Neerkol* case study, we heard from AYE who said, 'I struggle to be intimate with my wife. I can't stand anybody putting their arms around me, or even touching me. We don't even kiss. That is what the priests did to me and that is all I can think about'.²³⁸

BQF told us in Case Study 34: The response of Brisbane Grammar School and St Paul's School to allegations of child sexual abuse (Brisbane Grammar School and St Paul's School), 'I have never married. The sexual abuse has definitely been a catalyst. Whenever I become intimate, I just have flashbacks all the time. I have mates, but I guess I'm not close to them'.²³⁹

In another example, 'Eliza' told us in a private session:

I know this sounds awful and I feel bad, but I used to fake it with my husband, do you know what I mean? I wanted to be normal. I wanted to be normal and have kids. I always knew there was something wrong with me 'cause I didn't really have a sex drive.²⁴⁰

Sexual difficulties can also affect people's ability to have children. One survivor, who was sexually abused by an Anglican priest when he was 13, told us in evidence in the *Church of England Boys' Society* case study:

Child sexual abuse by priests ... made me question my sexuality, and made normal marital sexual relations impossible. I was traumatised by communal showers and toilets, something that remains embedded today. One of my two children had to be conceived by IVF as a result of my anxiety and distress. I am conscious of the toll that this has had on my wife and family, especially my children.²⁴¹

However, despite difficulties with intimacy being common among survivors, many victims also told Commissioners they have had satisfying and supportive long-term relationships, often citing their partner as their main support in helping them to cope with the effects of the abuse.

3.3.2 Parenting

Research suggests that child sexual abuse may affect the relationships that victims form with their children and their parenting behaviours. One study found that victims of child sexual abuse identified greater negativity in their relationships with their children and were less confident about their parenting ability compared with parents who had not been sexually abused as children. However, both groups derived the same level enjoyment from their relationships with their children. This suggests that the impact of sexual abuse on parenting may relate to parental confidence and communication skills.

Another perspective on the effects of child sexual abuse on parenting emphasises how the effects of trauma can impair survivors' parenting capabilities. One literature review suggested that sustained and overwhelming trauma in childhood has been shown to affect some women's experiences as mothers, despite their best underlying intentions to protect and nurture their children. For example, mental health issues, intimate partner violence and substance abuse, which may be outcomes of child sexual abuse, are understood to substantially affect parenting.²⁴⁵

Some survivors spoke in private sessions about the effects of child sexual abuse on their ability to parent, and on their views of parenting. Of the survivors in private sessions who discussed impacts, 13.5 per cent told us about their discomfort or lack of confidence with parenting, with slightly more females (17.8 per cent) than males (11.2 per cent) describing this as an outcome.

Most common among survivors who are now parents was a concern that they would not be 'good parents'. Some survivors who became parents said they were unable to show love or affection to their children. 'Hany', for example, spoke in a private session about being sexually abused multiple times by three older boys at a Catholic mission school in the mid-1960s. He described how he has had difficulties maintaining relationships with partners, children and his wider family as a result of the sexual abuse:

my inability to communicate with my kids, to show the affection, to show the love for my children ... It's almost like your emotional process has just completely switched down. You don't want to talk about nothing. You go out the back and you keep quiet ... I've lost my family. I think that this has affected me, because of my feelings, I don't know how to feel. I do feel compassion, but as a survivor I don't know how to feel for a family. And so I've got family members that I don't even see. 246

In the Australian Defence Force case study, CJA told us the abuse he experienced made it particularly difficult to show physical affection to his child:

My son is an affectionate child, and often wants to be physically close, and while I have no problems with this, I sometimes find it uncomfortable and have often had to remove myself from the situation. He sees this as rejection. However, because of his age, I can't explain to him why I have this reaction to him. I often worry about the long term effect of my behaviour towards him.²⁴⁷

Some survivors of sexual abuse in out-of-home care told us how the emotional toll of the abuse contributed to estrangement from their children, replicating their own separation from their family when they were children. 'Hayley' told us that she suffered physical and sexual abuse, as well as neglect, at home and in her foster care placements, later battling drug and alcohol addiction and mental health problems. She surrendered her first two children to care: 'I learned it was perfectly acceptable, even by the state's standards, to abandon your children ... and that's exactly what I did'. Becoming pregnant again when she was a lot older pushed 'Hayley' to get off drugs and stop drinking. She sought help for her mental health issues and left her violent relationship. 'Hayley' now dreams of having her own home — 'a safe place' for her and her child.²⁴⁸

Another survivor, 'Casey', who told us he was physically and sexually abused by a houseparent in a home run by the Aborigines' Inland Mission, said he is 'not very close' to his son and doesn't want to tell him what happened in the home. He told us his son would say, 'You're not my father, you don't come and look after me'. 'Casey' said, 'But that's in here, I think, not [having] that mother or father love when I was a child. Just never had any'. He also said, 'That's why I feel sorry for my son, but I don't want to tell him what happened in [the home] ... I wasn't a proper father, a father figure. I don't know how to be a father figure'.²⁴⁹

Some survivors described becoming overly protective and anxious when their children reached the age they were when they were first sexually abused. As 'Ramona' said, 'The worst thing I think I've done to my son is [that] I didn't treat him like a kid. I always treated him like an adult – I didn't want him to be a kid to get abused like me'.²⁵⁰

Some survivors in private sessions raised similar concerns about their ability to grandparent. The intergenerational impacts of child sexual abuse are discussed in Chapter 5.

A number of survivors spoke about a common myth that victims of sexual abuse will become perpetrators of child sexual abuse. They told us that they worried they might sexually abuse their own or other children because they had been sexually abused themselves, although they had no other reason to believe their children were in danger. For that reason, some said they avoided being physically affectionate with their children or engaging in nappy changing or bathing, and others said they had decided not to have children. This is consistent with research in which male survivors described themselves as being overprotective and nervous about physical contact with their children and fearful of becoming sexual abusers themselves.²⁵¹ 'Landon', who told us he was sexually abused at a boys' home in the 1970s, described how this fear affected the way he related physically to his son:

Every one of my mates has got a photo of them and their little boy in the shower – holding his boy in the shower or holding him up and he's wet, you know what I mean. I was robbed big time. I haven't got a photo like that where I can say, "That's me and my boy". That's a normal Daddy shot and I haven't got it. I was too scared to get it.²⁵²

'Dane' told us in a private session that when his first child was born, memories of the sexual abuse he experienced as a teenager in a government-run children's home came flooding back to him:

What happened, that set off post-traumatic stress disorder related to the sexual assault. Now, I didn't know that at the time ... I started to feel really uncomfortable when I was giving my daughter a bath, that type of stuff. ²⁵³

'Dane' worried that he may sexually abuse his daughter – even though there was no indication he may do so – 'because I'd heard this type of thing in the media, about survivors becoming offenders'.²⁵⁴

'Rosswell' told us he was first sexually abused when he was eight years old. 'Rosswell' married in his early 20s. He and his wife divorced 20 years later and didn't have children. He told us, 'We both had our own demons on that'. 'Rosswell's' fear, one that he knew he could 'rationally discount' was that he might become a perpetrator himself.²⁵⁵

Ms Andrea Lockhardt, a qualified social worker ²⁵⁶ and Senior Counsellor at the Ballarat Centre Against Sexual Assault (CASA), ²⁵⁷ gave evidence in the *Catholic Church authorities in Ballarat* case study about this common fear among male victims:

I'm running a support group, or co-facilitating ... and there's 18 to 20 men who haven't gone on to abuse others, we wouldn't be working with them. But it's such an impact. I remember one man who said his psychiatrist told him that he would definitely go on to abuse, and so that impacted on his ability to bathe his children, to even hold them, and so there was such a grief and loss around that. I would say most of the men in the group have that fear, because people out in the community believe it, so it actually prevents them from disclosing too and seeking help. It silences them.²⁵⁸

However, despite the common myth that victims of child sexual abuse become perpetrators, research shows that the vast majority of victims do not go on to commit sexual offences in adulthood, including child sex offences (See Section 3.8.3).

3.3.3 Other relationships

114

The ability to form and maintain friendships and other relationships outside the family can also be affected by the lack of trust many victims experience as a result of child sexual abuse. Some survivors told us they lost friendships because people did not know how to respond when they disclosed the sexual abuse. Others spoke of feeling socially anxious, not knowing what to say in social situations and not feeling they belonged. Travis' said, it sort of shuts you down socially ... You become more of a, not so much a recluse, but tend to prefer your own company'. In the *Sporting clubs and institutions* case study survivor BXI told us he was sexually abused by the volunteer groundskeeper, who was also a cricket coach, at the local cricket club. He told us he couldn't be around men, because he was unable to trust them:

I have struggled with men all my life. I have trouble trusting men. The abuse robbed me of my ability to show love to my father. I have always suppressed my emotions and feelings towards him. I struggle to urinate in a public urinal if other men are present. I find it awkward to be around other men's penises so I try to avoid having showers in a locker room after playing sport, or try to be the last one to shower.²⁶¹

Low self-esteem, discussed in Section 3.2.1, also damaged victims' ability to maintain friendships. In a private session, 'Cecilia' spoke about the devastating effects that being sexually abused by a priest and a Scoutmaster has had on her ability to maintain friendships:

Once people become my friends, I work out that they don't like me and they're going to use me, and if they actually knew what I was like they would hate me. So I always break it off, I always sort of finish everything. I never have friendships last more than a year or two.

Still, 'Cecilia' has managed to maintain two close friendships over the last four years: 'The longest friendships I've maintained. Mostly because I'm starting to come to some understanding of what has happened'. ²⁶²

Some survivors told us about the lonely lives they have led as a result of the sexual abuse, some because they were unable to maintain relationships, and others because they felt they could not trust anyone. 'Ronald' told us in a private session that he was raped and physically abused in a Salvation Army boys' home and now leads a lonely life, despite work being important to him, and having a psychologist he can now trust. 'I know I'm lonely. I don't trust anybody, but I open that door [at home] and I know nothing can hurt me'. 263

3.4 Physical health

Although physical health was not discussed as frequently as many other impacts, a number of survivors in private sessions and public hearings spoke about having adverse physical health effects of child sexual abuse, including:

- direct physical consequences of abuse, such as injuries and sexually transmitted infections
- symptoms, illnesses and injuries that they developed over their lives, as an indirect effect of the sexual abuse and their response to the abuse.

Research shows a relationship between adverse childhood experiences, including child sexual abuse, and poor health outcomes.²⁶⁴ A landmark epidemiological study in the United States found a relationship between adverse childhood experiences, such as abuse and family dysfunction and some leading causes of death, such as ischemic heart disease, cancer, chronic lung disease, skeletal fractures and liver disease, as well as poor self-rated health (the study did not look specifically at child sexual abuse).²⁶⁵ The study also found associations with health risk behaviours, including smoking, obesity, alcoholism, risky sexual behaviour and intravenous drug use. Adverse childhood experiences were strongly interrelated and persons with multiple categories of childhood exposure to adverse experiences were likely to have multiple health risk factors later in life.²⁶⁶

Another study found that child sexual abuse victims are more likely to suffer a range of chronic conditions, including heart disease, osteoarthritis, chronic back and neck pain, and frequent or severe headaches.²⁶⁷ Other inquiries on child sexual abuse in institutional contexts have also noted its relationship with adverse health outcomes, particularly sleep problems, impaired or lost hearing, and muscular and skeletal issues.²⁶⁸

3.4.1 Direct health effects

Injury

Many survivors who gave evidence in public hearings and 3.9 per cent of survivors who described impacts in private sessions said they had received a physical injury as a direct result of sexual abuse. The physically violent nature of some child sexual abuse is described in Volume 2, *Nature and cause*.

Survivors told us about injuries related to sexual penetration, such as damage to the genitals and anus. For example, 'Rory' told us he was raped anally several times by a priest at his Catholic boarding school when he was 15 years old in the late 1980's. He said the final incident

was particularly violent and left him bleeding. He said the injuries required surgery, which was organised and paid for by the school. 'Rory' told us his parents were never consulted or told.²⁶⁹ In another example, Mr David Owen, giving evidence in the *St Joseph's Orphanage, Neerkol* case study, told us that he was sodomised by a priest at least twice a week for two to three years when he was a young boy in the orphanage. On a number of occasions he bled as a result and, when he told one of the religious sisters about the injuries, was given a nappy to wear.²⁷⁰

'Ramona' told us that she was sexually abused by her female primary-school teacher in the late 1970s when she was four years old. She said the perpetrator would also hit and penetrate her with a ruler. As a result of the abuse, 'Ramona' required medical treatment. 'She ruined me down there, I couldn't walk', she said. At the time, 'Ramona' told her mother and the doctor that she sustained these injuries when she was playing in the schoolyard. She doesn't know whether they believed this.²⁷¹

Some survivors described violent assaults during the sexual abuse that caused injuries, including bruises and bleeding, to other parts of their bodies, such as the head, mouth and ears. VG, a child migrant from Malta, gave evidence about his sexual abuse at the hands of a Christian Brother, Brother Simon, in *Case Study 11: Congregation of Christian Brothers in Western Australia response to child sexual abuse at Castledare Junior Orphanage, St Vincent's Orphanage Clontarf, St Mary's Agricultural School Tardun and Bindoon Farm School (Christian Brothers).* He described being taken one night to Brother Simon's 'den' and penetrated anally. When VG got free he said he hit Brother Simon with a chair. Brother Simon responded by violently assaulting him with a strap.²⁷² The attack left VG unconscious and he ended up in hospital.²⁷³

We heard evidence in public hearings of the close association between physical punishment and sexual abuse in certain institutions. In *The Salvation Army boys' homes, Australia Eastern Territory* case study, many of the former residents gave evidence of the devastating effects of physical and sexual abuse throughout their lives. HP said that during his time at Riverview Training Farm, he was tripped and kicked unconscious when he tried to escape an officer who was trying to have sex with him. He said he awoke to find the officer raping him.²⁷⁴ We also heard from FT who told us that when Captain X17 called him to an office to tell him of a death, he was sexually abused. FT said he ran from the room and into a toilet block. X17 then took him into an area between the locker room and shower room where he kicked him with 'these large army boots' he wore until FT fell to the floor and was further kicked.²⁷⁵

While some injuries – such as bruising – faded over time, other injuries have persisted into adulthood, leaving victims with long-term chronic conditions that require ongoing management. 'Sean Michael' told us in a private session that he was sexually abused as a child while living in a government-run residential care facility. This followed his experience of multiple forms of abuse during his time in his grandmother's care and at a residential detention centre. 'Sean Michael' said he estimates that he was raped seven times as a child, and this has caused ongoing physical injuries. He told us:

Every time I go to the bathroom I am reminded of each and every one of those rapes. It still hurts to this day to use my bowels, I am too afraid to have the operation to fix it, I am too afraid to be unconscious for fear of being hurt again ... I have bad spinal problems from the kicking and beatings across my back ... that my spine is bent and hurts all the time.²⁷⁶

In the Australian Defence Force case study, survivor Mr Daryl James told us:

I suffer numerous ongoing detrimental issues relating to the abuse including back and joint injuries I sustained from 'arseholing' which are chronic and debilitating. The injuries sustained from being kicked in the testicles have, I believe, contributed to chronic and debilitating pain. The piles that flare up remind me of the 'Crab Night' torture. Playing percussion instruments is too painful to consider and teaching triggers flashbacks. I had to treat the pain with trusses and medication and in February 1982, I had corrective surgery to treat the testicular and anal injuries that I suffered. I need ongoing treatment and further surgery on my groin, back, and joints.²⁷⁷

This is similar to the experiences of a number of victims, who told us that the sexual abuse had left them with long-term bowel and bladder problems and rectal dysfunction, sometimes requiring surgery. 'Bert' told us of the long-term physical effects that resulted from being sexually abused at the age of six by a religious sister at the orphanage in which he lived in the 1960s. When he told the sister-in-charge about the incident, and said that he was in pain, 'Bert' was told not to lie and to go away. In his private session, 'Bert' said:

The physical results of this abuse was that the testicle had been so severely damaged by squeezing it so very hard that it swelled up for weeks, and I assume it was ruptured at this time, because it then shrivelled up to the size of a pea and never grew normally like my other testicle. I have lived my life with one testicle permanently damaged.²⁷⁸

In another example, Mr Troy Quagliata told us in the *Sporting clubs and institutions* case study how he has suffered from anal bleeding on and off for most of his life after being anally raped by his cricket coach when he was a boy. His doctors told him it was a result of tearing to the internal skin.²⁷⁹

One survivor told us he has had penile dysfunction all his life because a male nurse at the children's home where he was sent at the age of 12 performed a circumcision on him without his consent. He said that the nurse gave him an anaesthetic but it was the wrong dose. He said the nurse ended up slashing and mutilating his penis with a blade while he was still awake, and that they rushed him to a hospital where his penis was sewn back together. He told us that he has not been able to have sex or children as a result.²⁸⁰

We heard in private sessions of the shame and embarrassment felt by victims with long-term injuries that were sustained from the sexual abuse. These injuries have remained as daily and permanent reminders of what happened to them as children.

Sexually transmitted infections

Some victims told us that they contracted sexually transmitted infections they believed were the result of the sexual abuse, including herpes, syphilis, genital warts, hepatitis C and HIV. 'Clive', who described abuse in a boys' home when he was aged 11, told us he discovered some years after his abuse that he had latent syphilis. At the time of the diagnosis he had only had one sexual partner, and as he was told he had contracted the disease some time ago, he believed the perpetrator was the source of the disease. 'Clive' recalled that while at the boys' home he was taken to hospital after developing a rash, which staff were worried could be 'something more sinister'.²⁸¹

BXA, giving evidence in the *Sporting clubs and institutions* case study, told us how she was raped by her soccer coach when she was aged eight. She was diagnosed as HIV positive shortly before her 15th birthday, and believes she contracted it from the perpetrator.²⁸² BXA told us about the ongoing effects of the HIV:

The sexual abuse that I experienced has had a massive, irreversible impact on my life. I live with a constant reminder of the abuse every single day of my life because of the HIV. Sometimes I want to forget that I have HIV and I have gone through times where I have not taken my medication, which I now know would be a really easy way to kill myself ... You can get really sick from the HIV medication itself, although I am more used to it now. I can still get a bad reaction whenever they change the dose or the type of medication. My liver has been affected and I have had really bad itchy eyes like with conjunctivitis. I have also had to have brain tests because the chemicals affect your brain.²⁸³

Pregnancy

Of all survivors who discussed impacts in private sessions, child sexual abuse resulted in pregnancy for 3.2 per cent of female survivors.²⁸⁴ In *Case Study 7: Child sexual abuse at the Parramatta Training School for Girls and the Institution for Girls in Hay (Parramatta Training School for Girls)*, we heard from survivor Ms Wendy Patton, who told us that another victim confided in her that she had become pregnant as a result of sexual abuse by a deputy superintendent.²⁸⁵ In the *Youth detention centres, Victoria* case study, Katherine X gave evidence that she was sexually abused by her father. She disclosed the sexual abuse to social workers before being removed from parental care, and also to youth workers and other department staff at Winlaton Youth Training Centre. As a result of the sexual abuse, Katherine X gave birth to four children by her father and had two miscarriages.²⁸⁶ We also heard in private sessions that a child may have resulted from the sexual assault of a male victim. 'Corey', born with disability and using a wheelchair, told us how his foster carers forced him, at the age of 14, to have sex with his foster mother so she would fall pregnant. 'Corey' believes she had a child as a result.²⁸⁷

In some cases, victims gave birth to their babies, who were often removed from the victims' care. For example, 'Maggie', who was sent with her siblings to a foster care placement on a dairy farm when she was five years old, told us in a private session that she became pregnant at the age of 14, after being raped on multiple occasions over seven years by three foster boys at the placement. She said she was taken to a Salvation Army home for unwed mothers where she gave birth to a boy and nursed him for six weeks before the staff approached her with adoption papers. She said that despite her strong objections, a short while later a family adopted 'Maggie's' son and took him away.²⁸⁸

Other female victims told us in private sessions and public hearings about miscarriages and abortions.²⁸⁹ In the *Parramatta Training School for Girls* case study, Ms Janet Young Mulquiney, who was sent to Parramatta Girls in 1970, told us:

There were miscarriages in there after girls had been punched by male officers and mistreated. It was all whispered secretly and all that sort of thing. We just knew that if a girl was in sick bay, that was the reason, basically – a pregnant girl.²⁹⁰

The experience of birth, miscarriage or abortion at an early age caused physical damage and trauma for some victims. For example, 'Dee' told us in a private session that she gave birth at about age nine as a result of rape while she was a state ward and living at a children's home run by Catholic religious sisters. She recounted that the sisters made her have a normal birth which 'broke all of my hips'. Afterwards, she was 'in plaster all the way down' because of these injuries and had to wear callipers.²⁹¹

For some victims, the sexual abuse has affected their ability to have children. Ms Mary Farrell-Hooker, giving evidence in the *Parramatta Training School for Girls* case study, said she suffered 10 miscarriages because of the stress, trauma and bashings she received as a state ward. She told us that her doctor said she had a lot of scarring in her vagina, and Ms Farrell-Hooker attributed this to the sexual abuse she suffered as a child.²⁹² Another survivor, 'Christine', was told she would never be able to carry a baby to term, because of the damage she suffered from being raped as a young child on an Aboriginal mission and at subsequent foster homes. Despite this prognosis, 'Christine' defied the medical advice she had received and gave birth to her first child at the age of 18.²⁹³ What we were told by these survivors is consistent with a large-scale epidemiological study of the impacts of child sexual abuse, which indicated an association between child sexual abuse and higher rates of miscarriage and stillbirth.²⁹⁴

3.4.2 Indirect health effects

In addition to the direct effects on physical health, research indicates that the cumulative and compounding effects of child sexual abuse indirectly contribute to poor health outcomes for many victims. Health problems for victims may stem from a complex interrelationship between behavioural, social, emotional and cognitive factors,²⁹⁵ including factors related to stress,

risky behaviours such as alcohol and other drug taking, inadequate education and economic insecurity (discussed below).

While it is difficult for research to distinguish the influence of child sexual abuse from many other influences on a person's life, some research links sexual abuse with a variety of illnesses and disease.²⁹⁶ There is also increasing evidence of the detrimental effects of stress on the brain. As discussed in Section 3.1, there is a growing body of knowledge about the impacts of trauma from neglect and child abuse more broadly on brain development. Emerging research suggests that child sexual abuse may lead to changes in hormonal stress pathways and brain structure, function and cognition. It suggests that brain regions responsible for functions including processing emotion, learned fear, and planning and organising tend to be smaller for victims than for people who have not been sexually abused, although there is some criticism of this argument.²⁹⁷ Research confirms that excessive or protracted exposure to traumatic experiences causes profound alterations in the regulation and functioning of many bodily systems.²⁹⁸

Psychiatrist Dr Louise Newman, giving evidence in the *Nature, cause and impact of child sexual abuse* case study, explained how the stress of child abuse can relate to disease:

The impact on the immune system is something that's becoming apparent, that particularly when we're talking about early abuse, when the immune system is literally being set up and established, if those core processes are disrupted, that, as far as we understand currently, can set up ongoing deficits in immune regulation so that the body is exposed to high levels of stress related hormones and is likely to have a less efficient immune system which increases the risk of a whole range of disease processes. We see that across whole communities, sometimes, who have experienced high levels of stress. You can actually see increased rates of things like hyper-tension, diabetes, chronic diseases related to stress and trauma, and I would imagine this is exactly the same mechanisms, as far as we understand, happening in the abused population.²⁹⁹

In private sessions, survivors described various physical health conditions they believed were indirect results of child sexual abuse, particularly sleeping difficulties, overeating, hypertension, kidney and liver disease, headaches and migraines, stomach and digestion problems, musculoskeletal pain, heart attacks, chronic fatigue syndrome, and dental problems. One survivor, AOA, in the *Nature*, cause and impact of child sexual abuse case study, told us:

I am an example of the increased likelihood of survivors developing inflammatory diseases. Over the last five years I have had single or multiple episodes of the following inflammatory conditions: excruciatingly painful cystitis, urinary tract infections and prostatitis; pelvic inflammatory disease, including crippling spasms of the pelvic floor; irritable bowel syndrome; asthma; bronchitis; sinusitis; arthritis; oedema and pneumonia.³⁰⁰

Some survivors told us they believed they developed health conditions because of the stress of child sexual abuse, as well as the stress of disclosing, reporting and coping with the abuse. 'Amber' has experienced ongoing issues with her immune system:

I started to get stomach ulcers when I was 12 and I ended up being in so much pain, I couldn't concentrate at school, as well as the pain in my stomach all the time. At the end of '92, I had a nervous breakdown and then my health just never recovered from that, I ended up getting chronic fatigue, I had glandular fever multiple times, and that took well into my 20s to – my immune system was so low from probably stressing so much, that, yeah, it did take a very long time to recover from all of that.³⁰¹

We were also told by many survivors in private sessions that their behavioural response to the sexual abuse – such as avoiding certain situations that reminded them of the abuse – had damaged their health, as well as affecting their behaviour on a daily basis. For example, 'Clement Gerald' told us that he is unable to urinate if there are other people present, and attributes this to being sexually abused by a priest when he was about eight years old and then later by a teacher at the age of 12. As a result, 'Clement Gerald' has been unable to give a urine sample when required from prison officers, which has resulted in a breach of parole.³⁰²

A number of survivors told us how they were fearful of medical and dental examinations because they reminded them of the sexual abuse, which meant they avoided health checks or other medical procedures. The nature of the sexual abuse has led to a number of female victims avoiding seeking medical attention for sexual or gynaecological health involving internal examinations. This was especially the case if the sexual abuse they experienced as children involved invasive examinations. One survivor, 'Edyth', said she has never been able to have a pap smear because of the sexually abusive internal examinations she and her sister were subjected to at the shelter where they were placed at the ages of 11 and 14 respectively. She told us: 'He made me lay on the table and internally examined me. I didn't even know about that sort of [thing] ... I've never forgotten. It was horrible. I've never had a pap smear in my life because of that'. 304

In the *Youth detention centres, Victoria* case study, Ms Gabrielle Short told us that the physical, sexual and psychological abuse she suffered in her 17-and-a-half years in state-run institutions affected her greatly and still does.³⁰⁵ Because of the forced internal medical examinations that were part of the sexual abuse she suffered, Ms Short has not had a pap smear for over 20 years. She said in evidence, 'I would rather risk dying from cancer than to have to go through another one of these procedures and the memories of it. Anything medically wrong with my private body part/organ, I will not seek medical attention for'.³⁰⁶

Male survivors also told us about their fear of invasive examinations. 'Gene' said he was sexually abused by a worker in a remand home at the age of 16. 'Gene' disclosed this abuse as an adult when he was required to have a medical assessment, which he found extremely difficult:

To this day, I do not like people, including doctors, placing their hands on me. I have recently not been able to have a prostate examination and will now need to have this procedure done under a general anaesthetic.³⁰⁷

This fear of medical examinations has also affected survivors' ability to access other types of health care, which can lead to serious health consequences. AWD, giving evidence in *Case Study 27: The response of health care service providers and regulators in New South Wales and Victoria to allegations of child sexual abuse (Health care providers and regulators, New South Wales and Victoria)*, told us that he was sexually abused by a medical practitioner when he was 13 years old. As a result of this abuse, AWD said he was unable to see male doctors. After he attempted to take his own life as an adult, he was taken to hospital, where a male doctor attended. AWD became extremely angry and uncontrollable. Years later, when he had a heart attack, AWD suffered extreme pain for several hours at home before seeking help, and self-medicated for as long as possible to avoid doctors and hospitals.³⁰⁸

Another survivor, who told us she was sexually abused by a dentist at the local hospital several times a year from the time she was five until she was 10, several times a year, described a lifelong fear and avoidance of dentists. This has led to serious health problems with her teeth: 'I'd say the dentist changed my life, but he most certainly took my smile away. Forever'. 309

Similarly, the shame of the sexual abuse has meant that some survivors – both male and female – have avoided having their injuries seen to by doctors, even decades after the abuse.

3.5 Sexual identity, gender identity and sexual behaviour

Child sexual abuse in institutional and other settings has been associated in research with a range of sexual impacts on victims, in both childhood and adulthood, including confusion about sexual identity, difficulties with sexual functioning, sexual disorders and various risky sexual behaviours.³¹⁰

Of those survivors who discussed impacts in private sessions, 24.2 per cent experienced sexual impacts. In this section, we describe what we heard about the impacts on sexual identity, gender identity and sexual behaviours in adolescence and adulthood.

Sexual difficulties, such as problems with arousal or aversion to, and fear of, sex, are discussed in Section 3.3.1.

Developmentally inappropriate sexual behaviours, which may manifest in childhood as a result of trauma, are discussed in Section 3.1.4.

3.5.1 Sexual identity

Child sexual abuse often occurs in close proximity to a child's first awareness of their sexual orientation and therefore may complicate the development of their sexual identity. A number of survivors in private sessions told us that child sexual abuse affected their understanding of their developing sexual identity. In particular, the abuse caused them confusion about whether they were gay, lesbian, heterosexual, or bisexual. Of all survivors who described impacts in private sessions, 11.1 per cent of males and 2.1 per cent of female survivors described confusion about sexual identity.

Children who are developing a heterosexual orientation and children who are developing a homosexual orientation can experience confusion about their sexual identity, although the experience may differ and be more complicated for victims who identify as gay or lesbian. Child sexual abuse may also erase a sense of any sexual identity among some survivors.

A limited amount of qualitative research with victims of child sexual abuse confirms what survivors told us in private sessions – that the experience of child sexual abuse had caused them confusion in terms of their sexual development.³¹¹ Uncertainty about sexual identity following a childhood experience of sexual abuse may be especially pervasive among male victims of male perpetrators.³¹² This is particularly relevant for institutional child sexual abuse as most of the child sexual abuse described to the Royal Commission has been by male perpetrators abusing male victims (see Volume 2, *Nature and cause*). In one study, male victims of sexual abuse by male clergy described feeling conflicted and confused about their sexual desires or behaviour and their sexual identity.³¹³

As discussed in Chapter 2 and in Volume 4, *Identifying and disclosing child sexual abuse*, male survivors in private sessions described how social and institutional attitudes towards masculinity and homophobia led them to query whether they were homosexual or were perceived as homosexual. Historically, male-to-male sexual abuse has been regarded as a subset of homosexual behaviour,³¹⁴ and male child sexual abuse victims have also been labelled as homosexual.³¹⁵ Research suggests that some victims may have internalised some of these homophobic beliefs, including men who were abused by Catholic clergy who now question their sexuality, fearing they somehow attracted men.³¹⁶

In the *Catholic Church authorities in Ballarat* case study, child and family Dr Carolyn Quadrio described how being sexually abused by a male perpetrator can affect the developing sexual identity of boys:

Young boys who have been abused by a male offender over a long period of time are often very confused about their sexuality. They often assume that, because a male offender's had sex with them over a period of time, that that means that they're homosexual, where they may not be, so they become quite confused about their sexual orientation.³¹⁷

A number of male survivors in private sessions and case studies told us that as children they were concerned that being sexually abused by a male perpetrator meant that they were homosexual, would become homosexual, or were going to be labelled as homosexual.

Uncertainty about sexual identity is particularly relevant for boys who were sexually abused in their adolescent years, when their sexualities were developing. AWC, who gave evidence in the *Health care providers and regulators, New South Wales and Victoria* case study, explained that at 15 years old he felt too traumatised and embarrassed to speak about being sexually abused by his male doctor. As the abuse was his first sexual experience, AWC became concerned that he would become homosexual, despite not wanting to be. Among other impacts, the abuse also affected his relationships with girlfriends and his acceptance of sexual advances from partners.³¹⁸

We also heard from some male victims of male perpetrators and victims' families about how the sexual abuse had made them intensely homophobic. In the *Australian Defence Force* case study, CJC told us, 'I have come to rationalise the abuse by believing my perpetrator was homosexual. As a result, I have developed a dislike towards homosexuals. I'm sure I would not have become homophobic had my abuse not happened'.³¹⁹

A survivor in a private session described how he engaged in heterosexual behaviour to 'prove' that he was not homosexual. 'James Peter' told us:

I was terrified that I'd become a homosexual so I became quite promiscuous as a teenager after that ... I wanted to prove to myself that I wasn't a homosexual. I was terrified that I was going to become a homosexual. I had no idea how it worked.³²⁰

If a child's body responded physically during the sexual abuse (for example, ejaculation), this was frequently a further cause of shame, embarrassment and confusion. A child may interpret this reaction as evidence that they are enjoying themselves. Research suggests this is internalised and has an impact on the child's developing identity.³²¹ For example, WP described to us in the *Perth independent school* case study the effects of the sexual abuse he suffered when he was 10 years old:

I have been through a phase of totally questioning my sexuality. In particular, having an erection while being sexually abused by a male made me question my sexuality for many years growing up. I consider that the abuse has made me a very confused and upset individual. I have deep issues and fears with becoming sexually involved with women and I think I always will.³²²

WP went on to say, 'I have constant feelings of guilt about what happened to me, and that it was my own fault. I often think that I must have done something wrong to deserve what happened to me'. 323

Like WP, many other victims told us that they had internalised the sexual abuse, deciding not to disclose or blaming themselves, and subsequently grappling with their sexual identity for many years. 'Mathew', who told us he was abused by a trainee priest from when he was aged four into his early teens, described the self-blame associated with confusion about his sexual identity: 'And that, I think, had a lot to do with dealing with guilt. "Well, why, why did this happen if I'm not a homosexual?" I like girls but guys seem to like me'. 324

Similarly, 'Ross William' told us in a private session that he was sexually abused by a priest when he was an altar boy in the 1970s. He said the abuse started at the age of eight and continued for 18 months, when it abruptly stopped. In the aftermath, his behaviour changed, and he became a confident 'clown' to protect himself. As he got older, the impact of the abuse entwined with confusion about his sexual identity. 'I had thought that because I was gay I attracted this behaviour from the priest so I was at fault. There was guilt, there was shame involved with being who I was, so I spent a lot of time trying not to be who I was'. 325

Although in private sessions and case studies male survivors described this uncertainty and confusion more often than female survivors, Commissioners also heard from female survivors who had been sexually abused by female perpetrators, and who questioned their sexual identity as a result. For example, 'Abby' told us, '... I was abused by someone of the same sex so there's that additional thing of confusion. I was thinking, does that mean I'm a lesbian? I think there's an additional stigma attached to it and it makes it harder to say something. People don't think women abuse'. 326

In addition, some survivors – both male and female – spoke about an absence of sexual identity, which they perceived as stemming from the sexual abuse:

I have no knowledge of my sexual identity. I recall a young girl who was sporty, no interest in dolls, who kept animals. I never developed – I don't recall in my life having sexual attraction to anyone. I never developed a sexual identity. At this point in my life, I don't know that I ever will.³²⁷

We heard that victims of child sexual abuse may be reluctant to disclose the abuse out of fear of being stigmatised. As discussed in Volume 4, *Identifying and disclosing child sexual abuse*, some social views related to gender and sexuality can make it more difficult for victims to disclose, ³²⁸ including negative beliefs, attitudes and stigmas about homosexuality. ³²⁹ In some cases, we heard that when victims did disclose sexual abuse by an adult or child of the same gender, victims were subjected to homophobic attack and vilification. We also heard that victims were hesitant to report the sexual abuse because of misconceptions that homosexual victims attract sexual abuse.

In a private session 'Alan Eric' told us that when he was in Year 4 he was sexually abused by Brother 'Barnaby' at the Catholic school he attended, and two years later by 'Mr McGuiness' before being sexually abused again in his early teens by Brother 'Pember'. He told us that prevailing mindsets about homosexuality were obstacles to disclosure for him. 'I haven't been a willing participant in a gay act ... but yet I think I'll be judged that way and I'll go to hell', he said. He told us he was also aware that anyone who looked gay might have the 'absolute Christ' flogged out of them because 'poofter bashing' was a 'common thing' back then. Even 20 years later, in the 1990s, his fear of being labelled 'gay' stopped him from disclosing to police, who contacted him during an investigation into the school.³³⁰

3.5.2 Gender identity

Gender identity refers to one's deeply held internal and individual sense of gender, ³³¹ whether a man or woman, both, neither or in-between. Gender identity can correlate with assigned sex at birth, or can differ from it completely.

A few survivors in private sessions spoke about the ways in which child sexual abuse disrupted, at a critical stage in their development, a process of understanding their gender identity. Confusion about one's gender identity, while potentially related to a child's understanding of their sexuality in some instances, may also manifest as a distinct issue, and for this reason is discussed separately to sexual identity in this chapter.

While there is limited research on the impacts of child sexual abuse on gender identity, several survivors in private sessions described to us how child sexual abuse disrupted their process of understanding their gender identity, causing confusion for a time.³³²

'Xanthe' told us that a school principal had sexually abused her at the age of 14, at the same time as she was starting to realise she was not comfortable living as a boy. She started to internalise the abuse, and this affected her identity:

And I made this really strong link to my gender variation ... that because I was 'wrong' as a person this is why it had happened to me. And if I was 'right' as a person that it never would have happened ... And I think at that point in my life, right when I was really starting to discover a different sense of gender about myself, that at that point that the pathway that I could have followed in terms of developing a much healthier sense of who I was as a person, I was shunted off onto another track.³³³

Similarly, we heard from 'Emory' about the effects of being sexually abused from the age of eight by a Catholic Brother who was visiting from interstate. 'Emory' grew up as a boy, but now identifies as gender-neutral, using the pronouns 'they', 'them' and 'their' to reflect this understanding.

'Emory' told us that the Catholic Brother groomed them and made them feel special. At the time, they did not disclose the abuse, but rather started to dissociate, refusing to be touched or held, and exhibiting 'strange and angry behaviour'. 'Emory' was sent to a child psychiatrist in Grade 4. That same year, 'Emory's' older brother showed them pornography. 'Emory' said:

When I became more aware of sexuality and the roles associated to this, I questioned not really my sexuality, but gender and that I must be a girl due to the abuse and what it had done. I didn't talk to anyone about it. I was confused about what 'chicks' do and what had happened to me must be related.³³⁴

3.5.3 Sexual behaviour

Research suggests that child sexual abuse can have impacts on the sexual behaviour of victims, in both childhood and adulthood.³³⁵ These impacts may be developmentally specific, manifesting at certain stages of development, and sometimes subsiding or emerging later in adolescence or adulthood in a different form.³³⁶ This section discusses what we heard about sexual behaviours that manifest in adolescence or adulthood, including early and unprotected sex, multiple partners and engaging in sex work. Sexual impacts that survivors told us had affected their intimate relationships are discussed in Section 3.3.1.

We heard that risky sexual behaviours, such as unprotected sex, can be related to other risky behaviours common among survivors of child sexual abuse. Australian population-based research indicates that both men and women who have experienced sexual coercion are far more likely to have drunk alcohol in excess of official guidelines, to have injected drugs, and to have had a sexually transmitted infection, all of which are factors linked with high rates of unsafe or unprotected sex.³³⁷ These factors can lower inhibitions and expose victims of child sexual abuse to further victimisation by impairing their defences and risk perceptions, and increasing their likelihood of being targeted by perpetrators.

A number of survivors told us that they started engaging in early and unprotected sex in their adolescent years, as a consequence of the sexual abuse they experienced as children. For example, 'Bridget' told us the sexual abuse she experienced in childhood caused her to start using drugs and 'acting out' sexually in her teenage years. At the age of 14, 'Bridget' started exchanging sex for money.³³⁸

Some female survivors described to us becoming sexually active at an early age, because that was the way that they had come to understand relationships with men. 'Alana' told us she was sexually abused by her uncle, who was also a Christian Brother, numerous times from the age of seven. She said, 'I was sexually active at 16 and I think part of that was because I believed that pleasing men involved sex'.³³⁹ For others, they felt that the sexual abuse led to them being 'over-sexualised'. For example, 'Rebecca' told us, 'In my 20s, I was over-sexualised from the

experience and then, as I got older, I just found I had too many hurdles to jump quite often to just get to that happy place with sex'. Engaging in early and unprotected sex led to teenage pregnancy and motherhood for some survivors.

We heard that some victims engaged in sex in their search for affection because of their experiences of abuse. In her private session, 'Jacinta' told us that she was placed into a Catholic girls' home when she was almost 14 where she was mentally, physically and sexually abused. She told us that she eventually ran away to live with her stepmother which was okay until her grandfather attempted to have sex with her. 'Jacinta' said that when she disclosed this to her stepmother, she slapped 'Jacinta' and their relationship broke down. 'Jacinta' lived on the streets for a while, and became a 'people pleaser'. She told us that she felt obliged to have sex with anyone who was kind to her, and became pregnant at $18.^{341}$

Similarly, in the Australian Defence Force case study survivor CJD told us:

I had several relationships after breaking up with CJK that contributed to my already low self-esteem. Due to the years of being broken down by CJK, I sexualised affection and had to be in a relationship. I had sex on every date I went on because I thought it meant that the man loved me. I could not be alone. I would rather have been with the wrong person than be by myself.³⁴²

In private sessions, some survivors told Commissioners that their experience of child sexual abuse continued to affect their sexual behaviour into adulthood. They talked about having unprotected sex and multiple partners, and engaging in sex work and other types of risky sexual behaviours that they believed stemmed from being sexually abused as children. In *Case Study 21: The response of Satyananda Yoga Ashram at Mangrove Mountain to allegations of child sexual abuse by the ashram's former spiritual leader in the 1970s and 1980s*, we heard from Ms Alecia Buchanan, who was sexually abused by Akhandananda on multiple occasions between about 1982 and 1986. She told us she was 15 when she was first sexually abused:³⁴³

Throughout my 20s and 30s, I had a series of unsuccessful relationships with partners which were marred by my own behaviours and emotional damage stemming from my years being raised in the ashram and being sexually abused. Sometimes I engaged in promiscuous and unsafe sex, which caused myself and partners I cared for great pain. ... I exposed myself to the possibility of sexually transmitted diseases and experienced a number of occasions in which I was convinced I was either HIV positive or infected in some other way ... I was taught the only way I could secure love, affection and validation as a person was to rely on and exploit my sexuality. There was no other reason I could fathom why anyone would love me.³⁴⁴

AOA spoke in the *Nature, cause and impact of child sexual abuse* case study about the sexual impacts of the trauma of child sexual abuse, which he continues to experience. He told us, 'I have been in a secure, monogamous and happy relationship now for 22 years, yet I experience ongoing confusion between sex and affection with close friends. When I am triggered into a trauma cycle, I have intrusive fantasies of wanting a man in me and I become obsessed with romantic fantasies about other people'.³⁴⁵

A few survivors spoke in private sessions about other sexual behaviours that they believe stemmed from their experiences of sexual abuse as children. For example, 'Declan', who told us he was sexually abused by the school principal in primary school, became intensely interested in pornography in later years, which he managed to overcome. He told us, '... I guess I developed a porn thing, and that was kind of really sad. I'm over that now. Now I've got some help and I take medication now and settled down'. 346

For some, their experiences of child sexual abuse led them to engage in aggressive sexual behaviour or hyper-sexual behaviour. 'Wayne Christopher' told us that as a teenager he was confused about his sexuality and acted out. He describes himself as being an 'exhibitionist', who displayed multiple sexually inappropriate behaviours. 'I feel I was railing against the idea of being perceived as gay, and striking back in the only way I could. I had very little understanding of what I was doing, and very little control over it'.³⁴⁷

A small number of survivors told us in private sessions that they exchanged sex for money or other things at various times in their adolescence and early adult years, either as a one-off or occasional experience, or in other cases as a professional arrangement. Research suggests that some survivors of child sexual abuse go on to exchange sex for money, although it is clear that many factors besides the sexual abuse leads a victim to do so.³⁴⁸ One study examining the criminal consequences of child sexual victimisation by using official arrest records suggests that victims of child sexual abuse are at specific risk of arrest for prostitution as adults.³⁴⁹

We heard that sex work was the only means of survival for some victims without homes or income, and a way of funding the addictions that were a response to the trauma of child sexual abuse. As BQS told us in the *Brisbane Grammar School and St Paul's School* case study:

I have struggled with alcohol and drug addiction since 1987. I have spent time in a residential rehabilitation facility on a number of occasions. I have had approximately 30 admissions to hospital over the years as a result of my addiction and have been diagnosed with liver disease. In the past, on occasion, I engaged in male prostitution to fund my addiction. I find this ironic, given what happened to me as a child.³⁵⁰

In the *Parramatta Training School for Girls* case study, Ms Yvonne Kitchener described leaving Parramatta Girls at the age of 18 without any more government assistance and being 'dumped at the People's Palace Hotel in Sydney', which was 'a cheap hotel run by the Salvos'. Within a week of leaving, she ended up in sex work and with a drug habit. Ms Kitchener said in evidence:

When I arrived at the People's Palace Hotel, I immediately went to Kings Cross. I developed a big drug habit and worked as a prostitute. I didn't even want to be alive. I just thought, 'Well, what's the point? I've got nobody. It's just me. I'm not rehabilitated to fit into the outside world after being raped, bashed and God knows what else, and had my pride and sanity taken from me'.³⁵³

She also told the Royal Commission that 'it was a known fact that all the Parra girls went to the Cross. There was nowhere else for any of us to go'. 354

3.6 Connection to culture

Child sexual abuse in institutional contexts can also have cultural impacts. During this inquiry, we heard in particular from Aboriginal and Torres Strait Islander survivors and some survivors from culturally and linguistically diverse backgrounds, including child migrants, about the effects of child sexual abuse in an institutional context on their knowledge and practice of culture.

These cultural impacts included loss of connection to culture through physical separation or alienation from family, community and country. This frequently included deliberate denigration of culture and direct, institutional control of cultural practices. We also heard how community shame related to sexual abuse led some survivors to reject, or be rejected by, their community, which in turn further alienated them from their culture.

Volume 5, *Private sessions* illustrates further the experiences of Aboriginal and Torres Strait Islander survivors and survivors from culturally and linguistically diverse backgrounds.

3.6.1 Aboriginal and Torres Strait Islander experiences

The survival of Aboriginal and Torres Strait Islander cultures demonstrates resilience and adaptability. The songlines of Aboriginal and Torres Strait Islander peoples predate colonisation by tens of thousands of years. In this time Aboriginal and Torres Strait Islander cultures successfully managed trauma and traumatic events that affected communities. At colonisation, Aboriginal and Torres Strait Islander peoples were 'considered to be much healthier than many of the British people who arrived suffering a raft of afflictions including malnutrition, infectious diseases, alcoholism and violence'. However, the forced institutionalisation and widespread abuse, including sexual abuse of Aboriginal and Torres Strait Islander children has had profound cultural and collective impacts that have been transferred across generations. The song transferred across generations.

In private sessions, we heard from many Aboriginal and Torres Strait Islander survivors how the forced institutionalisation, and their maltreatment within that institution, including sexual abuse, intentionally disconnected them from family, culture and Country. In *Case Study 17:*

The response of the Australian Indigenous Ministries, the Australian and Northern Territory governments and the Northern Territory police force and prosecuting authorities to allegations of child sexual abuse which occurred at the Retta Dixon Home (Retta Dixon Home), survivor Mrs Lorna Cubillo gave evidence that she could not be initiated into tribal ways as a result of her institutionalisation and abuse, and her grief at the loss of her cultural heritage. The institutionalisation and abuse, and her grief at the loss of her cultural heritage. The institutionalisation and abuse, and her grief at the loss of her cultural heritage. The institutionalisation are survivor who gave evidence in the Retta Dixon Home case study, spoke of being disconnected from her Aboriginal identity and heritage, having been completely separated from her mother's family. The Parramatta Training School for Girls case study, survivor Ms Mary Farrell-Hooker recounted how being institutionalised affected her Aboriginal identity: she was isolated from her culture and lived between two different worlds. Took her six years to find her parents after being released from the institution, and she had trouble fitting back into her family. The bond had been broken between me and my mother. We were never mother and daughter again'. The bond had been broken between me and my mother.

Some survivors told us that, in addition to being sexually abused, any cultural connections they had were deliberately destroyed. In one private session, 'Carl' described being physically punished for practising his culture at the Methodist mission where he lived: '[At the mission] each time you spoke your language you were flogged. Any time you did something cultural, meaningful for yourself, you were flogged'. 'Carl' told us he was sexually abused by the cook at the mission, which continued until he was 10 or 11.³⁶² He said, 'We were there to learn, we were there to become Western, we were there to become Christian, and we were there to lose all our nativeness'.³⁶³ The abuse in this context is one of many harms experienced over generations that are part of the intentional 'destruction of cultural links'.³⁶⁴ As the *Bringing them home* report says, 'Culture, language, land and identity were to be stripped from the children in the hope that the traditional lore and culture would die by losing their claim on them and sustenance of them'.³⁶⁵

In Aboriginal and Torres Strait Islander community consultations, survivors spoke about the cultural impacts of being sexually abused in an institution to which they had been removed, including difficulties related to disclosure to their communities and attempts to reconnect with family.³⁶⁶ They told us that when they tried to return home as adults, it was common that family and community did not know about or understand the sexual and other forms of abuse that had occurred in the institutions. The survivors' sense of shame or fear of rejection or retribution alienated them from their community, which may otherwise have been a potential source of healing.³⁶⁷

We heard that the stigma of having been abused and myths that survivors could become perpetrators or cause trouble on communities stopped survivors from disclosing and fuelled this disconnection. Some survivors did not know how to communicate or 'fit in' with their community. Some spoke of feeling a sense of betrayal and confusion about how their families could have allowed them to be taken and abused, not understanding that their parents had no rights or power in the situation. This inability to reconnect with culture was a compounding trauma on top of the multiple traumas they had experienced in the institution.³⁶⁸

In a private session, 'Rainey' told us about sexual abuse by her boss and other men on a sheep station where she was sent from an Aboriginal mission to work as a housemaid. She became pregnant and her baby was taken from her. Returning to the mission, her community ostracised her. Because of the rape and pregnancy, her planned marriage could not go ahead. 'Rainey' said, 'Where I come from you marry something that's pure. You don't touch anything that's been used ... To them I was proper dead, although I was walking around'. It was a source of great shame to her family. It also meant her community would never approve a marriage. On the advice of her father, she left the community.³⁶⁹

In these ways, forcible separation from kin, Country and culture, as well as the shame of the abuse and its consequences within their communities stopped victims from gaining cultural knowledge and practice. These lost connections to cultural identity also affected the handing down of cultural practices and knowledge to future generations. A number of Aboriginal and Torres Strait Islander organisations made submissions to the Royal Commission that refer to cycles of abuse and intergenerational trauma as legacies of this shared history. These submissions stress that for some victims, recovery from this impact requires culturally safe and appropriate services that respond to the need for cultural healing and reconnection to cultural practices (see Volume 9, *Advocacy*, *support and therapeutic treatment services*).

3.6.2 Culturally and linguistically diverse backgrounds

Survivors from culturally and linguistically diverse backgrounds also described particular impacts related to culture and connection to their cultural community.

We heard, for example, how discussing sexual matters or engaging in sexual contact outside sanctioned relationships can attract taboos in certain cultural communities.³⁷¹ For some victims, these types of taboos and the stress of community shame led them to reject, or be rejected by, their community, which in turn disconnected them from their culture. 'Aisha' told Commissioners in a private session that she believed she and other victims of the perpetrator would be 'blacklisted' from their cultural and religious community if they reported the abuse to the police and if their identities became known.³⁷² Another private session attendee from the same community, 'Dinah', told Commissioners that people in small cultural communities 'have the need to be accepted and there's nowhere else for them to be accepted and particularly those who don't speak English. So they have no other choice, really'.³⁷³

We also learned in written submissions how victims who do disclose may be shunned or excluded from cultural practices or events by their community.³⁷⁴ Research suggests that cultural communities that place a high value on community cohesion and family honour are less likely to support disclosure or the reporting of child sexual abuse for fear of the negative impact on family and social harmony.³⁷⁵ We also heard that fears that the cultural community will experience marginalisation, racism or excessive scrutiny if abuse is made public can also make disclosure and reporting less likely.³⁷⁶

For child migrants, the sexual abuse may have occurred in an institutional context that had already alienated them from their original culture and language, further compounding their loss of identity. Not only were many of them completely cut off from their families who remained in their country of origin, but the orphanages and children's homes in which they lived were geographically and culturally isolated from Australian communities.³⁷⁷ Child migrants have spoken to us about their experiences of racism, being punished for speaking in their native language, and being targeted for sexual abuse because of their cultural background. 'Maria' told Commissioners in a private session that she tried to connect with her Maltese culture when she was older, but was unable to learn the language because of the physical, emotional and sexual abuse she experienced when she was a child migrant:

We went back to Malta ... it was absolutely chaos because we'd been belted not to talk our language, so we forgot that language ... You know, I want to live my culture ... I want to feel my culture, you know, because that's what I want ... there's nothing in the world that I really want except just to be Maltese and just to learn the language.³⁷⁸

3.7 Spirituality and religious involvement

One distinctive impact of child sexual abuse in institutional contexts relates to the spirituality and religious involvement of victims. Research suggests that child sexual abuse, irrespective of setting, can be associated with spiritual impacts – such as a decline in faith. But for those victims abused in religious contexts, and for whom spirituality and religion were especially important before the abuse, the spiritual impacts may be greater, leading in some cases to spiritual disengagement.³⁷⁹ As noted in Volume 2, *Nature and cause*, 58.6 per cent of survivors we heard from in private sessions told us they were sexually abused in an institution managed by a religious organisation.

Child sexual abuse by a person in religious ministry can have a profound impact. This is described in research in relation to the Catholic Church, which notes that:

The sexual exploitation of a child by one who has been privileged, even anointed as a representative of God, is a sinister assault on that person's psychological and spiritual wellbeing. The impact of such a violent betrayal is amplified when the perpetrator is sheltered and supported by a larger religious community.³⁸⁰

The betrayal by a person who is seen as a 'representative of God' and also sexually abuses children can be profoundly horrifying. As 'Justin James' told us in a private session, 'I had God raping me – in my mind, that's what I'm thinking'. The violation by a representative of an all-powerful God can result in a loss of trust in everything and everyone.

'Ervin' told us sexual abuse by multiple people in a religious context destroyed his spiritual relationship with God

'Ervin' described to us in a private session how being sexually abused as a child by a number of perpetrators — a lay teacher, a Catholic Brother and then a Catholic Father — destroyed his relationship with God. In his teens, he had wanted to be a Catholic Brother in the Carmelite Order, staying with them and learning their life. 'Ervin' said he had been left with a deep spiritual wound that he is unable to heal. He called it his 'enduring spiritual cancer'.

'Ervin' has scars that run very deep, as demonstrated in an excerpt from a short story he wrote about an imagined meeting with God, which he gave to us at his private session:

And now here we are, across a table, So close, but a chasm of uncertainty between us. I wonder if he feels the same urge I do - to reach across the table and touch. But I pull back ... that would be a step too far ... a reminder too raw of what we shared.³⁸²

Some 8.3 per cent of all survivors who described impacts in private sessions described a 'loss of faith' as a result of abuse. In a private session, 'Arnett' told Commissioners that his 'previous belief in the goodness and sanctity of the Roman Catholic Church had been extremely affected by my experiences. My belief system had been undermined and I turned away from religion'. Another survivor who gave evidence in the *Catholic Church authorities in Maitland – Newcastle* case study, Mr Gerald McDonald, told us he was sexually abused at altar boy practice by a priest, when he was in Year 5.384 He told the Royal Commission that he does not believe in God and has not since being abused.385 He never goes to church unless it is for a wedding or a funeral and feels agitated and uneasy when he does so.386

Similarly, 'Dino' described to us how being sexually abused by a Marist Brother at school caused him to lose his faith in God and his trust in the Catholic Church, and to abandon his plans to become a priest:

From Year 7, I was actively involved in [the Catholic Church] as an altar boy. The closeness and love I felt from God was amazing. I was heavily involved in religious education and studies in school. The Catholic Church was everything to me since I was a young boy. My family all believed that I would become a priest one day. After Year 10, everything changed. I felt like I didn't even have a place in the church. By the time I left high school, I stopped being involved ... For many years, I believed God was dead. This caused the greatest of conflicts between my parents and myself. Looking back now, I can see that I didn't want to believe or be associated with an institution that couldn't even protect its own children.³⁸⁷

Themes of distrust, betrayal and deep anger towards the religious institution, increasing discomfort with rituals, symbols or practices, rage at the institution for its perceived role in facilitating and concealing the abuse, and a crisis in faith, are also outcomes that are cited in studies of child sexual abuse in institutions with a religious affiliation.³⁸⁸

Some researchers have pointed out the unique impacts of sexual abuse by Catholic priests. Psychologist and former Catholic priest, Assistant Professor Joseph Guido, describes this form of child sexual abuse as a unique betrayal. He argues that because Catholics regard the priest as another Christ ('alter Christus') his violation of a child's body is also a violation of a sacred trust and worldview. Many children in Catholic institutions, especially small children, are also likely to have regarded clergy as God's representatives. Hence, the sexual abuse can leave a victim not only betrayed, but also spiritually alone and bereft of the ability to believe in God. The damage a member of the Catholic clergy can cause by sexually abusing a child with faith has also been described in research as soul destroying. Further, as discussed in Volume 16, Religious institutions, a number of victims of sexual abuse perpetrated by Catholic priests were abused in a sacramental setting, such as during confession, or as altar boys assisting with religious ritual in the sacristy, which accentuated the betrayal.

For example, AAP, giving evidence in the Marist Brothers case study, told us:

The confusion between violence, deception and religion that was presented on a daily basis was very hard for a 12 year old or 13 year old to try to reconcile, and especially after going through hundreds — well, multiple incidences of abuse, to try and have a relationship with God after you were abused by someone who was wearing their robes — for me, it was unpalatable, rejected.³⁹²

Others spoke of how the act of confession played a role in their feelings about the sexual abuse, either because they were abused³⁹³ or 'groomed' in the confessional box,³⁹⁴ or were turned away when they went to the confessional for help. 'Laurel' told us that when she went to confession, she thought she would find help. Instead, she said, the priest told her to give him 10 Hail Marys and 10 Our Fathers, and told her she was a disgusting girl who must not allow the perpetrator, another priest, to touch her any more. 'That's something I will never get out of my mind – him turning his back on me and not helping me. From that day onwards, I never told anyone, till I told 'Liana' [her twin sister] on our 52nd birthdays'.³⁹⁵

Survivors spoke about how some Catholic priests' reactions to their disclosure of sexual abuse, made within the confessional, made them feel as though the abuse was their fault. They were told to do their penance and not to do 'it' again, which they took to mean that they were at fault in some way.³⁹⁶ VG, a child migrant from Malta, giving evidence in the *Christian Brothers* case study, told us he disclosed the sexual abuse to a Maltese priest, who chastised him and accused him of lying. Shortly after, the perpetrator called him into the office and strapped him, calling him a troublemaker and a liar. He knew the perpetrator had been told about the confession.³⁹⁷

In the *St Joseph's Orphanage, Neerkol* case study, AYB gave evidence that after each occasion of sexual abuse by a priest she had to go to confession with him and confess [her] sin of impurity.³⁹⁸ She told us that when she went to confessions she would tell the priest that she had committed a sin.³⁹⁹

In some instances, we were told that perpetrators used confession to keep a child from reporting the sexual abuse to others. The role of the confessional in providing the setting for abuse, and for hiding the abuse within the Catholic Church, is discussed further in Volume 16, *Religious institutions*.

The psychological impact of using the confessional in this way is not confined to the Catholic faith. Mr DB Gould, giving evidence in the *Church of England Boys' Society* case study, explained the psychological impact of having the perpetrator, an Anglican priest, hear his confession after he had sexually abused him:

I felt that what occurred was wrong and I felt polluted. I became distressed, and he laid the blame on me for what happened and used the fact that I was upset as proof of this. Daniels said he would hear my 'confession' and then I would be forgiven. He said to me, 'We can fix the problem; God will absolve you. I am a priest and I can act for God in this way'. Daniels explained to me the theology of confession and that he was bound as a priest to keep it in confidence between me, him and God. Daniels then heard my 'confession' in his backyard. Daniels' tactic to silence me profoundly affected me. It put the moral responsibility on me. It meant the secret would stay with him and also guaranteed my silence as I felt bound to keep the contents of my confession confidential, just as he did. For many years, I shouldered the entire blame for the abuse. I felt intense resentment for this abuse of his authority as a priest. This, to me, is more significant than any physical abuse I suffered.⁴⁰⁰

While many survivors who described spiritual impacts of child sexual abuse were assaulted by a person in religious ministry in a Catholic institution, we heard the same themes of betrayal, loss of faith and loss of religious community from those sexually abused in other religious contexts. AVA said he rejected his Orthodox Jewish faith as a result of his experience of abuse:

I have separated from the Jewish community completely and lost complete faith in religion. It is too horrible and incomprehensible to think that there could be an omnipotent being when the world can be so nasty. My life spiralled out of control as a result of the sexual abuse.⁴⁰¹

Another survivor, 'Cleo', told us how she and her sister were sexually abused repeatedly from an early age by the priest in her Greek Orthodox Church. When 'Cleo' was about 12, she and her sister spoke to their mother about the abuse. Their mother believed them immediately, but told them to keep quiet about it, and physically punished 'Cleo', who was made to go back to Sunday school the following week. She said that this time, when the priest tried to touch her, 'Cleo'

grabbed the wooden ruler he was holding and struck him with it repeatedly. 'He just looked terrified. I was empowered. I just saw red, I was so angry. I was angry with God, I was angry with my mother, with her betraying me, angry with the Church, angry with him. I wasn't going to cop it anymore'. Although she continued attending church and Sunday school, he never touched her again, although as far as she knew her mother kept contact with him until he died. 'Cleo' feels the sexual abuse 'stole her life', including her faith in God and in the Greek Orthodox Church. 402

Some survivors expressed deep sadness at their loss of faith as a result of the sexual abuse. 'Edmond' said:

it has cost me my faith in religion. I was raised Catholic, which was with me right from the start so, you know, praying, talking to God, for me has always been a huge sense of strength, deep, deep strength. And yet I find it very, very difficult ... I don't know what to say to Him. 403

Victims also told us about the sadness they felt when they lost their religious community as a result of the abuse. 'Laurel' said, 'It's a huge loss. I loved it. Going to church and saying your prayers and being part of this big group was beautiful ... to me, the religion, it – it was just everything'.⁴⁰⁴

In the Jehovah's Witnesses case study, BCB told us:

Since reporting my story to the Royal Commission, I have stopped attending meetings at [the] congregation. I am sad about this, because it means that I no longer see a lot of my friends from the congregation. Telling my story to the Royal Commission has brought up a lot of feelings of anger in relation to what Bill did to me and the way I have been treated by the Jehovah's Witnesses.⁴⁰⁵

Some victims told us that members of their religious community isolated and ostracised them when the sexual abuse became known. As a result, they were no longer comfortable being involved in the institution (see Chapter 4).

However, we also heard from victims who kept their faith. Some found that it helped them to cope with the effects of the sexual abuse (see Chapter 2). For example, 'Carole' told us, 'The faith I don't think I ever really lost, so for me it was coming back to a religion. So, it was really just about reaching that point of, you don't throw the baby out with the bathwater'. Others sought out religion later in life as a way of helping them to process and gain perspective on what occurred. What we heard is consistent with a few academic studies on the spiritual impact of child sexual abuse in religious institutions. Women in these studies said that the abuse affected their involvement with religious organisations and behaviours, but not their experience of spirituality.

As Professor Sheila the Baroness Hollins, United Kingdom member of the Pontifical Commission for the Protection of Minors, pointed out in evidence in *Case Study 50: Institutional review of Catholic Church authorities*:

A number of people have said to me that they haven't lost their faith in God, but they find it very difficult to go into a church or to go to a service in a church where there's a priest officiating, particularly if nobody has helped them to try to make sense of the spiritual aspect of their experience and their attempt to come to terms with and to live with what happened for them. So many people will have stopped going to church, because they find it too painful to be in church.⁴⁰⁸

Baroness Hollins suggests that the spiritual needs of survivors have often been overlooked in the healing and care of survivors.⁴⁰⁹

The 'ripple effects' of child sexual abuse, such as the loss of faith and respect for the religious institution within families and communities, are discussed further in Chapter 5.

3.8 Interactions with society

Child sexual abuse can have adverse effects on victims' interactions with society, beyond interpersonal relationships. In Section 3.3, we discuss impacts on interpersonal relations with partners, friends and family. In this section, we examine what the available research suggests, and what survivors and others have told us in private sessions and public hearings about the impacts of child sexual abuse on victims' broader social interactions, whether at school, at work or in the general community. Although not discussed as frequently as the impacts of child sexual abuse on mental health, impacts on interactions with society are identified in research as associated with child sexual abuse for many victims. They include:

- distrust and fear of institutions and authority
- re-victimisation
- criminal behaviour.

3.8.1 Distrust and fear of institutions and authority

One of the distinctive impacts of child sexual abuse in institutional contexts that we heard about in private sessions is an ongoing distrust and fear of institutions and authority. Of the survivors who described impacts in private sessions, 32.7 per cent spoke about a lack of trust in authority as a result of the abuse.

In line with existing research, ⁴¹⁰ survivors who spoke to us in private sessions and public hearings about their lack of trust expressed a general distrust of institutions, including the institution where they were sexually abused. Others spoke about particular institutions or systems they found hard to trust and, therefore, to associate with. These included schools, police, social welfare, government authorities and aged care. 'Rea', who told us she was physically and sexually abused by her father as a child and later in two government-run girls' homes, said that she no longer trusted 'the system' that should protect children:

I still have quite an underlying deep distrust of what I see as 'the system' that is there to support and help people who are traumatised, and I think that's part of what motivates me even now is to help advocate for people in systems that let them down, so real systemic failures in the Department of Human Services, Child Services, family systems that just let people down. So, I still carry quite a deep distrust there.⁴¹¹

Notably, some older survivors told us in private sessions that they were distrustful and fearful of aged care institutions, because aged care facilities either reminded them of the abuse or of the institution where the abuse occurred. 'Glenys Maree' told us that the thought of residential aged care alarmed her, along with many other survivors who had been in homes and orphanages as children. 'That's the greatest fear for us homies, that we'll be re-abused in a nursing home'. 412

'Naomi', who told us she was sexually abused by religious sisters and a visiting religious brother, while in an orphanage, and later by a foster father, spoke of how staff at the aged care facility where she now lives did not appreciate her difficulty in being contained within four walls. She said:

They need people who can support us at the end of our time because we go into our shells. We get scared. I've run away a couple of times because in my head it was like being back in that home, and they don't understand. They just don't understand.

Another survivor told us that as she grew older, she worried about having to go into residential care. 'I do not want to be put back into an old person's home. To me that would be just, like, the start of my life and the finish of my life. It would be very traumatic'. 414

'Regina', now elderly, told us her biggest fear is having to live in a nursing home, as she is scared of being sexually abused again. She said she was aged six when she was placed in a Catholic girls' home where she was physically abused by religious sisters and sexually abused by an older girl. 'Regina' told us, 'I believe it's happening in nursing homes, is really bad ... I'd rather die than go to a nursing home, I really would'.⁴¹⁵

Others described their lack of trust as manifesting as a general resistance to and dislike of authority, often because they felt they had been betrayed by the institution where the abuse took place, as well as by the perpetrator. In the *Catholic Church authorities in Maitland – Newcastle* case study, we heard from Mr Michael Balk who was sexually abused at St Gabriel's,

a Marist Brothers School, when he was in 4th Form in 1967.⁴¹⁶ Reflecting on his diverse work history since leaving school, he said:

Looking back, I have moved around a fair bit between various work roles. I remained interested in radio and information technology. I have had a number of instances of unemployment and of being unsettled in my work. I didn't realise at the time, but I can identify that I have had, at times, problems with people in authority. In particular, if those authority figures fail to live up to my expectations.⁴¹⁷

For some, a distrust of authority involved an intense anger and hatred. 'John Robert' grew up in a violent household, and was made a ward of the state at the age of 10. He told us he was sexually abused in boys' homes and foster care, eventually ending up on the streets before going to jail at the age of 17. 'John Robert' has been in and out of jail ever since, but said he came to the Royal Commission because he was fed up with it all. He said, 'I want to know what me problem is. Why I do what I do. And I hate the law. I hate people with authority. … I'm at an age where I should know … what I do is right and what I do is wrong. Instead I've got that much hatred in me that I want to break down and cry'.⁴¹⁸

The lack of trust in and dislike of authority has implications for a survivor's employment. Many survivors in private sessions told us that their dislike of authority has meant that they have faced difficulties at work because they find it challenging to take orders from superiors (see Section 3.9.2). In the *Australian Defence Force* case study, CJU told us:

My work life has been a nightmare. Since leaving the Army I have worked in many different jobs but have struggled to keep a job as I leave when I see bullying or feel threatened. I find it hard to trust colleagues or people in senior positions. If I see people being threatened and bullied I get cold shakes and have to leave the room.⁴¹⁹

As discussed in Chapter 4, a negative or dismissive response by an institution to a disclosure of sexual abuse may compound this lack of trust in and dislike of institutions and authority.

3.8.2 Re-victimisation

Many survivors in private sessions told us that they were re-victimised after the initial incident of sexual abuse. We heard from victims who were sexually abused by different perpetrators at various times during their lives, in both childhood and adulthood. Survivors also told us that they were targeted for other types of abuse following sexual abuse, such as emotional and physical abuse. This adds weight to research suggesting that perpetrators outside the family context often target children who have previously been abused and neglected because they believe these children are less likely to report it (see Volume 2, *Nature and cause*). 420

Some survivors told us they were sexually abused several times, by different perpetrators, during their childhood, as is seen in the experiences of 'Clyde'.

What 'Clyde' told us about re-victimisation

'Clyde' told us in a private session that his first memory of the Anglican boys' home he was sent to at the age of six was watching a housemaster beat his older brother and being told that if he misbehaved, he would be beaten too. It was 1953 and set the scene for 'Clyde's' future years in institutions.

We heard that within a short time of arriving at the boys' home, 'Clyde' was sexually abused by a person who came into the dormitory and put their hands under his bed covers and fondled him. 'I remember seeing the silhouette dressed in sort of religious clothing like cassocks or what they used to wear in church and them sitting on the bed'. For years afterwards, 'Clyde' had to sleep with the blankets pulled up over his head.

When he was eight years old, 'Clyde' was transferred to a Salvation Army boys' home. He told us that while he was being fitted for clothing, one of the officers sexually assaulted him. He didn't report it to anyone. 'Who do you tell? Do you tell the abusers you're being abused?'

'Clyde' told us that in 1955, an older boy who was sent to live in the younger boys' dormitory raped him. When the assault became known, staff members assembled all residents and announced that 'Clyde' and the 13-year-old boy were having a homosexual relationship. 'The inference was that I was party to it'. Thereafter, 'Clyde' was picked on relentlessly, with the story carrying on into other institutions to which he was sent.

'Clyde' said that after the rape, he tried to abscond and his father became involved, taking his son to court to request he be charged with being 'uncontrollable'. From this time on, 'Clyde' became a ward of the state and was moved to several other government-run boys' homes.

At 14, 'Clyde' was sent to a boys' hostel to serve out a 14-month sentence. Conditions were harsh but improved greatly with the appointment of a new superintendent who cared about the boys and their living conditions. When his sentence was completed, 'Clyde' left and 'thumbed' his way around Australia. After he turned 18, a letter was sent to his father stating that 'Clyde' was now discharged from all care.

'Clyde' told us he got into a lot of trouble over the years, committing crimes and spending time in gaols, where he would often become a target after other inmates recognised him from the boys' homes. He told us that, on one occasion, he was raped in gaol.

'Clyde' said his life started to change in the late 1980s. 'I was an alcoholic. I am an alcoholic, but I suppose I woke up one day and thought there's got to be something better'. He started going to Alcoholics Anonymous and later sought counselling for a gambling addiction, during which memories of the sexual abuse came up.⁴²¹

Some survivors described how they were targeted because they had previously experienced sexual abuse, a few sexually abused by the very person they went to for help. 'Frances' told us a priest sexually abused her in the school toilet when she was nine years old. When she got home, she told her mother and stepfather about this abuse. Her stepfather put her on the kitchen table, and examined her for any sign of sexual assault. 'Frances' was told never to tell anyone in their community about the abuse, as it would bring shame to the family. She told us that from this time on, her stepfather began sexually abusing her:

My stepfather had never shown any signs of sexual behaviour or even interest towards me prior to the priest's sexual abuse. I believe the priest's sexual abuse gave my stepfather the perception that he had the right to do the same to me, as I had already been damaged.⁴²²

The abuse only stopped after 'Frances' had a breakdown when she was 14 and was sent to live with another family.

Survivors also described being victimised in non-sexual ways following the sexual abuse, such as being bullied at school or in other institutions. We heard in particular from male survivors who were bullied and called 'poofter' by other children because they were sexually abused, or thought to be sexually abused, by male perpetrators or other boys. 'Clement' told us in a private session that when he returned to school after the police caught the principal who sexually abused him, charged him and sent him to gaol, he was ostracised, ruthlessly teased and bashed by the other boys. He said, 'They'd torment me. Like, they'd sing songs like, "Clement's a poofter, he takes it up the arse He can't run because his bum's full of cum ..."'. 423

After being sexually abused, victims can withdraw and become isolated. 'Jordan' told us how he was relentlessly bullied at the age of 15, following an incident of sexual abuse at a school camp. 424 We heard that one night at the camp, 'Jordan' awoke to find a fellow student masturbating close by. 'Jordan' said the boy ejaculated onto his sleeping bag. He tried to ignore the incident and the next morning told a few friends, expecting them to be disgusted by the act. Their reaction was very different however. 'Jordan' said he became the target of persistent taunting and obscene name-calling by boys from all the years. A few weeks into the ordeal, 'Jordan' got into a fight with a former friend over bullying text messages. Called before his housemaster to explain, 'Jordan' told the teacher about the incident at the camp. 'Jordan' said the teacher did not report the sexual abuse, nor was much done to counter the bullying. 'Jordan's' time at school was becoming intolerable, with bullying happening '24/7'. 'People would just open my door, come in, yell stuff at me, close the door. They had no respect for me anymore'. He told us that matters came to a head four months after the school camp when 'Jordan' was taunted in front of a house meeting of students and teachers. He walked off the grounds and called his father to warn him he was coming home. After this, his parents learned about what 'Jordan' had been going through. However, 'Jordan's' school life did not improve. In the end, 'Jordan' withdrew from the school and enrolled in another school that provided a supportive response. 425 Volume 13, Schools discusses the impacts of child sexual abuse in a school setting in more detail.

Research suggests that victims of child sexual abuse are at increased risk of re-victimisation in adult life as well.⁴²⁶ One large-scale Australian study found that compared to people with no history of abuse, victims of child sexual abuse were more likely to be a victim of violence, more than four times as likely to experience threats of violence, and at five times greater risk of sexual assault.⁴²⁷ One large-scale prospective study found that prior victimisation, including abuse and neglect, increased the risk for future physical and sexual abuse.⁴²⁸ Evidence from other studies associates child sexual abuse involving greater force, and occurring over a long duration and with more frequency, with up to three times greater risk of re-victimisation.⁴²⁹

Another study on child sexual abuse in institutional contexts noted that women who had been sexually abused in institutional care settings were more likely than men to describe experiences of abuse persisting into adulthood. Some victims of child sexual abuse in institutions told government inquiries they accepted ongoing abuse as a feature of their adult relationships.

Female survivors, in particular, told us in private sessions about re-victimisation occurring within intimate relationships. For example, 'Dora' told us that she had been sexually abused at the age of seven by the chaplain at her Catholic school in Sydney's north-west, and at the age of 14 by the religious studies and sex education teacher at her Catholic high school. 'Dora' told us that these incidents had significant impacts on her mental health and believes they also made her more susceptible to later emotional and sexual violence. She had a number of abusive relationships as an adult and experienced a further rape. 'It made me vulnerable to certain types of people, who would take advantage of me ... There was almost a different side to me'. 432

While it is difficult to establish a direct causal relationship between the child sexual abuse and later victimisation, some survivors themselves made an association between the two. 'Rosalie', who told us she married a violent man who assaulted her, said, 'I call it a "scent of abuse". So you wear a bit of vulnerability all over you'. Some victims who have grown up in an environment of repeated and prolonged abuse may have low self-esteem and may not feel entitled to safety and respect. Perpetrators may target these victims because they think they are less likely to be believed or to disclose.

3.8.3 Criminal behaviour

A number of survivors in private sessions and public hearings described how the impacts of child sexual abuse had contributed to their criminal behaviour as adolescents and adults. Of the survivors who discussed impacts of abuse in private sessions, 22.7 per cent said they committed one or more types of criminal offence. These survivors described the complex pathways which led them to engage in criminal behaviour, telling us about various social, cultural, institutional and family factors in their lives at the time of abuse and following the abuse, including disadvantage, maltreatment and trauma.

Retrospective research studies indicate a higher rate of experiences of child sexual abuse amongst prisoners compared with the general population. As discussed in Volume 1, Our inquiry, we heard from 713 survivors who were incarcerated in adult prison when they participated in a private session. Many other survivors who spoke to us in private sessions, while not in prison at the time, had been in correctional centres once or several times for a variety of crimes. These survivors told us they were sexually abused as children in a range of institutional contexts, including out-of-home care and youth detention. Volume 5, Private sessions describes in more detail what adult prisoners told us in their private session about their experiences of child sexual abuse and their lives following the abuse.

There is a growing body of research that examines a potential relationship between child sexual abuse and subsequent criminal offending. While the vast majority of child sexual abuse victims do not go on to commit crimes, some research suggests a higher prevalence of offending than for people in the general community. According to one large-scale Australian study, child sexual abuse victims were almost five times more likely to be charged with an offence than their peers in the general population. Studies of victims of child sexual abuse in institutional contexts have also suggested increased rates of criminal behaviour, charge and conviction. However, a commissioned review of impacts literature suggests that research findings vary, with some highlighting that a direct relationship between abuse and later offending has not been definitively established.

Although the path from victim of child sexual abuse to criminal behaviour is complex and depends on a variety of factors, we heard of common patterns in the lives of those survivors who were involved in criminal behaviour. Some survivors in private sessions and public hearings told us that their behaviour deteriorated in the years following the sexual abuse, most commonly in their teens and early 20s – years that were marked by increased substance abuse and antisocial and rebellious behaviour, leading to criminal offending. For example, 'Rickey', who was sexually abused in residential care, told us:

Alcohol – I started off with alcohol and I ended up being charged with 37 charges of breaking and entering when I was 17 – just touching 17 … I got a general committal, which is until I was 18. I was locked up in a correctional centre … I came out of there and went to – you got charged with vagrancy in those days if you didn't have a dollar in your pocket. So I got locked up for vagrancy for two weeks and got caught then starting in the drugs – I got caught with about this much marijuana and got 12 months' jail … 440

As discussed in Volume 5, Private sessions, many survivors told us that they turned to drugs or alcohol at a young age and ended up getting involved in crime such as theft and drug offences.

Some male survivors who were sexually abused by male perpetrators told us in private sessions and public hearings that violence and rebellious behaviour were connected to their fears they would be seen as homosexual. Mr Andrew Collins, giving evidence in the *Catholic Church authorities in Ballarat* case study, told us that he started 'acting up' after being abused by Brother Peter Toomey at the age of 14:

I decided that I needed to show that I was tough so that everyone knew that I wasn't gay and that I wasn't weak. In sport, I went in hard and I started to get into fights. Before this, I was not violent at all. My marks dropped and I was caught shoplifting. It was the worst year of my life.⁴⁴¹

How a victim's criminal behaviour can impact on others was another issue highlighted in private sessions. Prisoners in particular spoke about the anger and violence they have inflicted on other people, and how the sexual abuse had led them to a hard, emotionless and numbing insensitivity to the feelings of others. We heard how feelings of rage and anger contributed to crimes of violence. A number of survivors told us about their violent offences, some in domestic situations, and often linked to alcohol and other drug use. 'Keith Michael' told us he was constantly fearful of others and used violence to protect himself. It was a strategy that quickly slipped from his control. He said drugs and alcohol inflamed his temper and he often flew into uncontrollable rages. Violence landed him in gaol and he ended up spending much of his life there. '442

Another reason that victims engaged in criminal behaviour was financial destitution. For example, we heard from survivors that they had to leave their out-of-home care placement when they turned 18, and had no means of financial support (see Volume 12, *Contemporary out-of-home care*). 'Paul Andrew' who left state care at 18 years of age, told us he ended up 'drifting' for the next few years. 'No direction. No wind in me sail, going against the tide of adversity, in and out of prisons. Shipwrecked on the rocks of prison, so to speak'.⁴⁴³

While some survivors spoke of putting their criminal activity behind them and moving forward with their lives as adults, others spoke of being caught in what one research project commissioned by us refers to as a cycle of criminal activity and imprisonment.⁴⁴⁴ 'Monica' described the difficulties of stopping the cycle of reoffending.

I need help with my offending. I know I have an anger problem ... and I'm trying to change my life around, but it's such a struggle with being on medication and doing that — a lot goes through my mind ... I just want you to know that I have been thinking a lot about my victims of my crimes. That's the main thing that impacts for me because I know what it's like to be hurt and that's something that I don't want [for] them.⁴⁴⁵

We heard from some survivors how the impacts of child sexual abuse played a role in their criminality, leading to incarceration. 'Kane Travis', attending a private session while in prison, told us he had suppressed memories of his sexual abuse for 30 years but memories surfaced in the early 2010s when the perpetrator suddenly appeared at his parents' house. 'Kane Travis' said that at the time, he was visiting his parents, who had come across the perpetrator and invited him back to their home, not knowing he had sexually abused their son as a child. When 'Kane Travis' suddenly came face to face with the perpetrator he was shocked. Although he remained composed at the time of the encounter, 'Kane Travis' 'fell apart' afterwards and started to use drugs for the first time in many years. He told us that while he was under the influence of drugs, he committed a string of offences resulting in his incarceration.⁴⁴⁶

A few survivors told us they had physically attacked the person who had sexually abused them as a child, or committed property offences against the institution where the abuse had occurred. Others attacked people for reasons related to the sexual abuse, for example a small number of male survivors who had committed murder told Commissioners that sex or the suggestion of sex had acted as a trigger to committing the crime (see Volume 5, *Private sessions*). These types of outcomes illustrate the profound impacts that child sexual abuse can have on victims' lives.

Some survivors – both male and female, although commonly male – told Commissioners they feared becoming sex offenders themselves. As discussed in Section 3.3.2, many survivors had a fear of being close to their own and other children for this reason. The fear that they could be seen as perpetrators may also inhibit some victims – especially males – from disclosing the details of the sexual abuse (see Volume 4, *Identifying and disclosing child sexual abuse*). However, research studies show that the vast majority of victims of child sexual abuse do not go on to commit sexual offences, including child sex offences. ⁴⁴⁷ One study of male victims of child sexual abuse found that three per cent of victims in the study went on to commit a sexual offence. ⁴⁴⁸

The potential relationship between being a victim of child sexual abuse and subsequent sexual offending is contested in research. Some research suggests there may be a weak relationship between sexual abuse and sexual offending (including child sexual abuse) for males, but not females. On the other hand, more recent research suggests that exposure to multiple types of maltreatment may be more significantly associated with later offending, including sexual offending, than child sexual abuse specifically. Research commissioned by us confirms what this research found – that is, childhood physical abuse and/or neglect may be stronger predictors of sexual offending than child sexual abuse. As discussed in Volume 2, Nature and cause, many perpetrators report a history of adverse childhood experiences, including physical, sexual and emotional abuse, and neglect. However, the vast majority of people with these adverse childhood experiences do not go on to sexually abuse children.

3.9 Education, employment and economic security

Many survivors told us that the child sexual abuse they experienced had long-term impacts on their education, employment and overall economic security. Of all survivors who described impacts in private sessions, more than half (55.7 per cent) told us about negative educational and economic outcomes.

3.9.1 Education

Of those survivors who discussed impacts in private sessions, 19.2 per cent described poor learning and educational outcomes. Survivors of child sexual abuse across all types of institutional contexts described impacts on their education, although there were some differences in these impacts depending on the context of the abuse.

As discussed in Section 3.1, the experience of child sexual abuse can affect a child's experience of learning. Often the educational impacts begin immediately after the abuse starts. Survivors told us about being forced to leave school early, being unhappy at school, avoiding school, and experiencing academic difficulties, an inability to concentrate and behavioural problems (such as anger or sexualised behaviour), all of which could lead to suspension or expulsion. These experiences were often accompanied by drug and alcohol use. Survivors told us that as children they were unable to concentrate or apply themselves to schoolwork because they were suffering the effects of the sexual abuse, including fear, anxiety, distrust and anger.

The negative effects of child sexual abuse on concentration, learning ability, self-esteem and educational performance are recognised in research. Chronic irritability, unexpected or uncontrollable anger and difficulties associated with expressing anger are also detailed as part of the experience of child abuse. All these impacts can influence educational outcomes. Research suggests that children who have experienced sexual abuse generally show reduced academic achievement, reduced cognitive function and reduced IQ scores compared with physically abused and non-abused children. Alack of educational engagement and resulting low school completion rates among victims of child sexual abuse were also observed in findings from a longitudinal study on child maltreatment more broadly.

Some survivors we heard from in private sessions and public hearings said that the effects of the sexual abuse meant they stammered or stuttered as children, and this in turn affected their schooling. 'Hany' told us, 'I couldn't get words together, there was no words coming together because there's so much emotion built up in my body that you can't breathe properly ... You've been sexually raped and you lose your sense of trying to express yourself'. When 'Hany' left the institution – a mission school – and went home again, he was treated as stupid or 'not all there', he said. The stammer lasted throughout his high school years and beyond. 458 CNQ told us in evidence in the *Catholic Church authorities in Maitland – Newcastle* case study that he developed a stutter at school after the sexual abuse started. 459

Survivors told us that because of the impacts of sexual abuse, their grades deteriorated. 'Jared Wayne' said, 'Everything suffered from that point on. I had reasonably good marks, I was either in the highs or the very highs. I wasn't just passing. After the incident, everything went to fail or just pass'. 460

Many survivors told us that as a result of the sexual abuse they refused to go to school – either playing truant or dropping out. Not graduating affected their subsequent educational and employment opportunities. 'Josh', who told Commissioners he was sexually abused by the school counsellor, told us:

I stopped even going to school much. Like, I would then not go to school or leave or on the weekends take off ... I just didn't go, I evaded, I ended up not even finishing my last exams. So it really affected my [final result], because I didn't do any of my last exams. 461

For victims who were sexually abused at school by a teacher, another staff member or a student, concentration and application to study were disrupted by fear and anxiety that they would be abused again, or have to face the abuser. Mr Troy Quagliata, giving evidence in the *Sporting clubs and institutions* case study, told of how the sexual abuse affected his schooling. Growing up in a small country town, it was difficult to avoid the perpetrator, Robert Ross, who was both the cricket coach at the local club and a groundsman at his school.⁴⁶²

I left school partly to avoid seeing [Robert Ross]. My grades suffered and everything else followed. I failed virtually every subject except 'phys. ed'., wood and metal work. Every morning I woke up and felt uncomfortable even going to school. Every time I saw him at school, I froze. I dropped out of school in Grade 11. I couldn't do Grade 12, I just couldn't face going back to school. I needed to get out of town. 463

'Ingrid', who has an intellectual disability, and her mother, 'Marla', told us how 'Ingrid's' experience of sexual abuse at the hands of other students at her government high school led her to be fearful and unhappy at school, eventually forcing her to drop out in Year 12. Because they lived in a small regional town, there was no other school that 'Ingrid' could go to. She told us she was sad she didn't complete Year 12. 'It's a huge change. I worked really hard all the way to Year 12. I wanted to do Year 12 and graduate'. She wanted to 'do all the girly stuff' – get dressed up, go in the car, have photos taken for graduation. 'It's very upsetting', she said. She also lost the opportunity to take up a school-based traineeship. 464

For some, the sexual abuse set them on a path to disruptive behaviour at school, including problematic sexual behaviours, getting into fights and drug and alcohol abuse. The behaviour sometimes resulted in suspensions and expulsion. 'Trina Jane' told us in a private session that she had deliberately vandalised teachers' cars so she would be expelled. 'Trina Jane' said she had been sexually abused by three male students in her art classroom when she was 12. She believed the teacher saw some of what happened so she waited a couple of weeks, but the issue of the abuse was never raised. In that time, her behaviour deteriorated. 'I didn't tell anybody and I had to do things to get out of that school which involved writing on the teachers' cars and I was expelled'. 'Trina Jane' said that as a result it was difficult to find another school that would take her, and she ended up using drugs to block it out. ⁴⁶⁵ She told us:

I started inhaling substances at home to deal with what had happened. I found that that blocked out everything for me and while I was under the influence of inhaling things I just was in a different place and everything was okay. And I started at a new school and there was kids there that inhaled as well and so I did that with them and I would pass out at the new school ... 466

Survivors spoke about how their lack of educational achievement affected their adult lives, preventing them going on to tertiary education or training. The sexual abuse also stopped them gaining the skills and attitudes needed to participate in the workforce and be financially secure (see Section 3.9.2).

Many victims, now adults, felt keenly that they had lost opportunities because of the impact that the sexual abuse had on their education. Older survivors who experienced sexual abuse in residential care – including child migrants, Aboriginal and Torres Strait Islander children removed into missions and homes, and children placed in orphanages and reformatories – told us that, in addition to suffering multiple forms of abuse in care, their educational needs were severely neglected. They were often forced into child labour instead.

We heard evidence in the *St Joseph's Orphanage, Neerkol* case study that the levels of education at the orphanage were 'lamentable'. Some children left the orphanage barely literate and there was a lack of special assistance for children even until the 1970s. ⁴⁶⁷ Survivors told us in private sessions about their poor literacy and numeracy, ⁴⁶⁸ which had long-term effects on their self-esteem, relationship with their children and employment opportunities. ⁴⁶⁹ This is supported by the findings of the Commission of Inquiry into Abuse of Children in Queensland Institutions (the Forde Inquiry) in 1999, which described educational deprivation as one of the most profound and enduring losses suffered by former residents of institutions. ⁴⁷⁰

For many older survivors of child sexual abuse, their lack of education has been a continuing cause of shame and disappointment. As a child migrant from Malta, Mr Raphael Ellul, told us in the *Christian Brothers* case study:

At Tardun, it took me more than three years to learn enough pidgin English to get by. I didn't receive any teaching in English language. I left Tardun aged 16. I still only spoke broken English. After approximately three years' schooling at Tardun (up to the age of 13), which was interrupted by stints of mallee-root picking, I had hardly any written skills. After leaving Tardun, I had little opportunity to improve my English skills. I am able to read basic English. I am unable to write in English. I am embarrassed by this and it has impacted my employment options.⁴⁷¹

Another child migrant from Malta, VG, who was also subjected to sexual and physical abuse at Tardun, gave evidence in the *Christian Brothers* case study about the shame surrounding his lack of education. His mother sent him to Australia after a priest told her he would get a free education and would be able to return home after he finished secondary school. Instead, VG was put to work. Eventually he went back to Malta, but was ashamed that he was uneducated.

Even when I was in Malta, I didn't feel free from what happened at Tardun because I always pretended that I was qualified, so my mother would never find out what really happened. My wife and I eventually returned to Australia in 1972 so that others would not find out that I was not qualified at all – that was including my wife.⁴⁷²

'Clarence' told us in a private session that he has 'done his best' to teach himself to read and write as an adult. 473

What 'Clarence' told us about illiteracy

One legacy of the sexual abuse 'Clarence' experienced as a child in institutional care in the 1960s is that for much of his life he has been illiterate.

When 'Clarence' was eight, his mother was found to be 'unfit' to look after him and his siblings, and they were removed from her care. 'Clarence' was told he was being sent away to be educated. The explanation made sense to him because in the small Victorian country town where his family lived it was common for children to go to boarding school. 'That was a standard thing in the country. People go to another place to get educated', 'Clarence' recalled. Instead, 'Clarence' and some of his siblings were placed in a children's home run by Catholic sisters. 'Now I know it was an orphanage. At the time I didn't. I thought I was going to boarding school'.

'Clarence' told us he suffered physical, psychological and sexual abuse from sisters at the home. One of his roles at the home was to act as a 'parlour boy'. 'That was someone who opened the door and said, "Welcome, come in, sit down and I'll go and get Sister so and so'', he told us. 'Clarence' would go and find the sister, then go to the kitchen and organise the tea tray and deliver it to the sister and her visitor. The job allowed him to roam the institution unsupervised, which meant he could secretly visit his brothers and sisters in the nursery. But it also presented an opportunity for others. He described the abuse that happened then as, 'older people took advantage of me. At the time I didn't tell anybody about it, because that was how it worked', said he.

In the early 2000s, 'Clarence' contacted the Catholic Church in his home town. 'I rang someone to say that things weren't right'. That call led to sessions with a counsellor, and eventually to a meeting with representatives of the Catholic Church.

'Clarence' went to the meeting very clear about what he wanted to get out of it.

'I said, "I need to learn to read and write, please". I said, "I'm illiterate". He was told he'd be given \$3000 to buy a computer. However, 'Clarence' also made it 'distinctly clear' that he needed money and support to help him look after his brothers and sisters. Like him, they have ongoing needs as a result of their time in care. 'I was reassured that that was okay. The \$3000 was to get me started so I would go off and learn to read and write ... They basically saw me out the door and said, "Don't worry Clarence, everything's going to be fixed up and it'll all be all right".

'Clarence' bought the PC and has learned to read and write. 'I've done my best', he said. 'I'm a self-taught person. Am I an educated person? No, I'm not'. But he didn't hear any more from the church and when he got in touch with them he was told, 'No, that's it'.

'Clarence' is now being supported by his own advocate who has contacted the church on his behalf. Six months after writing a letter asking for the matter to be revisited, they are yet to receive a response.

'I'd like my matter resolved before my mother dies', he told the Commissioner. 'That's important for me'. She is now in her 70s. 'Clarence' believes that she has also been treated unjustly. 'My mother wasn't unfit, and her name should be cleared'. As well, she was told her children would be given a good education and a better life. But the reality was very different.⁴⁷⁴

Academic underachievement meant that some victims were unable to pursue their childhood dreams. 'Carmen', who told us she was sexually abused by her female physical education teacher throughout her high school years, spoke of how the teacher's behaviour meant she was unable to concentrate on her studies during her final years at high school. She received poor grades and had to give up her childhood dream of becoming a doctor.⁴⁷⁵

what I really wanted to do when I was young ... all I wanted to do was be a doctor, and that's what my sister became. I loved the whole idea of medicine and I wanted to be a doctor and, if I couldn't be a doctor, then I wanted to be a psychologist. But of course, I didn't achieve any – and my Year 12 results were appalling because I didn't do any work. I was just being dragged around the school by this [perpetrator]. 476

A number of survivors told us they were sexually abused by teachers or coaches in specialist areas of study, such as music, dance or sport, where they had talent or a passion. Some found alternative teachers or a way to keep studying the subject after a disclosure of sexual abuse;⁴⁷⁷ some even found that their talent was an ongoing source of protection from the impacts of abuse. For others, the sexual abuse destroyed their special relationship with this activity.⁴⁷⁸ This type of impact is discussed further in Volume 14, *Sport, recreation, arts, culture, community and hobby groups*.

While a considerable number of survivors told us about their poor educational achievement, we also heard from many people who pursued an education or gained skills later in life, going on to have successful careers. For some, the pursuit of education and employment goals was a means of reasserting self-esteem, finding their identity or proving themselves in the face of adversity. 'Merle' told us she was sexually abused by a priest and placed in an orphanage where she was also physically and emotionally abused. In the years after leaving the orphanage, 'Merle' gained an education and learned about her Aboriginal culture, from which she had been alienated as a child. 'Merle' worked in areas where traditional laws and spiritual practices thrive, and as an adult has gained two university degrees to 'prove to people that I was not stupid'.⁴⁷⁹

3.9.2 Employment and income

In case studies and private sessions, survivors told us they believed being sexually abused as a child had led to them being unemployed, underemployed or lowly paid at various times in their lives.

Some survivors told us they were unable to hold down a job because of mental health issues, including anxiety, depression, panic attacks or low self-esteem. 'Rachel', who told us she was sexually abused by her high school science and guitar teacher, went on to complete graduate and post-graduate qualifications at university. However, maintaining consistent employment has been difficult due to her fluctuating physical and emotional health.⁴⁸⁰ Another survivor, 'Chris Patrick', who told us he endured regular sexual abuse by a priest from the age of 12, spoke of how he left school in Year 10 and was later forced to leave a job because of his difficulties with drugs and alcohol.⁴⁸¹

Other survivors such as 'John' spoke of being unable to work because of health issues, causing them to worry about their future. Of the survivors who described impacts of child sexual abuse in private sessions, 6.8 per cent told us they were unable to work. Many survived on a Disability Support Pension because of their physical or mental health problems.

'John' told us about the effects of child sexual abuse on his employability

'John' told us his abuse began in the family home in Melbourne. His father, a violent alcoholic, believed that 'John' was the result of a brief relationship his wife had had with another man. 'John' said that although his father was abusive to all his seven children, he reserved the worst of it for 'John', whom he considered a 'bastard'. 'He zeroed in on me with a lot of verbal abuse, physical abuse, and emotional abuse', 'John' recalled.

In the mid-1960s, when 'John' was aged eight or nine, he was put under a care and protection order. He had been running away from home and had been caught stealing.

Not long afterwards, 'John' had his first experience of institutionalisation. He was sent to a reception facility, where he spent three months waiting to be transferred to longer term residential care.

At the start of Grade 6, 'John' was sent to a Christian Brothers' orphanage just outside Melbourne. His first year 'wasn't too bad', he recalled. In his second year, he was put in a different dormitory, under the charge of Brother 'Fencher'. 'John' was a chronic bedwetter; the result of stress and anxiety. One night, 'John' wet the bed and went to ask 'Fencher' for help. 'John' said, 'He decided to have sex with me on his couch'. 'Fencher', we heard, also sexually assaulted 'John' in the shower when no one was around. 'Even today I have trouble having a shower', 'John' told the Commissioner. 'Showering on a regular basis is something I really have to push myself to do'.

'John' had placements at three institutions, and told us he experienced physical and sexual abuse at each. He returned home several times, but his father's violent behaviour made it impossible to stay. As a 14-year-old, 'John' lived on the streets, where he was sexually assaulted again.

'John' has suffered from anxiety and depression throughout his life. He tried to educate himself but could not complete courses and gain certification, and has not been able to stay in steady employment. He married and had three daughters, but his marriage ended some years ago.

Now living on a disability pension, 'John' worries about his future. 'I think my health issues at the moment is probably the most concerning thing', he said. 'I feel my body breaking down ... I think that's through anxiety and the constant stress'. 'John' spoke of his enduring financial hardship:

I remember once when I was working I had a Medibank Blue Ribbon account, and a couple of times I went into private places for depression. It gave me the safety net, that I could just ring up ... The depression would get that much – it gave me time out and you had good quality food and you could rest and then get back into it again. All those safety nets have been taken away, purely because I'm not making any money. I don't have a nest egg or a car or anything to run on ... So I feel that that type of thing at the moment is a ball and chain in my life. 482

Several survivors were unable to settle into a job or career path, often because they had difficulties working under authority as a result of their experiences of sexual abuse by a trusted adult (see Section 3.8.1). One survivor told us:

I think I've had about 33 different careers – never been sacked – quit every job. I won't tolerate being bullied or pushed around. If I'm working or something and the boss says something like ... 'Pick up that shit', I'll say, 'Pick it up your fucking self'. You know what I mean? I'm not walking over there to do that if you're standing beside it, that sort of thing ... It's that power thing, like to have the power over other people. I think that's probably why I left some of those jobs. 483

Although some survivors have been able to find work, a lack of educational qualifications – combined with their mental health issues – has, at various times, meant that some were only able to work part-time, or in lowly paid jobs.

Some expressed sadness that they did not reach their career potential. BQG, giving evidence in the *Brisbane Grammar School and St Paul's School* case study, spoke about how he failed to complete Year 12, missing out on a law career because of the impacts of sexual abuse by his former school counsellor:

My failure to complete Year 12 meant that I wasn't able to pursue studies in law, as I had hoped to. Looking back, it appears that I was running — North Queensland, New South Wales and then ultimately Perth. I have been lucky to have the successes I have had. However, I still feel inadequate that I have no degree. My siblings all have degrees and are quite successful in their chosen fields. The law is something I have always wanted to pursue. However, I feel that the time in my life when I could have been achieving that, I was running away.⁴⁸⁴

In the Retta Dixon Home case study we heard from Ms Sandra Joy Kitching, who said:

The worst part of it was that being in Retta Dixon left me without hope ... That was the worst, missing out on so many opportunities in life, because Retta Dixon just gave us no hope, no life skills and no self-confidence to do anything.⁴⁸⁵

Despite the profoundly negative impacts of child sexual abuse on their lives, numerous survivors told us that they were pursuing successful and fulfilling careers. In many cases, we heard that their desire for justice influenced their career choice. Some survivors became lawyers, academics and advocates, dedicated to helping child sexual abuse victims or other people who had suffered adversity in their lives. For example, 'Nancie' told us in a private session that she was physically and sexually assaulted at a government-run Aboriginal mission in the mid-1960s and then in a foster home. Eventually she ran away from her foster parents. 'Nancie' said she lived on the streets for a while, and had a baby when she was 18. However, 'Nancie' managed to finish her schooling and went on to become a successful sports player, with help from significant mentors. She is now a mentor herself. 'Well, I said I wouldn't go down that pathway, because

I seen it all my life. And the pathway was the alcohol, the drugs, the prostitution'. Instead, 'Nancie' decided to help children and young people, starting in a local neighbourhood. She now helps young people in the criminal justice system, and is studying towards a law degree. She also ran a group home for a while, looking after 30 street kids.⁴⁸⁶

3.9.3 Housing security

Survivors and their families told us that the flow-on effects of the sexual abuse, including mental health impacts, alcohol and other drug abuse, and poor education, had at times led to extreme economic hardship and homelessness and other housing problems. A recent Australian longitudinal study suggests a relationship between childhood trauma – such as sexual abuse – and homelessness, with nearly one-third (31.5 per cent) of those who had been homeless for four years saying that they had been sexually assaulted during childhood.⁴⁸⁷

A number of survivors described their experiences of homelessness as young people, after running away from school, home or foster care because of sexual and other abuse. 'Marcia', told us in a private session that she was sexually abused by a supervisor as a small child in an Anglican home where she was placed, ending up in a series of institutions. 'Marcia' began to get in trouble with police for shoplifting, avoiding school and running away. Eventually at age 14, she escaped from a hostel and started sleeping on the streets, in squats and on friends' couches. 'Marcia' told us that when she turned 16, her state wardship ended and she was able to get government benefits and find a place to live.⁴⁸⁸

Other survivors described episodes of homelessness throughout their adult lives. For some, homelessness also resulted from a combination of alcohol abuse, drugs and an inability to maintain employment. It was also often related to experiencing significant mental health issues (see Volume 5, *Private sessions*).

While not homeless, some adult survivors told us they lived in fear of losing their home or had been forced to live in low-cost rental accommodation, such as caravans. For example, 'Brendon' told us in a private session that he was sexually abused multiple times by a friend of his father's at Scouts. He said that he had, and continues to have, difficulty with people in positions of authority, which has caused him a lot of problems in his career and personal life. 'Brendon' said he suffers daily from anxiety caused by the abuse, with multiple health problems exacerbated by the stress of severe financial pressure. He can't pay his mortgage and fears losing his home. He told us he has considered taking his own life in the past. 490

'Dion' told us he was homeless until 45 after leaving institutional care

'Dion' told us in a private session that he spent his childhood in and out of a Catholic orphanage and youth detention facilities, and was sexually abused in both these settings. He 'acted out', was involved in petty crime, and was diagnosed as a sociopath. When he finally left institutional care, 'Dion' didn't know what to do:

I didn't go back home because I didn't have to. I was homeless then until I was 45. I used a lot of alcohol, a lot of drugs, but I made a commitment to myself that I would never, never get in trouble with the law. I didn't want to end up in gaol or anything like that. I would be able to do a bit of work, but I didn't have that ability to be able to retain or have consistency, or I had no trust or faith in myself.

In between living on the streets and staying in homeless shelters and refuges, 'Dion' occasionally set himself up in a flat, but said his drinking always led to him becoming homeless again. He also 'went bush' often and during one of these periods he became very sick:

I was six and a half stone. I was really thin. I needed a walking stick to walk. I had abscesses and sores all over my legs. So I guess I came back into society and I had a cathartic moment and I just realised that if I chose to start learning things, I can learn how to be okay, and I had a vision in my head ... Then all of a sudden I realised that all my life I've been fighting myself, you know, and all this devastation, all this destruction I caused fighting myself. And I didn't want to fight myself anymore.

'Dion' said this was a major turning point. Within a short time he applied for and was accepted into a university preparation course. In 2007 'Dion' completed a degree and was employed in a teaching position while furthering his education and in 2016 he completed his doctorate.⁴⁹¹

Endnotes

- 1 T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017; J Breckenridge & G Flax, Service and support needs of specific population groups that have experienced child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016.
- 2 Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study 36: The response of the Church of England Boys' Society and the Anglican Dioceses of Tasmania, Adelaide, Sydney and Brisbane to allegations of child sexual abuse, Sydney, 2016, pp 127-8.
- 3 Exhibit 36-0060, 'Statement of [BYC]', Case Study 36, STAT.0809.001.0001 R at 0011 R-0012 R.
- 4 D Roche, MG Runtz & MA Hunter, 'Adult attachment: A mediator between child sexual abuse and later psychological adjustment', Journal of Interpersonal Violence, vol 14, no 2, 1999; J Herman, Trauma and recovery: The aftermath of violence - From domestic abuse to political terror, Basic Books, New York, 1992, pp 110-4; BA van der Kolk, 'Developmental trauma disorder', Psychiatric Annals, vol 35, no 5, 2005.
- 5 T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 70.
- 6 C Kezelman & P Stavropoulos, Practice guidelines for treatment of complex trauma and trauma informed care and service delivery, Adults Surviving Child Abuse, 2012. For an overview of trauma-informed and recovery-oriented approaches, see L Wall & A Quadara, 'Acknowledging complexity in the impacts of sexual victimisation trauma', ACSSA Issues, no 16, 2014, pp 14-18.
- 7 J Herman, Trauma and recovery: The aftermath of violence - From domestic abuse to political terror, Basic Books, New York, 1992, p 33.
- 8 J Herman, Trauma and recovery: The aftermath of violence - From domestic abuse to political terror, Basic Books, New York, 1992, p 96.
- 9 BD Perry, Effects of traumatic events on children, The ChildTrauma Academy, 2003, pp 2–8.
- 10 A Quadara & C Hunter, Principles of trauma-informed approaches to child sexual abuse: A discussion paper, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 12.
- H Hart & K Rubia, 'Neuroimaging of child abuse: A critical review', Frontiers in Human Neuroscience, vol 6, no 52, 2012; 11 MH Teicher & JA Samson, 'Annual research review: Enduring neurobiological effects of childhood abuse and neglect', The Journal of Child Psychology and Psychiatry, vol 57, no 3, 2016; Child Welfare Information Gateway, Understanding the effect of maltreatment on brain development, USA, 2015.
- 12 H Bath, 'The three pillars of trauma-informed care', Reclaiming Children and Youth, vol 17, no 3, 2008, p 18.
- BA van der Kolk, 'Developmental trauma disorder', *Psychiatric Annals*, vol 35, no 5, 2005, p 403. BA van der Kolk, 'Developmental trauma disorder', *Psychiatric Annals*, vol 35, no 5, 2005, p 402. 13
- 14
- 15 RL Gaskill & BD Perry, 'Child sexual abuse, traumatic experiences, and their impact on the developing brain' in P Goodyear-Brown (ed), Handbook of child sexual abuse: Identification, assessment, and treatment, John Wiley & Sons, Hoboken, New Jersey, 2012, p 33.
- 16 S Boney-McCoy & D Finkelhor, 'Prior victimization: A risk factor for child sexual abuse and for PTSD-related symptomatology among sexually abused youth', Child Abuse & Neglect, vol 19, no 12, 1995.
- 17 P Ackerman, J Newton, W McPherson, J Jones & R Dykman, 'Prevalence of post traumatic stress disorder and other psychiatric diagnoses in three groups of abused children (sexual, physical, and both)', Child Abuse & Neglect, vol 22, no 8, 1998, p 759.
- American Psychiatric Association, Diagnostic and statistical manual of mental disorders Fifth edition, 5th (DSM-5) edn, 18 American Psychiatric Publishing, Washington, 2013, pp 271–80.
- 19 Transcript of C Quadrio, Case Study 28, 25 May 2015 at C8458:21-29.
- PT Ackerman, JEO Newton, WB McPherson, JG Jones & RA Dykman, 'Prevalence of post traumatic stress disorder and 20 other psychiatric diagnoses in three groups of abused children (sexual, physical, and both)', Child Abuse & Neglect, vol 22. no 8. 1998.
- 21 L Wall & A Quadara, 'Acknowledging complexity in the impacts of sexual victimisation trauma', ACSSA Issues, no 16, 2014.
- 22 L Wall & A Quadara, 'Acknowledging complexity in the impacts of sexual victimisation trauma', ACSSA Issues, no 16, 2014; J Herman, Trauma and recovery: The aftermath of violence - From domestic abuse to political terror, Basic Books, New York, 1992, pp 118–22. The term 'complex PTSD', was first proposed by psychiatrist Judith Herman in 1992 to describe the syndrome that follows prolonged, repeated trauma.
- 23 J Herman, Trauma and recovery: The aftermath of violence - From domestic abuse to political terror, Basic Books, New York, 1992, p 119.
- 24 C Kezelman & P Stavropoulos, Practice guidelines for treatment of complex trauma and trauma informed care and service delivery, Adults Surviving Child Abuse, 2012, p 49.
- 25 C Kezelman & P Stavropoulos, Practice guidelines for treatment of complex trauma and trauma informed care and service delivery, Adults Surviving Child Abuse, 2012, p 48.
- J Herman, Trauma and recovery: The aftermath of violence From domestic abuse to political terror, Basic Books, 26 New York, 1992, pp 118-19.
- 27 L Wall & A Quadara, 'Acknowledging complexity in the impacts of sexual victimisation trauma', ACSSA Issues, no 16, 2014, pp 7-8.
- 28 J Herman, Trauma and recovery - The aftermath of violence - From domestic abuse to political terror, Basic Books, New York, 1992, p 96.

- 29 Transcript of C Quadrio, Case Study 28, 25 May 2015 at C8461:25–43.
- 30 RL Gaskill & BD Perry, 'Child sexual abuse, traumatic experiences, and their impact on the developing brain' in P Goodyear-Brown (ed), *Handbook of child sexual abuse: Identification, assessment, and treatment*, John Wiley & Sons, Hoboken, New Jersey, 2012, pp 33–7.
- B Perry, RA Pollard, TL Blaicley, WL Baker & D Vigilante, 'Childhood trauma, the neurobiology of adaptation, and 'use-dependent' development of the brain: How "states" become "traits", *Infant Mental Health Journal*, vol 16, no 4, 1995, p 277.
- RL Gaskill & BD Perry, 'Child sexual abuse, traumatic experiences, and their impact on the developing brain' in P Goodyear-Brown (ed), *Handbook of child sexual abuse: Identification, assessment, and treatment*, John Wiley & Sons, Hoboken, New Jersey, 2012, pp 36–7; B Perry, RA Pollard, TL Blaicley, WL Baker & D Vigilante, 'Childhood trauma, the neurobiology of adaptation, and 'use-dependent' development of the brain: How "states" become "traits"', *Infant Mental Health Journal*, vol 16, no 4, 1995, p 276.
- RL Gaskill & BD Perry, 'Child sexual abuse, traumatic experiences, and their impact on the developing brain' in P Goodyear-Brown (ed), *Handbook of child sexual abuse: Identification, assessment, and treatment*, John Wiley & Sons, Hoboken, New Jersey, 2012, p 35.
- BD Perry, 'Examining child maltreatment through a neurodevelopmental lens: Clinicial applications of the neurosequential model of therapeutics', *Journal of Loss and Trauma*, vol 14, no 4, 2009, p 244.
- RL Gaskill & BD Perry, 'Child sexual abuse, traumatic experiences, and their impact on the developing brain' in P Goodyear-Brown (ed), *Handbook of child sexual abuse: Identification, assessment, and treatment*, John Wiley & Sons, Hoboken, New Jersey, 2012.
- M Barrera, L Calderón & V Bell, 'The cognitive impact of sexual abuse and PTSD in children: A neuropsychological study', Journal of Child Sexual Abuse, vol 22, no 6, 2013.
- 37 Transcript of C Quadrio. Case Study 28, 25 May 2015 at C8457:39–47 and C8458:1–6.
- Attachment theory, first developed in the 1970s by British psychoanalyst John Bowlby and further developed by Mary Ainsworth, Mary Main and others, conceptualises the foundations of human sociability and wellness in the complex neurobiological systems that cause infants to seek out caregivers when they are frightened or experiencing stress. See J Bowlby, A secure base: Parent-child attachment and healthy human development, Basic Books, USA, 1988.
- 39 C Kezelman & P Stavropoulos, *Practice guidelines for treatment of complex trauma and trauma informed care and service delivery*, Adults Surviving Child Abuse, 2012, pp 55–8.
- J Herman, *Trauma and recovery: The aftermath of violence From domestic abuse to political terror*, Basic Books, New York, 1992, p 261.
- J Herman, *Trauma and recovery: The aftermath of violence From domestic abuse to political terror*, Basic Books, New York, 1992, pp 110–14.
- J Cashmore & R Shackel, *The long-term effects of child sexual abuse*, Australian Institute of Family Studies, Melbourne, 2013, p 23; MT Lynskey & DM Fergusson, 'Factors protecting against the development of adjustment difficulties in young adults exposed to childhood sexual abuse', *Child Abuse & Neglect*, vol 21, no 12, 1997, pp 1177, 1186–7; M Domhardt, A Munzer, JM Fegert & L Goldbeck, 'Resilience in survivors of child sexual abuse: A systematic review of the literature', *Trauma Violence & Abuse*, vol 16, no 4, 2015, pp 488–9.
- 43 K Kendall-Tackett, L Williams & D Finkelhor, 'The impact of sexual abuse on children: A review and synthesis of recent empirical studies', *Psychological Bulletin*, vol 113, no 1, 1993, p 167.
- 44 Victorian Department of Human Services, *Child development and trauma specialist practice resource*, 2012, www.dhs.vic.gov.au/for-service-providers/children,-youth-and-families/child-protection/specialist-practice-resources-for-child-protection-workers/child-development-and-trauma-specialist-practice-resource (viewed 10 February 2017).
- 45 Transcript of AIQ, Case Study 19, 22 October 2014 at 10102:31–38; Exhibit 19-0006, 'Statement of AIQ', Case Study 19, STAT.0345.001.0001 R at 0006 R.
- 46 Victorian Department of Human Services, *Child development and trauma specialist practice resource*, 2012, www.dhs.vic.gov.au/for-service-providers/children,-youth-and-families/child-protection/specialist-practice-resources-for-child-protection-workers/child-development-and-trauma-specialist-practice-resource (viewed 10 February 2017).
- 47 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 32: The response of Geelong Grammar School to allegations of child sexual abuse of former students*, Sydney, 2015, p 25.
- 48 Victorian Department of Human Services, *Child development and trauma specialist practice resource*, 2012, www.dhs.vic.gov.au/for-service-providers/children,-youth-and-families/child-protection/specialist-practice-resources-for-child-protection-workers/child-development-and-trauma-specialist-practice-resource (viewed 10 February 2017).
- 49 Victorian Department of Human Services, *Child development and trauma specialist practice resource*, 2012, www.dhs.vic.gov.au/for-service-providers/children,-youth-and-families/child-protection/specialist-practice-resources-for-child-protection-workers/child-development-and-trauma-specialist-practice-resource (viewed 10 February 2017).
- Name changed, private session, 'Myles'.
- Victorian Department of Human Services, *Child development and trauma specialist practice resource*, 2012, www.dhs.vic.gov.au/for-service-providers/children,-youth-and-families/child-protection/specialist-practice-resources-for-child-protection-workers/child-development-and-trauma-specialist-practice-resource (viewed 10 February 2017).
- Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 22: The response of Yeshiva Bondi and Yeshivah Melbourne to allegations of child sexual abuse made against people associated with those institutions, Sydney, 2016, p 29; Transcript of M Waks, Case Study 22, 2 February 2015 at C6036:43–47; Exhibit 22-0003, 'Statement of Menahem Leib Waks', Case Study 22, STAT.0460.001.0001 R at 0016 R.

- Victorian Department of Human Services, *Child development and trauma specialist practice resource*, 2012, www.dhs. vic.gov.au/for-service-providers/children,-youth-and-families/child-protection/specialist-practice-resources-for-child-protection-workers/child-development-and-trauma-specialist-practice-resource (viewed 10 February 2017); S Hackett, *Children and young people with harmful sexual behaviours,* Research in Practice, Devon, UK, 2014, p 39.
- 54 S Hackett, Children and young people with harmful sexual behaviours, Research in Practice, Devon, UK, 2014, pp 39–40.
- Victorian Department of Human Services, *Child development and trauma specialist practice resource*, 2012, www.dhs.vic.gov.au/for-service-providers/children,-youth-and-families/child-protection/specialist-practice-resources-for-child-protection-workers/child-development-and-trauma-specialist-practice-resource (viewed 10 February 2017).
- Names changed, private session, 'Coralie' and 'Jack'.
- 57 Name changed, private session, 'Gil James'.
- M Proeve, C Malvaso & P DelFabbro, Evidence and frameworks for understanding perpetrators of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 29; K Richards, 'Misperceptions about child sex offenders', Trends and Issues in Crime and Criminal Justice, no 429, 2011, p 4; C Leach, A Stewart & S Smallbone, 'Testing the sexually abused-sexual abuser hypothesis: A prospective longitudinal birth cohort study', Child Abuse & Neglect, vol 51, 2016, p 150.
- 59 Victorian Department of Human Services, *Child development and trauma specialist practice resource*, 2012, www.dhs. vic.gov.au/for-service-providers/children,-youth-and-families/child-protection/specialist-practice-resources-for-child-protection-workers/child-development-and-trauma-specialist-practice-resource (viewed 10 February 2017).
- 60 Transcript of AVA, Case Study 22, 2 February 2015 at C5993:26-37; Exhibit 22-0001, 'Statement of AVA', Case Study 22, STAT.0461.001.0001 R at 0006 R.
- J Cashmore & R Shackel, *The long-term effects of child sexual abuse,* Australian Institute of Family Studies, Melbourne, 2013, p 7.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 50.
- 63 MC Cutajar, PE Mullen, JRP Ogloff, SD Thomas, DL Wells & J Spataro, 'Psychopathology in a large cohort of sexually abused children followed up to 43 years', *Child Abuse & Neglect*, vol 34, no 11, 2010, p 816.
- A Browne & D Finkelhor, 'Impact of child sexual abuse: A review of the research', *Psychological Bulletin*, vol 99, no 1, 1986, p 76; JN Briere & DM Elliott, 'Immediate and long-term impacts of child sexual abuse', *The future of children*, vol 4, no 2, 1994, pp 55, 63.
- T Lewis, E McElroy, N Harlaar & D Runyan, 'Does the impact of child sexual abuse differ from maltreated but non-sexually abused children? A prospective examination of the impact of child sexual abuse on internalizing and externalizing behavior problems', Child Abuse & Neglect, vol 51, 2016, pp 38–9.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 54.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 56.
- For example, see Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 17: The response of the Australian Indigenous Ministries, the Australian and Northern Territory governments and the Northern Territory police force and prosecuting authorities to allegations of child sexual abuse which occurred at the Retta Dixon Home, Sydney, 2015, pp 23–24.
- Name changed, private session, 'Jarrod'.
- 70 Name changed, private session, 'Lillian Jane'.
- 71 Name changed, private session, 'Chelsea'.
- 72 Name changed, private session, 'Janis'.
- 73 Name changed, private session, 'Lennox'.
- 74 Name changed, private session, 'Margo'.
- 75 Name changed, private session, 'Margo'.
- 76 Name changed, private session, 'Cecilia'.
- 77 Name changed, private session, 'Bridget'.
- 78 Name changed, private session, 'Dom'.
- 79 Transcript of AIQ, Case Study 19, 22 October 2014 at 10103:1–5; Exhibit 19-0006, 'Statement of AIQ', Case Study 19, STAT.0345.001.0001_R at 0006_R.
- Transcript of KR, Case Study 6, 17 February 2014 at Q39:28–32.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 56.
- 32 JP Wilson, B Drozdek & S Turkovic, 'Posttraumatic shame and guilt', Trauma, Violence & Abuse, vol 7, no 2, 2006, p 123.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 56.
- Name changed, private session, 'Davey'.
- Name changed, written account, 'Luke Anthony'.

- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 56.
- Name changed, private session, 'Roddick'.
- 88 Exhibit 12-0001, 'Statement of WP', Case Study 12, STAT.0256.001.0001_M_R at 0007_M_R.
- 89 Exhibit 39-0010, 'Statement of BXI', Case Study 39, STAT.0977.001.0001_R at 0003_R.
- 90 Transcript of K Menzies, Case Study 57, 31 March 2017 at 27822:20–26.
- 91 Transcript of K Menzies, Case Study 57, 31 March 2017 at 27822:24–26.
- Human Rights and Equal Opportunity Commission, *Bringing them home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families*, Human Rights and Equal Opportunity Commission, Sydney, 1997, pp 75, 165; H McGlade, *Our greatest challenge: Aboriginal children and human rights*, Aboriginal Studies Press, Canberra, 2012; P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, *Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings*, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 11–12.
- 93 Royal Commission consultation with Aboriginal and Torres Strait Islander community, 2014–2017.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, *Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 32.
- JP Wilson, B Drozdek & S Turkovic, 'Posttraumatic shame and guilt', *Trauma, Violence & Abuse*, vol 7, no 2, 2006, p 123; G Tremblay & P Turcote, 'Gender identity construction and sexual orientation in sexually abused males', *International Journal of Men's Health*, vol 4, no 2, 2005, p 141.
- 96 E Romano & RVD Luca, 'Male sexual abuse: A review of effects, abuse characteristics, and links with later psychological functioning', *Aggression and Violent Behavior*, vol 6, no 1, 2001, p 60.
- 97 S Craven, S Brown & E Gilchrist, 'Sexual grooming of children: Review of literature and theoretical considerations', Journal of Sexual Aggression, vol 12, no 3, 2006, p 296; E Romano & RVD Luca, 'Male sexual abuse: A review of effects, abuse characteristics, and links with later psychological functioning', Aggression and Violent Behavior, vol 6, no 1, 2001, p 61.
- 98 E Romano & RVD Luca, 'Male sexual abuse: A review of effects, abuse characteristics, and links with later psychological functioning', *Aggression and Violent Behavior*, vol 6, no 1, 2001, p 61.
- 99 Name changed, private session, 'Fletcher'.
- 100 Transcript of BVD, Case Study 35, 24 November 2015 at 13236:29–35.
- 101 Name changed, private session, 'Jarrett'.
- 102 Exhibit 29-0001, 'Statement of [BCB]', Case Study 29, STAT.0603.001.0001 R at 0015 R.
- 103 Exhibit 39-0010, 'Statement of [BXI]', Case Study 39, STAT.0977.001.0001 R at 0011 R-0012 R.
- E Romano & RVD Luca, 'Male sexual abuse: A review of effects, abuse characteristics, and links with later psychological functioning', *Aggression and Violent Behavior*, vol 6, no 1, 2001, pp 58–9.
- 105 Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study 26: The response of the Sisters of Mercy, the Catholic Diocese of Rockhampton and the Queensland Government to allegations of child sexual abuse at St Joseph's Orphanage, Neerkol, Sydney, 2016, p 45.
- 106 Exhibit 26-0001, 'Statement of AYB', Case Study 26, STAT.0531.001.0012_R at 0019_R.
- 107 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 189–91.
- 108 Name changed, private session, 'Lisa Michelle'.
- 109 Name changed, private session, 'Grant Lee'.
- 110 Transcript of Mr S M Hallett, Case Study 43, 7 September 2016 at C18127:32, C18129:40–41.
- 111 Name changed, private session, 'Dalton John'.
- 112 Exhibit 18-0001, 'Statement of AHA', Case Study 18, STAT.0367.001.0001_R at 0008_R.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 48.
- DM Fergusson, GFH McCleod & LJ Horwood, 'Childhood sexual abuse and adult developmental outcomes: Findings from a 30-year longitudinal study in New Zealand', *Child Abuse & Neglect*, vol 37, no 9, 2013, p 670.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 51–2.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 51–2.
- 117 Transcript of AWB, Case Study 27, 6 May 2015 at 14087:24–27.
- 118 Transcript of AWD, Case Study 27, 6 May 2015 at 14113:27–32.
- Name changed, private session, 'Nathan'.

160

- 120 Transcript of AAJ, Case Study 13, 11 June 2014 at ACT3004:39–43.
- 121 Transcript of AAJ, Case Study 13, 11 June 2014 at ACT3005:2–6.

- For example: Transcript of E J Fretton, Case Study 18, 10 October 2014 at 9477:18–22; Transcript of L P Knight, Case Study 19, 22 October 2014 at 10108:40–44; Transcript of ATQ, Case Study 23, 23 February 2015 at 11763:1–2; Transcript of S J Fisher, Case Study 36, 27 January 2016 at C15018:28–30.
- 123 JG Allen, Coping with trauma: Hope through understanding, American Psychiatric Publishing, 2005, p 154.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 50.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 50–1.
- 126 Name changed, private session, 'Ernst'.
- 127 Name changed, private session, 'Fiona'.
- 128 Name changed, private session, 'Sophia'.
- 129 Name changed, private session, 'Alanna'.
- 130 Exhibit 40-0002, 'Statement of Graeme Frazer', Case Study 40, STAT.0995.001.0001 at 0007.
- 131 Exhibit 40-0002, 'Statement of Graeme Frazer', Case Study 40, STAT.0995.001.0001 at 0010.
- 132 Exhibit 40-0002, 'Statement of Graeme Frazer', Case Study 40, STAT.0995.001.0001 at 0010.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 50–1.
- EO Paolucci, ML Genuis & C Violato, 'A meta-analysis of the published research on the effects of child sexual abuse', The Journal of Psychology, vol 135, no 1, 2001, p 28.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 51.
- American Psychiatric Association, *Diagnostic and statistical manual of mental disorders Fifth edition*, (DSM-5), American Psychiatric Publishing, Washington, 2013, pp 271–2.
- 137 Name changed, private session, 'Claire Grace'.
- 138 JG Allen, Coping with trauma: Hope through understanding, American Psychiatric Publishing, 2005, pp 174–5.
- 139 JG Allen, Coping with trauma: Hope through understanding, American Psychiatric Publishing, 2005, p 175.
- 140 Name changed, private session, 'Noeline'.
- Name changed, private session, 'Trina Beth'.
- Name changed, private session, 'Roy Leonard'.
- 143 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 70; L Wall & A Quadara, 'Acknowledging complexity in the impacts of sexual victimisation trauma', *ACSSA Issues*, no 16, 2014; J Herman, *Trauma and recovery: The aftermath of violence From domestic abuse to political terror*, Basic Books, New York, 1992, pp 118–22.
- JG Allen, Coping with trauma: Hope through understanding, American Psychiatric Publishing, 2005, p 208.
- 145 JG Allen, Coping with trauma: Hope through understanding, American Psychiatric Publishing, 2005, p 208.
- 146 Name changed, private session, 'Marjorie'.
- 147 Name changed, private session, 'Charlene'.
- 148 Exhibit 20-0005, 'Statement of AOE', Case Study 20, STAT.0401.001.0001_R at 0001_R, 0002_R and 0004_R.
- 149 Exhibit 20-0005, 'Statement of AOE', Case Study 20, STAT.0401.001.0001_R at 0007_R.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 152.
- 151 Commission to Inquire into Child Abuse, Final report of the Commission to Inquire into Child Abuse, Ireland, 2009, Volume 3, p 222.
- 152 CP Carr, CMS Martins, AM Stingel, VB Lembruber & MF Juruena, 'The role of early life stress in adult psychiatric disorders', *Journal of Nervous and Mental Disease*, vol 201, no 12, 2013, p 1018. Research studies showed associations between child sexual abuse and personality disorders (especially with borderline personality disorder), schizophrenia, anxiety disorders (especially with post-traumatic stress disorder, panic disorder, agoraphobia, and obsessive-compulsive disorder), substance abuse disorders, mood disorders (especially major depression and bipolar illness), disruptive behaviour disorders, eating disorders, and dissociative disorders.
- 153 CP Carr, CMS Martins, AM Stingel, VB Lembruber & MF Juruena, 'The role of early life stress in adult psychiatric disorders', *Journal of Nervous and Mental Disease*, vol 201, no 12, 2013, p 1007.
- MC Cutajar, PE Mullen, JRP Ogloff, SD Thomas, DL Wells & J Spataro, 'Psychopathology in a large cohort of sexually abused children followed up to 43 years', *Child Abuse & Neglect*, vol 34, no 11, 2010, p 820; CP Carr, CMS Martins, AM Stingel, VB Lembruber & MF Juruena, 'The role of early life stress in adult psychiatric disorders', *Journal of Nervous and Mental Disease*, vol 201, no 12, 2013, p 1018.
- See also AP Association, *Diagnostic and statistical manual of mental disorders, Fifth edition (DSM-5)*, Amercian Psychiatric Association, Washington, DC, 2013, pp 663–6.

- MC Cutajar, PE Mullen, JRP Ogloff, SD Thomas, DL Wells & J Spataro, 'Psychopathology in a large cohort of sexually abused children followed up to 43 years', *Child Abuse & Neglect*, vol 34, no 11, 2010, p 820.
- 157 Name changed, private session, 'Ryan Ross'.
- JG Allen, *Traumatic relationships and serious mental disorders*, John Wiley & Sons, Chichester, 2002, pp 163–7.
- 159 JG Allen, Traumatic relationships and serious mental disorders, John Wiley & Sons, Chichester, 2002, pp 174, 187.
- 160 JG Allen, Traumatic relationships and serious mental disorders, John Wiley & Sons, Chichester, 2002, p 178.
- J Herman, Trauma and recovery: The aftermath of violence From domestic abuse to political Terror, Basic Books, New York, 1992, p 100.
- B Perry, RA Pollard, TL Blaicley, WL Baker & D Vigilante, 'Childhood trauma, the neurobiology of adaptation, and 'use-dependent' development of the brain: How "states" become "traits", *Infant Mental Health Journal*, vol 16, no 4, 1995, p 282.
- Name changed, private session, 'Lloyd Edward'.
- J Herman, *Trauma and recovery: The aftermath of violence From domestic abuse to political terror*, Basic Books, New York, 1992, p 239.
- 165 C Kezelman & P Stavropoulos, *Practice guidelines for treatment of complex trauma and trauma informed care and service delivery*, Adults Surviving Child Abuse, 2012, p 102.
- 166 JG Allen, *Coping with trauma: Hope through understanding*, American Psychiatric Publishing, 2005, p 194. See also AP Association, *Diagnostic and statistical manual of mental disorders, Fifth edition (DSM-5)*, American Psychiatric Association, Washington, DC, 2013, pp 292–8.
- 167 Name changed, private session, 'Dion'.
- 168 JG Allen, Coping with trauma: Hope through understanding, American Psychiatric Publishing, 2005, p 197.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 53; K Kendall-Tackett, 'The health effects of childhood abuse: Four pathways by which abuse can influence health', *Child Abuse & Neglect*, vol 26, no 6, 2002, p 718.
- 170 Name changed, private session, 'Edna'.
- For example: Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland, Sydney, 2015, p 8; Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 17: The response of the Australian Indigenous Ministries, the Australian and Northern Territory governments and the Northern Territory police force and prosecuting authorities to allegations of child sexual abuse which occurred at the Retta Dixon Home, Sydney, 2015, p 23; Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 21: The response of Satyananda Yoga Ashram at Mangrove Mountain to allegations of child sexual abuse by the ashram's former spiritual leader in the 1970s and 1980s, Sydney, 2016, p 33; Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 32: The response of Geelong Grammar School to allegations of child sexual abuse of former students, Sydney, 2015, pp 21–22, 25.
- 172 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland*, Sydney, 2015, p 37.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland,* Sydney, 2015, p 8.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland,* Sydney, 2015, p 8.
- 175 Name changed, private session, 'Renny'.
- 176 JG Allen, Traumatic Relationships and Serious Mental Disorders, John Wiley & Sons, Chichester, 2002, p 219.
- 177 JG Allen, Coping with Trauma: Hope through Understanding, American Psychiatric Publishing, 2005, p 210.
- 178 Exhibit 28-0006, 'Statement of BAS', Case Study 28, STAT.0570.001.0006_R at 0006_R.
- 179 Transcript of JH, Case Study 10, 2 April 2014 at 6971:35–39; Exhibit 10-0011, 'Statement and Annexures of JH', Case Study 20, STAT.0206.001.0001 M R at 0003 M R and 0008 M R.
- Australian Human Rights Commission, *Children's Rights Report 2014*, Australian Human Rights Commission, Sydney, 2014, pp 74, 76–7.
- Australian Human Rights Commission, *Children's Rights Report 2014*, Australian Human Rights Commission, Sydney, 2014, p 78.
- 182 JG Allen, Coping with Trauma: Hope through Understanding, American Psychiatric Publishing, 2005, p 210.
- For example: Name changed, private session, 'Kate'; Name changed, private session, 'Ilana'; Name changed, private session, 'Dominik'.
- 184 Name changed, private session, 'Lucy'.
- 185 JG Allen, Traumatic Relationships and Serious Mental Disorders, John Wiley & Sons, Chichester, 2002, p 205.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 50–1.
- DM Fergusson, JM Boden & LJ Horwood, 'Exposure to childhood sexual and physical abuse and adjustment in early adulthood', *Child Abuse & Neglect*, vol 32, 2008, p 617.
- HA Bergen, G Martin, AS Richardson, S Allison & L Roeger, 'Sexual abuse and suicidal behavior: A model constructed from a large community sample of adolescents', *Journal of the American Academy of Child and Adolescent Psychiatry*, vol 42, no 11, 2003, p 1304.

- MC Cutajar, PE Mullen, JR Ogloff, SD Thomas, DL Wells & J Spataro, 'Suicide and fatal drug overdose in child sexual abuse victims: A historical cohort study', *Medical Journal of Australia*, vol 192, no 4, 2010.
- 190 Transcript of M Cutajar, Case Study 57, 30 March 2017 at 27735:38–27736:1.
- 191 Name changed, private session, 'Adeline'.
- 192 Name changed, private session, 'Simeon Paul'.
- 193 Exhibit 29-0006, 'Statement of [BCG]', Case Study 29, STAT.0590.001.0001_R at 0018_R.
- 194 Exhibit 28-0007, 'Statement of BAP', Case Study 28, STAT.0578.001.0001 R at 0006 R-0007 R.
- 195 Transcript of A P Nash, Case Study 43, 6 September 2016 at C18042-40-C18043-10.
- BE Molnar, LF Berkman & SL Buka, 'Psychopathology, childhood sexual abuse and other childhood adversities: Relative links to subsequent suicidal behaviour in the US', Psychological Medicine, vol 31, no 6, 2001; MC Cutajar, PE Mullen, JR Ogloff, SD Thomas, DL Wells & J Spataro, 'Suicide and fatal drug overdose in child sexual abuse victims: A historical cohort study', Medical Journal of Australia, vol 192, no 4, 2010.
- MC Cutajar, PE Mullen, JR Ogloff, SD Thomas, DL Wells & J Spataro, 'Suicide and fatal drug overdose in child sexual abuse victims: A historical cohort study', *Medical Journal of Australia*, vol 192, no 4, 2010, p 185.
- 198 Transcript of P Nagle, Case Study 28, 19 May 2015 at C8141:6–9; Transcript of BAC, Case Study 28, 19 May 2015 at C8151:40–43.
- 199 Transcript of PF Nagle, Case Study 28, 19 May 2015 at C8141:6–9.
- 200 Exhibit 28-0011, 'Statement of BAV', Case Study 28, STAT.0584.001.0001_R at 0011_R.
- 201 Transcript of CQT, Case Study 43, 6 September 2016 at C18033:44–47-C18034:1–13.
- J Cashmore & R Shackel, *The long-term effects of child sexual abuse*, Australian Institute of Family Studies, Melbourne, 2013, p 13.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 59.
- J Cashmore & R Shackel, *The long-term effects of child sexual abuse*, Australian Institute of Family Studies, Melbourne, 2013, p 13.
- MC Cutajar, PE Mullen, JR Ogloff, SD Thomas, DL Wells & J Spataro, 'Suicide and fatal drug overdose in child sexual abuse victims: A historical cohort study', *Medical Journal of Australia*, vol 192, no 4, 2010, p 186.
- Family and Community Development Committee, *Betrayal of trust: Inquiry into handling of child abuse by religious* and other non-government organisations, Victoria, 2013, vol 1, p 75; T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 60.
- MC Cutajar, PE Mullen, JRP Ogloff, SD Thomas, DL Wells & J Spataro, 'Psychopathology in a large cohort of sexually abused children followed up to 43 years', *Child Abuse & Neglect*, vol 34, no 11, 2010.
- 208 Transcript of CJV, Case Study 40, 23 June 2016 at 19570:40–43.
- 209 Transcript of CJV, Case Study 40, 23 June 2016 at 19570:40–43.
- 210 Transcript of AW Ollerenshaw, Case Study 23, 5 March 2015 at 12718:26–30.
- Name changed, private session, 'Zahara'.
- 212 Transcript of AIH, Case Study 19, 22 October 2014 at 10093:28–31; Exhibit 19-5, 'Statement of AIH', Case Study 19, STAT.0343.001.0001 R at 5:27.
- 213 Transcript of AIH, Case Study 19, 22 October 2014 at 10094:8–11; Exhibit 19-5, 'Statement of AIH', Case Study 19, STAT.0343.001.0001_R at 5:32.
- Name changed, private session, 'Gabriel'.
- 215 Exhibit 43-0018, 'Statement of CNQ', Case Study 43, STAT.1128.001.0001_R at 0002_R.
- 216 Name changed, private session, 'Odie'.
- 217 Name changed, private session, 'Odie'.
- 218 Name changed, private session, 'Norm'.
- 219 M Griffiths, Submission to the Royal Commission into Institutional Responses to Child Sexual Abuse, *Issues paper No 10: Advocacy and support and therapeutic treatment services*, 2015.
- J Cashmore & R Shackel, *The long-term effects of child sexual abuse,* Australian Institute of Family Studies, Melbourne, 2013, p 2.
- 221 Transcript of B Perry, Case Study 57, 30 March 2017 at 27708:2–13.
- J Cashmore & R Shackel, *The long-term effects of child sexual abuse,* Australian Institute of Family Studies, Melbourne, 2013, p 14.
- J Cashmore & R Shackel, *The long-term effects of child sexual abuse,* Australian Institute of Family Studies, Melbourne, 2013, p 14.
- See, for example, Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 3:
 Anglican Diocese of Grafton's response to child sexual abuse at the North Coast Children's Home, Sydney, 2014, p 14;
 Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 5: Response of The
 Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland, Sydney, 2015, p 8; Royal
 Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 8: Mr John Ellis's experience of
 the Towards Healing process and civil litigation, Sydney, 2015, p 21; Royal Commission into Institutional Responses to
 Child Sexual Abuse, Report of Case Study No 17: The Response of the Australian Indigenous Ministries, the Australian and
 Northern Territory governments and the Northern Territory police force and prosecuting authorities to allegations of child
 sexual abuse which occurred at the Retta Dixon Home, Sydney, 2015, pp 22, 24, 28.

- Transcript of R J Cummings, Case Study 30, 18 August 2015 at C8950:46–C8951:2.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 56–7.
- O Rumstein-McKean & J Hunsley, 'Interpersonal and family functioning of female survivors of childhood sexual abuse', Clinical Psychology Review, vol 21, no 3, 2001; T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 56–7.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 55–6.
- 229 Name changed, private session, 'Sean Peter'.
- 230 Exhibit 34-0002, 'Statement of BQG', Case Study 34, STAT.0723.001.0001_R at 0011 R.
- 231 Exhibit 28-0014, 'Statement of BAB', Case Study 28, STAT.0582.001.0001 R at 0007 R.
- Name changed, private session, 'Vittorio'.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 57–8.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 57.
- DA Wolfe, KJ Francis & AL Straatman, 'Child abuse in religiously-affiliated institutions: Long-term impact on men's mental health', Child Abuse & Neglect, vol 30, no 2, 2006, p 209; T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 57–8.
- Name changed, private session, 'Francesca'.
- Name changed, private session, 'Cedric'.
- 238 Exhibit 26-0014, 'Statement of AYE', Case Study 26, STAT.0532.001.0013 R at 0013 R.
- 239 Exhibit 34-0012, 'Statement of [BQF]', Case Study 34, STAT0732.002.0001_R at 0008_R.
- Name changed, private session, 'Eliza'.
- 241 Exhibit 36-0012, 'Statement of David Gould', Case Study 36, STAT.0823.001.0001 R at 0015 R.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 58–9.
- R Roberts, T O'Connor, J Dunn, J Golding & TAS Team, 'The effects of child sexual abuse in later family life: Mental health, parenting and adjustment of offspring', *Child Abuse & Neglect*, vol 28, 2004; T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 58.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 58–9.
- 245 C Tarczon, 'Mothers with a history of childhood sexual abuse: Key issues for child protection practice and policy', ACSSA Research Summary No. 2, 2012.
- Name changed, private session, 'Hany'.
- 247 Exhibit 40-0001, 'Statement of CJA', Case Study 40, STAT.0993.001.0001_R at 0015_R.
- Name changed, private session, 'Hayley'.
- 249 Name changed, private session, 'Casey'.
- Name changed, private session, 'Ramona'.
- 251 R Price-Robertson, Fathers with a history of child sexual abuse, Australian Institute of Family Studies, Canberra, 2012, p 1.
- Name changed, private session, 'Landon'.
- Name changed, private session, 'Dane'.
- Name changed, private session, 'Dane'.
- Name changed, private session, 'Rosswell'.
- 256 Transcript of A Lockhart, Case Study 28, 21 May 2015 at C8358:39–40.
- 257 Transcript of A Lockhart, Case Study 28, 21 May 2015 at C8358:15–17.
- 258 Transcript of A Lockhart, Case Study 28, 21 May 2015 at C8361:6–18.
- 259 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 176.
- Name changed, private session, 'Travis'.
- Exhibit 39-0010, 'Statement of [BXI]', Case Study 39, STAT.0977.001.0001_R at 0014_R. See also: Name changed, private session, 'Earl'.
- Name changed, private session, 'Cecilia'.
- Name changed, private session, 'Ronald'.

- J Cashmore & R Shackel, *The long-term effects of child sexual abuse*, Australian Institute of Family Studies, Melbourne, 2013, p 18; VJ Felitti, RF Anda, D Nordenberg, DF Williamson, AM Spitz, V Edwards, MP Koss & JS Marks, 'Relationship of childhood abuse and household dysfunction to many of the leading causes of death in adults: The adverse childhood experiences (ACE) study', *American Journal of Preventive Medicine*, vol 14, no 4, 1998; VJ Felitti & RF Anda, 'The relationship of adverse childhood experiences to adult medical disease, psychiatric disorders and sexual behavior: Implications for healthcare' in RA Lanius, E Vermetten and C Pain (eds), *The impact of early life trauma on health and disease: The hidden epidemic*, Cambridge University Press, New York, 2010.
- VJ Felitti, RF Anda, D Nordenberg, DF Williamson, AM Spitz, V Edwards, MP Koss & JS Marks, 'Relationship of childhood abuse and household dysfunction to many of the leading causes of death in adults: The adverse childhood experiences (ACE) study', *American Journal of Preventive Medicine*, vol 14, no 4, 1998, p 250.
- VJ Felitti, RF Anda, D Nordenberg, DF Williamson, AM Spitz, V Edwards, MP Koss & JS Marks, 'Relationship of childhood abuse and household dysfunction to many of the leading causes of death in adults: The adverse childhood experiences (ACE) study', *American Journal of Preventive Medicine*, vol 14, no 4, 1998, p 251.
- D Runyan, C Wattam, R Ikeda, F Hassan & L Ramiro, 'Child abuse and neglect by parents and other caregivers' in EG Krug, L Dahlberg, JA Mercy, AB Zwi and R Lozano (eds), *World report on violence and health*, World Health Organisation, Geneva, 2002, p 69.
- For a list of these inquiries, see T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings*, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 53.
- 269 Name changed, private session, 'Rory'.
- 270 Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study 26: The response of the Sisters of Mercy, the Catholic Diocese of Rockhampton and the Queensland Government to allegations of child sexual abuse at St Joseph's Orphanage, Neerkol, Sydney, 2016, p 52.
- Name changed, private session, 'Ramona'.
- 272 Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 11: Congregation of Christian Brothers in Western Australia response to child sexual abuse at Castledare Junior Orphanage, St Vincent's Orphanage Clontarf, St Mary's Agricultural School Tardun and Bindoon Farm School, Sydney, 2014, p 23.
- 273 Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 11: Congregation of Christian Brothers in Western Australia response to child sexual abuse at Castledare Junior Orphanage, St Vincent's Orphanage Clontarf, St Mary's Agricultural School Tardun and Bindoon Farm School, Sydney, 2014, p 23.
- 274 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland*, Sydney, 2015, pp 8, 23.
- 275 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland,* Sydney, 2015, p 31.
- Name changed, private session, 'Sean Michael'.
- 277 Exhibit 40-0015, 'Statement of Daryl James', Case Study 40, STAT.1001.001.0001 at 13.
- Name changed, private session, 'Bert'.
- 279 Exhibit 39-0009, 'Statement of Troy Quagliata', Case Study 39, STAT.0975.001.0001_R at 0012_R.
- 280 Private session attendee, 2015.
- Name changed, private session, 'Clive'.
- 282 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No. 39: The response of certain football (soccer), cricket and tennis organisations to allegations of child sexual abuse,* Sydney, 2017, p 26.
- 283 Exhibit 39-0005, 'Statement of [BXA]', Case Study 39, STAT.0970.001.0001_R at 0016_R-0017_R.
- For example: Name changed, private session, 'Kirsty Simone'; Name changed, private session, 'Melita'; Name changed, private session, 'Rainey'.
- 285 Transcript of W P Patton, Case Study 7, 26 February 2014 at 4923:23–24.
- Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 30: The response of Turana, Winlaton and Baltara, and the Victoria Police and the Department of Health and Human Services Victoria to allegations of child sexual abuse, Sydney, 2016, p 69.
- Name changed, private session, 'Corey'.
- 288 Name changed, private session, 'Maggie'.
- For example: Name changed, private session, 'Melita'; Name changed, private session, 'Hazel'; Name changed, private session, 'Miranda'; Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 30: The response of Turana, Winlaton and Blatara, and the Victoria Police and the Department of Health and Human Services Victoria to allegations of child sexual abuse, Sydney, 2016, p 69; Transcript of J Y Mulquiney, Case Study 7, 27 February 2014 at 4978:31.
- 290 Transcript of J Y Mulquiney, Case Study 7, 27 February 2014 at 4981:4–8.
- Name changed, private session, 'Dee'.
- Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 7: Child sexual abuse at the Parramatta Training School for Girls and the Institution for Girls in Hay, Sydney, 2014, p 31. See also: Name changed, private session, 'Miranda'.
- 293 Name changed, private session, 'Christine'.

- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 52.
- J Cashmore & R Shackel, *The long-term effects of child sexual abuse*, Australian Institute of Family Studies, Melbourne, 2013, p 18.
- VJ Felitti & RF Anda, 'The relationship of adverse childhood experiences to adult medical disease, psychiatric disorders and sexual behavior: Implications for healthcare' in RA Lanius, E Vermetten and C Pain (eds), *The impact of early life trauma on health and disease: The hidden epidemic*, Cambridge University Press, New York, 2010.
- E McCrory, SA De Brito & E Viding, 'Research review: The neurobiology and genetics of maltreatment and adversity', Journal of Child Psychology and Psychiatry, vol 51, no 10, 2010, pp 1083–5.
- 298 RL Gaskill & BD Perry, 'Child sexual abuse, traumatic experiences, and their impact on the developing brain' in P Goodyear-Brown (ed), *Handbook of child sexual abuse: Identification, assessment, and treatment*, John Wiley & Sons, Hoboken, New Jersey, 2012, p 34.
- 299 Transcript of L Newman, Case Study 57, 30 March 2017 at 27777:41–27778:10.
- 300 Transcript of AOA, Case Study 57, 30 March 2017 at 27766:40–27767:1.
- 301 Name changed, private session, 'Amber'; I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 179.
- 302 Name changed, private session, 'Clement Gerald'.
- 303 For example: Name changed, private session, 'Hallie'.
- Name changed, private session, 'Edyth'; See also survivor descriptions of internal medical examinations at Winlaton Youth Training Centre in Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study 30: The response of Turana, Winlaton and Baltara, and the Victoria Police and the Department of Health and Human Services Victoria to allegations of child sexual abuse,* Sydney, 2016, p 39, and Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 7: Child sexual abuse at the Parramatta Training School for Girls and the Institution for Girls in Hay*, Sydney, 2014, pp 15, 30.
- Exhibit 30-0023, 'Statement of Gabrielle Rose Short', Case Study 30, STAT.0647.001.0001_M_R at 0009_M_R.
- 306 Exhibit 30-0023, 'Statement of Gabrielle Rose Short', Case Study 30, STAT.0647.001.0001_M_R at 0004_M_R-0005_M_R and 0010_M_R.
- 307 Name changed, private session, 'Gene'.
- 308 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 27: The response of health care service providers and regulators in New South Wales and Victoria to allegations of child sexual abuse,* Sydney, 2016, p 27.
- 309 Name changed, private session, 'Noeline'.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 57–8.
- D Lisak, 'The psychological impact of sexual abuse: Content analysis of interviews with male survivors', *Journal of Traumatic Stress*, vol 7, no 4, 1994, pp 534–5; PJ Isely, P Isely, J Freiburger & R McMackin, 'In their own voices: A qualitative study of men abused as children by Catholic clergy', *Journal of Child Sexual Abuse*, vol 17, no 3–4, 2008, pp 205, 209; J Cashmore & R Shackel, 'Gender differences in the context and consequences of child sexual abuse', *Current Issues in Criminal Justice*, vol 26, no 1, 2014, pp 77; E Romano & RVD Luca, 'Male sexual abuse: A review of effects, abuse characteristics, and links with later psychological functioning', *Aggression and Violent Behavior*, vol 6, no 1, 2001, pp 64–5; T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings*, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 78.
- 312 M Eskin, H Kaynak-Demir & S Demir, 'Same-sex sexual orientation, childhood sexual abuse, and suicidal behaviour in university students in Turkey'. *Archives of Sexual Behaviour*, vol 34, no 2, 2005.
- PJ Isely, P Isely, J Freiburger & R McMackin, 'In their own voices: A qualitative study of men abused as children by Catholic clergy', *Journal of Child Sexual Abuse*, vol 17, no 3–4, 2008, pp 205, 209.
- D Sauvage & P O'Leary, 'Child sexual abuse in faith-based institutions: Gender, spiritual trauma and treatment frameworks' in Y Smaal, A Kaladelfos and M Finnane (eds), *The sexual abuse of children: Recognition and redress,* Monash University Publishing, Clayton, Victoria, 2016, p 147.
- 315 S Crome, *Male survivors of sexual assault and rape*, Australian Institute of Family Studies, Melbourne, 2006; M Kia-Keating, FK Grossman, L Sorsoli & M Epstein, 'Containing and resisting masculinity: Narratives of renegotiation among resilient male survivors of childhood sexual abuse', *Psychology of Men and Masculinity*, vol 6, no 3, 2005, p 179.
- PJ Isely, P Isely, J Freiburger & R McMackin, 'In their own voices: A qualitative study of men abused as children by Catholic clergy', *Journal of Child Sexual Abuse*, vol 17, no 3–4, 2008, p 205.
- 317 Transcript of C Quadrio, Case Study 28, 25 May 2015 at C8465:39–45.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 27: The response of health care service providers and regulators in New South Wales and Victoria to allegations of child sexual abuse,* Sydney, 2016, p 26.
- 319 Transcript of CJC, Case Study 40, 23 June 2016 at 19546:42–47.
- 320 Name changed, private session, 'James Peter'.

- 321 S Craven, S Brown & E Gilchrist, 'Sexual grooming of children: Review of literature and theoretical considerations', Journal of Sexual Agaression, vol 12, no 3, 2006, p 296.
- 322 Transcript of WP, Case Study 12, 19 May 2014 at WA2335:20–26.
- 323 Transcript of WP, Case Study 12, 19 May 2014 at WA2335:39–42.
- 324 Name changed, private session, 'Mathew'.
- 325 Name changed, private session, 'Ross William'.
- 326 Name changed, private session, 'Abby'.
- 327 Transcript of B Manning, Case Study 21, 4 December 2014 at 11079:35–40.
- 328 S Murray & A Powell, Sexual assault and adults with a disability: Enabling recognition, disclosure and a just response, Australian Centre for the study of Sexual Assault, Melbourne, Australia, 2008, p 5.
- 329 For example: Name changed, private session, 'John Peter'; Name changed, private session, 'Rod Paul'.
- 330 Name changed, private session, 'Alan Eric'.
- Australian Human Rights Commission, *Sexual orientation, gender identity and intersex rights: Snapshot report,* Australian Human Rights Commission, Sydney, 2014, p 1.
- 332 For example: Name changed, private session, 'Emory'; Name changed, private session, 'Lexi'.
- 333 Name changed, private session, 'Xanthe'.
- Name changed, private session, 'Emory'.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 57–8; TE Senn, MP Carey & PA Vanable, 'Childhood and adolescent sexual abuse and subsequent sexual risk behavior: Evidence from controlled studies, methodological critique, and suggestions for research', *Clinical Psychology Review*, vol 28, 2008.
- K Kendall-Tackett, L Williams & D Finkelhor, 'The impact of sexual abuse on children: A review and synthesis of recent empirical studies', *Psychological Bulletin*, vol 113, no 1, 1993, p 168.
- 337 RO de Visser, PB Badcock, C Rissel, J Richters, AMA Smith, AE Grulich & JM Simpson, 'Experiences of sexual coercion in a representative sample of adults: The second Australian study of health and relationships', *Sexual Health*, vol 11, 2014, p 478.
- 338 Name changed, private session, 'Bridget'.
- 339 Name changed, private session, 'Alana'.
- Name changed, private session, 'Rebecca'.
- 341 Name changed, private session, 'Jacinta'.
- 342 Exhibit 40-0028, 'Statement of CJD', Case Study 40, STAT.1004.001.0001_R at 0009_R.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 21: The response of the Satyananda Yoga Ashram at Mangrove Mountain to allegations of child sexual abuse by the ashram's former spiritual leader in the 1970s and 1980s*, Sydney, 2016, p 32.
- 344 Exhibit 21-0002, 'Statement of Alecia Buchanan', Case Study 21, STAT.0425.001.0001_R at 0023 R.
- 345 Transcript of AOA, Case Study 57, 30 March 2017 at 27767:8–13.
- Name changed, private session, 'Declan'.
- Name changed, private session, 'Wayne Christopher'.
- 348 CS Widom & AM Ames, 'Criminal consequences of childhood sexual victimization', *Child Abuse & Neglect*, vol 18, no 4, 1994, p 311.
- 349 CS Widom & AM Ames, 'Criminal consequences of childhood sexual victimization', *Child Abuse & Neglect*, vol 18, no 4, 1994, p 311.
- 350 Exhibit 34-0004, 'Statement of BQS', Case Study 34, STAT.0716.001.0001_R at 0008_R.
- 351 Transcript of Y Kitchener, Case Study 7, 28 February 2014 at 5113:20–24.
- Transcript of Y Kitchener, Case Study 7, 28 February 2014 at 5132:41–45.
- 353 Transcript of Y Kitchener, Case Study 7, 28 February 2014 at 5113:28–34.
- 354 Transcript of Y Kitchener, Case Study 7, 28 February 2014 at 5133:5–7.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 19. See also S Zubrick, C Shepherd, P Dudgeon, G Gee, Y Paradies, C Scrine & R Walker, 'Social determinants of social and emotional wellbeing' in P Dudgeon, H Milroy and R Walker (eds), Working together: Aboriginal and Torres Strait Islander mental health and wellbeing principles and practice (2nd edition), Commonwealth of Australia, Canberra, 2010, p83; L Funston, 'Aboriginal and Torres Strait Islander worldviews and cultural safety transforming sexual assault service provision for children and young people', International Journal Environmental Research and Public Health, vol 10, 2013, p 3818.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 19, citing Funston 2013.
- J Atkinson, J Nelson, R Brooks, C Atkinson & K Ryan, 'Addressing individual and community transgenerational trauma' in P Dudgeon, H Milroy and R Walker (eds), Working together: Aboriginal and Torres Strait Islander mental health and wellbeing principles and practices, Commonwealth of Australia, Canberra, 2014; P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 19–25.

- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 17: The response of the Australian Indigenous Ministries, the Australian and Northern Territory governments and the Northern Territory police force and prosecuting authorities to allegations of child sexual abuse which occurred at the Retta Dixon Home,* Sydney, 2015, p 21.
- Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 17: The response of the Australian Indigenous Ministries, the Australian and Northern Territory governments and the Northern Territory police force and prosecuting authorities to allegations of child sexual abuse which occurred at the Retta Dixon Home, Sydney, 2015, p 22.
- 360 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 17: The response of the Australian Indigenous Ministries, the Australian and Northern Territory governments and the Northern Territory police force and prosecuting authorities to allegations of child sexual abuse which occurred at the Retta Dixon Home,* Sydney, 2015, p. 32.
- 361 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 17: The response of the Australian Indigenous Ministries, the Australian and Northern Territory governments and the Northern Territory police force and prosecuting authorities to allegations of child sexual abuse which occurred at the Retta Dixon Home,* Sydney, 2015, p 31.
- 362 Name changed, private session, 'Carl'.
- 363 Name changed, private session, 'Carl'.
- Human Rights and Equal Opportunity Commission, *Bringing them home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, p 175.
- Human Rights and Equal Opportunity Commission, *Bringing Them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, p 175.
- 366 Royal Commission consultation with Aboriginal and Torres Strait Islander community 2014–2017.
- 367 Royal Commission consultation with Aboriginal and Torres Strait Islander community 2014–2017.
- 368 Royal Commission consultation with Aboriginal and Torres Strait Islander community 2014–2017.
- 369 Name changed, private session, 'Rainey'.
- Northern Territory Stolen Generations Aboriginal Corporation, Submission to the Royal Commission into Institutional Responses to Child Sexual Abuse, *Issues paper No 6: Redress schemes*, 2014, p 5; Secretariat of National Aboriginal and Islander Child Care, National Aboriginal and Torres Strait Islander Legal Service, Aboriginal Child, Family and Community Care State Secretariat, New South Wales, SA Aboriginal Family Support Services, Queensland Aboriginal and Torres Strait Islander Child Protection Peak & Victorian Aboriginal Child Care Agency, Submission to the Royal Commission into Institutional Responses to Child Sexual Abuse, *Issues paper 4: Preventing sexual abuse of children in out-of-home care*, 2013, p 2; Aboriginal Child, Family and Community Care State Secretariat, New South Wales, Submission to the Royal Commission into Institutional Responses to Child Sexual Abuse, *Issues paper No 10: Advocacy and support and therapeutic treatment services*, 2015, p 2.
- 371 Royal Commission multicultural public forums, 2016; Transcript of Jatinder Kaur, Case Study 57, 29 March 2017 at 27636:24–27639:1
- Name changed, private session, 'Aisha'.
- 373 Name changed, private session, 'Dinah'.
- Australian Muslim Women's Centre for Human Rights, Private submission in response to Royal Commission Multicultural Forums, April 2016, p 3.
- 375 SC Taylor & C Norma, 'The ties that bind: Family barriers for adult women seeking to report childhood sexual assault in Australia', *Women's Studies International Forum*, vol 37, 2013; MC Kenny & AG McEachern, 'Racial, ethnic, and cultural factors of childhood sexual abuse: A selected review of the literature', *Clinical Psychology Review*, vol 20, no 7, 2000; P Sawriker, *Working with ethnic minorities and across cultures in western child protection systems*, Routledge, London, 2017, p 103.
- 376 Royal Commission multicultural public forums, 2016.
- 377 Commonwealth of Australia, *Lost Innocents: Righting the record Report on child migration,* Commonwealth of Australia, Canberra, 2001.
- 378 Name changed, private session, 'Maria'.
- 379 T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 69.
- R McMackin, T Keane & P Kline, 'Introduction to special issue on betrayal and recovery: Understanding the trauma of clergy sexual abuse', *Journal of Child Sexual Abuse*, vol 17, no 3–4, 2008, p 198.
- 381 Name changed, private session, 'Justin James'.
- Name changed, private session, 'Ervin'.
- 383 Name changed, private session, 'Arnett'.
- 384 Transcript of G J McDonald, Case Study 43, 31 August 2016 at C17593:9 and C17593:27–33.
- Transcript of G J McDonald, Case Study 43, 31 August 2016 at 17601:17–18.
- 386 Transcript of G J McDonald, Case Study 43, 31 August 2016 at 17601:22–23.
- Name changed, private session, 'Dino'.

- 388 T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 69.
- 389 JJ Guido, 'A unique betrayal: Clergy sexual abuse in the context of the Catholic religious tradition', Journal of Child Sexual Abuse, vol 17, no 304, 2008.
- 390 JJ Guido, 'A unique betrayal: Clergy sexual abuse in the context of the Catholic religious tradition', Journal of Child Sexual Abuse, vol 17, no 304, 2008, p 261.
- 391 J Cornwell, The dark box: a secret history of confession, Basic Books, New York, 2014, pp 169–70.
- 392 Transcript of AAP, Case Study 13, 11 June 2014 at ACT3048:27-33.
- 393 Historian John Cornwell, in his history of confession, states that 'when the original grooming or confession occurs in the circumstances of confession, the auspices of the sacrament aggravate the harm. Sexual abuse of children linked to the confessional has not only been widespread but is known to be especially destructive to the children involved'. J Cornwell, The dark box: a secret history of confession, Basic Books, New York, 2014, p 70.
- 394 For example: Name changed, private session, 'Ned'; Name changed, private session, 'Alfred'; Name changed, private session, 'Elliott'.
- 395 Names changed, private session, 'Laurel' and 'Liana'.
- 396 Name changed, private session, 'Vince'.
- 397 Exhibit 11-0007, 'Statement of VG', Case Study 11, STAT.0229.001.0001_R at 0010_R.
- Transcript of AYB, Case Study 26, 14 April 2015 at C7316:19-26. 398
- 399 Transcript of JA Kiernan, Case Study 26, 15 April 2015 at C7315:45-C7316:2.
- Exhibit 36-0012, 'Statement of David Gould', Case Study 36, STAT.0823.001.0001 R at 0005 R.
- 401 Exhibit 22-0001, 'Statement of AVA', Case Study 22, STAT.0461.001.0001 R at 0006 R.
- 402 Name changed, private session, 'Cleo'.
- 403 Name changed, private session, 'Edmond'.
- 404 Names changed, private session, 'Laurel' and 'Liana'.
- 405 Transcript of BCB, Case Study 29, 27 July 2015 at 15168:2-9.
- 406 Name changed, private session, 'Carole'.
- 407 T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 68, citing Flynn 2008 and Collins et al. 2014.
- 408 Transcript of Baroness Hollins, Case Study 50, 23 February 2017 at 25950:26–35.
- 409 Transcript of Baroness Hollins, Case Study 50, 23 February 2017 at 25950:25–35.
- 410 T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 87.
- 411 Name changed, private session, 'Rea'.
- 412 Name changed, private session, 'Glenys Maree'.
- Name changed, private session, 'Naomi'. 413
- 414 Name changed, private session, 'Selina'. See also: Name changed, private session, 'Cordelia'.
- 415 Name changed, private session, 'Regina'.
- Exhibit 43-0015, 'Statement of Michael Balk', Case Study 43, STAT.1129.001.0001 at 0002–0004. Exhibit 43-0015, 'Statement of Michael Balk', Case Study 43, STAT.1129.001.0001 at 0006. 416
- 417
- 418 Name changed, private session, 'John Robert'.
- Exhibit 40-0017, 'Statement of CJU', Case Study 40, STAT.0987.001.0001 R at 0011 R. 419
- B Gallagher, 'The extent and nature of known cases of institutional child sexual abuse', British Journal of Social Work, 420 vol 30, no 6, 2000, pp 807–9; P O'Leary, E Koh & A Dare, Grooming and child sexual abuse in institutional contexts, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 15–16.
- 421 Name changed, private session, 'Clyde'.
- 422 Name changed, private session, 'Frances'.
- 423 Name changed, private session, 'Clement'.
- 424 Name changed, private session, 'Jordan'.
- 425 Name changed, private session, 'Jordan'.
- 426 T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 60-1.
- 427 JR Ogloff, MC Cutajar, E Mann & P Mullen, Child sexual abuse and subsequent offending and victimisation: A 45 year follow-up study, The Australian Institute of Criminology, Canberra, 2012, p 4.
- 428 T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 60; CS Widom, SJ Czaja & MA Dutton, 'Childhood victimization and lifetime revictimization', Child Abuse & Neglect, vol 32, no 785-796, 2008, p 790.
- 429 T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 60–1, citing Fleming et al. 1999, Nelson et al. 2002, Swanston et al. 2003.

- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 61; N Spröber, T Schneider, M Rassenhofer, A Seitz, H Liebhardt, L König & JM Fegert, 'Child sexual abuse in religiously affiliated and secular institutions: A restrospective descriptive analysis of data provided by victims in a government-sponsored reappraisal program in Germany', *BMC Public Health*, vol 14, no 282–294, 2014.
- 431 Commission to Inquire into Child Abuse, Final report of the commission to inquire into child abuse, Ireland, 2009, volume III, para 11.13; Community Affairs References Committee, Forgotten Australians: A report on Australians who experienced institutional or out-of-home care as children, Parliament House, Canberra, 2004, para 6.11.
- 432 Name changed, private session, 'Dora'.
- 433 Name changed, private session, 'Rosalie'.
- J Cashmore & R Shackel, *The long-term effects of child sexual abuse,* Australian Institute of Family Studies, Melbourne, 2013, p 17; T Butler, B Donovan, J Felming, M Levy & J Kaldor, 'Childhood sexual abuse among Australian prisoners', *Venereology*, vol 14, no 3, 2001, p 1.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 61–2.
- 436 JR Ogloff, MC Cutajar, E Mann & P Mullen, *Child sexual abuse and subsequent offending and victimisation: A 45 year follow-up study,* The Australian Institute of Criminology, Canberra, 2012; CS Widom & MA Ames, 'Criminal consequences of childhood sexual victimization', *Child Abuse & Neglect*, vol 18, no 4, 1994.
- 437 JR Ogloff, MC Cutajar, E Mann & P Mullen, *Child sexual abuse and subsequent offending and victimisation: A 45 year follow-up study,* The Australian Institute of Criminology, Canberra, 2012, pp v, 36, 48.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 62.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 61.
- 440 Name changed, private session, 'Rickey'.
- Transcript of A D Collins, Case Study 28, 21 May 2015 at C8379:32–37.
- Name changed, private session, 'Keith Michael'.
- Name changed, private session, 'Paul Andrew'.
- 444 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 184.
- Name changed, private session, 'Monica'.
- Name changed, private session, 'Kane Travis'.
- The percentage of victims of child sexual abuse that have a criminal offence history varies depending on the methodology of the research study. A small sample of recent research studies suggests a range from approximately one to 12 per cent of the study population. See C Leach, A Stewart & S Smallbone, 'Testing the sexually abused-sexual abuser hypothesis: A prospective longitudinal birth cohort study', Child Abuse & Neglect, vol 51, 2016, pp 144, 148, 152; JR Ogloff, MC Cutajar, E Mann & P Mullen, Child sexual abuse and subsequent offending and victimisation: A 45 year follow-up study, The Australian Institute of Criminology, Canberra, 2012, pp 47–8; D Salter, D McMillan, M Richards, T Talbot, J Hodges, A Bentovim, R Hastings, J Stevenson & D Skuse, 'Development of sexually abusive behaviour in sexually victimised males: A longitudinal study', The Lancet, vol 361, no 9356, 2003, p 473. The estimate will vary significantly between self-report and administrative data.
- 448 C Leach, A Stewart & S Smallbone, 'Testing the sexually abused-sexual abuser hypothesis: A prospective longitudinal birth cohort study', *Child Abuse & Neglect*, vol 51, 2016, p 150.
- JR Ogloff, MC Cutajar, E Mann & P Mullen, *Child sexual abuse and subsequent offending and victimisation: A 45 year follow-up study,* The Australian Institute of Criminology, Canberra, 2012, pp 3–4.
- 450 M Proeve, C Malvaso & P DelFabbro, Evidence and frameworks for understanding perpetrators of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 29–31.
- 451 M Proeve, C Malvaso & P DelFabbro, Evidence and frameworks for understanding perpetrators of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 31.
- 452 M Proeve, C Malvaso & P DelFabbro, *Evidence and frameworks for understanding perpetrators of institutional child sexual abuse*, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 29–30.
- K Kaufman & M Erooga, *Risk profiles for institutional child sexual abuse: A literature review,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 34. See also C Leach, A Stewart & S Smallbone, 'Testing the sexually abused-sexual abuser hypothesis: A prospective longitudinal birth cohort study', *Child Abuse & Neglect*, vol 51, 2016, pp 148, 152; D Salter, D McMillan, M Richards, T Talbot, J Hodges, A Bentovim, R Hastings, J Stevenson & D Skuse, 'Development of sexually abusive behaviour in sexually victimised males: A longitudinal study', *The Lancet*, vol 361, no 9356, 2003, p 473.

- 454 T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 62–3; C Porter, JS Lawson & ED Bigler, 'Neurobehavioral sequelae of child sexual abuse', Child Neuropsychology, vol 11, no 2, 2005; C Shakeshaft, Educator sexual misconduct: A synthesis of existing literature, US Department of Education, 2004, p 42.
- 455 JN Briere & DM Elliott, 'Immediate and long-term impacts of child sexual abuse', The Future of Children, vol 4, no 2, 1994, p 58.
- 456 T Blakemore, JL Herbert, F Arney & S Parkinson, Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 62–3; A Sadeh, RM Hayden, J McGuire, H Sachs & R Civita, 'Somatic, cognitive and emotional characteristics of abused children in a psychiatric hospital', Child Psychiatry and Human Development, vol 24, 1994; M Barrera, L Calderón & V Bell, 'The cognitive impact of sexual abuse and PTSD in children: A neuropsychological study', Journal of Child Sexual Abuse, vol 22, no 6, 2013, pp 626-7.
- 457 JP Mersky & J Topitzes, 'Comparing early adult outcomes of maltreated and non-maltreated children: A prospective longitudinal investigation', Children and Youth Services Review, no 32, 2009, p 1092.
- 458 Name changed, private session, 'Hany'.
- 459 Exhibit 43-0018, 'Statement of CNQ', Case Study 43, STAT.1128.001.0001_R at 0002_R.
- 460 Name changed, private session, 'Jared Wayne'.
- 461 Name changed, private session, 'Josh'.
- Exhibit 39-0009, 'Statement of Troy Quagliata', Case Study 39, STAT.0975.001.0001_R at 0002_R. 462
- Exhibit 39-0009, 'Statement of Troy Quagliata', Case Study 39, STAT.0975.001.0001 R at 0011 R. 463
- 464 Names changed, private session, 'Ingrid' and 'Marla'.
- 465
- Name changed, private session, 'Trina Jane'. Name changed, private session, 'Trina Jane'. 466
- 467 Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 26: The response of the Sisters of Mercy, the Catholic Diocese of Rockhampton and the Queensland Government to allegations of child sexual abuse at St Joseph's Orphanage, Neerkol, Sydney, 2016, p 42.
- 468 For example: Name changed, private session, 'Neville'.
- 469 For example: Name changed, private session, 'Lucian'; Name changed, private session, 'Seth'.
- L Forde, Report of the Commission of Inquiry into Abuse of Children in Queensland Institutions, Government of 470 Queensland, Queensland, 1999, p vi.
- 471 Transcript of R J Ellul, Case Study 11, 29 April 2014 at WA1647:20-30; Exhibit 11-0010, 'Statement of Raphael Ellul', Case Study 11, STAT.0235.001.0001_R at 0004_R.
- 472 Transcript of VG, Case Study 11, 29 April 2014 at WA 1609:22–29.
- 473 Name changed, private session, 'Clarence'.
- 474 Name changed, private session, 'Clarence'.
- 475 Name changed, private session, 'Carmen'.
- 476 Name changed, private session, 'Carmen'.
- 477 For example: Transcript of CAD, Case Study 37, 3 March 2016 at 16611:35-44.
- 478 For example: Exhibit 39-0010, 'Statement of [BXI]', Case Study 39, STAT.0977.001.0001_R at 0012_R.
- 479 Name changed, private session, 'Merle'.
- 480 Name changed, private session, 'Rachel'.
- Name changed, private session, 'Chris Patrick'. 481
- Name changed, private session, 'John'. 482
- Name changed, private session, 'Daniel Paul'. 483
- Exhibit 34-0002, 'Statement of BQG', Case Study 34, STAT.0723.001.0001 R at 0009 R. 484
- Exhibit 17-0002, 'Statement of Sandra Kitching', Case Study 17, STAT.0325.001.0001_R at 0011_R. 485
- 486 Name changed, private session, 'Nancie'.
- 487 R Scutella, A Chigavazira, E Killackey, N Herault, G Johnson, J Moschion & M Wooden, Findings from Waves 1 to 4: Special topics, Melbourne Institute of Applied Economic and Social Research, Melbourne, 2014, pp 81–2.
- Name changed, private session, 'Marcia'.
- 489 Name changed, private session, 'Murphy'.
- 490 Name changed, private session, 'Brendon'.
- 491 Name changed, private session, 'Dion'.

4 Impacts of institutional responses

I have come to terms with my abuse a long time ago ... what I have never really been able to come to terms with was the part society played — or didn't play, I guess, being the point. You know, the people turning the blind eye, the people not recognising things when they were in a position that they should have been educated to recognise. People not wanting to listen. People putting their own businesses, or money, or schools, above the health and wellbeing of a child. These are the things that I find really hard to forgive. ... the people that had the ability to do something about it and chose not to, they are the people that I have the more anger for.¹

The way institutions respond to child sexual abuse – including reactions to disclosure, action taken following abuse, and broader prevention and protection measures – has profound effects on victims. Institutional responses have the potential to either significantly compound or help alleviate the impacts of the abuse.² Responses include: those of the institution where the abuse took place; those of institutions with authority over, or responsibility for, that institution; and those of the police, criminal justice system, complaint and oversight bodies, support services, and health services. Often we heard about instances where two or more institutions did not respond appropriately to the sexual abuse of a child. For example, we were told about children sexually abused in out-of-home care who were failed by both their care provider and the child protection agency. Some victims were affected by multiple poor responses across a whole system. The effects on victims of institutional responses to child sexual abuse received limited attention before the Royal Commission. Previous research tended to focus on the impacts of the abuse itself (see Chapter 3).

Most people who came forward to the Royal Commission described how institutions responded to their experiences of abuse in ways that affected them negatively. For instance, Commissioners were often told about how an institution's response contributed to silence, shame and secrecy. However, survivors' experiences of institutional responses were not universally bad, with some telling us that the responses were a source of justice and support. Others spoke of mixed experiences of institutional responses and the subsequent effect on them. It was clear that victims' recovery and healing was tied to the nature of the response they received to disclosure and in seeking redress.

This chapter examines the impacts of institutional responses on victims of child sexual abuse. It discusses how poor responses by institutions can constitute institutional betrayal. It then discusses other commonly reported impacts of such poor responses, namely:

- continuation of sexual abuse
- re-traumatisation
- fear, distrust and contempt
- ostracism.

It also sets out what Commissioners were told about the outcomes for victims who felt supported when institutions responded appropriately to child sexual abuse.

Chapter 5 examines the impacts of child sexual abuse and the institutional responses to that abuse on secondary victims, such as family members and whistleblowers.

4.1 Institutional betrayal

Many survivors of child sexual abuse who came forward to the Royal Commission described the important role played in their lives by the institutions in which the sexual abuse occurred. For some, their involvement in an institution meant community and family belonging, educational opportunities, recreational activities, or religious or spiritual guidance. Some valued being connected to prestigious institutions with good reputations.³ For others, the institution had been granted legal guardianship by the state. Many survivors told us about their unquestioning trust in institutions before abuse occurred.⁴ They expected institutions to fulfil their duty of care to children – and, indeed, many institutions were required by law to do so.

This was not the case for all survivors. For those with prior experiences of discrimination and mistreatment by institutions, this trust had already been undermined. This was the case for many Aboriginal and Torres Strait Islander people, people from culturally and linguistically diverse backgrounds, people with refugee backgrounds and people who were care-leavers (spent time in out-of-home care as a child).

Some survivors who did trust in institutions at the time of the abuse told Commissioners they expected the institution's responses to be effective, compassionate and just. We were often told about the importance of being listened to and believed, and of the institution acknowledging and accepting responsibility for failing to protect children in their care. The prevention of further abuse and protection of other children was essential. Many survivors wanted perpetrators to face consequences, including criminal convictions, for their actions. For others, a meaningful apology, compensation and redress from the institution were critical (see our *Redress and civil litigation* report). Counselling and support were also key for many survivors (see Volume 9, *Advocacy, support and therapeutic treatment services*).

Yet a great number of survivors spoke about being betrayed by the very institutions they once trusted. This 'institutional betrayal' — when trusted and powerful institutions act in ways that harm those dependent on them for safety and wellbeing — can exacerbate the trauma of abuse. It can also have its own distinct impacts. Dr Brett Ferguson, psychiatrist, described in Case Study 57: Nature, cause and impact of child sexual abuse in institutional contexts (Nature, cause and impact of child sexual abuse) how a poor institutional response adds another layer to the impacts of child sexual abuse:

My experience has been that [institutional child sexual abuse] tends to add a different — an extra layer to the adverse impact, because it is not just the individual person who abused them to take into account, but also the institutional response, which is sometimes, unfortunately, a lack of response in terms of denial, a lack of acknowledgment, a covering up of, and that is just added to the sense of mistrust and anger and the other emotional aspects that go with having been sexually abused and not being believed and taken seriously.⁸

Institutional betrayal is a concept reflected in many of the experiences described by survivors who shared their stories with us. Some survivors told us how institutions failed to protect them from abuse in the first place, or did not respond appropriately when it occurred. Others described how institutions responded in ways that were actively damaging, for example, by perpetuating the abuse or punishing them for disclosing it. Responses that fail victims exacerbate a sense of injustice and betrayal, and foster resentment and anger. Many survivors described how institutional betrayal aggravated their anxiety, trauma and dissociation resulting from the abuse. 'Johnny', who told us he was abused as a schoolboy in the 1970s by the principal at his Catholic school, said the Catholic Church's response many years later led him 'to question my worth again, like I did when I was a child'.9

Many survivors of historical abuse described how, when they were young, it was commonly accepted that 'children should be seen but not heard'. This meant they were silenced or disbelieved when they disclosed the sexual abuse, exacerbating their feelings of helplessness, anger and shame. Many who tried to disclose the abuse when they were children were told that they were liars, and some were punished as a result. We heard this response frequently in private sessions and public hearings. Volume 4, *Identifying and disclosing child sexual abuse* discusses in further detail the impacts of disclosure.

One survivor told us 'most survivors just want to be listened to, because when they were a child and tried to tell they didn't have a voice'.¹¹ Research suggests that victims' experiences of being believed may have changed over time; children are now more likely to be believed when disclosing, particularly by their parents, although this is not a universal experience.¹²

Poor institutional responses can also create a significant barrier to disclosure.¹³ In the *Nature,* cause and impact of child sexual abuse case study, survivor Ms Shelly Braieoux told us that she was ostracised and silenced from speaking about her abuse to anyone. As a result of not being able to deal with it at the time, she said she was forced to internalise the pain, grief, anger, betrayal, worthlessness and shame she felt:

There was no support from the religion, my family or friends, not one person would stand by my side to help, protect or advocate for me. In fact, it is the injustice of this treatment that drives me and it is a very big part of the reason why I have come forward today. It was a very callous way to be treated and, in my opinion, it has been far worse and caused more damage to me than the abuse itself. No survivor of abuse should be made to feel this way.¹⁴

Commissioners heard of many instances where victims were blamed for the abuse, or the significance of the abuse was minimised. Survivors reflected that this led to feelings of hopelessness, confusion and self-blame. Research indicates responses to disclosure can enhance or hinder recovery depending on the level of victim validation and attribution of blame to the perpetrator. We heard from 'Maree', who told us that her early life and marriage were ruined by her childhood sexual abuse at the hands of the local Catholic curate. Although the perpetrator had said that nobody would believe her, 'Maree' told us she did try to disclose her abuse at confession sometime in the late 1950s or early 1960s:

I told the priest what had happened and he screamed at me and he said it was my fault. He blamed me. I came out of the confessional box and I remember, with just my head in my hands ... I vowed that I would never, ever speak of it again.¹⁶

Other survivors described being told that they had sinned and needed to make a confession when they disclosed abuse,¹⁷ or to stop wearing 'revealing clothing' around the perpetrator.¹⁸ Some survivors told us the institution where the abuse occurred asked them to forgive the perpetrator. 'Jack Eric' told us he was abused by a church assistant as a child in the 1960s. He said that he wrote to the Anglican Primate when the perpetrator became a priest to tell him about what had happened. At a meeting arranged between 'Jack Eric', the perpetrator, and the Primate, 'Jack Eric' requested that the perpetrator be defrocked. In a later letter to 'Jack Eric', the Primate said he needed to forgive the perpetrator because he had been convicted by a court, confessed his sin to God and received absolution – and that 'nothing bad had happened' since the perpetrator was ordained.¹⁹ In *Case Study 13: The response of the Marist Brothers to allegations of child sexual abuse against Brothers Kostka Chute and Gregory Sutton,* we heard that a senior official within the Marist religious order minimised the sexual aspect of Kostka Chute's conduct by describing it as 'inappropriate' but 'not extreme' and 'not even genital'.²⁰

Many survivors told us they wanted the institution to accept responsibility for not protecting the children in their care. However, institutions had often refused to acknowledge the abuse or denied culpability. Survivors described the frustration, anger, powerlessness and hurt they experienced as a result. 'Sergio' told us he was abused as an adolescent in the 1960s by the school chaplain at his Catholic boys' school. Years after seeking compensation from the Catholic Church, 'Sergio' told us he saw some letters from one church representative to another concerning his abuse. He became angry because of the 'dismissive, contemptuous tone of the letter'. During his private session, 'Sergio' reflected on the impact on victims of the church's response:

... [W]hen the Church is honest and humble, and admits the bad effects of the sexual abuse of boys and young men by priests, and says sorry and tries to compensate, then it gives freedom and health back to the victim. When it tries to avoid responsibility, and make the victim feel guilty about telling their story or confronting their abuser, and uses tricks to steer them away from obtaining an apology because it fears the cost of compensation, then it hurts itself.²¹

Another victim, CK, who told us he was abused as a boy during the 1950s, gave evidence in *Case Study 3: Anglican Diocese of Grafton's response to child sexual abuse at the North Coast Children's Home* (*North Coast Children's Home*) of how the Anglican Church's denial of responsibility affected him and others who experienced abuse as children:

It hurt a lot because we knew the truth. We were there for years. We knew what went on and what happened. And that really hurt. It's not anything to do with the money. It's them taking responsibility and ownership of the results and the mental state of all those children in that home.²²

Many survivors told us that institutions prioritised protecting their reputations over the welfare of the victim. Mr Guy Lamond, a victim of abuse in the 1980s at Knox Grammar School, gave evidence during Case Study 23: The response of Knox Grammar School and the Uniting Church in Australia to allegations of child sexual abuse at Knox Grammar School in Wahroonga, New South Wales (Knox Grammar School) that his school put its reputation above the safety and wellbeing of the children in its care:

I thought the response was impersonal and it was disbelieving; Knox were terrible. I believe the school knew about the abuse but tried to cover it up and ignore it. The school was more interested in protecting its reputation than protecting kids.²³

We have accepted the evidence of Dr Ian Paterson, headmaster of Knox Grammar at the time, that, in relation to the allegations against music teacher Barrie Stewart and the teacher ARZ, he was involved in a cover-up of those allegations. Dr Paterson did not notify the parents of boys who had made allegations against staff members. Through these actions, Dr Paterson failed to prioritise the welfare of the boys at Knox over the reputation of the school.²⁴

We heard from survivors who felt that institutions prioritised their reputation over protecting victims of child sexual abuse by fostering a culture of silence. In *Case Study 12: The response* of an independent school in *Perth to concerns raised about the conduct of a teacher between 1999 and 2009 (Perth independent school)*, WP told us:

When the school was told about the abuse, I didn't feel supported by the school like I should have been. I began to feel isolated, like a troublemaker for the school's reputation that was best kept quiet. ... I felt at times they were just protecting themselves and everything that was said or written was very carefully prepared.²⁵

Many survivors told us that redress, including compensation, was an important element of addressing the injustice they experienced (see our *Redress and civil litigation* report). However, for some survivors, compensation was often accompanied by a sense of anger and disappointment at how the abuse they experienced was reduced to monetary terms. Many survivors told us they needed professional counselling and support to help them deal with the impacts of the abuse they experienced. They called for this to be separate from other aspects of the response (that is, compensation). Our *Redress and civil litigation* report includes recommended elements and principles for redress.²⁶

4.1.1 Other institutions with a responsibility

Survivors described how other institutions, such as child protection agencies, the police, the criminal justice system, complaint and oversight bodies, support services and health services, had also failed them by not believing them, ignoring their claims or not protecting the child. For example, one survivor who told us she was sexually abused in out-of-home care as a child and as an adolescent asked, '... aren't [the police] supposed to protect me?' She explained that when the police did not protect her, her attitude toward them became, 'well you don't give a crap'.²⁷ Criminal justice responses to institutional child sexual abuse are discussed in detail in our *Criminal justice* report.

Some survivors told us they felt betrayed by the criminal justice system when a charge against the perpetrator did not go to court, or if there was a verdict of 'not guilty'. They described how in some cases these outcomes compounded their feelings of powerlessness and betrayal. In *Case Study 46: Criminal justice (Case Study 46 on the criminal justice consultation paper)*, FAA explained how the perpetrator's trial affected him:

At the conclusion of the trial, I was informed by the detective that FAD was found not guilty. I just cried. I was devastated. There was no discussion of an appeal. I believe that the DPP [Department of Public Prosecutions] just accepted the verdict. By that stage, I had had enough and just wanted it to be finished.²⁸

We heard in private sessions that the sense of betrayal when a perpetrator fails to be convicted can be intensified when a victim has been poorly treated during the court process. For example, we heard from some survivors who told us they were questioned in a demeaning way by the perpetrator's defence barrister, or that the prosecution did not keep them updated with information.

Several victims and their families told us in private sessions that they were refused help or information from institutions that would help them pursue criminal or civil proceedings in cases of historical child sexual abuse. This increased victims' feelings of powerlessness against institutions and their resources. Some had difficulty finding information about the perpetrator or gaining access to their personal files from the institution, delaying their attempts to achieve justice, and find answers for what had happened to them.²⁹

4.2 Continuation of abuse

The way that an institution responds to child sexual abuse can also contribute to a child continuing to be sexually abused by a perpetrator. We heard institutions failed to provide an ongoing safe environment for children, and failed to protect children from abuse, despite information about sexual abuse.³⁰ Not only can these poor responses place a victim at risk of repeated and prolonged abuse, they continue to expose other children to an unsafe environment. Many survivors told Commissioners they disclosed child sexual abuse in an attempt to stop it, both for themselves and other children. One survivor, 'Matilda', told us, 'I could have just left it and done nothing, but I didn't because I was concerned for other people'.³¹ Research shows that protecting others from abuse may be a motivation to disclose for some victims.³² Motivations for disclosing child sexual abuse are explored in further detail in Volume 4, *Identifying and disclosing child sexual abuse*.

We heard how, in some instances, victims made multiple disclosures to adults within and outside the institution where the abuse was taking place. We also heard from victims' parents, whistleblowers and others who indicated they made reports in an attempt to stop the abuse. However, we also heard that many institutions failed to respond appropriately to this information about abuse, instead disbelieving victims, dismissing the reports, playing down the seriousness of the abuse or seeking to protect the perpetrator and the institution. The consequences of this lack of intervention were significant. Most notably, they meant victims continued to be sexually abused or that other children were placed at risk of abuse. 'Alexander Martin' told us how, as a victim of sexual abuse at an orphanage in the 1960s, his mind went 'into total lockdown' when his sexual abuse by a male staff member continued after he disclosed it to a priest. He told us that, at the time, he felt 'very angry' and 'deeply hurt' that he was punished when he disclosed the abuse. He told us, 'The response by the persons from whom I sought solace and help had a profound lifelong lasting impact on the private and working life of my beliefs, my views, my interactions ...'³³

Some research has noted an association between poor responses to disclosure and victims' re-abuse and the risk of further abuse.³⁴ Victims of child sexual abuse who are not believed or supported when they disclose describe feeling cynical, helpless and isolated, as well as having trust issues, increased anxiety, dissociative symptoms and trauma.³⁵ As discussed in Chapter 2, there can be profound psychological outcomes when children are exposed to cumulative harm, as a result of multiple and ongoing episodes of maltreatment. In particular, disclosures and attempted disclosures by children with disability who have been sexually abused are often not recognised, or are ignored, resulting in cumulative harm (see Chapter 2, and Volume 4, *Identifying and disclosing child sexual abuse*). The children's behaviours that are reactions to abuse can be misinterpreted as being part of their disability – and allow the abuse to continue. In one private session, the parents of 'Josh', who has a moderate to severe intellectual disability and limited descriptive speech, said he began displaying sexualised behaviours, including exposing himself and grabbing at other people's genitals, having never previously done so.

'Josh' was frequently in trouble at school for touching other students inappropriately. He began to grind his teeth while awake and asleep. The school and others put this behaviour change down to his disability and did not explore it further. His parents now believe it was learned behaviour from the sexual abuse he experienced.³⁶

We heard that many victims were called liars by parents, police, those in authority and other adults when they disclosed. In some cases they were punished for doing so. As a result, some victims stopped telling people about ongoing abuse or tried alternative strategies to prevent it, such as running away.³⁷

Research suggests that when child sexual abuse is reported and the perpetrator is not brought to justice (for example, if the perpetrator is moved to another location and commits further acts of abuse), the victim can experience further self-blame.³⁸ Ineffective institutional responses can inhibit victims from making future disclosures or seeking support.

We received evidence that some institutions – particularly religious institutions – moved perpetrators between locations, which enabled perpetrators to have continued access to children and put children at increased risk of abuse (see Volume 16, *Religious institutions*).

In some cases, representatives of institutions responded to reports of sexual abuse by abusing the victim themselves. In *Case Study 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland (The Salvation Army boys' homes, Australia Eastern Territory)*, 'ES' said he was sexually abused by a Salvation Army officer after disclosing sexual abuse by an older boy in a boys' home to that officer.³⁹ Another victim, 'Ethel Anne', told us in a private session of being sexually abused while away in holiday care from the Anglican children's home in Victoria where she lived in the 1950s. During confession, she told a priest about the abuse. She told us that he responded by asking her to show him what the man did to her, and that he then sexually abused her too. 'Ethel Anne' said the priest also told the religious sisters what she confessed, and she was beaten for being a 'bad girl'.⁴⁰

Many survivors expressed anger and outrage that an institution that advocates high moral standards could betray them by protecting and supporting perpetrators over children. The daughter of one victim, attending a private session on behalf of her father who had died, said:

I hold the institution of the Catholic Church responsible. They afforded not only protection to paedophiles, but they enabled them to gain access to generations of victims and in some cases form networks to facilitate this. Protected by the shield of their cloth, they raped and abused children, and no one would or could protect these children. Families held these abusers up as people to be admired and respected.⁴¹

Sexual abuse sometimes continued when an organisation with devolved responsibility for the day-to-day management and delivery of services at an institution failed to detect or prevent it. In *The Salvation Army boys' homes, Australia Eastern Territory* case study, we heard evidence that abuse at the homes continued between 1965 and 1977 because state social services secretaries employed by the Salvation Army did not conduct regular inspections of its boys' homes in New South Wales and Queensland, and because children were denied access to the visiting officers. ⁴² The failure of institutions to detect or prevent abuse in historical residential institutions is discussed further in Volume 11, *Historical residential institutions*.

4.2.1 Other institutions with a responsibility

We also heard evidence about the role of other institutions in failing to protect children from the ongoing sexual abuse. If criminal justice and oversight organisations fail to respond appropriately to indications of child sexual abuse or a disclosure of abuse, there is a risk the sexual abuse will continue, or that other children will be abused. In *The Salvation Army boys' homes, Australia Eastern Territory* case study we found that staff at the Queensland Department of Children's Services monitored Indooroopilly and Riverview homes in Queensland, but were slow to act on their knowledge of sexual abuse. The cursory reporting by staff at the (then) New South Wales Department of Child Welfare enabled child sexual abuse and other forms of maltreatment to persist at the Gill and Bexley homes throughout the 1970s. ⁴³ During a private session, 'Rosemary', who told us she was sexually abused as a child by her foster father, described visits from an inspector while she was in care. She said the foster mother always ensured she was present during these visits and would punish her if she 'said anything out of place', meaning 'Rosemary' was forced to conceal the abuse and the inspector did not identify it. As a consequence, she remained with the foster family and the abuse continued. ⁴⁴

Similarly, some victims of sexual abuse in contemporary out-of-home care told us that the abuse continued because child protection authorities failed to act on a complaint. For example, in a private session 'Milo' recalled that he began to complain about sexual abuse by his brother 'Stephen' when he was about 10 years old. 'I told Mum and Dad about it, I told my stepmother about it ... I told DOCS [the Department of Community Services] on numerous occasions'. DOCS workers told him they would act. 'We'll follow it up, we'll follow it up'. But they never did, to 'Milo's' knowledge. He said:

They just kept putting us back in that situation. DOCS knew that we were getting abused. They knew we were being physically, emotionally and sexually abused and they still kept putting us back into that predicament.⁴⁵

We also heard from some survivors in private sessions that they believed the police had allowed the sexual abuse of children to continue. Some survivors told us they ran away from the institution where they were being sexually abused, but were returned there by the police, even after they had disclosed the abuse. 'Howard' told us he was abused by clergy at a Catholic boys'

orphanage in the 1960s. He ran away a couple of times, and was found by the police. He told us that he tried to explain the abuse to them, but they did not believe him. He said he 'got a kick in the "you-know-what", [was] sent back, and got dobbed in to the orphanage again ...' which meant he was in trouble again. Some survivors said the police further sexually abused them when they sought help.

4.3 Re-traumatisation

Negative responses to disclosures of child sexual abuse can re-traumatise victims, compounding the effects of abuse and affecting their mental health and ability to adjust to social situations.⁴⁷ Research has suggested an adverse response can even constitute a secondary traumatic experience for victims.⁴⁸

Consistent with the available research, many survivors in private sessions and public hearings described being re-traumatised by the responses of institutions. Institutional responses can also re-traumatise adults, years after they were abused. Reflecting on her experiences in 1996 in reporting sexual abuse to the Anglican Church, 'Joann' told us in a private session, 'The Church should have begun the healing but they just exacerbated the wounds'.⁴⁹

In Case Study 29: The response of the Jehovah's Witnesses and Watchtower Bible and Tract Society of Australia Ltd to allegations of child sexual abuse, we heard from BCB who was groomed and sexually abused by Mr Bill Neill from 15 years of age. 50 BCB eventually disclosed her abuse to her husband in 1989 and to a Jehovah's Witness acquaintance in 1991 when she was 23 or 24 years old. 51 Shortly after her disclosure, an elder in her congregation convened two meetings with BCB and Mr Neill, among others. The meeting was at her house. 52 BCB said that during the first meetings Mr Neill had asked her, with reference to his alleged conduct, 'Don't you think I was joking?' BCB said that she did not feel comfortable 'talking while Bill was in the room' and felt unable to report the full extent of Mr Neill's abuse at the meeting. 54 BCB told the Royal Commission that:

Throughout the meeting, Bill looked at me defiantly. I felt like he was challenging me to tell the full story of what he had done. I felt uncomfortable and could not bring myself to tell the elders everything that had happened. I felt like I was still Bill's victim. I was still so scared of saying anything that would get me or Bill into trouble.⁵⁵

BCB told the Royal Commission that she did not feel supported and that the elders were testing her credibility.⁵⁶ She told us in evidence, 'I found the experience of reporting my abuse to a room full of men, including the man who had abused me, very distressing'.⁵⁷ The elders did not explain to BCB the purpose of their investigation or the meetings with her which left her confused and disempowered.⁵⁸

Others told us how they found it traumatising when an institution failed to do anything in response to their disclosure, in particular allowing the perpetrator to remain in a position of power. One survivor, 'Claira', described the damage done to her when her church failed to do anything in response to a disclosure of sexual abuse.

'Claira' told us that an inadequate response by the institution can be damaging

'Claira' told us she was sexually abused over four years by 'Mr Daniels', a senior member of her local church. 'Claira' said she disclosed her abuse to her parents after a 'stranger danger' course was conducted at her school by police officers. It was then that she started to think again about her contact with 'Mr Daniels', which motivated her to speak to one of the officers privately. She was encouraged to tell her parents and then come back to the police.

'Claira' disclosed the abuse to her parents that night. She said it was very difficult because she didn't think that they would believe her. Her parents then took her to speak to the leader of the congregation. She told us: 'He looked like he was going to do something about it ... [However] he wasn't in his position for much longer, he was extradited from the church ... There was a rumour that he had broken confidentiality ... It was implied that it was about my situation'.

'Mr Daniels' denied that he abused 'Claira', saying that she was flirting with him. At the time 'Claira' had no idea what this meant but she was upset. She went to a female elder of the congregation for help, but found that the elder believed 'Mr Daniels' rather than her. The church did nothing and 'Mr Daniels' remained in his position of power.

Throughout her teens and adulthood, 'Claira' has been upset with the way the church disregarded the abuse. She never finished her education. Her relationship with her family is severely strained. 'Claira' finds it difficult to trust people, especially males in authority in her church

When she was in her 30s, 'Claira' arranged a mediated meeting with 'Mr Daniels' and several church elders. She wanted to try and forgive 'Mr Daniels' and move on. It upset 'Claira' when 'Mr Daniels' admitted to kissing her once. She also learned that 'Mr Daniels' had been nominated for a higher role in the church. 'They wanted clarification if I was going to vote against him [for a higher position]'.

She said: 'I love my church and it does a lot of good in the world. But I'm feeling that the leadership are doing some damage as well. I feel quite damaged'.⁵⁹

Some survivors also told the Commission about the re-traumatising effects of seeking redress from the institution. We heard these processes were often adversarial in nature and pitted the word of the victim against that of the institution, rather than adopting a compassionate approach. 'Clifford' told us about his experiences of seeking redress from the Anglican Church in 2013 for abuse at a Sunday school in the early 1950s. During his private session 'Clifford' described the redress process as slow, overcomplicated and misleading. Discussing the impact of this process, 'Clifford' told us:

back in [the 1950s] I was molested by one Sunday school teacher, now I'm being molested by the whole bloody Church. [My psychologist] said to me that it's almost certainly undone 98 per cent of the work she's done with me in the last 20 sessions.⁶⁰

In the *North Coast Children's Home* case study, we found that the Anglican Diocese of Grafton did not follow its own policies when responding to abuse victims. It caused further distress for victims of child sexual abuse by denying responsibility for the institution and offering inconsistent support and compensation.⁶¹ We found Bishop Keith Slater did not follow the processes recommended in the Anglican Church's Pastoral Care and Assistance Scheme, and sought to protect the interests of the former members of the committee.⁶²

In Case Study 8: Mr John Ellis's experience of the Towards Healing process and civil litigation, we examined how the Catholic Church maintained a position of non-admission of the sexual abuse by Father Aidan Duggan in the 1970s. This was despite a Church-appointed assessor previously finding that, on the balance of probabilities, Mr Ellis was abused as alleged.⁶³ We found that Cardinal George Pell accepted legal advice to vigorously defend the claim brought by Mr Ellis to discourage other prospective plaintiffs from litigating claims of abuse.⁶⁴ This resulted in the victim being subjected to a lengthy cross-examination in circumstances that were 'hurtful and painful' to him: this was accepted with regret and apology by the Church parties.⁶⁵

4.3.1 Other institutions with a responsibility

The response from oversight institutions could also be re-traumatising. In *Case Study 27:* The response of health care service providers and regulators, New South Wales and Victoria to allegations of child sexual abuse (Health care providers and regulators, New South Wales and Victoria), survivor AWH gave evidence about being re-traumatised by a request from an officer of the NSW Health Care Complaints Commission (HCCC) to provide detailed information about alleged abuse, within 14 days. 66 He reflected:

This was a very frustrating experience for me. The HCCC correspondence was happening at a time in my life when I was mentally very upset. I had around this time attempted to commit suicide and was experiencing suicidal thoughts all the time. The level of technicality, procedures and processes the HCCC was asking from me was overwhelming.⁶⁷

When AWH did not provide the information within the time frame, the HCCC terminated the investigation. We found that HCCC's approach was insensitive given the nature of the allegations and the survivor's personal circumstances.⁶⁸ We found that it was important that any body with responsibility to receive and respond to complaints has staff who are capable of responding appropriately to complainants.⁶⁹ This is supported by research that highlights the need for professionals to be conscious of victims' potential to disclose child sexual abuse and to be able to respond effectively.⁷⁰

A number of survivors who came forward to the Royal Commission said they were retraumatised by the responses of the police. Some described the process of disclosing to the police, and the subsequent investigation, as upsetting and invalidating. They said that either the police did not follow up on the disclosure, or they did not follow up in the way that the victims anticipated they would, creating an overall negative experience for the victims. A decision by the police not to proceed with criminal charges against the alleged perpetrator was often a difficult outcome for victims to process, exacerbating their feelings of injustice. However, we also heard in private sessions of more appropriate police responses, which empowered victims (see Section 4.6.1).

For victims of child sexual abuse with a history of confrontation with police and a criminal record, reporting to police could be particularly difficult. It was often the first time they had provided a full account of the abuse. We also heard that reporting to the police can be especially difficult for children. ⁷² Some survivors recalled in private sessions how the process took a negative toll on them emotionally, with some struggling to speak in detail of the abuse. One survivor noted that she felt her wellbeing was not a priority when she made a formal report to the police. She did not receive the counselling support she expected:

I'd never spoken the story before, you know; never verbalised, told people, 'Yes, I've been sexually abused'. I'd never gone into the intimate details of how it actually had played out ... I found the whole thing just horrendous...⁷³

Some survivors described the experience of engaging with the legal system as traumatic, triggering the impacts of the abuse even after many years. Mr Adrian Steer, a survivor of child sexual abuse at Knox Grammar School, described during the *Knox Grammar School* case study how participating in criminal proceedings precipitated a breakdown and undermined his faith in the legal system:

I found my involvement in the criminal proceedings to be very traumatic. It was the beginning of an emotional downward spiral in my life. I stopped eating properly. I couldn't focus at work. I would say I was working at home but just stared at the screen and did nothing. I couldn't sleep properly ... I have no faith in the legal process after my involvement in the criminal process.⁷⁴

In Case Study 46 on the criminal justice consultation paper, we heard evidence from Ms Clare Leaney from the In Good Faith Foundation, an independent advocacy service which assists victims and survivors of clergy, religious and lay abuse. She reflected on the experiences of clients in an adversarial system and the impact this could have. She spoke about how the power imbalance that victims felt could reflect their experiences of their initial abuse and leave them feeling traumatised:⁷⁵

If they don't feel like they have control of a situation, if they feel like they are once again removed from their experience – and it is their experience – then that can really put someone off and become such a large barrier to appropriately reporting and appropriately engaging with a police or criminal justice process that it may be something that someone is never able to overcome.⁷⁶

Some survivors told us that specific actors within the criminal justice system did not treat them with respect or compassion. Others related that the adversarial nature of the court process was traumatic for them. One survivor, who told us he was sexually abused as a child in the 1980s, said:

I've got to say, they might be compassionate, but I don't think judges have a clue what a victim is going through in this. And what they might consider acceptable behaviour in the courtroom and acceptable tactics, isn't, because – and I appreciate there are false claims and all those sorts of things in this kind of space – the focus should be on getting an outcome and the care of the victims. The victims get re-traumatised through this whole thing, and [judges] don't seem to get that.⁷⁷

Criminal justice responses to institutional child sexual abuse are discussed in detail in our *Criminal justice* report.

We also heard that mental health services intended to support victims of child sexual abuse can instead have profound negative effects if they respond inappropriately to the abuse. This could result in victims feeling unable to speak about their experience of abuse. This could result in victims feeling unable to speak about their experience of abuse. This could result in victims feeling unable to speak about their experience of abuse. This could result in victims feeling unable to speak about their experience of abuse. The council to the to the bishop and the same spoke to a priest and he told her it was out of his jurisdiction. She spoke to the bishop and he said that 'Father Stevenson' was trying to help out a troubled girl, and that he would probably have done the same. Tara' told us she encountered a similarly dismissive attitude when she sought help from a counsellor. Instead of reporting the crime or providing her with coping strategies, the counsellor told 'Tara' to come up with her own plan. She said, 'I just found that she was telling me things I needed to do, and I was so overwhelmed'. We also heard from 'Beck', who told us in a private session that he was sexually abused in the 1980s at a community sporting organisation when he was about nine or 10. He said he was extremely upset that his first counsellor told him that, because of his age at the time of his abuse, he would be permanently affected by severe post-traumatic stress disorder — 'that I'm broken, like there's no way of fixing it'. **

Care-leavers are often dependent on institutions for information about their time in institutional care. However, they can be vulnerable to re-traumatisation from institutions' responses to their requests for access to the information. Access to records can aid the healing process for care-leavers who were abused in care. They can be a critical source of evidence of abuse and become pivotal in civil or criminal proceedings. As we heard in case studies, many institutions did not provide records or provided only limited access to records, records that were heavily redacted or records marred by inaccurate or missing information as to why they were in care.⁸¹ During Case Study 24: Preventing and responding to allegations of child sexual abuse occurring in out-of-home care, we heard evidence from Ms Tash Dale, a youth consultant with CREATE Foundation, that young people in care may 'start blaming themselves' for their situation when not given accurate information.⁸² Care-leavers can also be re-traumatised by reading records that contain insensitive or judgemental language about them. 83 Records can also contain information that triggers past experiences of trauma, neglect, and physical and sexual abuse. Survivors told us they did not receive accompanying support to manage these impacts.⁸⁴ Volume 8, Recordkeeping and information sharing discusses the importance of recognising individuals' rights to access records about themselves.

4.4 Fear, distrust and contempt

Many survivors described how subsequently they had generalised distrust of institutions, hierarchy and people in positions of authority, and difficulties in engaging with institutions because of the institutional betrayal they experienced. Research has shown that victims' interactions with institutions can be coloured by feelings of fear, distrust and contempt as a result of abuse, the process of disclosure, and subsequent responses.⁸⁵

We heard that police responses to child sexual abuse can frighten children. 'Zahara' told us she was molested by a leader at the mosque she and her family attended in the early 1990s. After disclosing to her mother, she was taken to speak to a female officer from the police Child Protection Unit. 'Zahara' said she felt scared of the officer as she wore a gun and warned her she could get into a lot of trouble if she was telling lies. Because she felt intimidated and frightened, 'Zahara' said she did not go back to the police, and is unsure if what she told them was considered a formal statement.⁸⁶

Some survivors attributed their inability to hold down a job or participate in society to the responses they received to the abuse. For some, the responses fostered feelings of contempt and hatred for institutions. Mr Coryn Tambling, a victim of child sexual abuse at Knox Grammar School, gave evidence about his lifelong rejection of authority because of the abuse and the school's response to it:

I have acted out of my deep-seated desire not to allow anyone ever to 'do me over'. I have fought every single authority figure or system that I have thought, whether real or imagined, has sought to control me or acted inappropriately ... I have attempted desperately at times to prove to myself that I am beyond the realms of society and all its institutions. I have held contempt, disregard and disrespect for the conventions of our community ... It is more than simply the sexual nature of this crime. It is all the forms of abuse itself: the breach of trust, the cover-ups, the silence, the punishments, the fear of recriminations, the lack of being able to be heard or to express oneself when feeling unsafe, unsure and violated.⁸⁷

In Chapter 3, we examined how one of the particular effects of child sexual abuse in institutional contexts is a fear and distrust of institutions and authority. A poor institutional response can compound this loss of trust, in specific and generalised ways. For example, one study of male adult survivors suggested that if a victim receives a poor response when disclosing to medical practitioners, or being treated for child sexual abuse, medical care could risk re-enacting the abuse of power that characterised the sexual abuse. We heard from survivors who had been negatively affected in this way by inappropriate responses from practitioners in the health system. In the *Health care providers and regulators, New South Wales and Victoria* case study, a survivor attested that the abuse and the response she received to it from the Royal Children's Hospital, Melbourne, had multiple adverse impacts. These included an enduring fear of hospitals, medical treatment and medical staff. So She said:

Over the years, I have suffered from bouts of depression, frustration and anger over the past, having all-consuming thoughts of my sexual abuse and how I was treated by the Royal Children's Hospital. I am now dependent on antidepressants and sleeping tablets to deal with the anxiety, sleep and the feelings I have about the abuse and reporting the abuse.⁹⁰

Some survivors told us they avoided accessing services because they distrusted institutions that provided them. They did not access support services intended to help them manage abuse-related trauma, which could mean impeded recovery and exacerbated poor outcomes for health, wellbeing and financial stability. One survivor of childhood abuse in a Catholic-run residential care home in the 1950s described his attempts to avoid institutions, including medical services and the police, throughout his life. He said: 'I was about 45 when I first applied for my Medicare card. Up until then I had never been on the dole and never went to the authorities for assistance. My fear of them and the police was still with me ...'91

A number of victims had to maintain ongoing relationships with the institution where the abuse happened, even when the response to the abuse was inappropriate or ineffective. We heard that victims did not want to access services connected to the institutions where they were abused. In research we commissioned, 'Sonya', a victim of child sexual abuse at her family's church, described being drawn back into interacting with the same institution later in life due to her mother's ill health. Throughout her childhood, 'Sonya's' mother was heavily involved with the institution, which had ignored 'Sonya's' attempts to disclose and stop the abuse. 'Sonya' had tried to cut ties with the institution but found that it was the only one providing the services her mother required. She said: 'We're doing an aged care assessment on Mum because she's got early stage dementia. So I'm having to now ring the [religious institution] direct to ask them about details to do with Mum'. '92 She described a profound ambivalence and unease at having no alternative but to seek care from an institution that had responded inadequately to her own abuse.

Research has highlighted the specific effects that abuse in institutional settings can have, including a general distrust of authority figures.⁹³ This distrust can compound victims' difficulties in accessing therapy and support because they see all institutional representatives, including counsellors, as untrustworthy and potentially abusive. It may also extend to government agencies and institutions that provide services as a whole (see Volume 9, *Advocacy, support and therapeutic treatment services*).

As a result of their particular experiences of injustice, Aboriginal and Torres Strait Islander survivors told us they distrusted institutions and authority. We heard through roundtables, consultations and private sessions that Aboriginal and Torres Strait Islander peoples' ongoing experiences of racism and discrimination compound these feelings. 'Geoffrey' told us that when he was sexually abused by the driver of his local school's bus, he didn't tell the police, as he did not trust them. He said there were curfews and restrictions for Aboriginal people at the time: 'We always used to run when we'd see the police, that was just a natural thing for Koori kids at the time. ... I just knew if I went to the police they wouldn't believe me anyway'. Research for the Royal Commission suggests that the memory of forced child removals is so recent and so traumatic that Aboriginal and Torres Strait Islander peoples live with the fear that engaging with an institution will result in a child being removed from the community. What we heard about the service needs of Aboriginal and Torres Strait Islander peoples is discussed in Volume 9, Advocacy, support and therapeutic treatment services.

People with disabilities, who may be particularly reliant on therapeutic or support services, can also experience fear and distrust of institutions, as well as fear that support services could be removed.⁹⁶

4.5 Ostracism

Survivors often spoke of being ostracised by the institution where the abuse occurred. Many told us they were rejected by the broader community in response to identification or disclosure of child sexual abuse. Some said they were alienated, humiliated and stigmatised as a result of coming forward. Family members and whistleblowers also frequently experienced, or feared experiencing, similar treatment. Their experiences are covered in Chapter 5 of this volume.

ARY, who told us he was sexually abused at Knox Grammar School in the 1970s, gave evidence in the *Knox Grammar School* case study that he was reluctant to disclose because he feared being ostracised. The headmaster was a known disciplinarian, with an 'intimidating and unapproachable character' who would deliver humiliating punishments in front of other students.⁹⁷ Having witnessed others being victimised for disclosing abuse, ARY feared receiving the same response:

I had seen other boys who had mentioned instances of sexual abuse by teachers become victimised and ostracised in the boarding house. They were seen as weak and they became everybody's bitch. They never lived the abuse down and it was seen as a stain against their character and reputation that followed them forever. I did not want to be seen as a victim and suffer this fate. The culture was very much that of blaming the victim, and I thought that if I made a complaint against a teacher, that I would get into more trouble than the teacher.⁹⁸

WP, who told us he was sexually abused by his schoolteacher, gave evidence during the *Perth independent school* case study that after disclosing the abuse he felt increasingly excluded from the school community. He felt that the school prioritised its reputation over his wellbeing. This treatment extended to his family including a younger brother who also attended the school, and his mother, who taught there occasionally: 'As time went on, it felt to me like it was the school versus me and my family. I didn't not feel believed, but the focus was on the school's reputation'.⁹⁹

WQ, the mother of WP, gave evidence of her fear of being stigmatised if she disclosed her concerns that WP was being abused by a teacher at their school. She said: 'I was fearful of the consequences of being a possible whistleblower or being perceived as interfering'. We were satisfied that during the time in question, there was evidence of a culture at the school where some staff members and one parent felt that, if they raised concerns about another staff member, they could be ostracised by parts of the school community. 101

In research we commissioned, several survivors described perpetrators being supported by the institution, even when their guilt was confirmed. This was usually at the expense of the victim and their family, who experienced personal and social consequences for exposing the abuse. One survivor, who told us he was abused by his teacher in the 1990s, said: 'He got the good life, he's come out of it smelling like roses, and I'm still scum, you know. That's how I feel I was treated, I was treated like scum'. 103

As a result of this ostracism, some victims told us that they left their community, moving to a different town, or even another country, leaving behind personal networks. For example, BAV gave evidence in *Case Study 28: Catholic Church authorities in Ballarat* of being sexually abused when he was in sixth grade. He told us that there were negative consequences of him speaking out about the abuse. He was excluded from social events, experienced a form of bullying in the community and believed that some people refused to hire him because he had spoken out against the Catholic Church. When he was 21 he left the Ballarat area and moved to Queensland, moving around from job to job. He told us: 'I had bouts of depression throughout this period, but I felt better for being in Queensland. It gave me an opportunity to remove myself from the Ballarat community and try a fresh start'. 105

Beyond the institution, the response of the broader community can contribute to the isolation experienced by victims of child sexual abuse. Research suggests that the strong community support for some institutions held in high esteem, together with the resources and power of the institution itself, may thwart a victim's efforts to disclose. The size of a community may also affect victims' efforts to disclose, with small, tight-knit communities sometimes 'posing formidable resistance' to public exposure or scandal. We heard from survivors that the fear or actual experiences of ostracism could be worse in smaller communities, such as rural towns. During the *Sporting clubs and institutions* case study, BXI gave evidence that he was abused as a boy by a volunteer cricket coach. He found it difficult to report the abuse because of the size of the community. He said:

For a very long time I did not tell anyone about my abuse, I think the fact that I lived in a small town made it hard to report my abuse ... I felt that if I report the abuse or told anyone what was happening, me and my family would be vilified or scrutinised. 108

While many survivors from a range of communities described the fear of being shamed or blamed by the institution in which the abuse occurred, it appeared to be a particular concern in communities that promoted an identity strongly associated with group membership. ¹⁰⁹ In Case Study 21: The response of Satyananda Yoga Ashram at Mangrove Mountain to allegations of child sexual abuse by the ashram's former spiritual leader in the 1970s and 1980s, the Royal Commission heard how children were separated from their families and encouraged to feel a sense of belonging within the ashram. One survivor, APK, gave evidence that residents understood that if they defied the leader they would be cast out of the ashram. ¹¹⁰

For many victims of child sexual abuse in religious institutions, the poor response to the abuse by the institution and the broader religious community led to their eventual disengagement from their religion. BCG, who told us she was abused by her father, a Jehovah's Witnesses elder, spoke in the *Jehovah's Witnesses* case study of how she left the church because she could not bear the hypocrisy any longer. She and her family were then shunned by the congregation.¹¹¹

During a private session, we heard from 'Julia Maree' that she was sexually abused by her father, from the age of five, who also arranged for her to be sexually abused by other male and female Jehovah's Witnesses. As an adult she sought help from members of the institution, who told her there was nothing they could do. She told the Royal Commission, 'I was also shunned by most of them too, by saying "I don't believe your story" or the condescending, re-traumatisation of "Oh well, if you believe it then it's true"'. She said that when she wrote to the congregation elders and told them what her father had done, her father retaliated by telling the elders that 'Julia Maree' had made up the story as revenge on him for cutting her out of his will. The institution sided with her father and dropped the matter. When eventually 'Julia Maree' asked for her baptism to be annulled, she was branded an apostate and ostracised.¹¹²

4.6 Appropriate responses and empowerment

When institutions respond appropriately to child sexual abuse, it can ensure children are safe. It can also promote healing for victims, by helping them to manage the effects of the abuse. However, only occasionally did we hear in private sessions from survivors who experienced positive impacts from appropriate institutional responses. Most survivors in private sessions and public hearings described negative impacts from poor institutional responses, in particular from the institution where the abuse took place.

We acknowledge that survivors who experienced appropriate institutional responses might have felt less need to share their experience with the Royal Commission in private sessions than those who experienced inappropriate responses. This might mean that we were more likely to have heard from survivors about inappropriate rather than appropriate institutional responses. Regardless, we have learned much less about the impacts of appropriate responses than we have about poor institutional responses. There is also a paucity of research on impacts of appropriate responses to child sexual abuse.

Within these limitations, we did hear from some survivors in private sessions about institutional responses they felt were helpful and supportive. These included responses from institutions where the sexual abuse occurred, as well as other institutions responsible in some way for responding to child sexual abuse and child safety, including police, the legal system and welfare organisations. In some instances, a supportive response came from certain individuals within an institution, rather than from the institution as a whole. Sometimes institutions gave a supportive response only after significant time had passed, or in response to intervention by a third party.

Although appropriate institutional responses can be instrumental in assisting victims, the profound and adverse impacts of child sexual abuse mean that victims may find any involvement with institutions difficult. Following child sexual abuse, victims may have to disclose and report to many different institutions, including the institution where the abuse occurred, institutions with authority over, or responsibility for, that institution, the police, complaint and oversight bodies, support services, and health services. This process can be traumatic and anxiety-provoking, regardless of how appropriately these institutions respond. However, if the risk of adding further trauma is to be minimised, institutions need to respond to child sexual abuse appropriately.

We heard that appropriate responses by institutions where the abuse occurred demonstrate some common features: compassion, transparency and accountability. Survivors described to us the importance of being heard and validated.

Survivors described other key features of appropriate institutional responses as:

- being systemic not confined to the institution in which the sexual abuse occurred, but promptly activating involvement of other relevant institutions, such as the police, as well as independent, professional counsellors
- involving ongoing communication between the survivor and the institution where the abuse occurred
- being humane and trauma-informed characterised by a sympathetic and compassionate attitude towards the survivor and including provision of traumainformed support
- being restorative founded on a meaningful apology and offering the survivor longterm psychological and practical support, as well as reparation for the damage done.

The experience of 'Evan' illustrates the positive impacts of a supportive institutional response. 113 'Evan' told us he was abused as a child by his school teacher in the 1970s but did not disclose at the time because he was too ashamed. He was driven to seek help later on in life after his marriage broke down and he had recurring thoughts about the abuse. 'Evan' told us that he found helpful elements in the responses from his old school, his church, the police and the legal system. He has also been helped by counselling.

'Evan' told us that a supportive response can help start the healing

'Evan' first reported his abuse to the Catholic Education Office. Through it, he was put in touch with a clinical psychologist, whom he has been seeing regularly ever since. 'He's been trying to get me back on track so I can decide what I'm going to do with my life in terms of work and things like that', 'Evan' said. He found the sessions, paid for by the Catholic Education Office, very useful.

The Catholic Education Office also gave 'Evan' information about other steps he could take, and he said he had been following these up. He has been to the principal of his old school, to the Order that ran it and to the police. He has found them all to be helpful and sympathetic.

'Evan' told us that contact with police led to the discovery that 'Ken Hooton', a lay teacher at a Sydney Catholic boys' school who sexually abused 'Evan' for three years, was already facing charges for offences against other victims. 'Evan' agreed to be a witness in the court case, which had had five adjournments at the time 'Evan' attended a private session. He believes 'Ken Hooton', in his 80s, will do everything he can to prevent the case going to trial. 'He's going to try to delay it till he falls off the perch', he said.

Though he said giving his statement to police was 'harrowing' and he doesn't look forward to giving evidence in court, 'Evan' told us he is positive about his dealings with the criminal justice system. Both the Director of Public Prosecutions and the police in charge of the case keep him well informed about what was happening, and the police officer he dealt with initially made the process as easy as possible. At 'Evan's' request, the officer came to his home to take his statement. 'He just said, "It doesn't matter what you tell me, I've heard it all before". He was an absolute professional'.

In the end, 'Evan' said speaking to the police was a relief. 'Even here today, it's cathartic, you know? Just to tell your story and somebody's prepared to listen. Because I couldn't tell my story. Here I am, 57 years old, and it's finally come out'. 114

As 'Evan's' example illustrates, sensitive, timely and clear communication is an essential part of a supportive response from an institution. Survivors described feeling empowered by knowing what was happening in their case and being included in decision-making.

Some survivors told us that a genuine and meaningful apology from the institution was a critical element of an appropriate institutional response. Some told us that it contributed to their healing process. The Royal Commission's *Redress and civil litigation* report discusses how a genuine apology can play an important therapeutic and restorative role for victims.¹¹⁵

In private sessions, we heard from 'Ridley', who told us he was abused in a Catholic orphanage in the 1960s when aged 10, and later in a Christian Brothers' school. Despite a difficult start to engaging with the Catholic Church's *Towards Healing* process in the early 2000s, 'Ridley' described meeting a leader from the Christian Brothers as 'a beautiful experience'. 'I got a beautiful apology from that guy', 'Ridley' said. '... he actually had showers in his eyes. And I went to rescue him ... I said, "Mate, you didn't do it". He said, "No, but people representing us did"'. 116

We heard examples of institutions responding well by taking immediate and appropriate action on abuse. 'Jay' and his mother, 'Leigh' described one such response during their private session. In 2014 some of 'Jay's' school friends disclosed that they were being sexually abused by a teacher. 'Jay' told his parents, who contacted the police. The school was supportive of 'Jay' and encouraged him to continue speaking out.¹¹⁷

'Jay' (son) and 'Leigh' (mother): impact of an immediate and transparent response

We heard how, in 2014, several of 'Jay''s Year 7 friends mentioned that 'Mr Blake', the sports master, had 'felt' them. 'I thought it was very strange', Jay said.

I knew that 'Mr Blake' had a very good relationship with those boys — he liked them a lot. They would talk a lot with 'Mr Blake' and joke around with him and everything and apparently they used to go on runs with him and stuff. Apparently — I'm not sure. I couldn't really believe it, but then I thought they can't be lying about something like that, that's not something you lie about. I thought, I better tell my mum. Then I told them I was going to tell my mum and they said, 'Yeah, that's good'. Then it all got sorted out.

That night 'Jay' told his mother, 'Leigh', what the boys had said. She recounted the conversation to 'Jay's' father, who recommended they call the police. 'Jay' initially disagreed:

I thought maybe we should tell the school first because I didn't want to hurt the school's reputation, maybe it could be dealt with in the school via the teacher. Then I thought, 'oh no, we should go to the police'.

'Leigh' said she also thought of calling the school first, but knew she'd made the right decision when Queensland Police acted swiftly. The school's response was also impressive, she said. After 'Mr Blake' was charged, he was immediately sacked from the school. The principal sent a text message and email to parents informing them of events and called an assembly of students and parents to discuss the matter. He put into practice the school's child protection and traumatic events policies and thoroughly briefed all teachers. The school counsellor was supported by external specialists and met with students, teachers and parents.

The principal also congratulated 'Jay' privately for the courage he'd shown in reporting the abuse.

'Jay' told the Commissioner that he'd learned from Year 4 classes with the school counsellor to tell someone if something was going wrong. 'You need five people you can talk to', he said. Students were told this could include the counsellor, their parents, teachers, the police or the kids' helpline.

'Leigh' said the abuse by 'Mr Blake' sent shockwaves through the school community. No one except the principal and the boys who'd disclosed the abuse knew that 'Jay' had reported 'Mr Blake', but she worried that it might be found out. Most people responded to the revelations with the safety of their sons uppermost in mind, however one family in particular refused to believe it.

'Leigh' told us: 'When [the principal] spoke at the assembly there was a real openness in what he said... It was very authentic – you could see there was a genuineness because there was an action behind it. So when he said the children were the most important, there was something behind it'.

She was reassured that school staff hadn't tried to deny or distance themselves from what had happened. 'It was seamless, there was no panic', she said.

'Leigh' said it was difficult to separate a system from an individual, but the principal's leadership drove the system response.

When that combines it's a perfect situation. I just think the way he shapes his members under him too in terms of the opportunities he gives his staff for leadership reflects through to the boys as well ... His leadership flows through the school.¹¹⁸

4.6.1 Other institutions with a responsibility

Survivors also described instances of appropriate and helpful responses to child sexual abuse by other institutions, such as police, other criminal justice and legal institutions, and others responsible for child safety.

Police responses can have a significant impact on a victim. Research conducted for the Royal Commission involving qualitative and quantitative analysis of private sessions found that survivors who related positive experiences of reporting to the police as adults tended to describe the police as compassionate and professional. They felt that action had been taken on their behalf. For many, the perpetrator was charged and convicted. When this was not the case, survivors said that they felt that the police had done as much as they could. Huon' told us he was abused by a teacher at his school in the 1990s. After he disclosed to the police, around 20 years later, the case went to trial. Huon' told us: The police said this was a very strong case, the strongest case they had seen in years. They weren't promising an outcome, but they were confident'. Huon' told us that the perpetrator was found not guilty. Despite the outcome, Huon' was impressed by the way the police and the Department of Public

Prosecutions handled the case, and the way they dealt with him personally. 'Just the way they handled it, and their sensitivity and communication, all that sort of stuff. It all made the process smooth'. The Department of Public Prosecutions' solicitor was 'fantastic ... very, very positive'.¹²¹

We heard from 'Ignatius', who told us he was sexually abused as a child by 'Father Blayney' on multiple occasions in the 1970s and targeted by 'Father Clarence' who attempted to fondle him when he was 13. He was approached by the police in the early 2000s, then investigating allegations against 'Blayney' and 'Clarence'. 'Ignatius' made a formal statement in a process he said he found daunting but positive. He described his experience of engaging with the police:

I went in and they were exceptional, fantastic ... They gave me the space to say what I was 100 per cent sure had happened, what I was 100 per cent sure didn't happen, and what were the grey areas. ... I was incredibly impressed with how it was conducted and the manner in which it was done. 122

We also heard about the positive impact of supportive legal representation. Giving evidence in *Case Study 19: The response of the State of New South Wales to child sexual abuse at Bethcar Children's Home in Brewarrina, New South Wales),* Ms Kathleen Biles told us that while she found the civil litigation process overwhelming and often could not fully understand what was going on and why, her solicitors were helpful in explaining what was happening.¹²³ She said: 'The only good thing about the civil litigation was my legal representation, because they supported me and I knew that they were there for me. ... It was my lawyers who got me through it all'.¹²⁴

For some victims, participation in the criminal justice system can form a part of their personal recovery. For example, in *Case Study 38: Criminal justice issues relating to child sexual abuse in an institutional context,* Mr Kevin Whiteley, who told us he was sexually abused from the age of eight until he was 16 years old, reflected on his experience of reporting his abuse to police and participating in the criminal justice system:

I have discovered, as a result of my own journey through this process, that it is vitally important for victims to tell their story, for their own benefit. Even if doing so does not result in the prosecution of their predator, it marks the beginning of their journey of recovery and growth. With support and expert help and treatment, it is possible for victims to recognise and repair the emotional scars they have carried through their lives, so that they can lead a normal and happy life. 125

Another survivor, 'Luke', told us in a private session that while he considers his perpetrator's sentence 'a bit of a joke', he still found the criminal justice process 'one of the most freeing things I've ever done ... I felt six inches taller, I just felt like this huge weight had been lifted'. Criminal justice responses to institutional child sexual abuse are discussed in detail in our Criminal justice report.

Other institutions, such as schools, and child protection and health institutions, can play a positive role for victims by responding to their needs appropriately, and with compassion and understanding. 'Jordan' and his parents said that 'Jordan' had a positive institutional response when he started attending a new school after being sexually abused by another pupil at his former school, and experiencing a poor institutional response. 'They really did such a wonderful job with him from the first meeting', his mother recalled. 'It was all about 'Jordan'. The first thing they said to him was, "Jeez mate, you've had a tough time"'. Rumours about 'Jordan' reached the other students, but there was no tolerance for bullying in the new school. 'They addressed the situation immediately', 'Jordan' remembers. 'They established solutions before I even got to the school'. He also found a counsellor he learned to trust at the school. He stayed and managed to complete his final exams.¹²⁷

'Graham' told us in a private session of the positive effect of the response of his GP to his disclosure of sexual abuse:

About a month later something just hit me like never before. My chest was so tight I thought I was going to have a heart attack, and a ball of knot in my stomach. I couldn't get any sleep and it just felt like having a nervous breakdown, which I'd never had before so I went to see my family GP. And he checked me over and physically fine. I told him the whole story. For the first time. He was so good with it all. After many tissues later I felt so relieved. I walked out of his office like a ton weight had been lifted off me. 128

Some institutions play a positive role for victims by providing a compassionate and authoritative forum for disclosure. For example, many survivors described their experience with the Royal Commission as an important part of their healing process, and how they felt empowered by being encouraged to share their experience and speak out. 'Leo' told us it had taken him two years to ring the Royal Commission, but he was glad he did:

For me the most important aspect of all of this is having someone listen to me, someone in authority without judgment, and [for them to] say, 'You're not lying. It is feasible that all these things happened and what you've experienced is what the offender is more likely to have done'. The main box I've ticked is to have you listen to me.¹²⁹

Endnotes

- 1 Name changed, private session, 'Meghan'.
- 2 CP Smith & JJ Freyd, 'Institutional betrayal', *American Psychologist*, vol 69, no 6, 2013, pp 576, 578; K Healy, 'Remembering, apologies, and truth: Challenges for social work today', *Australian Social Work*, vol 65, no 3, 2012, p 292; R Campbell, 'What really happened? A validation study of rape survivors' help-seeking experiences with the legal and medical systems', *Violence and Victims*, vol 20, no 1, 2005.
- A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 71.
- 4 A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 78–9.
- See also Family and Community Development Committee, *Betrayal of trust: Inquiry into the handling of child abuse* by religious and other non-government organisations, Family and Community Development Committee, Victoria, 2013, pp 88, 108.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 74; CP Smith & JJ Freyd, 'Institutional betrayal', *American Psychologist*, vol 69, no 6, 2013; CP Smith & JJ Freyd, 'Institutional betrayal', *American Psychologist*, vol 69, no 6, 2013.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 74; CP Smith & JJ Freyd, 'Institutional betrayal', *American Psychologist*, vol 69, no 6, 2013; DA Wolfe, PG Jaffe & JL Jette, 'The impact of child abuse in community institutions and organizations: Advancing professional and scientific understanding', *American Psychological Association*, vol 10, no 2, 2003, p 184.
- 8 Transcript of B Ferguson, Case Study 57, 30 March 2017 at 27706:10–19.
- 9 Name changed, private session, 'Johnny'.
- For example: Transcript of AYL, Case Study 26, 15 April 2015 at C7440:40–43.
- A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 85.
- 12 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 11.
- 13 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 64–5.
- Transcript of S Braieoux, Case Study 57, 30 March 2017 at 27762:10–19.
- The Chouliara, T Karatzias & A Gullone, 'Recovering from childhood sexual abuse: A theoretical framework for practice and research', *Journal of Psychiatric and Mental Health Nursing*, vol 21, no 1, 2014.
- Name changed, private session, 'Maree'.
- 17 Name changed, private session, 'Alexander Martin'.
- 18 Name changed, private session, 'Maxine'.
- 19 Name changed, private session, 'Jack Eric'.
- 20 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 13: The response of the Marist Brothers to allegations of child sexual abuse against Brothers Kostka Chute and Gregory Sutton*, Sydney, 2015, pp 7, 52.
- 21 Name changed, private session, 'Sergio'.
- Transcript of CK, Case Study 3, 18 November 2013 at 1636:1–6.
- 23 Transcript of GA Lamond, Case Study 23, 23 February 2015 at 11749:38-42; Exhibit 23-0006, 'Statement of Guy Lamond', Case Study 23, STAT.0462.002.0001 R at 0003 R.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 23: The response of Knox Grammar School and the Uniting Church in Australia to allegations of child sexual abuse at Knox Grammar School in Wahroonga, New South Wales*, Sydney, 2016, p 69.
- Transcript of WP, Case Study 12, 19 May 2014 at WA2332:30–39; Exhibit 12-0001, 'Statement of WP', Case Study 12, STAT.0256.001.00007_M_R at 0006_M_R.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Redress and civil litigation*, Sydney, 2015, pp 127–135.
- Name changed, private session, 'Tonya'.
- Transcript of FAA, Case Study 46, 28 November 2016 at 23828:42–23829:1; Exhibit 46-0001, 'Statement of FAA', Case Study 46, STAT.1278.001.00007_R at 0007_R.
- I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 161.
- For example: Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 32: The response of Geelong Grammar School to allegations of child sexual abuse of former students*, Sydney, 2015, pp 12, 14 and 52; Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 34: The response of Brisbane Grammar School and St Paul's School to allegations of child sexual abuse*, Sydney, 2015, p 68.
- Name changed, private session, 'Matilda'.

- D Tener & S Murphy, 'Adult disclosure of child sexual abuse: A literature review', *Trauma, Violence & Abuse*, vol 16, no 4, 2015, p 396; P Schaeffer, JM Leventhal & AG Asnes, 'Children's disclosures of sexual abuse: Learning from direct inquiry', *Child Abuse & Neglect*, vol 35, no 5, 2011, p 348; TA Rosler & TW Wind, 'Telling the secret: Adult women describe their disclosures of incest', *Journal of Interpersonal Violence*, vol 9, no 3, 1994, p 333.
- Name changed, private session, 'Alexander Martin'.
- I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 65.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 79; I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 65.
- Names changed, private session, 'Coralie' and 'Jack'.
- For example: Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 19: The response of the State of New South Wales to child sexual abuse at Bethcar Children's Home in Brewarrina, New South Wales, Sydney, 2015, p 12.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 79; DA Wolfe, PG Jaffe & JL Jette, 'The impact of child abuse in community institutions and organizations: Advancing professional and scientific understanding', *American Psychological Association*, vol 10, no 2, 2003, p 184.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland,* Sydney, 2015, p 22.
- 40 Name changed, private session, 'Ethel'.
- Names changed, private session, 'Becky and Gretel'.
- 42 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland*, Sydney, 2015, pp 69–70.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 5: Response of The Salvation Army to child sexual abuse at its boys' homes in New South Wales and Queensland*, Sydney, 2015, pp 10–11.
- Name changed, private session, 'Rosemary'.
- Name changed, private session, 'Milo'.
- Name changed, private session, 'Howard'.
- 47 SE Ullman, Social reactions to child sexual abuse disclosures: A critical review', Journal of Child Sexual Abuse, vol 12, no 1, 2002, pp 103, 109–10; SD Easton, 'Disclosure of child sexual abuse among adult male survivors', Clinical Social Work Journal, vol 41, no 4, 2013, p 346.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 79; J Astbury, *Child sexual abuse in the general community and clergy-perpetrated child sexual abuse*, Australian Psychological Society, 2013, p 12.
- 49 Name changed, private session, 'Joann'.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 29: The response of the Jehovah's Witnesses and Watchtower Bible and Tract Society of Australia Ltd to allegations of child sexual abuse*, Sydney, 2016, p 49.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 29: The response of the Jehovah's Witnesses and Watchtower Bible and Tract Society of Australia Ltd to allegations of child sexual abuse*, Sydney, 2016, p 49.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 29: The response of the Jehovah's Witnesses and Watchtower Bible and Tract Society of Australia Ltd to allegations of child sexual abuse*, Sydney, 2016, p 50.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 29: The response of the Jehovah's Witnesses and Watchtower Bible and Tract Society of Australia Ltd to allegations of child sexual abuse*, Sydney, 2016, p 50.
- 54 Exhibit 29-0001, 'Statement of [BCB]', Case Study 29, STAT.0603.001.0001 R at 0011 R.
- 55 Transcript of BCB, Case Study 29, 15 May 2015 at 15164:46–47; 15165:1–5.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 29: The response of the Jehovah's Witnesses and Watchtower Bible and Tract Society of Australia Ltd to allegations of child sexual abuse*, Sydney, 2016, p 51; Transcript of BCB, Case Study 29, 27 July 2015 at 15176:45-15177:4.
- 57 Transcript of BCB, Case Study 29, 15 May 2015 at 15169:9–15; Exhibit 29-0001, 'Statement of [BCB]', Case Study 29, STAT.0603.001.0001 R at 0016 R.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 29: The response of the Jehovah's Witnesses and Watchtower Bible and Tract Society of Australia Ltd to allegations of child sexual abuse*, Sydney, 2016, p 52.
- Name changed, private session, 'Claira'.

- 60 Name changed, private session, 'Clifford'.
- Royal Commission into Institutional Responses to Child Sexual Abuse, Report of Case Study No 3: Anglican Diocese of Grafton's response to child sexual abuse at the North Coast Children's Home, Sydney, 2014, p 4.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 3: Anglican Diocese of Grafton's response to child sexual abuse at the North Coast Children's Home*, Sydney, 2014, pp 35–7.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 8: Mr John Ellis's* experience of the Towards Healing process and civil litigation, Sydney, 2015, p 13.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 8: Mr John Ellis's* experience of the Towards Healing process and civil litigation, Sydney, 2015, p 75.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 8: Mr John Ellis's* experience of the Towards Healing process and civil litigation, Sydney, 2015, p 85.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 27: The response of health care service providers and regulators in New South Wales and Victoria to allegations of child sexual abuse*, Sydney, 2016, p 47.
- 67 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 27: The response of health care service providers and regulators in New South Wales and Victoria to allegations of child sexual abuse*, Sydney, 2016, p 47.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 27: The response of health care service providers and regulators in New South Wales and Victoria to allegations of child sexual abuse*, Sydney, 2016, p 47.
- 69 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 27: The response of health care service providers and regulators in New South Wales and Victoria to allegations of child sexual abuse*, Sydney, 2016, p 19.
- K McGregor, M Glover, J Gautam & S Julich, 'Working sensitively with child sexual abuse survivors: What female child sexual abuse survivors want from health professionals', *Women & Health*, vol 50, no 8, 2010; K Havig, 'The health care experiences of adult survivors of child sexual abuse: A systematic review of evidence on sensitive practice', *Trauma*, *Violence & Abuse*, vol 9, no 1, 2008; LS Friedman, JH Samet, MS Roberts, M Hudlin & P Hans, 'Inquiry about victimization experiences: A survey of patient preferences and physician practices', *Archives of Internal Medicine*, vol 152, no 6, 1992.
- 71 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 141–42.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Consultation paper: Criminal justice,* Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016.
- 73 Name changed, private session, 'Rebecca'.
- 74 Transcript of A D Steer, Case Study 23, 23 February 2015 at 11730:3–34; Exhibit 23-0005, 'Statement of Adrian Steer', Case Study 23, STAT.0491.001.0001 R at 0005 R.
- 75 Transcript of G Davies/C Leaney, Case Study 46, 28 November 2016 at 23858:19–23.
- 76 Transcript of G Davies/C Leaney, Case Study 46, 28 November 2016 at 23858:26–33.
- 77 Name changed, private session, 'Remy'.
- 1 Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 157.
- 79 Name changed, private session, 'Tara'.
- Name changed, private session, 'Beck'.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 30: The response of Turana, Winlaton and Baltara, and the Victoria Police and Human Services Victoria to allegations of child sexual abuse,* Sydney, 2016, pp 86–89.
- Transcript of T Dale, Case Study 24, 29 June 2015 at 14648:5.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Consultation paper: Institutional responses to child sexual abuse in out-of-home care*, Sydney, 2016, p 119.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 30: The response of Turana, Winlaton and Baltara, and the Victoria Police and the Department of Health and Human Services Victoria to allegations of child sexual abuse*, Sydney, 2016, p 89; Young Persons Panel 1, Case Study 24, 29 June 2015 at 14651:17; Royal Commission into Institutional Responses to Child Sexual Abuse, *Consultation paper: Institutional responses to child sexual abuse in out-of-home care*, Sydney, 2016, p 119.
- DA Wolfe, PG Jaffe & JL Jette, 'The impact of child abuse in community institutions and organizations: Advancing professional and scientific understanding', *American Psychological Association*, vol 10, no 2, 2003, p 186.
- Name changed, private session, 'Zahara'.
- 87 Transcript of C Tambling, Case Study 23, 24 February 2015 at 11822:6–47 and 11823:1–5; Exhibit 23-0011, 'Statement of Coryn Tambling', Case Study 23, STAT.0499.001.0001_R at 0015_R-0016_R.
- L Gallo-Silver, CM Anderson & J Romo, 'Best clinical practices for male adults survivors of childhood sexual abuse: "Do no harm", *The Permanente Journal*, vol 18, no 3, 2014, p 84.
- 89 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 27: The response of health care service providers and regulators in New South Wales and Victoria to allegations of child sexual abuse,* Sydney, 2016, p 69.
- 90 Transcript of AWI, Case Study 27, 12 May 2015 at 14405:2–8; Exhibit 27-0025, 'Statement of AWI', Case Study 27, STAT.0558.001.0001 R at 0024 R.
- 91 Name changed, private session, 'Norris'.
- 92 A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 78.

- DA Wolfe, PG Jaffe & JL Jette, 'The impact of child abuse in community institutions and organizations: Advancing professional and scientific understanding', *American Psychological Association*, vol 10, no 2, 2003, pp 185–6.
- 94 Name changed, private session, 'Geoffrey'.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, *Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 31.
- J Breckenridge & G Flax, Service and support needs of specific population groups that have experienced child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 46–7.
- 97 Transcript of ARY, Case Study 23, 23 February 2015 at 11697:17–18; Exhibit 23-0002, 'Statement of [ARY]', Case Study 23, STAT.0483.001.0001 R at 0002 R.
- 98 Transcript of ARY, Case Study 23, 23 February 2015 at 11696:25–34; Exhibit 23-0002, 'Statement of [ARY]', Case Study 23, STAT.0483.001.0001 R at 0007 R.
- 99 Transcript of WP, Case Study 12, 19 May 2014 at WA2332:30–36; Exhibit 12-0011, 'Statement of WP', Case Study 12, STAT.0256.001.0001 M R at 0006 M R.
- 100 Transcript of WQ, Case Study 12, 19 May 2014 at WA2358:17–18; Exhibit 12-0004, 'Statement of WQ', Case Study 12, STAT.0257.001.0001 M R at 9:43.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 12: The response of an independent school in Perth to concerns raised about the conduct of a teacher between 1999 and 2009*, Sydney, 2015, pp 9–10.
- 102 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 160.
- Name changed, private session, 'Jared Wayne'.
- 104 Transcript of BAV, Case Study 28, 20 May 2015 at C8236:42–47; Exhibit 28-0011, 'Statement of BAV', Case Study 28, STAT.0584.001.0001_R at 0012_R.
- 105 Transcript of BAV, Case Study 28, 20 May 2015 at C8229:39–44; Exhibit 28-0011, 'Statement of BAV', Case Study 28, STAT.0584.001.0001_R at 0004_R.
- DA Wolfe, PG Jaffe & JL Jette, 'The impact of child abuse in community institutions and organizations: Advancing professional and scientific understanding', *American Psychological Association*, vol 10, no 2, 2003, pp 182–3.
- DA Wolfe, PG Jaffe & JL Jette, 'The impact of child abuse in community institutions and organizations: Advancing professional and scientific understanding', *American Psychological Association*, vol 10, no 2, 2003, p 183.
- Transcript of BXI, Case Study 39, 5 April 2016 at 18589:40–47 and 18590:1–4; Exhibit 39-0010, 'Statement of [BXI]', Case Study 39, STAT.0977.001.0001_R at 0006_R.
- D Palmer, *The role of organisational culture in child sexual abuse in institutional contexts,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 52.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 21: The response of the Satyananda Yoga Ashram at Mangrove Mountain to allegations of child sexual abuse by the ashram's former spiritual leader in the 1970s and 1980s*, Sydney, 2016, p 53.
- 111 Transcript of BCG, Case Study 29, 28 July 2015 at 15292:29–35.
- 112 Name changed, private session, 'Julia Maree'.
- 113 Name changed, private session, 'Evan'.
- Name changed, private session, 'Evan'.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Redress and civil litigation*, Sydney, 2015, p 140.
- 116 Name changed, private session, 'Ridley'.
- Names changed, private session, 'Leigh' and 'Jay'.
- Names changed, private session, 'Leigh' and 'Jay'.
- 119 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 141.
- 120 I Katz, A Jones, BJ Newton & E Reimer, *Life journeys of victim/survivors of child sexual abuse in institutions: An analysis of Royal Commission private sessions,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 141.
- 121 Name changed, private session, 'Huon'.
- 122 Name changed, private session, 'Ignatius'.
- Transcript of K M Biles, Case Study 19, 22 October 2014 at 10066:21–25; Exhibit 19-0002, 'Statement of Kathleen Monica Biles', Case Study 19, STAT.0381.001.0001_R at 0017_R.
- Transcript of K M Biles, Case Study 19, 22 October 2014 at 10066:41–45; Exhibit 19-0002, 'Statement of Kathleen Monica Biles', Case Study 19, STAT.0381.001.0001_R at 0017_R.
- Transcript of K J Whiteley, Case Study 38, 15 March 2016 at 17459:5–16; Exhibit 38-0007, 'Statement of Kevin James Whiteley', Case Study 38, STAT.0914.001.0001 R at 0009 R.
- Name changed, private session, 'The Peterson Family'.
- 127 Name changed, private session, 'Jordan'.
- 128 Name changed, private session, 'Graham'.
- 129 Name changed, private session, 'Leo'.

5 Ripple effects

Child sexual abuse not only affects the victim. It has ripple effects that reverberate to a wider network of people. These ripple effects can continue over time, affecting subsequent generations. Those affected can include the victim's family, carers and friends, as well as other children and staff in the institution, the community and wider society.

The experiences of parents, carers, siblings, partners and children ('secondary victims') are different from those of primary victims, but the effects of child sexual abuse and institutional responses can also be significant for them.¹ Secondary victims can suffer adverse impacts on their mental health, relationships, family functioning, employment, financial security and social connectedness. 'Secondary trauma' or 'vicarious trauma' is the term used to describe the emotional effects for an individual when they hear firsthand about the traumatic experiences of another.² Research on trauma suggests that secondary traumatic stress can be similar to the symptoms of post-traumatic stress disorder (PTSD).³

In private sessions and public hearings we heard about the way that child sexual abuse can significantly affect the lives of people other than the victim. In private sessions, survivors often spoke about the impact of the child sexual abuse on their parents, partners and children. Family members (including parents, partners, siblings, children and other relatives), whistleblowers and others also attended their own private sessions to talk about the sexual abuse of their family member or a child they knew, sometimes speaking about the impacts on their own lives. Family members of the victim made up 6.4 per cent (442 private sessions), and whistleblowers associated with the institution made up 0.3 per cent (21 private sessions) of all those who attended private sessions (see Volume 2, *Nature and cause*).

In public hearings, we heard evidence directly from parents or family members of victims about the far-reaching impacts of child sexual abuse. We also heard from staff members in the institution where a child had been sexually abused, including whistleblowers.

This chapter outlines what we learned about the ripple effects of child sexual abuse in institutional contexts and the responses of institutions. It details some of the commonly identified impacts on:

- victims' families
- · other individuals associated with the institution where the child sexual abuse occurred
- · communities, such as religious or cultural groups
- Australian society.

5.1 Victims' families

Victim's families are often deeply affected by child sexual abuse. Even when they are not aware that the victim has been abused, because the sexual abuse has not been disclosed, they may still be significantly affected by the physical and mental health and behaviour of the victim.

We most commonly heard about impacts on:

- parents
- partners (in adulthood)
- siblings
- children of victims.

However, we also heard that extended family members (such as aunts, uncles, cousins and grandparents), carers and friends can be significantly affected by child sexual abuse in an institutional context. The references in research on the impacts of child sexual abuse on these individuals are sparse. Some participants in qualitative research we commissioned on disclosure and family relationships, expressed concern about how extended family members, such as elderly grandparents, would cope with the emotional shock.⁴

In private sessions, survivors told us that extended family members and carers can also be directly affected by the difficulties associated with caring for the victim after the abuse. Many of the survivors we heard from had been cared for during childhood by members of their extended family, such as grandparents, or by foster carers and kinship carers. Others had received significant support and care over the course of their lives from friends. These individuals are also likely to have experienced some of the impacts described in this section.

5.1.1 Parents

I think about what happened to my son all of the time; I feel so terrible for him that he has carried this on his own for so long.⁵

Parents can experience devastating effects when they discover their child has been sexually abused. Shock, disbelief, confusion and anger are common feelings when they first discover that their child has been sexually abused. Commissioners were also often told about intense feelings of fear, anxiety and stress, shame and guilt, and grief and sadness among parents of victims. Hearing about the sexual abuse of their child can also be so overwhelming that parents may experience denial.

Parents can experience impacts on their:

- mental health
- parenting confidence
- relationship with their child
- relationship with their partner
- employment and financial security
- social connection
- trust
- faith and religious involvement.

Both parents of victims who disclose as children and parents of victims who disclose as adults can suffer these impacts. However, according to qualitative research we commissioned on disclosure and family relationships, there are some differences between the impacts on the two groups of parents. The main issues faced by parents whose child disclosed in childhood were anxiety about the parental role and changing their parenting practices in light of their child's experience. In contrast, parents whose child disclosed in adulthood were mainly concerned for the victim's wellbeing and supporting their needs. The support of the victim's wellbeing and supporting their needs.

Parents of victims who disclose as children

We heard that when a child discloses sexual abuse, it can lead to an immediate sense of crisis in parents, who may become anxious and concerned about the safety of their child. The fear and anxiety for the safety of their child could lead to over-protectiveness and hyper-vigilance.¹¹ 'Kelly' described her reaction following her son's disclosure of sexual abuse at the age of 10:

When it happened, it was halfway through Year 6 and I didn't want him to go to school. I kept him at home. I just — I actually felt like I wanted to wrap him in cotton wool and just keep him close to me. So I kept him at home for quite a few days. We went to movies, we just spent the time together and I think it was a counsellor that we'd seen who said this isn't good, you've got to let him go back to school and you can't keep him at home forever. I just didn't want him out of my sight.¹²

Some parents told us in private sessions that finding out that their child had been sexually abused led to a crisis in confidence in their ability to parent, in some cases changing the nature of their relationship with their child. The qualitative research on disclosure and family relationships found that some parents who participated in the study experienced doubts about their parenting skills and judgment upon learning about their child's sexual abuse. A father of a survivor who participated in the research revealed, You feel you failed as a parent. Because the signs were there. I keep asking myself, "How didn't I recognise the signs?"

This feeling of guilt and failure as a parent can be intensified where parents believe they were groomed by the perpetrator in order to enable the sexual abuse of their child. For example, in *Case Study 12: The response of an independent school in Perth to concerns raised about the conduct of a teacher between 1999 and 2009 (Perth independent school),* we heard from WQ about the sexual abuse of her son, WP, by a teacher, YJ, at his primary school. She told us that while watching an episode of Four Corners on ABC, it dawned on her that WP and the family 'may have been groomed by YJ and that my increasing fears and suspicions about YJ's attentions toward WP and our family may have been something more sinister than I dared to believe'. She said:

The sexual abuse suffered by WP has had a dreadful impact on our family. I have had terrible bouts of guilt that I should have done something sooner and acted on my initial suspicions. I have suffered deep bouts of depression, loneliness, fear and mistrust ... I focus on my failings as a parent, and the entire process has put an enormous strain on my marriage, especially in the first four years after WP told us about the abuse. My husband and I attended marriage counselling in an effort to stay together. It was really difficult. 16

We also heard in private sessions that the way parents and families responded to disclosures of child sexual abuse could lead to relationship problems between the survivor and their parents. When family members reacted to a disclosure with disbelief, minimised the experience of the sexual abuse, or blamed the victim for the abuse, it could increase feelings of isolation and victimhood and could create strained relationships within the family.¹⁷ During private sessions, we heard of several cases where division was created by victims and parents not sharing the same view of whether to pursue a response from the institution, with the victim wishing to let the issue go.¹⁸

The tasks, responsibilities and roles associated with parenting a child who has been sexually abused can also affect relationships between partners. Participants in the research on disclosure and family relationships spoke about how the anxiety about parenting and a changed view of their child's future became sources of conflict and tension within the family, leading to disagreements between partners.¹⁹

Parents told us that hearing about the sexual abuse of their child, and dealing with the effects of the abuse, can significantly affect their own mental health. For example, we heard that parents can experience depression, anxiety and PTSD as a result. For some parents, hearing that their child has been sexually abused brings up memories of their own childhood abuse, triggering them to re-experience its trauma. 'Francene', who told us her children were sexually abused as she had been in her own childhood, said in a private session:

I'd love to have some help. I need closure. I know it will never close. I know that I'm going to go to my grave with it, but if I can get some help to cope with it – because I know I'm not coping. People will come over and I'll talk about it and I shouldn't be. I was doing pretty well until my children were abused and that was my downhill, you know. I just went down. That happened, I just thought, 'Why?' Three people in one family. All of us and it's so unfair.²⁰

Not only are parents affected by learning about the sexual abuse of their child, they can be affected by coping with the ongoing negative impacts the sexual abuse has on their child, such as trauma-related behavioural issues, avoidance of school and mental health impacts. In the *Nature, cause and impact of sexual abuse* case study, Ms Susan Campbell told us about the impacts on her, when her teenage daughter took her own life after struggling to cope with the effects of sexual abuse:

For myself, the effects on my own mental health have been hard to accept and even harder to deal with. A diagnosis of post-traumatic stress disorder is certainly not a key to enhanced prospects. It is not easily managed. It comes back at night with grief and uncontrollable tears; by simply stopping one from staying asleep, so one wakes often during the night, agitated and restless, and so begins the next day tired and drained.²¹

In Case Study 45: Problematic and harmful sexual behaviours of children in schools (Harmful sexual behaviours of children in schools), EAL spoke about how their daughter's (CLF), behaviour as a result of sexual abuse at Shalom Christian College, Queensland affected her and her partner: 'CLF changed after what happened at Shalom. So did our family. CLF began drinking and drugging when we got her back from Shalom. I think she was trying to escape what happened'.²²

Dealing with the impacts of their child's sexual abuse led to loss of employment and financial disadvantage for some parents. For example, 'Helen Christine' told us that her son 'Robert', who had a disability, started to exhibit sexualised behaviours after being sexually abused while living in a care home. When he returned from the care home, he behaved inappropriately with a younger sibling. We heard that in the wake of this incident, child protection services mandated that 'Robert' had to be supervised by two adults at all times. This meant that 'Helen Christine's' partner 'Brian' had to quit his job, which put the family in financial strife. She told us 'Brian' hated being unemployed and it was a stressful time.²³

The pursuit of justice or redress for their child can also take a toll on parental employment and family finances. Findings from a study of 39 parents of children sexually abused in a day care setting found that many families in the study relocated after the abuse in response to media coverage and the legal process. Parents in the study described taking extensive time off work due to stress and losing social connections.²⁴

Many parents told us that they felt betrayed and let down by institutions because of the sexual abuse of their child, who they had entrusted into the institution's care, but also because of the poor response to the disclosure of child sexual abuse. Some had consequently lost trust in institutions more broadly which had an ongoing effect on how they accessed services for themselves and their families. Aboriginal and Torres Strait Islander families in particular have experienced systemic injustice from institutions over many generations. Further experiences of harm, such as having a child sexually abused in an institution, adds to the cumulative fear and mistrust of institutions, and makes it more difficult for these parents to access services, such as mental health treatment, for themselves or their children. As Dr Marshall Watson, child adolescent forensic psychiatrist, told us in the *Nature*, cause and impact of child sexual abuse case study:

families have become mistrustful of services. What happens is that things get held in and dealt with within families. As a result, Aboriginal people present end-stage or in mental health crisis with a significant burden of illness and disease already.²⁵

Some parents whose children were abused in religious institutions described a crisis or loss of faith as a result of the abuse and the way that the institution responded. We heard about the impact on the faith of a Catholic woman, 'Deidre Catherine', when she discovered her son had been sexually abused at her church. Her ex-husband 'Ross' told us, 'It cut her in half. She needed ... to be a Catholic. She needed to go to church – for strength. And when we found out what happened to [their son]. Zzzt! She's cut off at the guts!'²⁶

Parents also spoke about their determination to provide strength and support for their child. As EAL in the *Harmful sexual behaviours of children in schools* case study told us: 'At the end of the day, EAM (her partner) and I have to stay strong for [their daughter]. As parents you bring kids into the world, you have to look after them, have to be there for them. She didn't ask for what happened to her at Shalom'.²⁷

We heard many stories of parents who have spent years supporting their children and pursuing justice on their behalf, and in some cases on behalf of others who have suffered child sexual abuse. For example, Mr Anthony Foster and Mrs Christine Foster, whose daughters Emma and Katie were sexually abused by Kevin O'Donnell, the parish priest, when they were pupils at the Sacred Heart Catholic Primary School at Oakleigh in the late 1980s or early 1990s, dedicated many years of their lives to pursuing redress for their daughters and their family.²⁸ They also devoted themselves to achieving justice for other victims of child sexual abuse and their families, whilst managing their own grief.²⁹

Parents of victims who disclose as adults

Parents of victims who disclosed as adults can also experience feelings of shock and disbelief, anger, sadness and grief when they hear that their child was sexually abused, in many cases years previously. Guilt was an emotion frequently mentioned by parents of victims who disclose as adults. Ms Marija Radovejic, giving evidence in the *Nature, cause and impact of child sexual abuse* case study, told us of the deep guilt she felt about the sexual abuse of her son at his school, which he disclosed some years after the abuse. Ms Radovejic described how she and her husband 'knew that something critical was happening, but still had no idea that he was being sexually abused'. They believed that the Christian school they had sent him to was 'a cocoon' and safe, like their home. Ms Radovejic said:

We didn't want to 'rock the boat' too much with the HSC being so soon. Our attempt to discuss this with our son met with calculated silence. Finally, after the HSC, when pushed about his sexuality, our son declared that he had slept with two men, one of whom had been one of his teachers. I immediately told him I would go to the Police at which point he declared that if I did he would deny it all by saying that he had been the initiator all along! We were then totally helpless. Many years of self-destructive behaviour later, he did admit to having been abused, but did not name his perpetrator. He told me that his perpetrator had given him money during his senior years and even after he had left school. We are now left to deal, not only with our endless feelings of loss and trauma but also with the deep guilt about our naive ignorance.³⁰

The extent to which Ms Radovejic and her husband had been groomed by the perpetrator only became clearer to them after their son's death.³¹

In private sessions and public hearings, parents also expressed grief about the victim's loss of childhood. A mother, ATU, whose son was sexually abused at school, gave evidence in *Case Study 23: The response of Knox Grammar School and the Uniting Church in Australia to allegations of child sexual abuse at Knox Grammar School in Wahroonga, New South Wales (Knox Grammar School)*. She spoke of her shattered dreams for her now adult child: 'As a mother, I have hopes and dreams for my children. My dreams for [my son] were stripped down to mere survival'.³²

In private sessions, parents of victims who disclosed as adults described to us their anxiety about their child's wellbeing and safety. Instead of stepping back from active parenting after their children reached adulthood, disclosures of child sexual abuse sometimes meant that parents intensified or re-ignited their involvement in their adult children's lives, taking on responsibility for emotionally and financially supporting them, while having less control and influence over their adult child's safety.³³ Participants in the research on disclosure and family dynamics said that their lives were made unpredictable by their adult child's mental health issues which they attributed to the impact of child sexual abuse, trouble maintaining employment, trouble with police, and alcohol and other drug abuse.³⁴ Some parents told us in private sessions and public hearings that they effectively became carers for their adult children, looking after them on a daily basis.³⁵ Providing this support can overwhelm other

aspects of family life, as we heard from 'Vicki', the mother of 'Page', who was sexually abused by a Salvation Army captain as a child. From about age 12, 'Page' tried to talk about the abuse but was not able to articulate what had happened to her until she was 18, when she started experiencing flashbacks. 'Vicki' told us:

The effects on our family was that there was nothing normal ... 'Page' couldn't be left alone at all. She couldn't sleep in a room by herself. I either slept on her bedroom floor or she slept in our room for year after year after year, night after night. She would wake up in the night crying, upset. I don't think I had a whole night sleep for over 10 years. Life was a nightmare.³⁶

Parents worried about the risks of their adult child attempting to take their own lives, perceiving they had less control than when their children were young. In *Case Study 36: The response of the Church of England Boys' Society and the Anglican Dioceses of Tasmania, Adelaide, Brisbane and Sydney to allegations of child sexual abuse (Church of England Boys' Society)*, we heard from one mother, BYD, whose adult son made several attempts to take his own life. BYD spoke of the constant worry: 'Whenever I cannot contact him I think the worst. This is a heavy burden for me to carry'.³⁷ For some parents, these fears became a reality – parents told us of their immense grief at losing their adult child to suicide.³⁸

Parents of adult children also spoke to us about the damage done to their relationship with the institution within which their child was sexually abused, which had, before the disclosure, been an important part of their lives. In the qualitative research mentioned above, parents of victims who disclosed as adults told researchers that their long-held assumptions and beliefs about the institution were deeply affected.³⁹

In some cases, the effects on parents of a poor institutional response to their child's abuse continue for many years. In Case Study 43: The response of Catholic Church authorities in the Maitland – Newcastle region to allegations of child sexual abuse by clergy and religious (Catholic Church authorities in Maitland – Newcastle), Ms Audrey Nash, 90 years old, told us how she felt devastated by the poor response by the Catholic Church to the sexual abuse of her sons (CQT and Andrew) at a Marist Brothers school in the 1970s, as well as its response to other people affected by child sexual abuse:

I spent my whole life committed to my Church and working for the Catholic Church. At my time of greatest need, after Andrew died, the only pastoral response I got from the Church was [a Father] telling me that sexual abuse of boys has been going on for thousands of years. I don't go to Church now. I still have my beliefs, but I am appalled at the lack of empathy, the lack of support and the lack of concern for all of the people affected by child sexual abuse. I am disgusted by the efforts of the Church to cover up the abuse and to protect the abusers. I have been devastated by what happened to CQT and Andrew, and my children and I have been just as devastated by the reaction of all of the members of the Catholic Church. I have been left feeling empty. I also feel so stupid that I used to fear and revere these people and that I used to respect them and look up to them.⁴⁰

The loss of connection with the community can be particularly shattering for parents from minority cultural and linguistic groups, given that community members may rely on the associated religious community to meet their social, cultural and spiritual needs. Giving evidence for *Case Study 22: The response of Yeshiva Bondi and Yeshivah Melbourne to allegations of child sexual abuse made against people associated with those institutions*, Mr Zephaniah Waks, whose son Mr Manny Waks was sexually abused by a staff member at Yeshivah College Melbourne, described how, as a member of the Chabad movement, he and his wife relied on the Yeshivah religious community for their social life.⁴¹ After the abuse of Manny was disclosed publicly, he and his wife were shunned by members of the community.⁴² Mr Waks said:

After Manny's public statement, I experienced intimidation and isolation from the Yeshivah community. I felt that leaders, members and many then-friends from the Yeshivah Centre community talked negatively about me behind my back and were very unapproachable. I was excluded from Chabad customs, refused spiritual blessings by senior Yeshivah leaders, physically assaulted in the synagogue by a member of the Yeshivah community and I lost many people that I considered to be good friends ... I am no longer a part of the Yeshivah community.⁴³

5.1.2 Partners

We heard in private sessions and public hearings that many victims first disclosed to their partner, who was often one of their main sources of strength and support (discussed in Chapter 2).

Research for the Royal Commission found that disclosure can bring relief for partners, as it can help explain certain behaviours demonstrated by the victim. A number of partners of victims told us that the disclosure explained things they had wondered about in their partner's behaviour, such as avoiding physical intimacy or suffering from depression or anxiety. In some cases the disclosure also helped partners develop an improved understanding of the victim and help them avoid situations that may act as a trigger for trauma responses. We heard how disclosures of child sexual abuse can evoke empathy and understanding in a partner. Michael', whose wife experienced sexual abuse as a child in an orphanage, described how learning about the sexual abuse helped him better support her:

It's made it easier. I know why I've got to peel the spuds now. I never realised how hard it was for her peeling potatoes after the millions of spuds she'd peeled at the orphanage as punishment ... And I know it's a little thing, but it's a big thing. We understand each other a lot better now, I think.⁴⁶

Partners can also be vicariously affected by the abuse, dealing with its effects on the victim or helping a victim to report the abuse and seek justice and redress.⁴⁷ They can be affected heavily by caring for and supporting a survivor with profound mental health issues, including PTSD, depression and suicidality. Some of the emotional effects of child sexual abuse may be challenging for a partner to live with on a daily basis, and can include violence and abusive behaviour. We heard repeatedly from partners about the stress and demands of supporting victims, especially through the process of reporting abuse and seeking justice and redress. In a private session, we heard from 'Joanne', the wife of 'Brett', a survivor of child sexual abuse. 'Joanne' described her experience of attending a support group for partners of survivors of child sexual abuse:

Those women (partners) at that meeting were struggling to survive with their lives, they were weighed down with responsibility, grief and sorrow. It then occurred to me that they were as broken as their partners. That their lives had been as implicitly affected as had their partners. It's similar to a drug addiction and how that affects the user and their loved ones and family. The drug/sexual abuse affects all who come into contact with it. The victim does not stand alone in it, he sheds its effects. We, the partners, experience it and we carry it, the 'dirty little secret', as surely as does the victim.⁴⁸

In private sessions, partners told us how emotionally distressing disclosures of child sexual abuse can be for them, causing intense feelings of shock and grief. Some partners told us the disclosure evoked complicated feelings of insecurity and uncertainty, by challenging previous beliefs that there were no secrets in the relationship. Others spoke about the burden of carrying the knowledge of the abuse without being able to speak about it, or being confronted by negative reactions from the extended family when they did so. 'Millie' talked about the burden of initially keeping the abuse of her husband a secret, as he wanted, and her sense of disappointment in people's reactions when he disclosed more widely. 'People did not know how to react, and our sense of isolation within our pain began snowballing', she said.⁴⁹

Partners often raised in their discussions the difficulties of maintaining a relationship with the victim. Research into the impacts of child sexual abuse has found that 'the marital and intimate partner relationships of adult victims of child sexual abuse have often been characterised ... as being unstable and unhealthy'. As discussed in Chapter 3, in some cases these relationship difficulties led to relationship breakdown and separation. 'Leland', who told us three years of sexual abuse by a priest began in Year 8, met his wife during a time when he was very isolated, taking drugs, drinking alone and not mixing with other people. He told us his wife suffered because of his neglect of her and she left him for another man. ⁵¹

In some cases, partners spoke of having to take on full responsibility for parenting of their children, when victims had a reduced capacity from psychological, physical or social dysfunction. For example, one wife of a survivor described how her husband became incapacitated by mental and physical illness as a result of being sexually abused as a child migrant in a

Christian Brothers institution. In 1985, her husband was classed as having a total and permanent disability, and could not continue working. His wife told us that she had to find work, as well as care for both her husband and a child with disability, and her other children:

The day was never long enough for me there was physiotherapy that needed to be done for [my son], he needed to be fed, as he could no longer raise his hands, and [my husband] was so drugged out with his medication that I felt I had two disabled people to look after. Many days I would come home and find [my husband] crying telling me that he is a burden on the family and he does not want to live anymore. This created further worry to me I was constantly scared that he may harm himself.⁵²

As with parents, partners experienced financial difficulties in dealing with the impacts of child sexual abuse, including the impact of financially supporting the survivor who is unable to work, costs associated with accessing health services and mental health support services, and the financial costs of pursuing redress from institutions. We heard that some partners of survivors struggled to maintain employment and financial security because of the impact on their physical and mental health. Some partners were unable to continue working because they were caring for the survivor. One survivor, 'Edmond' told us that when he began counselling he quickly realised that he couldn't afford to keep it up long term. He told us his wife was already working 'twice as many hours as she should be' to financially support him and their young children. It was then that he decided to sue the Catholic Church.⁵³

5.1.3 Siblings

Siblings can also be greatly affected by the abuse of their brother or sister. We heard in private sessions and public hearings that in some instances siblings were the first person the victim disclosed to. We heard that siblings are often a source of support and strength for victims.

Siblings can be affected in different ways, depending on the existing relationship between siblings and other family dynamics, and on whether the disclosure of abuse happened when the victim and siblings were children or adults. A range of impacts on the complex relationships between siblings were examined as part of research on family relationships and disclosure conducted for the Royal Commission.⁵⁴ They included:

- siblings taking on a supportive or protective role, particularly when the victim is younger than their sibling
- perceptions of resentment or jealousy between victims and siblings as parents struggle to balance support for victims with support for other siblings
- the breakdown of sibling relationships, especially in adulthood.⁵⁵

Although research in this area is sparse, one research review of literature describes how siblings can be significantly affected by the sexual abuse of their brother or sister. ⁵⁶ It suggests siblings can experience anxiety, anger, sadness, guilt and depression following a disclosure of child sexual abuse by the victim. Older siblings in particular may feel guilty for a variety of reasons, including for not having been able to protect the victim. It also suggests that differential treatment by parents between a sibling and the victim, which may have resulted from a parent's need to support the victim, can lead to sibling conflict and antagonism. Family stress because of the sexual abuse can also impact on a sibling, as can the disruption to daily life and family finances when families pursue redress or criminal justice measures.

In private sessions and public hearings, we heard that child sexual abuse could lead to tension, division and estrangement between victims and their siblings in childhood. These tensions could often continue into adult life. ATU, the mother of a survivor, gave evidence in the *Knox Grammar School* case study on the difficult relationship between her children:

While my other two children dearly love [the survivor], they are unable to have a close relationship with him due to his unpredictable behaviour and the trauma they witnessed surrounding his mental health. The impact also affects my extended family, particularly my parents, [the survivor's] maternal grandparents.⁵⁷

In the *Catholic Church authorities in Maitland – Newcastle* case study, CNE described the impact of his experiences of child sexual abuse on his relationships with his family:

Due to everything that has happened, my family is very broken. I only speak to about two of my siblings, my mum and my dad ... It is not just me who has suffered, but also those close to me who are also victims.⁵⁸

'Doris' told us how the sexual abuse of her son by two teachers in primary school in the 1970s negatively affected the relationship between her two sons. Formerly close, the victim and his brother began to fight, which 'Doris' believed was a result of the victim 'acting out' his pain on his younger brother through bullying. The brothers became estranged in adulthood and 'Doris's' relationship with both of her sons was affected, with her younger son believing that she did not protect him from her oldest son:

The consequences of what happened are sadly still being played out ... I'm here today because I want the Commission to understand the incredibly malignant effect [abuse] can also have on the families of these children. It can destroy the very fabric within which a healthy family operates.⁵⁹

The impacts on a survivor can be scary and alienating for a young sibling. 'Chantelle Elise' told us about the devastating effects of a priest's sexual abuse of both of her sisters, 'Belinda' and 'Angela'. Her oldest sister, 'Belinda', became very unwell and over a period of years was admitted many times to hospital with mental and physical health problems, and for

drug rehabilitation. The other sister, 'Angela', became withdrawn and depressed, and started drinking heavily. 'I was so young, it was all very scary', 'Chantelle Elise' said. She recalled once being with her mother and 'Belinda' when they started arguing. 'Because I'd never seen such family discord, I started crying, then ['Belinda'] felt bad. Still again, she was caring of me, which was really nice. She was always caring'. As 'Belinda's' health deteriorated and her drug use increased, 'Chantelle Elise' found it difficult to be with her. After 'Belinda's' death in the late 2000s, 'Chantelle Elise' found it difficult to continue studying. By that time, 'Angela's' health had also deteriorated and a sudden accident left her with significant injuries. 'Chantelle Elise' became one of the people caring for her. 'Chantelle Elise' is now married with children who adore 'Angela'. 'Chantelle Elise' told the Royal Commission that in comparison to events in her life, other things often seemed 'trivial'. She found it hard to answer questions from new people about her childhood and upbringing, and she wondered what might have been had her sisters not been sexually abused by that priest. 'Chantelle Elise' said:

In my current world I have nice friends, kind friends, and wholesome friends, but they don't know the world 'Belinda' was from and henceforth they don't know the world I'm really from, the world I grew up in. I'm a fraud with a foot in each camp and not really at home anywhere. I put up a huge wall, block people out from getting to know me. I ignore their efforts to connect with texts and phone calls despite enjoying their company and thinking it would be good to get to know them. Despite my best intentions my wall works against them and I block them out.⁶¹

We also heard from families in which siblings were sexually abused by the same person, meaning victims were dealing with the trauma of their own abuse as well as the abuse of their sibling. Sisters, 'Laurel' and 'Liana', now aged in their early 60s, told us that they had both been sexually abused by the same Catholic priest as teenagers. Neither knew about the other's abuse until they disclosed to each other when they were in their 50s. The revelation many years later that they had both been sexually assaulted by the same priest came as a terrible shock. 'Laurel' said: 'That just absolutely broke my heart. Because 'Father Holmes' had said to me that if I let him do things to me, that he would never touch my sister'. She felt that her relationship with her sister was hindered because she kept the abuse secret throughout most of their adult life. Having now disclosed, they support each other, and they attended their private session together.

In some cases, siblings were witnesses to the sexual abuse of their brother or sister. For example, 'Calista' told us that sometimes when she was being sexually abused by her foster father, 'Ken', late at night, she would reach out for the only friend she had – her little brother. She said:

He would sit at the door. I would be like that, and 'Ken' would be doing 'Ken's' thing, and we would sometimes just touch fingers. And he'd be hiding behind the door. ... He was tiny. And there was nothing he could do.⁶³

Siblings also can be affected by the way institutions respond to the sexual abuse of their sister or brother. We heard from survivors for whom the process of seeking an apology or redress from an institution, or seeking criminal justice, consumed their whole family, diverting attention from other siblings in need of parental care. Loss of quality family time can be a consequence not only of the effects of sexual abuse on a victim, but also of the various actions that families collectively have to go through to deal with the abuse, ⁶⁴ such as seek redress and criminal justice.

For example, in the *Harmful sexual behaviours in children in schools* case study, EAA, giving evidence about the abuse of his son, CLA, at Trinity Grammar School in the 1990s, told us that the abuse also affected his daughter:

She is 18 months younger than [CLA] and they are very close. She harboured a lot of anger for quite a few years about what happened to [CLA]. We were so preoccupied with what happened to [CLA], the criminal proceedings and the lawsuit, that we did not have as much time for her. She is a wild spirit who was left to her own devices for a few years.⁶⁵

We also heard that siblings of adult survivors seeking redress or criminal justice were the primary source of support to the survivor. Similar to many parents and partners, siblings told us they often found this role affected their health, work and personal lives. 'Phyllida' told us that her younger brother, 'Mark', was physically and sexually abused at the Catholic school he attended in Melbourne. 'Phyllida' has provided the most support for 'Mark', writing numerous letters on his behalf to high-level members of the Church seeking justice for him, accompanying him to medical and legal appointments, and paying for medical, dental, optical and financial services, and for lawyers for his court appearances. When the bank was about to foreclose on 'Mark's' mortgage, 'Phyllida' bought his house and let him continue living there rent free. She pays for all repairs, insurance and rates and has, over time, also provided furniture and clothes. Now nearly 70 years old, 'Phyllida' told us she has been doing this for nearly 40 years, at an estimated cost of over a million dollars. 'Phyllida' told us she has always had the support of her husband and regularly sees a psychologist to help manage the stress, but standing by her brother has taken its toll:

I've lived the life of a secondary victim, suffering in silence, stressed and fatigued, hyper-vigilant, sad and lonely. Who could you talk to about these things? This is critical. You lose friends as they no longer wish to hear or share your concerns. They just do not understand, it's beyond their experience and in some ways it seems exaggerated ... Not only is there a material cost, a financial cost, there's an enormous psychological cost to your life and, like, I'm nowhere near the person I used to be. I'm now anxious all the time and it's changed my life.⁶⁶

5.1.4 Children of survivors

I could go on and on about the intergenerational impact, because in some way we, as children of these victims, actually lived through it. We are systematically victims ourselves. To me, this is disturbing, that our generation was and still is so greatly affected in its own way by something that was so damaging to the generation before our own.⁶⁷

We learned that child sexual abuse has intergenerational effects, affecting children and grandchildren of victims. Most of what we heard about the effects of child sexual abuse on children – in private sessions and public hearings – came from survivors of abuse, speaking as adults about their fears for their own children. However, we also heard from some children of survivors, now adults, about how the impacts of the child sexual abuse affected them and their children. Further, some survivors came to private sessions supported by their adult children.

The intergenerational impact of child abuse is also described by the children of victims of abuse in the Parliament of Australia report, *Forgotten Australians: A report on Australians who experienced institutional or out-of-home care as children*. They spoke of difficulties victims have in showing warmth and affection. They also described dysfunctional family environments characterised by alcoholism and addiction, mental health issues, and unresolved anger.⁶⁹

We also heard in private sessions and public hearings that children have been exposed to the debilitating effects of trauma on their parent's lives, including mental illness, alcohol and drug abuse, family breakdown, violence and suicidal behaviour. As described in Chapter 3, child sexual abuse is associated with a range of mental health impacts on victims, including depression, PTSD and other disorders, substance abuse and poor physical health on a daily basis, all of which can affect the ability of a victim and their family to provide a stable and secure environment for children.

However, we also heard from many survivors for whom caring for their children provides their key source of strength and meaning (see Chapter 2). While dealing with the profound impacts of child sexual abuse in their lives, many survivors have raised their children with love, as well as a determination to protect their own children from experiencing the same harm that they suffered as children.

Mostly we heard from survivors who were concerned that their children were negatively affected by the various impacts of abuse on them. They told us they worry about their children's exposure to their own distressing behavior and mental illness. 'Leland', who told us he was sexually abused by two Catholic priests in the 1960s, said, 'I'm supposed to be someone [my son] looks up to and he looks at me with sorrow'.⁷⁰

Some children, whose parents' mental and physical health had been significantly affected by the abuse, had to take on a caring role for their parent at an early age as a result. A disclosure of abuse can result in a shift in the relationship between these children and their parent. Mr Mark King, giving evidence during the *Church of England Boys' Society* case study, told us how his disclosure of abuse exposed his daughter to his distress and affected the dynamic between his children and himself:

The memories were triggered while I was watching a woman on television tell her story of abuse. I broke down and wailed in front of my teenage daughter, who didn't know what was going on. I then disclosed to my wife what had happened to me. This became a very difficult time for our family. There was a big shift in my relationship with my children because they effectively became parents to me – still are. I feel that I was abused by the Church and that my family – the one I grew up in and the one I made – suffered too because of that abuse.⁷¹

Survivors – both male and female – told us in private sessions they were concerned that their children were affected by their inability to be nurturing, or provide emotional security, due to the impacts of abuse. As part of his private session, 'Cameron John' provided a written statement to the Royal Commission that stated: 'I can see the terrible cost for my children from having a father who was emotionally unavailable and physically absent for much of their childhood ... It hurts me deeply to know that my children still carry my burden, which has become their burden'. The Children of survivors also noted this as an impact. For example, 'Benita', who told us her father 'Max' was abused as a child in a Catholic orphanage in the 1950s, described the effects of the abuse on her relationship with her father. 'The hardest part for me is the lack of emotion he shows ... I have been robbed of having "shows of affection" towards me', she said. The hardest part for me is the said.

'Benita' also thought that the lack of parenting skills exhibited by her father, abused in care, would affect her own children:

How I parent them and how I bring them up and so on and so forth ... It will continue to go on and on, until, you know, perhaps people stop hurting from what happened, if that's even possible. Or until that generation ... pass on, and the people that inflicted that pain pass on.⁷⁴

Some survivors in private sessions and public hearings were anxious that their lack of parenting skills might impact on their children's development. This lack of parenting confidence described to us by survivors in private sessions is consistent with some research conducted on the effects of childhood abuse on the later life course of victims and their relationship with their children. In some cases, child sexual abuse can later significantly affect victims' abilities to parent in line with what society expects of them. For example, during Case Study 7: Child sexual abuse at the Parramatta Training School for Girls and the Institution for Girls in Hay (Parramatta Training School for Girls), Ms Yvonne Kitchener said she did not have a close relationship with some of

her children and believed they hate her. She thought she did not have the necessary parenting skills because of the abuse she experienced, saying 'Because of the generational effect from me, it's gone on to them and they've both got mental health problems'.⁷⁷

Some children, now adults, described their childhood as marked by their survivor parent's violence or substance abuse. 'Becky' and 'Gretel' told us in a private session how their father had been sexually abused by the parish priest as a child, which led to him being extremely distrustful of authority and having trouble at school. As he grew up, he began taking drugs and drinking heavily. His marriage was marred by violence towards his wife and children, and he had trouble expressing his love for them. 'Becky' told us she realises now why this was the case:

A child interrupted by abuse reverberates through generations and I still feel its impact upon myself and our family. I still feel outrage and anger ... We will never know what type of person our father would've been if his childhood had not been so abruptly changed.⁷⁸

As discussed in Section 3.3, child sexual abuse can take an enormous toll on intimate relationships between a victim and their partner, contributing to family breakdown and conflict, with implications for children in the family.

In some cases, the cumulative effects of abuse and social disadvantage have contributed to victims' children being placed in out-of-home care, continuing the pattern of institutionalisation across generations. The *Bringing them home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families (Bringing them home)* report which remains the most comprehensive inquiry into the Stolen Generations, concluded that the many negative consequences of past child removal policies and practices have directly affected subsequent generations and 'increased their likelihood of institutionalisation'.⁷⁹ One Aboriginal survivor of abuse, Ms Robin Kitson, gave evidence during the *Parramatta Training School for Girls* case study that she believed her children had suffered as a result of her experiences of abuse:

I have had to give my children to my mother and father when they were little because the welfare threatened to take them from me, just like they took me. I didn't raise my children up the way I wanted. I never had the chance to explain my story to them. I never cuddled them. I never loved them the way a mother should love her children. This was a consequence of the way I have been hurt and the things that were done to me.⁸⁰

In 1910, when Aboriginal and Torres Strait Islander children were being removed into institutions where they were vulnerable to child sexual abuse, the Commissioner of Police heard how Aboriginal and Torres Strait Islander peoples 'have as much love and affection for their children as the white people have, and they will fight for the sake of their children'.⁸¹ We heard from Stolen Generations survivors, like 'Vernon' who continued this fight to prevent their children from being removed into out-of-home care. 'No white bastard's going to take my kids away from me'.⁸²

Savannah Szoredi: the intergenerational impacts of child sexual abuse

Ms Savannah Szoredi, age 24, gave evidence in the *Nature*, cause and impact of child sexual abuse case study about the intergenerational impacts on her of the child sexual abuse experienced by her mother in out-of-home care. Ms Szoredi told us she admired her mother for the work she did with the Commission of Inquiry into Abuse of Children in Queensland Institutions (The Forde Inquiry) and with the Forgotten Australians parliamentary inquiry and the fight that she had within her that she carried with her until the end. 'She was a survivor and she was proud of that, as am I'. While the impacts of her mother's emotional issues and poor health were significant, Ms Szoredi told us that her mother undoubtedly wanted the best for her and tried her hardest to give her some sense of a normal life:

I was raised by my mother on her own and cannot recall exactly when I found out about the abuse or how I felt. I can say that pretty early on I worked out that things weren't the ideal notion of a normal family home, due to the abuse in question and the absence of family and my father ... The past traumas that would come up for my mother daily often made things difficult for us. I still deal with some of these issues personally but have learned to overcome them. I have psychology reports dating as far back as 1998 that describe me as being, at the time, an extremely emotionally disturbed and anxious young child, who had experienced a lot of death and was living with a single parent who had a hard time distinguishing between parenting and friendship, which in turn means I had to grow up very early. The reports go on to talk about our relationship being very dependent on one another for support and very isolated. They also talk about our relationship being difficult and problematic at times. This was all very true and continued on throughout my life.

My mum sometimes had poor anger management skills and would have irrational outbursts of anger that you could see were coming from a deep place of pain, and at times she did lack the correct skill set to be the best parent she would have liked to have been. Subsequently it was often very volatile and dysfunctional and what I would describe to be as a tumultuous relationship between us both. Although there was a lot of love, it was often hard for us to always express it correctly because of miscommunication, anger and resentment.

The second hardest issue for me growing up, and explains some of the resentment, was witnessing my mother have quite a severe nervous breakdown over a lengthy period of time. I have described this in great detail in my original submission to the Commission. This happened when I was around 12 and I did not speak about it to anybody because I feared, if I did, I would end up a ward of the state myself. As I was well aware of what had happened to my mum in her time in care, this was not an option I wanted to take. I wanted to stay with her and keep her safe, as my loyalty to her ran deep. I believe this nervous breakdown had a lot to do with the traumatic and painful life events that she lived and the experience of it affected me severely. I should also add that it affected my schooling substantially.

Growing up, we also lived quite poorly due to my mum's health problems from her time in institutions. The flat we lived in when I was younger would flood drastically and had concrete floors. Throughout my childhood, I had constant respiratory issues, which I believe were because of the mould and the concrete. I grew up feeling quite isolated at times from other families and the community because of our struggle with poverty, and as I did not see us as a normal family I would often avoid bringing friends over ...

My mother's health was affected tremendously due to her time in state care. The psychological, emotional, physical and sexual trauma she experienced impinged on and deteriorated her health greatly. I blame her time in youth detention and their systems of medicating children for her eventual nervous breakdown and drug habit. Today I feel sad because it could have been so different without all the trauma and my life could have been normal if my mother had been given the opportunity to do it all without the neurological effects of sexual and physical abuse and drugs. Some of us children of these victims don't have relationships with our parents. Some still worry, will today be the day my parent will commit suicide. And some of us have already lost our parents far before their time, like myself.⁸³

5.2 Other affected individuals

5.2.1 Children who witnessed abuse

Studies of child sexual abuse in institutional contexts have reported on the adverse effects on children who live in environments where the abuse occurs, even if they are not abused themselves. In these studies, children in these living situations described being co-opted into maintaining silence, or living in a continual state of fear of becoming the next victim, all of which contributed to secondary trauma. Exposing a child to the abuse of another child, or sexual material involving the abuse of another child, can be abusive in itself.

In private sessions and public hearings, we heard about the effects of witnessing sexual and physical abuse of other children, in particular within closed or 'total' institutions that were isolated and had an atmosphere of generalised fear and ill-treatment. Mr Frank Golding, Vice President of Care Leavers Australasia Network, giving evidence in the *Nature*, cause and impact of child sexual abuse case study, explained how he had frequently been a witness to the sexual abuse of other children when he was a child in institutional care. Witnessing abuse was terrifying, as well as eliciting particular feelings of shame and guilt:

I had a really bad night last night because I was thinking about today, and I had a flashback to an occasion when, two beds down from me, there was a violent rape on a boy. There was no need to conceal it or to make it happen in secret, because who would you talk to? So we were often faced with this feeling of collective shame that this happened ... But there was also a feeling of guilt, in a sense, which is very strange now that I think about it: a feeling that somehow or other you'd escaped; it could have been you.⁸⁵

A number of survivors told Commissioners that, in addition to their own sexual abuse at the institution, they had witnessed the sexual abuse of other children, which compounded their own trauma. Their experiences highlight the damaging impact on children generally of witnessing other people being abused. For instance, 'Aden' told us that he was on an overnight excursion with classmates from his Catholic school in 1984 when one of the Catholic brothers came into the dormitory and sexually abused him. The perpetrator then moved to the boy in the next bed and did the same to him. 'Aden' said the sight of his friend being abused still haunted him:

It was probably the worst thing of what's happened ... It's what I saw, it's not what happened to me. That to me has had more of an impact on me, the visual, I think. When it's up close it's all a bit fuzzy but he was in the bed over here and he was probably one or two metres away from me, and it was clear as [day].⁸⁶

'Sean Michael' described to us in a private session the physical, emotional and sexual abuse he suffered in government-run residential care facilities in New South Wales. During his time in the homes, 'Sean Michael' said he witnessed many other children being abused in different ways. 'Sometimes it's not about yourself either, it's about what you've seen happen to other people as well'.⁸⁷

Survivors told us of their attempts to try to save other children being sexually abused. One survivor, 'Les', told us of repeated abuse by a religious brother at a Catholic boys' home in the 1970s. 'Les' told us he was viciously punished by the Brother for resisting the abuse. 'Les' also tried to save the other boys by yelling every time he saw the Brother coming into the dorm at night. This worked until the Brother started waiting for him to fall asleep before entering. The Brother targeted one boy, 'Eric', in particular. 'Les' believes that 'Eric' was raped by the brother dozens of times. 'Les' later discovered that 'Eric' had taken his own life, which 'Les' says he can understand: since escaping from the home in his mid-teens, 'Les' has also often thought of killing himself.⁸⁸

Survivors also told us of ongoing feelings of guilt and remorse because they had not been able to prevent the sexual abuse of other children, or felt they had been complicit in some way. In *Case Study 21: The response of Satyananda Yoga Ashram at Mangrove Mountain to allegations of child sexual abuse by the ashram's former spiritual leader in the 1970s and 1980s*, APK spoke of the impact of witnessing other children being abused, including one boy who she believed was physically abused because of her close relationship with him. ⁸⁹ 'The trauma of seeing the people I cared about being beaten and humiliated has been, for me, the worst part of my experience'. ⁹⁰

In Case Study 40: The response of the Australian Defence Force to allegations of child sexual abuse, we heard that victims were forced to witness the sexual and physical abuse of other junior recruits, and that witnessing these incidents adversely affected them. ⁹¹ Mr Glen Greaves told us of the effects of witnessing a fellow recruit being physically abused by junior recruits who were in their final three months of training and referred to as 'top shits'. ⁹²

I witnessed the bashing of Shane Connolly ... Connolly was bashed by four top shits. He ended up with a black eye, a split lip, and had blood pouring out of his nose. I feel terrible because I didn't step in to help him. I was too scared. We all were.⁹³

5.2.2 Staff in the institution

We heard that staff at an institution – including teachers, carers, and people in religious ministry – can be profoundly disturbed when they suspect or discover that child sexual abuse has occurred in the institution in which they work and, in some cases, to which they have dedicated their lives. For many, their work caring for children is a vocation to which they are strongly committed. They can feel betrayed and distressed when they discover that a colleague has been sexually abusing children in the institution. We also heard, in private sessions and public hearings, of cases in which the perpetrator's offending behaviour was not addressed, and the sexual abuse continued, which led staff to be disgusted and disillusioned with the institution as a whole. In *Case Study 2: YMCA NSW's response to the conduct of Jonathan Lord*, Ms Danielle Lockwell, employed at the YMCA, told us that she saw her role 'first and foremost as protecting children', and that Jonathan Lord was doing the 'very opposite'.⁹⁴ She felt betrayed by Lord and said in evidence, 'He made me believe that we had this amazing centre where the children were safe and happy, and he came across to parents as a lovely guy. So I just felt betrayed that he could do the things he did'.⁹⁵

5.2.3 Whistleblowers

Whistleblowers – usually staff members of an institution where children are sexually abused – can be psychologically affected by the sexual abuse, and in particular by an inadequate response from the institution. They are susceptible to being victimised, ostracised or penalised by an institution that tries to avoid or deny that child sexual abuse has occurred. We heard from or about people who were threatened with dismissal or labelled as troublemakers for identifying and trying to prevent child sexual abuse.

One of the impacts commonly described in private sessions and public hearings by whistleblowers was a loss of employment or involvement in the institution. For example, the Royal Commission heard evidence from WG, WF and WH in our *Perth independent school* case study about the circumstances in which they raised concerns about the perpetrator and the difficulties they experienced in doing so. Each teacher was concerned that they would be subjected to rejection, ostracism or bullying/harassment from some staff if they were identified as whistleblowers or complainants. ⁹⁶ For example, when WH was asked in evidence whether she thought that writing a confidential letter regarding her concerns about the perpetrator's conduct towards a student would compromise her position at the school, she said:

Being a reasonably recent staff member into the school, I had already gained my permanency after two years ... so I didn't feel threatened for my job in that way, but I did feel that it may have some repercussions on my standing with the community, as YJ was very popular with the community, and if it was found to be gossipy or 'tell-taley', or whatever, then that could ruin my reputation for the years that I had intended to spend at the school.⁹⁷

Ms Ann Ryan, former teacher at St Colman's School in Mortlake, Victoria, giving evidence in *Case Study 28: Catholic Church authorities in Ballarat (Catholic Church authorities in Ballarat)*, described how she was threatened with dismissal because she tried to uncover the sexual abuse perpetrated by the priest, Gerald Ridsdale, in the early 1980s. As a result of the poor response to the abuse, she lost her trust in the Catholic Church and in general:

I no longer practise the Catholic faith or participate in any church-related activities. This is because of my loss of trust in church authorities and what I perceive to be their double standards. Finding out about child sexual abuse in the Catholic Church, and the response of the Church, has affected my ability to trust in general, and especially trust authority and authority figures. I am much more cynical than I used to be and I am regularly concerned when I see adult men with younger children.⁹⁸

Some whistleblowers described being abused and ostracised by other people in the community when the sexual abuse was reported. For instance, 'Karla', a teacher's aide at a local school, told us she was unable to go out shopping at one stage, because she was abused by locals who blamed her for the demise of the school's fortunes.⁹⁹

5.2.4 Family members of people who have sexually abused children

The families of people who have sexually abused children can also be affected by the disclosure of child sexual abuse, especially if they have an ongoing relationship with the institution where the abuse took place. For instance, 'Francie', the partner of a perpetrator of child sexual abuse within a religious institution, told us about the impacts on the whole family when they learned that her husband – the father of her children – had sexually abused children. Not only did she and some of her children suffer from depression and anxiety, but she says they were also shunned by the religious community. 'Francie' told us that she was forced out of a job she loved, which was connected to the institution. We also heard about the shock and devastation experienced by family members of alleged perpetrators when the alleged perpetrator is named in the media, especially when it is the first time that the family has heard the allegations.

We are unaware of any research to date on the extent of the impacts on families of perpetrators of child sexual abuse in institutional settings. However, a qualitative study on the experiences of, and impact on, the non-offending partners of people involved with child exploitation material suggested that they can experience significant trauma, ostracism and isolation, with a distinct lack of support. Women interviewed in the study demonstrated distress and turmoil arising from their partner's conduct. Most participants identified that they experienced mental health issues, many consistent with PTSD. Some of the participants described losing contact with family and not being able to make friends for fear of further alienation if the issue was discussed. They felt judged and tainted by association rather than being understood as a collateral victim to their partner's criminal conduct. 102

Qualitative research found that common reactions from parents who discover their child's harmful sexual behaviour include shock, confusion, disbelief and minimisation – responses which serve as a defence mechanism to protect parents from the negative personal implications of total acceptance of the child's behaviour.¹⁰³ Finding out about their child's behaviour can be a profoundly difficult and isolating experience for parents that can prompt secondary post-traumatic responses.¹⁰⁴

5.3 Communities

Child sexual abuse doesn't just tear individuals and families apart. In my experience, its claws reach into the community as well, whether they know it or not. 105

Victims of child sexual abuse belong to many different types of geographical, cultural or spiritual communities. These communities can also include schools, sporting clubs and even groups of victims from the same institution. People from these communities understand and respond to child sexual abuse in different ways, and some can be severely affected.

Some trauma theorists have recognised the impacts of serious and terrible events such as child sexual abuse on community well-being, using the term 'cultural trauma' to describe the effects on group identity, social cohesion, and group safety. Previous government inquiries have also identified that child sexual abuse in institutional contexts can have significant community impacts. The report: Betrayal of trust: Inquiry into the handling of child abuse by religious and other non-government organisations by the Victorian Parliament noted the centrality of many of the religious institutions to the fabric of communities, both historically and regionally. As a consequence of this centrality, disclosures of child sexual abuse have led to deep divisions in communities. Research notes that community-level impacts remain a relatively unexplored consequence of institutional child sexual abuse, particularly in regional, remote and rural communities where power dynamics associated with institutions may be more pronounced. One of the sexual abuse have be more pronounced.

Information from private sessions, public hearings and research suggests that community connectedness can be shattered by the revelation of child sexual abuse, especially when the perpetrator is well liked or the institution is respected or trusted. We heard that the breakdown of community cohesion can be intensified if large-scale sexual abuse is revealed, or if attempts by the institution to conceal the abuse are discovered. As we heard from some survivors, this can create deep division and mistrust, and sometimes fragment the community.

Yet we also note that many communities have demonstrated an incredible capacity to come together in the face of child sexual abuse and institutional betrayal, to generate support and foster healing. We have seen many examples of communities demonstrating their support of, and solidarity with, victims of child sexual abuse.

5.3.1 Aboriginal and Torres Strait Islander communities

Aboriginal and Torres Strait Islander communities represent the oldest continuing cultures in history. However, like other indigenous peoples in colonised nations, Aboriginal and Torres Strait Islander peoples have experienced collective trauma and grief on a magnitude that would compromise the resources of any community to manage the impacts. Volume 1, Our inquiry and Volume 2, Nature and cause discuss how for Aboriginal and Torres Strait Islander victims child sexual abuse in institutional contexts has been part of the collective and intergenerational trauma of colonisation.

In Aboriginal and Torres Strait Islander community consultations, survivors and secondary victims told us the experience of institutionalisation and dehumanisation, including widespread sexual abuse of multiple generations of Aboriginal and Torres Strait Islander children in those institutions, has impacts that ripple through communities and are transmitted across generations. ¹¹² In the *Nature, cause and impact of child sexual abuse* case study, expert witnesses stressed that this experience has affected the wellbeing and capacity of Aboriginal and Torres Strait Islander communities. ¹¹³ Survivors and expert witnesses spoke about the destruction of kinship structures, family and community roles, the breakdown of collective child-rearing practices, and loss of language, cultural values and norms, leaving communities grieving and overwhelmed. ¹¹⁴

Research on collective trauma suggests that when a whole group experiences trauma, such as is the case for Aboriginal and Torres Strait Islander peoples, this trauma can overwhelm the resources that community would ordinarily used to heal or come terms with traumatic events. In the *Nature, cause and impact of child sexual abuse* case study, Ms Karen Menzies, Lecturer in Indigenous Health, University of Newcastle, illustrated the loss of any protective buffer in Aboriginal and Torres Strait Islander communities as a result of collective trauma:

[Among] non-Aboriginal people, there might be 12, 15 people in the extended family and there might be one person who has experienced child sexual assault, another person who has experienced domestic violence or another person who has had a horrific car accident. But by and large the extended family is a healthy structure to support those individuals. Unfortunately what we see in many Aboriginal families and in many Aboriginal communities, that collective trauma impacts on every individual – because there's varying degrees – some where people are quite seriously unwell through mental illness, but some are high-functioning also.¹¹⁶

In some Aboriginal communities all children were removed in a sudden event that led to mass trauma. For example, on a single day in 1955, all the children of Moola Bulla in the Kimberley were taken away on a truck. Research published by the Royal Commission suggests that authorities compounded the trauma of child removals by failing to protect large numbers of children from sexual abuse in the institutions they were taken to, and by intentionally preventing the transmission of language and culture, which further disrupted the social fabric of the community. The community impacts related to culture are also discussed in Chapter 3.

Community as a source of resilience for Aboriginal and Torres Strait Islander survivors is discussed in Chapter 2.

In community consultations, survivors said the removal from Country and culture not only made the institutional sexual abuse possible, it also denied them access to the material resources and strong networks of family support necessary to provide for their own children. In these ways, imposed poverty, ill health and disadvantage can become entrenched. They described this as a 'cycle' of continuing harm from the original abuse, which runs across generations. 119

Aunty Lorraine Peeters, Ms Shaan Hamann and Ms Kerrie Kelly, writing about the Marumali Journey of Healing program for Aboriginal survivors, provide a helpful explanation of the intergenerational cycle of cumulative harm:

- The primary burden of trauma has been borne by those who directly experienced forcible removal during the years from 1910 to 1972 (first generation).
- The secondary burden of trauma lies with those other than the individuals forcibly removed, such as their families including their children (second generation) and grandchildren (third generation) and communities.
- The future burden is the ongoing legacy of not adequately addressing the burden
 of trauma in the population of people who directly experienced it, and the
 transgenerational transmission of social, emotional and spiritual wellbeing problems as
 a result of connections that were severed or attenuated by past government policies.¹²⁰

The *Bringing them home* report states: 'not one Indigenous family ... escaped the effects of forcible removal ... Most families have been affected in one or more generations'. ¹²¹ According to the report, 'Children in every placement were vulnerable to sexual abuse and exploitation'. ¹²² Although witnesses were not asked about sexual abuse, almost one in ten men and more than one in ten women who gave evidence to that inquiry said they had experienced sexual abuse in a 'children's institution'. ¹²³ These figures could substantially underestimate the numbers of children sexually abused in institutional settings (see Volume 4, *Identifying and disclosing child sexual abuse*). Aboriginal and Torres Strait Islander children were frequently sent from children's institutions into other institutional settings as domestic servants in private homes or as agricultural labourers on pastoral stations. Elsewhere in the *Bringing them home* report, it states that 7.7 per cent of boys and 17.0 per cent of girls reported sexual assaults. ¹²⁴

Thus, in many Aboriginal and Torres Strait Islander communities, a large number of adult survivors are dealing with the impacts of sexual abuse in institutions and an even greater proportion are living with trauma from broader experiences of loss and grief. Professor Muriel Bamblett, Chief Executive Officer, Victorian Aboriginal Child Care Agency, in the *Nature*, cause and impact of child sexual abuse case study, pointed out how this intergenerational trauma continues to affect the lives of Aboriginal and Torres Strait Islander children today:

In 2009 the State of Victoria's Children reported huge amounts of grief and loss and trauma that Aboriginal children are exposed to on a daily basis, the fact that they have to go to numerous funerals that they're involved in, you know, the deaths, suicides and criminalisation, and youth suicides, so a lot of those things impact on the children that we're working with, particularly in child welfare.¹²⁵

Research and past inquiries have consistently pointed to a clear link between the impacts of past legislation, policies and practices, including widespread sexual abuse, ongoing experiences of dispossession and marginalisation, and the full range of social and economic problems that face Aboriginal and Torres Strait Islander communities today. This has a cumulative impact for communities and families in a cycle of re-traumatisation, and implications for the vulnerability of future generations to institutionalisation. Volume 9, Advocacy, support and therapeutic treatment services considers the implications of these community-level impacts on the therapeutic service needs of Aboriginal and Torres Strait Islander survivors.

5.3.2 Religious communities

When incidents of child sexual abuse occur within religious institutions, there can be negative outcomes for the cohesion and social connectedness of religious communities. For example, in private sessions, community forums and meetings in Ballarat, we heard how revelations of child sexual abuse by Catholic clergy have affected the community. The effects were compounded by the ensuing suicides of some victims and inadequate responses from the Catholic Church. The impacts in Ballarat are also described in the Victorian *Betrayal of trust* report. The impacts in Ballarat are also described in the Victorian *Betrayal of trust* report. The impacts in Ballarat case study, told us:

Ballarat is a very Catholic town and the Catholic community is very closed. The Catholic culture is very strong. Coming forward and talking publicly about child sex abuse in Catholic institutions not only has repercussions at the family level, but also at the business and social level in Ballarat. It is these impacts that stop other victims from coming forward. Some of the little towns outside Ballarat are also extremely Catholic. Sometimes the only institutions in these towns are a Catholic Church and a Catholic school. I know of survivors in these towns that have spoken out about child sexual abuse. They have told me that after speaking out they were stood down from clubs where they were lifelong members. It is like they have literally been wiped out of these communities.¹³⁰

Yet Ballarat has also demonstrated how religious communities can come together to support the victims and survivors of child sexual abuse. One highly visible manifestation of this is the 'Loud Fence' movement, where parishioners and community members tied ribbons to the fences of institutions where child sexual abuse had been perpetrated, to demonstrate their solidarity and support.

The sheer number of suicides and suicide attempts by a number of victims in one community can have an overwhelming effect. In the *Catholic Church authorities in Maitland – Newcastle* case study, we heard of multiple perpetrators within the Catholic clergy who sexually abused many children over a period of time in the Maitland – Newcastle region. As discussed in Chapter 3, we heard from people who believed that the sexual abuse was linked to a number of suicides by boys and men who had attended Marist Brothers Hamilton. CQT, who gave evidence of his own sexual abuse at the hands of Marist Brothers, and the death by suicide of his younger brother Andrew, said that the number of suicides has had a major impact on the community.¹³¹

Mr Gerard McDonald gave evidence about his sexual abuse by a priest in the *Catholic Church* authorities in Maitland – Newcastle case study. He spoke about how the Newcastle Catholic community was divided after the abuse came to light:

The sexual abuse by [the priest] and all of the other Catholic clergy in Newcastle has split the community. There are those who believe the abuse and support the survivors. There are also those who think we are making it up.¹³²

BAP, in the *Catholic Church authorities in Ballarat* case study, reflected on similar impacts of institutional child sexual abuse on the community. He said:

I think institutional child sexual abuse has split the Ballarat community into factions. Some people are starting to talk to survivors about it and are starting to understand the impact. There are others in the community that block it out completely. It is like it is not even in their vocabulary and they cannot understand.¹³³

In some religious communities, it may be frowned upon to seek support from mainstream practitioners and institutions that do not share the same religious beliefs or values. 134 Victims and their families might feel that their reputation, or the reputation of their community, would be ruined if the sexual abuse was known, or even that disclosure would be contrary to their beliefs. Because of the imperative to keep the story of the abuse within the community, disclosure by an individual or their family could be perceived as 'breaking ranks', or betrayal of the community. This can lead to division and disharmony. 135

The impacts on religious communities are discussed further in Volume 16, Religious institutions.

5.3.3 Culturally and linguistically diverse communities

We heard that child sexual abuse can have specific ripple effects in culturally and linguistically diverse communities. In public consultations with multicultural stakeholders, we heard how some members of multicultural communities can react to child abuse within their community by refusing to accept that abuse could occur or encouraging victims not to talk openly about their experience of abuse. We heard that these reactions from community members, particularly community leaders, can be for a number of reasons, such as:

- fears the community will experience marginalisation, racism or excessive scrutiny if abuse is made public, 137 including fears that outsiders (particularly the media) will incorrectly perceive the abuse to be culturally condoned or widespread within the community. 138
- false perceptions that strong moral or religious codes adhered to by the whole community mean child abuse is unlikely to occur within the community. 139
- expectations of a negative or unhelpful response from government authorities, including removal from Australia,¹⁴⁰ which can be particularly strong among refugee communities, especially those who have spent time in held immigration detention.¹⁴¹
- pressures to preserve community harmony at all costs, particularly in communities
 with strong collectivist values where whole-of-community interests can be perceived
 as more important to protect than individual interests.¹⁴²

We were told that these reactions can be polarising for multicultural communities and cause internal division, particularly following publicised prosecution or sentencing decisions. We also heard that these reactions can mean some multicultural communities become closed or secretive about abuse over time, harmfully reinforcing beliefs that child abuse does not occur within the community. During the *Nature, cause and impact of child sexual abuse* case study, we heard how gatekeepers in some multicultural communities can actively lead this process by 'taking it upon themselves to allow information or not allow information within the community'. 145

5.3.4 Other communities

We heard that child sexual abuse, and a poor institutional response to the abuse, can have a destructive effect on other communities centred around the institution where the sexual abuse occurred. These communities are made up of parents, children, staff, volunteers and others with shared interests. They can be centred on schools, sport and recreation clubs and associations, and arts and cultural institutions.

For many, the sporting institutions that children attend play a major role in family life. When child sexual abuse in these institutions is disclosed, the families of victims can be ostracised. Some might leave the institution and community, which can affect other families' trust in the institution and the cohesion of the community as a whole.

In recreational institutions, such as Scouts, community and hobby groups, or dance studios, parents and other family members can also be heavily involved. We were told in private sessions and public hearings of parents and staff establishing strong social networks and a shared commitment to the institution. In some cases these bonds better enabled perpetrators to use grooming to disguise child sexual abuse. In some cases the fracturing of social networks following the discovery of child sexual abuse caused distress to members of the community, including staff, parents and children, and particularly those who were highly invested in the institution. In the instance of RG Dance studio, examined in *Case Study 37: The response of the Australian Institute of Music and RG Dance to allegations of child sexual abuse*, the institution closed after the child sexual abuse was discovered. Ms Jennifer Davies, a former teacher at RG Dance, and the sister-in-law of the perpetrator, gave evidence during the public hearing:

I really thought that RG Dance was a wonderful place, that there were so many benefits for the children being part of this community of friends and dancers. And I find it so horrible that so many of the children didn't have that experience, that they – I find it so horrible and sad that behind all of that were things that I didn't see. 147

Volume 14, *Sport, recreation, arts, culture, community and hobby groups* discusses the community impacts of child sexual abuse in sport and recreation institutions in further detail.

Schools and day care centres can also operate as communities, bringing together children, families, teachers and others. They can reflect an investment of time and energy by these community members and bonds formed between community members over many years.

In Case Study 32: The response of Geelong Grammar School to allegations of child sexual abuse of former students, we heard in evidence from Mrs Catherine Parsons, matron at the school, of the impact of the sexual abuse perpetrated by Philippe Trutmann on the school:

There has been a lot of sadness in the Geelong Grammar School community relating to the actions of Philippe Trutmann. There is also sadness and disbelief regarding the responses from those in charge. There are many honourable and good people who have dedicated many years of their lives to Geelong Grammar School which is essentially a most positive and nurturing community and one which has already introduced many changes to try and ensure this never happens again.¹⁴⁸

As outlined in Chapter 4, if an institution fails to address child sexual abuse, the perpetrator or child with harmful sexual behaviours may continue to sexually abuse children. We heard in private sessions and public hearings that parents and carers felt betrayed and deeply disillusioned when the institution failed to deal appropriately with the abuse, or failed to adequately inform the community about their responses.

5.4 Australian society

I have at times felt stigmatised and vilified by the government of my country as I, and people like me, on disability support pensions, have been described essentially as a huge drain on the country, 'leaners, not lifters' ... I'm well aware of the huge financial burden I am ... There is a huge human capital cost to the nation with regard to all who are sexually abused in childhood ... But I also think I am what happens when as a society we turn a blind eye to what is happening to the most vulnerable in our communities. This is the price paid when we don't treasure and protect our infants and children. 149

Child sexual abuse in institutions is likely to have affected at least tens of thousands, if not hundreds of thousands, of children over the past decades. The number is even greater for those who were sexually abused as children outside of institutional contexts. The extent of child sexual abuse in institutional contexts in Australia is discussed further in Volume 2, *Nature and cause*.

As these victims develop into adults who experience, in many cases, debilitating symptoms of trauma, the effects multiply and reach outwards to families, friends, institutions and communities. For many people, mental health impacts can be severe, and include PTSD, suicidal behaviour and drug and alcohol abuse. These needs are likely to require significant intervention in order to be addressed.

Through our private sessions, public hearings and research it has been made clear that child sexual abuse has occurred in a broad range of institutional contexts, right across Australia and spanning many decades. It is likely that the ripple effects of child sexual abuse in institutions have not been limited to particular communities, but have directly and indirectly affected the lives of countless people in Australia today.

Moreover, child sexual abuse in institutions, and child sexual abuse more broadly, has intergenerational effects for the future health and wellbeing of the Australian community. We heard numerous cases where the children of victims were affected by the sexual abuse experienced by their parents. ¹⁵⁰ In some cases, the effects can span generations, perpetuating cycles of disadvantage and trauma. We heard from victims who experienced sexual abuse many decades ago, but who are only now realising how it has affected their own children. We also heard from those who have experienced sexual abuse more recently, who need help to recover and heal. The impacts of child sexual abuse must be urgently addressed to prevent this debilitating cycle.

5.4.1 Social impacts

We have discussed in previous chapters the impacts that child sexual abuse has on victims, and on their families, communities and institutions. Although difficult to quantify, child sexual abuse in institutional contexts also has broader, pervasive social impacts. Media coverage of child sexual abuse in institutions and the work of the Royal Commission has furthered the public's understanding of the impacts of institutional child sexual abuse.

Child sexual abuse can affect our social trust, in particular our trust in institutions and adults who care for children. As we become more aware of the many instances of child sexual abuse in institutions, and the extent of institutional failings, the level of trust we invest in institutions that care for children is diminished. A lack of trust in religious institutions, in particular, which provide a wide range of services for children and other vulnerable populations, has implications for service delivery and for membership of religious organisations.

Our knowledge about the nature and extent of child sexual abuse can also influence society's perception of men. In private sessions 95.2 per cent of survivors who were abused by an adult told us about sexual abuse by a male (see Volume 2, *Nature and cause*). This is consistent with research that suggests that the majority of perpetrators of child sexual abuse are male, regardless of the setting in which abuse occurs.¹⁵¹ What we hear about male perpetrators can influence how society sees men, and how society perceives their ability to be trusted as carers of children. However, as discussed in Volume 2, gender is not predictive of whether or not a person will become a perpetrator of child sexual abuse. The overwhelming majority of men are not perpetrators of child sexual abuse.

The impact of child sexual abuse on victims can also affect their contribution to their family and community later in life. Their behaviours have wider implications for both society and the economy, by impacting on each individual's ability to participate in family and community life and by placing significant demands on the national health system.

Further, many survivors spoke to us of what happened in their lives after being sexually abused as children – they spoke of their anger and violence, the crimes they have committed and the harm they have caused others. One Australian study found that 24 per cent of child sexual abuse victims had a recorded offence against them.¹⁵² This increased likelihood of criminal behaviour can be costly for the individual, and represents an additional cost for members of the community who become victims of crime by survivors of child sexual abuse.

Another broader social impact extends to mental health professionals who work with sexual abuse victims. They are at risk of sustaining vicarious trauma,¹⁵³ which can be triggered by working with traumatised clients and traumatic subject matter.¹⁵⁴ Other professionals exposed to sexual abuse victims or traumatic material may also be affected. One study has found that lawyers who work on matters concerning child sexual abuse demonstrate high levels of secondary trauma and burnout.¹⁵⁵ In another study, law enforcement professionals

working on child sexual abuse cases reported high levels of psychological distress, trauma and PTSD symptoms.¹⁵⁶ Researchers working in the field of sexual assault can also be negatively affected.¹⁵⁷ It is important to recognise the potential harm of other forms of work involving the trauma of child sexual abuse to appreciate the extent of the ripple effects of this abuse.

While it is difficult to measure the myriad ways that Australian society has been affected by child sexual abuse, we know it is likely to have had a broad and pervasive impact.

5.4.2 Economic impacts

What is the economic factor? Which sounds terrible, but the economic and intellectual loss of this kind of stuff to society ... The people, the shining people we could have ... My main reason is to say these things, about the killing of souls, and the killing of potential.¹⁵⁸

Survivors carry the heaviest burden of child sexual abuse, both economic and otherwise. As discussed in Chapter 3, survivors can experience a range of impacts with long-term economic consequences, including poor educational, health and employment outcomes. Survivors who have difficulty keeping a job may also have difficulty securing a reliable and adequate income, accumulating assets, finding a safe place to live, maintaining relationships, and meeting their costs of living. Not all survivors will experience these economic problems, but for those who do, economic insecurity can have broader impacts on survivors' families and the entire economy. The cost of child sexual abuse to the Australian economy is difficult to quantify, but it is likely to be in the billions of dollars each year.

Survivors' families may also experience negative economic impacts as a result of child sexual abuse. For example, if survivors remain jobless for a long time, their children could experience poorer educational and employment outcomes than other children. Survivors and their families who are economically disadvantaged might also live in areas without high-quality housing, transport, education, health and other forms of infrastructure. These circumstances can make it even harder for survivors and their families to escape the cycle of poverty. As discussed in Section 5.1.4, some children of survivors told us they experienced poverty growing up because their parent was affected by child sexual abuse.

When survivors and their families live more economically-disadvantaged lives than they otherwise would have, all of society may be affected. There will be fewer people in jobs and with less income to spend or invest than there would have been. The economy is smaller than it could have been and there are fewer economic opportunities for everyone.

Society also faces the additional costs of resolving some of the economic consequences stemming from child sexual abuse. Survivors and their families who experience economic disadvantage may need additional support from government in the form of transfer payments (such as unemployment benefits, parenting payments or the disability pension), and greater

access to government-funded services such as the justice system,¹⁶⁵ homeless shelters, and health and counselling services.¹⁶⁶ Governments always need to provide and fund these services, but some of these costs could be avoided by preventing child sexual abuse and addressing short and long term impacts.

Some international studies have attempted to quantify the economic costs to society of child sexual abuse. A United Kingdom study examined the increased likelihood of experiencing health problems, coming into contact with the criminal justice and child protection systems, and being unemployed, as a result of child sexual abuse. Using these increased likelihoods, the study estimated that child sexual abuse cost the United Kingdom's economy £3.2 billion in 2012–13. The majority of associated costs related to the increased risk of unemployment and survivors being less likely to achieve their earning potential.

Another study assessed the economic costs of child sexual abuse to the Asia-Pacific region. The study estimated the costs to society associated with survivor's physical and mental health problems, due to their experience of child sexual abuse. The cost was estimated to be between US\$31 billion and US\$43 billion in 2004. 168

These studies illustrate the high economic costs borne by society. However, they may not be directly applicable to the Australian economy as they consider much larger economies and, in some cases, different socio-economic environments.

Although the cost of child sexual abuse is unknown in Australia, there has been some research into the cost of child maltreatment more broadly. A recent Australian study by Pegasus Economics found that the combined costs of child sexual, physical and emotional abuse to the Australian Government and state and territory governments is \$6.8 billion per year. The study estimated the costs to government budgets from alcohol abuse, obesity, anxiety and depression, and suicide or attempted suicide related to the experience of childhood trauma. Given that the categories included in the study represent only some of the potential impacts, these estimates are likely to underplay the overall cost to the economy.

Another Australian study estimated that the cost of child abuse (including all types of abuse) and neglect in 2007 for all people ever abused was between \$3.5 billion and \$5.5 billion. The study indicates that the largest costs to society relate to government expenditure on care and protection, and crime.¹⁷²

As these studies suggest, the economic impacts of child sexual abuse should not be underestimated.

Most importantly, in light of all of the impacts of child sexual abuse described throughout this volume, it is clear that governments must do everything in their power to prevent child sexual abuse in our community. The impacts go beyond individual victims and survivors — they are a significant social and economic cost for Australian society, both now and in the future.

Endnotes

- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 65.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 65.
- A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 43.
- 4 A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 54–5.
- 5 Name changed, private session, 'Eoin'.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 65–6.
- A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 36.
- 8 Transcript of G Gee, Case Study 57, 31 March 2017 at 27814:17–18.
- 9 A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 58.
- A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 11.
- A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 44.
- A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 45–6.
- A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 50.
- A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 50.
- Transcript of WQ, Case Study 12, 19 May 2014 at WA2357:35–42; Exhibit 12-0004 'Statement of WQ', Case Study 12, STAT.0257.001.0001_M_R at 0008_M_R.
- Transcript of WQ, Case Study 12, 19 May 2014 at WA2366:47–WA2367:1–25; Exhibit 12-0004 'Statement of WQ', Case Study 12, STAT.0257.001.0001 M R at 0018 M R.
- A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 37–9, 48.
- For example: Name changed, private session, 'Rudy'; Private session attendee, 2014.
- A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 59.
- Name changed, private session, 'Francene'.
- 21 Exhibit 57-0001 'Statement of Susan Campbell', Case Study 57, STAT.1332.001.0005, pp 4–5.
- 22 Transcript of EAL, Case Study 45, 2 November 2016 at 22656:8–11; Exhibit 45-0046 'Statement of EAL', Case Study 45, STAT.1206.001.0001 R at 0010 R.
- Name changed, private session, 'Helen Christine'.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 65.
- 25 Transcript of M Watson, Case Study 57, 30 March 2017 at 27721:40–44.
- Name changed, private session, 'Deidre Catherine'.
- 27 Transcript of EAL, Case Study 45, 2 November 2016 at 22657:5–8; Exhibit 45-0046 'Statement of EAL', Case Study 45, STAT.1206.001.0001_R at 0010_R.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 16: The Melbourne Response*, Sydney, 2015, pp 5, 13–16.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Vale Anthony Foster*, media release, Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 27 May 2017.
- 30 Exhibit 57-0001 'Statement of Marija Radovejic', Case Study 57, STAT.1340.001.0001 at 0004.
- 31 Exhibit 57-0001 'Statement of Marija Radovejic', Case Study 57, STAT.1340.001.0001 at 0004.
- 32 Transcript of ATU, Case Study 23, 24 February 2015 at 11837:8–10; Exhibit 23-0012, 'Statement of ATU', Case Study 23, STAT.0510.001.0001_R at 0011_R.
- A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 59–60.
- A Quadara, M Stathopoulos & R Carson, Family relationships and the disclosure of institutional child sexual abuse, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 60.
- 35 Transcript of BQR, Case Study 34, 3 November 2015 at C12050:3–14.
- Names changed, private session, 'Vicki' and 'Page'.

- 37 Transcript of BYD, Case Study 36, 5 February 2016 at C15822:31–32; Exhibit 36-0061, 'Statement of BYD, Case Study 36, STAT.0810.001.0001 R at 0012 R.
- For example: Exhibit 40-0026, 'Statement of Susan Campbell', Case Study 40, STAT.0999.001.0001 at 0008; Transcript of A P Nash, Case Study 43, 6 September 2016 at C18040:7–8.
- A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 59–60.
- Transcript of AP Nash, Case Study 43, 6 September 2016 at C18044:23–41.
- Transcript of Z Waks, Case Study 22, 3 February 2015 at C6070:40–47.
- 42 Transcript of M Waks, Case Study 22, 3 February 2015 at C6039:11–19.
- Transcript of Z Waks, Case Study 22, 3 February 2015 at C6109:36–47 and C6110:1–2; Exhibit 22-005, 'Statement of Zephaniah Waks, Case Study 22, STAT.0469.001.0001 R at 0007 R.
- 44 A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 53–4.
- A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 54.
- 46 A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 54.
- 47 Research has found that intimate partners of victim/survivors may experience secondary trauma as a result of the sexual assault of their partner. See T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 65.
- 48 Name changed, private session, 'Joanne'.
- 49 Name changed, private session, 'Millie'.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 56.
- Name changed, private session, 'Leland'.
- 52 Written account, 2014.
- Name changed, private session, 'Edmond'.
- A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 51.
- A Quadara, M Stathopoulos & R Carson, *Family relationships and the disclosure of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 51.
- A Schreier, JK Pogue & DJ Hansen, 'Impact of child sexual abuse on non-abused siblings: A review with implications for research and practice', *Aggression and Violent Behaviour*, vol 34, 2017; A Schreier, JK Pogue & DJ Hansen, 'Impact of child sexual abuse on non-abused siblings: A review with implications for research and practice', *Aggression and Violent Behaviour*, vol 34, 2017.
- 57 Transcript of ATU, Case Study 23, 24 February 2015 at 11836:17–22; Exhibit 23-0012, 'Statement of ATU', Case Study 23, STAT.0510.001.0001 R at 0010 R.
- Transcript of CNE, Case Study 43, 1 September 2016 at C11709:31–38; Exhibit 43-0006, 'Statement of CNE', Case Study 43, STAT.1167.001.0001 R at 0010 R.
- Name changed, private session, 'Doris'.
- Name changed, private session, 'Chantelle Elise'.
- Name changed, private session, 'Chantelle Elise'.
- Names changed, private session, 'Laurel' and 'Liana'.
- Name changed, private session, 'Calista'.
- Transcript of D Higgins, Case Study 57, 31 March 2017 at 27820:4–8.
- Transcript of EAA, Case Study 45, 20 October 2016 at 21685:12–19; at Exhibit 45-003, 'Statement of EAA', Case Study 45, STAT.1218.001.0001 R at 0010 R.
- Name changed, private session, 'Phyllida'.
- 67 Transcript of S Szoredi, Case Study 57, 31 March 2017 at 27808:1–7.
- Name changed, private session, 'Ebony'.
- 69 Community Affairs References Committee, Forgotten Australians: A report on Australians who experienced institutional or out-of-home care as children, Parliament House, Canberra, 2004, pp 149–53.
- 70 Name changed, private session, 'Leland'.
- 71 Transcript of D M King, Case Study 36, 2 February 2016 at C15483:7–18; Exhibit 36-0043, 'Statement of Mark King', Case Study 36, STAT.0813.001.0001 at 0003.
- Name changed, private session, 'Cameron John'.
- 73 Name changed, private session, 'Benita'.
- Name changed, private session, 'Benita'.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 58, 65. See also R Roberts, T O'Connor, J Dunn, J Golding & TAS Team, 'The effects of child sexual abuse in later family life: Mental health, parenting and adjustment of offspring', *Child Abuse & Neglect*, vol 28, no 5, 2004.

- C Tarczon, 'Mothers with a history of childhood sexual abuse: Key issues for child protection practice and policy', ACSSA Research Summary No. 2, 2012, p 3.
- 77 Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 7: Child sexual abuse at the Parramatta Training School for Girls and the Institution for Girls in Hay,* Sydney, 2014, p 32.
- 78 Names changed, private session, 'Becky' and 'Gretel'.
- Human Rights and Equal Opportunity Commission, *Bringing them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, p 255.
- Transcript of R J Kitson, Case Study 7, 26 February 2014 at 4918:8–16.
- Human Rights and Equal Opportunity Commission, *Bringing them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, p 258.
- Name changed, private session, 'Vernon'.
- Transcript of S Szoredi, Case Study 57, 31 March 2017 at 27805–27808.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 66; DA Wolfe, PG Jaffe & JL Jette, 'The impact of child abuse in community institutions and organizations: Advancing professional and scientific understanding', *American Psychological Association*, vol 10, no 2, 2003, p 186.
- Transcript of F Golding, Case Study 57, 31 March 2017 at 27823: 22–37.
- Name changed, private session, 'Aden'.
- Name changed, private session, 'Sean Michael'.
- Name changed, private session, 'Les'.
- 89 Transcript of APK, Case Study 21, 3 December 2014 at T10975:25–31; Exhibit 21-0006; 'Statement of APK', Case Study 21, STAT.0429.001.0001_R at 0011_R.
- 90 Transcript of APK, Case Study 21, 3 December 2014 at 10975:4–6; Exhibit 21-0006, 'Statement of APK', Case Study 21, STAT.0429.001.0001_R at 0018_R.
- 91 Transcript of CJA, Case Study 40, 21 June 2016 at 19285:13–17.
- 92 Transcript of G T Greaves, Case Study 40, 21 June 2016 at T19330:34–35.
- 93 Transcript of G T Greaves, Case Study 40, 21 June 2016 at 19332:17–21.
- Transcript of D L Lockwell, Case Study 2, 22 October 2013 at 651:25–27.
- 95 Transcript of D L Lockwell, Case Study 2, 22 October 2013 at 651:11–15.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 12: The response of an independent school in Perth to concerns raised about the conduct of a teacher between 1999 and 2009*, Sydney, 2015, p 31.
- 97 Transcript of WH, Case Study 12, 20 May 2014 at WA2504:18–26.
- 98 Transcript of A Ryan, Case Study 28, 16 December 2015 at 14961:30–40; Exhibit 28-0122, 'Statement of Ann Ryan', Case Study 28, STAT.0743.001.0001 R at 0014-R.
- 99 Name changed, private session, 'Karla'.
- 100 Name changed, private session, 'Francie'.
- M Liddell & SC Taylor, Women's experiences of learning about the involvement of a partner possessing child abuse material in Australia, RMIT University, Victoria, 2015, p 3.
- M Liddell & SC Taylor, Women's experiences of learning about the involvement of a partner possessing child abuse material in Australia, RMIT University, Victoria, 2015, p 8.
- 103 Y Duane, A Carr, J Cherry, K McGrath & D O'Shea, 'Experiences of parents attending a programme for families of adolescent child sexual abuse perpetrators in Ireland', *Child Care in Practice*, vol 8, no 1, 2002.
- 104 S Hackett, Children and Young People with Harmful Sexual Behaviours, Research in Practice, Devon, UK, 2014, p 60.
- Transcript of A D Collins, Case Study 28, 21 May 2015 at C8385: 39–41; Exhibit 28-0025, 'Statement of Andrew Collins', Case Study 28, STAT.0568.001.0001 R at 0010 R.
- 106 K McPhillips, "'Unbearable knowledge": Managing cultural trauma at the Royal Commission', *Psychoanalytic Dialogues*, vol 27, no 2, 2017, p 134.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 67; Family and Community Development Committee, *Betrayal of Trust: Inquiry into handling of child abuse by religious and other non-government organisations,* Victoria, 2013, volume 1, pp 80–1; Law Commission of Canada, *Restoring dignity: Responding to child abuse in Canadian institutions,* Minister of Public Works and Government Services, Canada, 2000, pp 65–6.
- Family and Community Development Committee, *Betrayal of trust: Inquiry into handling of child abuse by religious and other non-government organisations*, Victoria, 2013, p 80.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 67.
- Australia, House of Representatives, 13 February 2008, Speech, p 167 (Kevin Rudd, Prime Minister).
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 19.

- 112 Royal Commission Aboriginal and Torres Strait Islander community consultations 2014–2017.
- Transcript of M Bamblett, Case Study 57, 29 March 2017 at 27628:31–45. Transcript of G Gee, Case Study 57, 31 March 2017 at 27812:23–27, Transcript of K Menzies, Case Study 57, 31 March at 27832:25–37.
- Royal Commission Aboriginal and Torres Strait Islander community consultations 2014–2017; Transcript of M Bamblett, Case Study 57, 29 March 2017 at 27628:31–45; Transcript of K Menzies, Case Study 57, 31 March at 27832:25–37.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, *Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 18.
- 116 Transcript of K Menzies, Case Study 57, 31 March at 27832:25–37.
- 117 P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, *Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 21.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 19. See also J Atkinson, Trauma trails, recreating song lines: The transgenerational effects of trauma in Indigenous Australia, Spinifex Press, North Melbourne, 2002; K Erikson, Everything in its path: Destruction of community in the Buffalo Creek flood, Simon and Schuster, New York, 1976, p 47.
- 119 Royal Commission Aboriginal and Torres Strait Islander community consultations, 2014–2017.
- L Peeters, S Hamann & K Kelly, 'The Marumali Program: Healing for Stolen Generations' in P Dudgeon, H Milroy & R Walker (eds), *Working Together: Aboriginal and Torres Strait Islander Mental Health and Wellbeing Principles and Practice*, 2nd edn, Commonwealth of Australia, Canberra, 2014, p 496.
- Human Rights and Equal Opportunity Commission, *Bringing them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, p 37.
- Human Rights and Equal Opportunity Commission, *Bringing them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, p 162.
- Human Rights and Equal Opportunity Commission, *Bringing them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, pp 162–3.
- Human Rights and Equal Opportunity Commission, *Bringing them Home: Report of the National Inquiry into the Separation of Aboriginal and Torres Strait Islander Children from Their Families,* Human Rights and Equal Opportunity Commission, Sydney, 1997, p 162.
- 125 Transcript of M Bamblett, Case Study 57, 29 March 2017 at 27628:1–8.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 24. See also Royal Commission into Aboriginal Deaths in Custody, Royal Commission into Aboriginal Deaths in Custody: National Report, Australian Government Publishing Service, Canberra, 1991; C Tilbury, 'The over-representation of Indigenous children in the Australian child welfare system', International Journal of Social Welfare, vol 18, no 1, 2009; H McGlade, Our Greatest Challenge: Aboriginal Children and Human Rights, Aboriginal Studies Press, Canberra, 2012, p 33; J Atkinson, J Nelson, R Brooks, C Atkinson & K Ryan, 'Addressing individual and community transgenerational trauma' in P Dudgeon, H Milroy and R Walker (eds), Working Together: Aboriginal and Torres Strait Islander Mental Health and Wellbeing Principles and Practices, Commonwealth of Australia, Canberra, 2014, p 292.
- P Anderson, M Bamblett, D Bessarab, L Bromfield, S Chan, G Maddock, K Menzies, M O'Connell, G Pearson, R Walker & M Wright, *Aboriginal and Torres Strait Islander children and child sexual abuse in institutional settings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, pp 26–27.
- 128 Transcript of P McClellan, Case Study 28, 19 May 2015 at C8103:25–36.
- 129 Family and Community Development Committee, *Betrayal of trust: Inquiry into handling of child abuse by religious and other non-government organisations,* Victoria, 2013, pp 80–82.
- Transcript of A D Collins, Case Study 28, 21 May 2015 at C8385:3–18; Exhibit 28-0025, 'Statement of Andrew Collins', Case Study 28, STAT.0568.001.0001_R at 0010_R.
- Transcript of CQT, Case Study 43, 6 September 2016 at C18033:44-C18034:13; Exhibit 43-0022, 'Statement of CQT', Case Study 43, STAT.1173.001.0001_R at 0009_R.
- Transcript of GJ McDonald, Case Study 43, 31 August 2016 at C17601:28–36; Exhibit 43-0001, 'Statement of Gerard McDonald', Case Study 43, STAT.1168.001.0001 at 0012_R.
- 133 Transcript of BAP, Case Study 28, 19 May 2015 at C8179:9–14; Exhibit 28-0007, 'Statement of BAP', Case Study 28, STAT.0578.001.0001 R at 0012 R.
- Royal Commission into Institutional Responses to Child Sexual Abuse, *Report of Case Study No 29: The response of the Jehovah's Witnesses and Watchtower Bible and Tract Society of Australia Ltd to allegations of child sexual abuse*, Sydney, 2016. p 57.
- See K Kaufman & M Erooga, *Risk profiles for institutional child sexual abuse: A literature review,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, pp 48–51.

- 136 Royal Commission multicultural public forums, 2016.
- 137 Royal Commission multicultural public forums, 2016; P Sawriker, *Working with ethnic minorities and across cultures in western child protection systems*, Routledge, London, 2017, p 104.
- 138 Royal Commission multicultural public forums, 2016.
- 139 Royal Commission multicultural public forums, 2016.
- 140 Royal Commission multicultural public forums, 2016.
- 141 Royal Commission multicultural public forums, 2016.
- Royal Commission multicultural public forums, 2016; P Sawriker, *Working with ethnic minorities and across cultures in western child protection systems*, Routledge, London, 2017, pp 102–3.
- 143 Royal Commission multicultural public forums, 2016.
- Royal Commission multicultural public forums, 2016.
- 145 Transcript of J Kaur, Case Study 57, Wednesday 29 March 2017 at 27639:1–7.
- 146 Exhibit 37-0026, 'Statement of Rebecca Davies', Case Study 37, STAT.0867.001.0001 R at 0012 R and 0020 R-0021 R.
- 147 Transcript of J Davies, Case Study 37, 10 March 2016 at 17211:47 and 17212:1–6.
- 148 Transcript of C Parsons, Case Study 32, 3 September 2015 at C10237:4–15.
- 149 Name changed, private session, 'Myra Heather'.
- For example: Transcript of S Szoredi, Case Study 57, 31 March 2017 at 27805–27808.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors:*A rapid review of research findings, report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 39; M Proeve, C Malvaso & P DelFabbro, *Evidence and frameworks for understanding perpetrators of institutional child sexual abuse,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2016, p 18.
- T Blakemore, JL Herbert, F Arney & S Parkinson, *Impacts of institutional child sexual abuse on victims/survivors: A rapid review of research findings,* report prepared for the Royal Commission into Institutional Responses to Child Sexual Abuse, Sydney, 2017, p 61.
- L McCann & LA Pearlman, 'Vicarious traumatization: A framework for understanding the psychological effects of working with victims', *Journal of Traumatic Stress*, vol 3, no 1, 1990; Transcript of M Watson, Case Study 57, 28 March 2017 at 27503:12–15.
- J Dunkley & TA Whelan, 'Vicarious traumatisation: Current status and future directions', *British Journal of Guidance & Counselling*, vol 34, no 1, 2006.
- 155 P Levin & S Greisberg, 'Vicarious trauma in attorneys', Pace Law Review, vol 24, no 1, 2003.
- V Follette, M Polusny & K Milbeck, 'Mental health and law enforcement professionals: Trauma history, psychological symptoms, and impact of providing services to child sexual abuse survivors', *Professional Psychology: Research and Practice*, vol 25, no 3, 1994, p 279.
- SM Wasco & R Campbell, 'Emotional rections of rape victim advocates: A multiple case study of anger and fear', Psychology of Women Quarterly, vol 26, no 2, 2002, p 121, citing Alexander et al. 1989.
- 158 Name changed, private session, 'Roslyn'.
- There is no nationally representative data on the prevalence of child sexual abuse in Australia from which estimates of the cost impact can be drawn. Moreover, child sexual abuse is not consistently defined within research and the impact of child sexual abuse includes a range of outcomes that are difficult to measure in monetary terms.
- 160 X Fang, DS Brown, CS Florence & JA Mercy, 'The economic burden of child maltreatment in the United States and implications for prevention', *Child Abuse & Neglect*, vol 36, no 2, 2012; P Taylor, P Moore, L Pezzullo, J Tucci, C Goddard & LD Bortoli, *The cost of child abuse in Australia*, Australian Childhood Foundation and Child Abuse Prevention Research Australia, Melbourne, 2008.
- 161 R McLachlan, G Gilfillan & J Gordon, *Deep and persistent disadvantage: Productivity Commission staff working paper*, Canberra, 2013, p 169.
- R McLachlan, G Gilfillan & J Gordon, *Deep and persistent disadvantage: Productivity Commission staff working paper*, Canberra. 2013. p 171.
- R McLachlan, G Gilfillan & J Gordon, *Deep and persistent disadvantage: Productivity Commission staff working paper*, Canberra, 2013, pp 169–70.
- 164 R McLachlan, G Gilfillan & J Gordon, *Deep and persistent disadvantage: Productivity Commission staff working paper*, Canberra, 2013, pp 169–70.
- Those who are the victims of crime associated with economic disadvantage may face other economic costs including having to repair damaged property or installing security systems. R McLachlan, G Gilfillan & J Gordon, Deep and persistent disadvantage: Productivity Commission staff working paper, Canberra, 2013, p 173.
- Other costs borne by society include the administrative costs associated with governments raising revenue and the impact of taxes on people's incentives to work and invest (known as deadweight cost). For example, people may choose to work less than they otherwise would have if they are faced with having to pay a higher rate of taxation to fund more government services. In Australia, the deadweight cost of taxation has been estimated to range from 8 cents per dollar of revenue (for the Goods and Services Tax) to 67 cents (for insurance taxes). KPMG Econtech, CGE analysis of the current Australian tax system, prepared for Department of Treasury, Excess Burden of Australian Taxes, 2010.
- 167 A Saied-Tessier, Estimating the costs of child sexual abuse in the UK, NSPCC, UK, 2014, p 19.

- Countries included in the study are: Indonesia, Thailand, Democratic People's Republic of Korea, Myanmar, Brunei Darussalam, Japan, Singapore, Cambodia, China, Cook Islands, Fiji, Kiribati, Lao People's Democratic Republic, Malaysia, Republic of Marshall Islands, Federated States of Micronesia, Mongolia, Nauru, Niue, Palau, Papua New Guinea, Philippines, Republic of Korea, Samoa, Solomon Islands, Tonga, Tuvalu, Vanuatu, and Vietnam. X Fang, DA Fry, DS Brown, JA Mercy, MP Dunne, AR Butchart, PS Corso, K Maynzyuk, Y Dzhygyr, Y Chen, A McCoy & DM Swales, 'The burden of child maltreatment in the East Asia and Pacific region', *Child Abuse & Neglect*, vol 42, 2015, pp 154, 159.
- This study estimated the number of survivors facing negative life outcomes as a result of maltreatment they experienced as children. Of the estimated 3.7 million survivors of child sexual, physical and emotional abuse, Pegasus Economics calculated that 1.04 million survivors would not have faced negative life outcomes had they not been maltreated as children. The remaining survivors either experienced positive life outcomes or would have had negative life outcomes even if they were not maltreated as children. To obtain the total cost to governments, the average annual cost to governments are calculated and applied to the estimated number of survivors who face negative life outcomes as a result of child maltreatment. C Kezelman, N Hossack, P Stavropouls & P Burley, *The cost of unresolved childhood trauma and abuse in adults in Australia,* Adults Surviving Child Abuse and Pegasus Economics, Sydney, 2015, p 41.
- 170 C Kezelman, N Hossack, P Stavropouls & P Burley, *The cost of unresolved childhood trauma and abuse in adults in Australia*, Adults Surviving Child Abuse and Pegasus Economics, Sydney, 2015, p 36.
- 171 Not included in the study's calculation but also considered to place high costs on society and governments are: relationship breakdowns, which impose significant costs on governments due to increased costs associated with single parent families, child support and the cost of divorce; smoking; and work impairment due to childhood trauma and abuse, which impacts on the government budget through lost revenue and welfare supports.
- P Taylor, P Moore, L Pezzullo, J Tucci, C Goddard & LD Bortoli, *The cost of child abuse in Australia,* Australian Childhood Foundation and Child Abuse Prevention Research Australia, Melbourne, 2008, p 135.



Commonwealth of Australia

Royal Commission into Institutional Responses to Child Sexual Abuse

ISBN 978-1-925622-54-6 Published December 2017